47-0772104

(I.R.S. Employer Identification Number)

UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

Post-Effective Amendment No. 1

Form S-4 REGISTRATION STATEMENT UNDER THE SECURITIES ACT OF 1933

ACI WORLDWIDE, INC.

(Exact Name of Registrant as Specified in its Charter)

7372

(Primary Standard Industrial Classification Code Number)

120 Broadway, Suite 3350 New York, New York 10271

 $\textbf{Tel.: (646) 348-6700} \\ \textit{(Address, including zip code, and telephone number, including area code, of registrant's principal executive offices)} \\$

Dennis P. Byrnes, Esq.

Executive Vice President, General Counsel and Secretary ACI Worldwide, Inc.

6060 Coventry Drive Elkhorn, Nebraska 68022

(402) 778-2118
(Name, address, including zip code, and telephone number, including area code, of agent for service)

Copies to:

Robert A. Profusek, Esq. Jones Day 222 East 41st Street New York, New York 10017 Tel.: (212) 326-3939

Approximate date of commencement of proposed sale of securities to the public: As soon as practicable after the effective date of this Registration Statement.

If the securities being registered on this Form are being offered in connection with the formation of a holding company and there is compliance with General Instruction G, check the following box o

If this Form is filed to register additional securities for an offering pursuant to Rule 462(b) under the Securities Act, check the following box and list the Securities Act registration statement number of the earlier effective registration statement for the same offering o

If this Form is a post-effective amendment filed pursuant to Rule 462(d) under the Securities Act, check the following box and list the Securities Act registration statement number of the earlier effective registration statement for the same offering o

Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer, or a smaller reporting company. See the definitions of "large accelerated filer," "accelerated filer" and "smaller reporting company" in Rule 12b-2 of the Exchange Act

Large accelerated filer ☑ Accelerated filer o Non-accelerated filer o Smaller reporting company o (Do not check if a smaller reporting company)

If applicable, place an X in the box to designate the appropriate rule provision relied upon in conducting this transaction:

Exchange Act Rule 13e-4(i) (Cross-Border Issuer Tender Offer) o Exchange Act Rule 14d-1(d) (Cross-Border Third-Party Tender Offer) o

Delaware (State or Other Jurisdiction of Incorporation or Organization)

The Registrant hereby amends this Registration Statement on such date or dates as may be necessary to delay its effective date until the Registrant shall file a further amendment which specifically states that this Registration Statement shall thereafter become effective in accordance with Section 8(a) of the Securities Act of 1933 or until the Registration Statement shall become effective on such date as the Commission, acting pursuant to said Section 8(a),

EXPLANATORY NOTE

This Post-Effective Amendment No. 1 to the Registration Statement on Form S-4 (File No. 333-176557) is filed primarily for the purpose of amending and restating the Prospectus/Offer to Exchange, the Letter of Election and Transmittal and the other materials related to the exchange offer to reflect the revised terms and conditions of the previously announced exchange offer pursuant to the Transaction Agreement, dated as of October 3, 2011, among ACI Worldwide, Inc., Antelope Investment Co. LLC and S1 Corporation.

The information in this prospectus/offer to exchange may change. The registrant may not complete the Exchange Offer and issue these securities until the registration statement filed with the Securities and Exchange Commission is effective. This prospectus/offer to exchange is not an offer to sell these securities and ACI and Antelope Investment Co. LLC are not soliciting an offer to buy these securities in any state or jurisdiction in which such offer is not permitted.



Offer to Exchange

Each Outstanding Share of Common Stock

of

S1 CORPORATION

for

0.3148 of a Share of ACI Common Stock

or \$10.00 in Cash,

subject to the proration procedures described in this prospectus/offer to exchange and the related letter of election and transmittal,

by

ANTELOPE INVESTMENT CO. LLC

a wholly-owned subsidiary of

ACI WORLDWIDE, INC.

Antelope Investment Co. LLC ("Offeror"), a Delaware limited liability company and a wholly-owned subsidiary of ACI Worldwide, Inc., a Delaware corporation, which we refer to as "ACI" or "we," "us" or "our," is offering, upon the terms and subject to the conditions set forth in this prospectus/offer to exchange and in the accompanying letter of election and transmittal, to exchange for each issued and outstanding share of common stock of S1 Corporation ("S1"), par value \$0.01 per share (the "S1 Shares"), validly tendered pursuant to the Exchange Offer and not properly withdrawn either of the following:

- 0.3148 of a share of ACI common stock (the "ACI Shares"), par value \$0.005 per share (the "Stock Consideration"); or
- \$10.00 in cash, without interest (the "Cash Consideration"),

subject to the proration procedures described in this prospectus/offer to exchange and the related letter of election and transmittal (together, as each may be amended, supplemented or otherwise modified from time to time, the "Exchange Offer").

The Exchange Offer is being made pursuant to the Transaction Agreement, dated as of October 3, 2011, among ACI, Offeror and S1. Pursuant to the Transaction Agreement, after the Exchange Offer is completed, subject to the approval of the S1 stockholders if required by applicable law, Offeror will merge with and into S1 (the "Second-Step Merger"). The S1 Board has unanimously (1) determined that the transactions contemplated by the Transaction Agreement are fair to, and in the best interests of, S1 and the S1 stockholders; (2) approved the transactions contemplated by the Transaction Agreement; and (3) determined to recommend that the S1 stockholders accept the Exchange Offer and tender their S1 Shares to Offeror pursuant to the Exchange Offer. The S1 Board unanimously recommends that the S1 stockholders accept the Exchange Offer by tendering their S1 Shares in the Exchange Offer.

This prospectus/offer to exchange amends and supersedes information included in the prospectus/offer to exchange dated September 21, 2011.

You should be aware that the \$10.00 per share Cash Consideration will have a value greater than the 0.3148 per share Stock Consideration if market prices for ACI Shares are less than \$31.77 per share. Furthermore, as explained below, if more than 66.2% of S1 Shares elect to receive cash, the proration procedures will result in some of those shares receiving stock. Conversely, if more than 33.8% of S1 Shares elect to receive stock, the proration procedures will result in some of those shares receiving stock. Conversely, if more than 33.8% of S1 Shares elect to receive stock, the proration procedures will result in some of those shares receiving as on October 12, 2011, the last trading

day prior to the date of this prospectus/offer to exchange and assuming the 33.8% Stock Consideration and the 66.2% Cash Consideration were allocated pro rata among all S1 Shares, which we refer to herein as "full proration", the blended value of the Cash Consideration and the Stock Consideration (together, the "Cash-Stock Consideration") as of the close of trading on October 12, 2011 was \$9.68 per S1 Share.

If market prices for ACI Shares upon consummation of the Exchange Offer are less than \$41.48, the Stock Consideration may be taxable to you, and would be taxable based on the trading price for ACI Shares on October 12, 2011, the last trading day prior to the date of this prospectus/offer to exchange. You are urged to obtain current trading price information prior to making any decision with respect to the Exchange Offer.

The equity capital markets have been highly volatile and market prices for ACI Shares and S1 Shares have fluctuated and can be expected to continue to fluctuate. S1 stockholders are urged to obtain current trading price information prior to making any decision with respect to the Exchange Offer.

S1 stockholders electing either the Cash Consideration or the Stock Consideration will be subject to proration so that 66.2% of S1 Shares will be exchanged for the Cash Consideration in the Exchange Offer. S1 stockholders who do not participate in the Exchange Offer and whose shares are acquired in the Second-Step Merger will receive \$6.62 in cash, without interest, and 0.1064 of an ACI Share (the "Proration Amount of Cash and Stock Consideration"). The elections of other S1 stockholders will affect whether a tendering S1 stockholder electing the Cash Consideration or the Stock Consideration receives solely the type of consideration elected or if a portion of such S1 stockholder's tendered S1 Shares is exchanged for another form of consideration. S1 stockholders who otherwise would be entitled to receive a fractional ACI Share will instead receive cash in lieu of any fractional ACI Share such holder may have otherwise been entitled to receive based on ten-day volume weighted average trading prices. See "The Exchange Offer — Elections and Proration" for a description of the proration procedure and "The Exchange Offer — Cash In Lieu of Fractional ACI Common Stock" for a description of the treatment of fractional ACI Shares.

ACI is not asking you for a proxy and you are not requested to send a proxy to ACI pursuant to the Exchange Offer.

THE EXCHANGE OFFER AND THE WITHDRAWAL RIGHTS WILL EXPIRE AT 5:00 P.M., EASTERN TIME, ON MONDAY, OCTOBER 31, 2011, OR THE "EXPIRATION TIME," UNLESS EXTENDED. THE EXCHANGE OFFER COULD BE SUBJECT TO MULTIPLE EXTENSIONS OF THE EXPIRATION TIME IF ALL OF THE CONDITIONS TO THE EXCHANGE OFFER ARE NOT SATISFIED OR WAIVED BY OFFEROR PRIOR TO THE MOST RECENT EXPIRATION TIME. BECAUSE CERTAIN CONDITIONS ARE OUTSIDE OUR CONTROL, THERE CAN BE NO ASSURANCE AS TO WHEN OR IF THE EXCHANGE OFFER MY BE WITHDRAWN AT ANY TIME PRIOR TO THE EXPIRATION TIME, BUT NOT DURING ANY SUBSEQUENT OFFERING PERIOD.

ACI Shares are listed on The NASDAQ Global Select Market under the ticker symbol "ACIW." S1 Shares are listed on The NASDAQ Stock Market under the ticker symbol "SONE."

FOR A DISCUSSION OF RISKS AND OTHER FACTORS THAT YOU SHOULD CONSIDER IN CONNECTION WITH THE EXCHANGE OFFER, PLEASE CAREFULLY READ THE SECTION OF THIS PROSPECTUS/OFFER TO EXCHANGE TITLED "RISK FACTORS" BEGINNING ON PAGE 25.

Offeror's obligation to accept S1 Shares for exchange and to exchange any S1 Shares for ACI Shares is subject to conditions, including (1) a condition that S1 stockholders shall have validly tendered and not withdrawn prior to the Expiration Time at least that number of S1 Shares that, when added to the S1 Shares then owned by ACI or any of its subsidiaries, constitutes a majority of the then-outstanding number of S1 Shares on a fully diluted basis (the "Minimum Tender Condition") and (2) a condition that the waiting period under the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended (the "HSR Act") has expired or terminated. The Exchange Offer is not conditioned on financing. The conditions to the Exchange Offer are described in the section of this prospectus/offer to exchange titled "The Exchange Offer — Conditions of the Exchange Offer."

Neither ACI nor Offeror has authorized any person to provide any information or to make any representation in connection with the Exchange Offer other than the information contained or incorporated by reference in this prospectus/offer to exchange and the accompanying letter of election and transmittal, and if any person provides any of this information or makes any representation of this kind, that information or representation must not be relied upon as having been authorized by ACI.

Neither the SEC nor any state securities commission has approved or disapproved of these securities or passed upon the adequacy or accuracy of this prospectus/offer to exchange. Any representation to the contrary is a criminal offense.

The date of this prospectus/offer to exchange is October 13, 2011.

	Page
QUESTIONS AND ANSWERS ABOUT THE EXCHANGE OFFER	v
SUMMARY OF THE EXCHANGE OFFER	1
The Companies	1
The Exchange Offer	2
The Second-Step Merger	2
Recommendation of the S1 Board	3
Reasons for the Exchange Offer and the Second-Step Merger	3
Conditions of the Exchange Offer	5
Ownership of ACI After the Exchange Offer	6
Comparative Market Price and Dividend Information	6
Interest of Executive Officers and Directors of ACI in the Exchange Offer	7
Interest of Executive Officers and Directors of S1 in the Exchange Offer	7
Source and Amount of Funds; Financing	8
Non-Solicitation by S1	8
Termination of the Transaction Agreement	9
Termination Fees	10
Appraisal/Dissenters' Rights	10
Certain Material Federal Income Tax Consequences	10
Accounting Treatment	11
Regulatory Approval and Status	11
<u>Listing of ACI Shares to be Issued Pursuant to the Exchange Offer and the Second-Step Merger</u>	12
Comparison of Stockholders' Rights	12
Expiration Time of the Exchange Offer	12
Extension, Termination and Amendment	13
Procedure for Tendering Shares	13
Withdrawal Rights	13
Acceptance for Exchange and Exchange of S1 Shares; Delivery of Exchange Offer Consideration	14
Cash in Lieu of Fractional ACI Shares	14
Elections and Proration	14
Risk Factors	14
SELECTED HISTORICAL CONSOLIDATED FINANCIAL DATA OF ACI	15
SELECTED HISTORICAL CONSOLIDATED FINANCIAL DATA OF S1	17
SUMMARY SELECTED UNAUDITED PRO FORMA COMBINED FINANCIAL INFORMATION	19
HISTORICAL AND PRO FORMA PER SHARE INFORMATION	22
COMPARATIVE MARKET PRICE AND DIVIDEND INFORMATION	23
RISK FACTORS	25
Risk Factors Relating to the Exchange Offer and the Second-Step Merger	25
Risk Factors Relating to S1's Businesses	28
Risk Factors Relating to ACI's Businesses	28
Risk Factors Relating to ACI Following the Exchange Offer	28
THE COMPANIES	30
<u>ACI</u>	30
<u>Offeror</u>	30

	Page
<u>S1</u>	30
THE ACQUISITION, BACKGROUND AND REASONS FOR THE EXCHANGE OFFER	31
Background of the Exchange Offer	31
Reasons for the Exchange Offer	39
THE TRANSACTION AGREEMENT	42
The Exchange Offer	42
Recommendation of the S1 Board	42
S1 Stockholder Meeting	42
Top-Up Option	43
Appraisal Rights/Dissenting Shares	43
Directors After the Acceptance Time	43
Second-Step Merger; Effect on Capital Stock	44
Merger Without Meeting of Stockholders; Special Meeting	44
Treatment of Stock Options; SARs; Restricted Stock	44
Representations and Warranties of the Parties in the Transaction Agreement	45
Conduct of Business Pending the Closing of the Second-Step Merger	47
Non-Solicitation by S1 of S1 Acquisition Proposals; Board Recommendation	48
Employee Matters	49
Director's and Officer's Indemnification and Insurance	50
Regulatory Filings; Efforts to Close	50
Other Agreements of the Parties	52
Conditions to Closing of the Second-Step Merger	52
Termination of the Transaction Agreement	53
Termination Fees	53
<u>Amendment</u>	54
Remedies; Specific Performance	54
THE EXCHANGE OFFER	55
<u>Overview</u>	55
Expiration Time of the Exchange Offer	55
Extension, Termination, Waiver and Amendment	56
Acceptance for Exchange and Exchange of S1 Shares; Delivery of Exchange Offer Consideration	57
Cash In Lieu of Fractional ACI Shares	58
Elections and Proration	58
Consequences of Tendering with No Election	59
Procedure for Tendering	59
Withdrawal Rights	62
Announcement of Results of the Exchange Offer	63
Ownership of ACI After the Exchange Offer	63
Material Federal Income Tax Consequences	63
Purpose and Structure of the Exchange Offer	66
Second-Step Merger	67
Appraisal/Dissenters' Rights	67
Plans for S1	67

	Page
Effect of the Exchange Offer on the Market for S1 Shares; NASDAQ Listing; Registration Under the Securities Exchange Act of 1934; Margin Regulations	68
Conditions of the Exchange Offer	69
Source and Amount of Funds	70
Interest; Letter of Credit Fees; Unused Commitment Fees	71
Certain Legal Matters; Regulatory Approvals	77
Certain Relationships With S1 and Interests of ACI in the Exchange Offer	79
Interest of Executive Officers and Directors of S1 in the Exchange Offer	80
Fees and Expenses	80
Accounting Treatment	81
DESCRIPTION OF ACI CAPITAL STOCK	82
COMPARISON OF STOCKHOLDERS' RIGHTS	84
Authorized Capital	84
Number of Directors	84
Structure of Board of Directors; Term of Directors	84
Removal of Directors	84
Vacancies on the Board of Directors	84
Special Meetings of Stockholders	85
Stockholder Proposals	86
Stockholder Nominations	88
Amendment of Certificate of Incorporation	89
Amendment of By-Laws	90
Limitations on Director Liability	90
<u>Dividends</u>	90
Stockholder Rights Plan	90
Restrictions on Business Combination	90
UNAUDITED CONDENSED COMBINED PRO FORMA FINANCIAL INFORMATION	91
FORWARD-LOOKING STATEMENTS	101
LEGAL MATTERS	101
<u>EXPERTS</u>	101
ADDITIONAL NOTE REGARDING THE EXCHANGE OFFER	102
WHERE YOU CAN FIND MORE INFORMATION	102
APPENDIX A DIRECTORS AND EXECUTIVE OFFICERS OF ACI	A-1
APPENDIX B DIRECTORS AND EXECUTIVE OFFICERS OF OFFEROR	B-1
APPENDIX C STOCK TRANSACTIONS IN THE PAST 60 DAYS	C-1
APPENDIX D TRANSACTION AGREEMENT	D-i
EX-23.1	
EX-23.2	
EX-233	
EX-991	
<u>EX-992</u>	
EX-993	
EX-99.4	

THIS PROSPECTUS/OFFER TO EXCHANGE INCORPORATES IMPORTANT BUSINESS AND FINANCIAL INFORMATION ABOUT ACI AND S1 FROM DOCUMENTS FILED WITH THE SEC THAT HAVE NOT BEEN INCLUDED IN OR DELIVERED WITH THIS PROSPECTUS/OFFER TO EXCHANGE. THIS INFORMATION IS AVAILABLE AT THE INTERNET WEBSITE THE SEC MAINTAINS AT HITTP://WWW.SEC.GOV, AS WELL AS FROM OTHER SOURCES. PLEASE SEE THE SECTION OF THIS PROSPECTUS/OFFER TO EXCHANGE TITLED "WHERE YOU CAN FIND MORE INFORMATION." YOU ALSO MAY REQUEST COPIES OF THESE DOCUMENTS FROM ACI, WITHOUT CHARGE, UPON WRITTEN OR ORAL REQUEST TO ACI'S INFORMATION AGENT AT ITS ADDRESS OR TELEPHONE NUMBER SET FORTH BELOW AND ON THE BACK COVER OF THIS PROSPECTUS/OFFER TO EXCHANGE. IN ORDER TO RECEIVE TIMELY DELIVERY OF THE DOCUMENTS, YOU MUST MAKE YOUR REQUEST NO LATER THAN OCTOBER 24, 2011, OR FIVE BUSINESS DAYS PRIOR TO THE EXPIRATION TIME, WHICHEVER IS LATER.

S1 STOCKHOLDERS WILL BE ABLE TO OBTAIN A FREE COPY OF ANY FILING CONTAINING INFORMATION ABOUT THE PARTIES FROM THE SEC'S WEB SITE AT HTTP://www.sec.gov. documents filed by Aci, if and when available, may also be obtained for free from Aci's web site at http://www.aciworldwide.com or upon written or oral request to the information agent at innisfree M&A inc., 501 madison avenue, 20th floor, New York, New York 10022, S1 STOCKHOLDERS MAY CALL TOLL-FREE AT (888) 750-5834, And Banks and Brokerage firms may call collect (212) 750-5833.

The information agent for the Exchange Offer is:

Innisfree

501 Madison Avenue, 20th Floor New York, New York 10022 Stockholders May Call Toll Free: (888) 750-5834 Banks and Brokers May Call Collect: (212) 750-5833

OUESTIONS AND ANSWERS ABOUT THE EXCHANGE OFFER

Below are some of the questions that you as a holder of S1 Shares may have regarding the Exchange Offer and answers to those questions. The answers to these questions do not contain all the information relevant to your decision whether to tender your S1 Shares in the Exchange Offer, and ACI urges you to read carefully the remainder of this prospectus/offer to exchange and the letter of election and transmittal circulated with this prospectus/offer to exchange.

Why was the Exchange Offer amended?

On July 26, 2011, ACI proposed to acquire S1 (the "Original ACI Merger Proposal") for a combination of cash and stock having a blended value of \$9.50 per share, assuming full proration and based on the closing market price for ACI Shares on July 25, 2011. On August 30, 2011, ACI commenced the exchange offer (the "Original ACI Exchange Offer") to acquire S1 for a combination of cash and ACI Shares that, on a blended basis and assuming full proration, had a value of \$9.44 per S1 Share based on the closing sales price for ACI Shares on August 29, 2011. On September 2, 2011, the Board of Directors of S1 (the "S1 Board") rejected the Original ACI Exchange Offer and recommended that S1 stockholders not tender their S1 Shares pursuant to the Original ACI Exchange Offer. Between August 30, 2011 and October 3, 2011, senior managers and representatives of S1 and ACI had discussions regarding ACI's revised acquisition proposal, conducted due diligence of the companies' respective businesses and operations, and then negotiated the terms of a Transaction Agreement that was entered into by ACI, Offeror and S1 on October 3, 2011.

The Exchange Offer is being made pursuant to that agreement where, among other things, the cash offer price was increased by \$0.42 per share, assuming full proration, and the conditions of the Original ACI Exchange Offer were modified as described in this prospectus/offer to exchange.

Who is making the Exchange Offer?

The Exchange Offer is being made by ACI, a Delaware corporation, through its wholly owned subsidiary, Antelope Investment Co. LLC, a Delaware limited liability company. ACI develops, markets, installs and supports a broad line of software products and services primarily focused on facilitating electronic payments. In addition to ACI's own products, it also distributes, or acts as a sales agent for, software developed by third parties. These products and services are used principally by financial institutions, retailers and electronic payment processors, both in domestic and international markets. Most of ACI's products are sold and supported through distribution networks covering three geographic regions — the Americas, Europe/Middle East/Africa and Asia/Pacific. Each distribution network has its own sales force that it supplements with independent reseller and/or distributor networks. ACI's products are marketed under the ACI Worldwide and ACI Payment Systems brands.

What is Offeror seeking for exchange in the Exchange Offer?

Offeror seeks to acquire all of the issued and outstanding S1 Shares.

Is there an agreement governing the Exchange Offer?

Yes. On October 3, 2011, ACI and Offeror entered into the Transaction Agreement with S1 as a means to acquire all of the outstanding S1 Shares.

Does the S1 Board support the Exchange Offer?

Yes. The S1 Board has unanimously (1) determined that the transactions contemplated by the Transaction Agreement are fair to, and in the best interests of, S1 and the S1 stockholders; (2) approved the transactions contemplated by the Transaction Agreement; and (3) determined to recommend that the S1 stockholders accept the Exchange Offer and tender their S1 Shares to Offeror pursuant to the Exchange Offer. The S1 Board unanimously recommends that S1 stockholders accept the Exchange Offer by tendering their S1 Shares into the Exchange Offer. Information about the recommendation of the S1 Board is more fully described in Amendment No. 2 to S1's Solicitation/Recommendation Statement on Schedule 14D-9, which is being mailed to S1 stockholders together with this prospectus/offer to exchange and is incorporated herein by reference.

What will I receive for my S1 Shares in the Exchange Offer?

ACI is offering to exchange for each issued and outstanding S1 Share validly tendered pursuant to the Exchange Offer and not properly withdrawn either of the following:

- · 0.3148 of an ACI Share (Stock Consideration); or
- \$10.00 in cash, without interest (Cash Consideration),

subject to the proration procedures described in this prospectus/offer to exchange and the related letter of election and transmittal.

You should be aware that the \$10.00 per share Cash Consideration will have a value greater than the 0.3148 per share Stock Consideration if market prices for ACI Shares are less than \$31.77 per share. Furthermore, as explained below, if more than 66.2% of \$11\$ Shares elect to receive cash, the proration procedures will result in some of those shares receiving stock. Conversely, if more than 33.8% of \$51\$ Shares elect to receive stock, the proration procedures will result in some of those shares receiving stock as the closing sales price for ACI Shares on October 12, 2011, the last trading day prior to the date of this prospectus/offer to exchange and assuming the 33.8% Stock Consideration and the 66.2% Cash Consideration were allocated pro rata among all \$11\$ Shares, which we refer to herein as "full proration", the blended value of the Cash-Stock Consideration as of the close of trading on October 12, 2011 was \$9.68 per \$11\$ Shares.

S1 stockholders who do not participate in the Exchange Offer and whose shares are acquired in the Second-Step Merger will receive the Proration Amount of Cash and Stock Consideration. The elections of other S1 stockholders will affect whether a tendering S1 stockholder electing the Cash Consideration or the Stock Consideration receives solely the type of consideration elected or if a portion of such S1 stockholder's tendered S1 Shares is exchanged for another form of consideration. S1 stockholders who otherwise would be entitled to receive a fractional ACI Share such holder may have otherwise been entitled to receive based on ten-day volume weighted average trading prices. See "The Exchange Offer — Elections and Proration" for a detailed description of the proration procedure and "The Exchange Offer — Cash In Lieu of Fractional ACI Shares" for a detailed description of the treatment of fractional ACI Shares.

The equity capital markets have been highly volatile and market prices for ACI Shares have fluctuated and will fluctuate, and could be higher or lower than the price of ACI Shares at or after the Expiration Time. Accordingly, S1 stockholders are urged to obtain current trading price information for ACI Shares prior to deciding whether to tender shares pursuant to the Exchange Offer, whether to exercise withdrawal rights as provided herein and, with respect to the election, whether to receive the Cash Consideration or the Stock Consideration or some combination thereof.

Solely for purposes of illustration, the following table indicates the value of the Cash Consideration, the Stock Consideration and the blended value of the Cash-Stock Consideration based on different assumed prices for ACI Shares:

	Assuming No	Proration	Assuming Full Proration				
Assumed ACI Share Price	Value of Stock Consideration	Value of Cash Consideration	Value of Stock Consideration	Value of Cash Consideration	Value of Cash-Stock Consideration		
\$37.93(1)	\$11.94	\$10.00	\$4.04	\$6.62	\$10.66		
\$35.70(2)	\$11.24	\$10.00	\$3.80	\$6.62	\$10.42		
\$30.49(3)	\$ 9.60	\$10.00	\$3.24	\$6.62	\$ 9.86		
\$27.54(4)	\$ 8.67	\$10.00	\$2.93	\$6.62	\$ 9.55		
\$28.77(5)	\$ 9.06	\$10.00	\$3.06	\$6.62	\$ 9.68		
\$22.70(6)	\$ 7.15	\$10.00	\$2.42	\$6.62	\$ 9.04		

⁽¹⁾ Represents highest sales price for ACI Shares in the 52 weeks ending October 12, 2011, the last trading day prior to the date of this prospectus/offer to exchange (the "52-Week Period").

⁽²⁾ Represents closing sales price for ACI Shares on July 25, 2011, the last trading day prior to the announcement of the Original ACI Merger Proposal.

- (3) Represents closing sales price for ACI Shares on August 29, 2011, the last trading day prior to the commencement of the Original ACI Exchange Offer.
- (4) Represents closing sales price for ACI Shares on September 30, 2011, the last trading day prior to the announcement of the Transaction Agreement.
- (5) Represents closing sales price for ACI Shares on October 12, 2011, the last trading day prior to the date of this prospectus/offer to exchange.
- (6) Represents the lowest sales price for ACI Shares in the 52-Week Period.

The prices of ACI Shares used in the above table, and the assumptions regarding the mix of cash and/or stock a hypothetical S1 stockholder would receive, are for purposes of illustration only. The value of the Stock Consideration will change as the price of ACI Shares fluctuates during the Exchange Offer period and thereafter, and may therefore be higher or lower than the prices set forth in the examples above at the expiration of the Exchange Offer and at the time you receive the ACI Shares. S1's stockholders are encouraged to obtain current market quotations for the ACI Shares and the S1 Shares prior to making any decision with respect to the Exchange Offer. S1 stockholders should also consider the potential effects of proration and should obtain current market quotations for ACI Shares and the S1 Shares before deciding whether to tender pursuant to the Exchange Offer and before electing the form of consideration they wish to receive. Please also see the section of this prospectus/offer to exchange entitled "Risk Factors."

Will I be taxed on the ACI Shares and cash I receive?

Based on closing trading prices of ACI Shares as of October 12, 2011, the Exchange Offer would be taxable to you because the integrated transaction would not qualify as a reorganization. If the integrated transaction does not qualify as a reorganization, you may be taxed on your exchange of \$1 Shares for the Stock Consideration in the Exchange Offer or the Second-Step Merger, depending on the surrounding facts. In general in this case, you will recognize a capital gain or a capital loss to the extent of the difference between your adjusted tax basis in your shares and the sum of the Cash Considerations and the fair market value of the Stock Consideration you receive.

If the Exchange Offer and the Second-Step Merger qualified as component parts of an integrated transaction that constitutes a reorganization within the meaning of Section 368(a) of the Internal Revenue Code of 1986, as amended (the "Internal Revenue Code"), your exchange of S1 Shares for the Stock Consideration should be tax free, except to the extent that you also receive cash, as discussed below. Whether or not such transactions will so qualify is dependent on whether certain factual requirements are met, including that the Exchange Offer and Second-Step Merger are "interdependent" (that is, ACI would not undertake the Exchange Offer without the intention and expectation of completing the Second-Step Merger). In addition, there must be a "continuity of interest" of holders of S1 Shares in the combined company. ACI believes that this test should be satisfied if the total value of the Stock Consideration represents at least 40% of the total value of the consideration received by holders of S1 Shares, and may be satisfied at a slightly lower percentage. If market prices for ACI Shares upon consummation of the Exchange Offer are less than \$41.48, the Stock Consideration would represent less than 40% of the total value of the Exchange Offer consideration. You are urged to obtain current trading price information prior to making any decision with respect to the Exchange Offer. We cannot provide any assurance as to whether these conditions will be satisfied at this time, since it may be affected, among other things, by the total value of the Stock Consideration at the time of the consummation of the Exchange Offer and the Second-Step Merger.

If the integrated transaction constitutes a reorganization within the meaning of Section 368(a) of the Internal Revenue Code, any gain (but not loss) you realize on the transaction will be treated as a taxable capital gain or dividend in an amount equal to the lesser of (1) the excess of the sum of the Cash Consideration and the fair market value of the Stock Consideration you receive in the transaction over your basis in your shares and (2) the amount of cash you receive in the transaction, including any cash you receive in lieu of a fractional ACI Share, depending on your circumstances. For more information, please see the section of this prospectus/offer to exchange titled "The Exchange Offer — Certain Material Federal Income Tax Consequences."

ACI urges you to contact your own tax advisor to determine the particular tax consequences to you as a result of the Exchange Offer and/or the Second-Step Merger.

What is the Exchange Offer worth today?

The value of the Exchange Offer depends in part on market prices for ACI Shares. You should be aware that the \$10.00 per share Cash Consideration will have a value greater than the 0.3148 per share Stock Consideration if market prices for ACI Shares are less than \$31.77 per share. As of the close of trading on October 12, 2011, the most recent date prior to the date of this prospectus/offer to exchange, the blended value of the Cash-Stock Consideration, assuming full proration, was \$9.68 per S1 Share. When we say "full proration", we mean that the 33.8% Stock Consideration and the 66.2% Cash Consideration were allocated pro rata among all S1 Shares. As explained herein, if more than 66.2% of S1 Shares elect to receive cash, the proration procedures will result in some of those shares receiving cash.

What is the purpose of the Exchange Offer?

The Exchange Offer is intended to allow ACI, through Offeror, to acquire all of the issued and outstanding S1 Shares. We intend, as promptly as possible after completion of the Exchange Offer, to consummate the Second-Step Merger of S1 with and into Offeror pursuant to the General Corporation Law of the State of Delaware, as amended (the "DGCL"). The purpose of the Second-Step Merger is for ACI to acquire all outstanding S1 Shares that are not acquired in the Exchange Offer. In this Second-Step Merger, each remaining S1 Share (other than S1 Shares held in treasury by S1 or owned by ACI or its wholly owned subsidiaries, certain restricted S1 Shares converted into restricted ACI Shares pursuant to the Transaction Agreement and S1 Shares held by S1 stockholders who properly exercise applicable dissenters' rights under Delaware law) would be cancelled and exchanged for the Proration Amount of Cash and Stock Consideration. After this Second-Step Merger, ACI would own all of the issued and outstanding S1 Shares. Please see the sections of this prospectus/offer to exchange titled "The Exchange Offer — Purpose and Structure of the Exchange Offer"; "The Exchange Offer — Second-Step Merger," and "The Exchange Offer — Plans for S1."

What is the "Top-Up Option" and when will it be exercised?

S1 has granted to Offeror an irrevocable option (the "Top-Up Option"), for so long as the Transaction Agreement has not been terminated, to purchase from S1 up to the number of authorized and unissued S1 Shares equal to the lowest number of S1 Shares that, when added to the number of S1 Shares owned by ACI, Offeror or any subsidiary of ACI at the time of the exercise of the Top-Up Option, constitutes at least one S1 Share more than 90% of the S1 Shares (after giving effect to the issuance of S1 Shares to be issued upon exercise of the Top-Up Option (such S1 Shares to be issued upon exercise of the Top-Up Shares")).

The Top-Up Option may be exercised by Offeror only once, in whole but not in part, at any time during the two-business day period following the Expiration Time, or if the Exchange Offer is extended, during the two-business day period following the expiration date of such Subsequent Offering Period, and only if Offeror owns as of such time more than 50% but less than 90% of S1 Shares outstanding. Please see the section of this prospectus/offer to exchange titled "The Transaction Agreement — Top-Up Option." If the Top-Up Option is exercised, Offeror will be able to complete the Second-Step Merger as a "short-form" merger under the DGCL without an S1 stockholder vote.

When do you expect the Exchange Offer to be completed?

We intend to complete the Exchange Offer as soon as we can. The completion of the Exchange Offer is subject to the satisfaction or waiver of the conditions to the Exchange Offer. As discussed in "The Exchange Offer — Extension, Termination, Waiver and Amendment", the Transaction Agreement provides that Offeror will extend the Expiration Time if such conditions are not satisfied or waived at such time. There can be no assurance when or whether these conditions will be satisfied or waived.

What are the conditions of the Exchange Offer?

The Exchange Offer is conditioned upon, among other things, the following:

- S1 stockholders shall have validly tendered and not properly withdrawn prior to the Expiration Time at least that number of S1 Shares (together with the S1 Shares then owned by ACI, Offeror or any of ACI's other subsidiaries), shall constitute a majority of the S1 Shares issued and outstanding on a fully diluted basis. We refer to this condition as the "Minimum Tender Condition."
- The registration statement of which this prospectus/offer to exchange is a part shall have been declared effective under the Securities Act of 1933, as amended (the "Securities Act"), and no stop order suspending the effectiveness of the registration statement shall have been issued and no proceedings for that purpose shall have been initiated or threatened by the SEC, and ACI shall have received all necessary state securities law or "blue sky" authorizations.
- Any applicable waiting period under the HSR Act, and, if applicable, any agreement with the Federal Trade Commission (the "FTC") or the Antitrust Division of the U.S. Department of Justice (the "Antitrust Division") not to accept S1 Shares for exchange in the Exchange Offer, shall have expired or shall have been terminated prior to the Expiration Time (the "HSR Condition").
- Any clearance, approval, permit, authorization, waiver, determination, favorable review or consent of any Governmental Authority, other than the HSR Condition, shall have been obtained and such approvals shall be in full force and effect, or any applicable waiting periods for such clearances or approvals shall have expired, except for any failures that would not reasonably be expected to have a material adverse effect on ACI or S1.
- · Any of the following fail to be true:
 - (1) the representations and warranties of the S1 relating to organization, standing and power, authority, capital structure, absence of certain changes or events, brokers and vote required, as set forth in the Transaction Agreement (the "Fundamental S1 Corporate Representations") were true and correct as of October 3, 2011 and will be true and correct on and as of the Expiration Time with the same force and effect as if made at the Expiration Time (in either case other than those representations and warranties which address matters only as of a particular date, which representations and warranties shall have been true and correct as of such particular date), except in either case contemplated by this clause (1) for de minimis inaccuracies and (2) the other representations and warranties of S1 set forth in the Transaction Agreement were true and correct as of October 3, 2011 and will be true and correct on and as of the Expiration Time with the same force and effect as if made on the Expiration Time (in either case other than those representations and warranties which address matters only as of a particular date, which representations shall have been true and correct as of such particular date), except in either case contemplated by this clause (2) where the failure of such representations and warranties to be true and correct (disregarding all qualifications or limitations as to materiality, material adverse effect or words of similar import set forth therein) has not had and would not reasonably be expected to have a material adverse effect on S1:
 - S1 has performed or complied in all material respects with all agreements and covenants required by the Transaction Agreement to be performed or complied with by it on or prior to the
 Expiration Time; and
 - since October 3, 2011, there shall not have occurred any material adverse change in the business, financial condition or continuing results of S1 and its subsidiaries, taken as a whole (excluding certain events specified in the Transaction Agreement).

The Exchange Offer is not conditioned on financing. Subject to applicable law and the terms of the Transaction Agreement, we may waive the foregoing conditions, other than the Minimum Tender Condition.

Do I need to grant proxies to ACI if I wish to accept the Exchange Offer?

No. ACI is not asking you for a proxy in this prospectus/offer to exchange and you are not requested to send a proxy to ACI pursuant to the Exchange Offer.

Can ACI acquire S1 without completing the Exchange Offer?

ACI may only complete the Second-Step Merger if it purchases the S1 Shares pursuant to the Exchange Offer. However, the Transaction Agreement gives ACI the right to require that S1 convene a stockholder meeting to approve a merger in which the S1 stockholders would have the right to receive the Proration Amount of Cash and Stock Consideration as a result of a merger of Offeror and S1 instead of the Exchange Offer. The terms and conditions of such a transaction would be substantially the same as the terms and conditions of the Exchange Offer. ACI had not determined whether to exercise this right as of the date of this prospectus/offer to exchange.

Will I have to pay any fee or commission to exchange S1 Shares?

If you are the record owner of your S1 Shares and you tender your S1 Shares in the Exchange Offer, you will not have to pay any brokerage fees, commissions or similar expenses. If you own your S1 Shares through a broker, dealer, commercial bank, trust company or other nominee and your broker, dealer, commercial bank, trust company or other nominee tenders your S1 Shares on your behalf, your broker, dealer, commercial bank, trust company or other nominee may charge a fee for doing so. You should consult your broker, dealer, commercial bank, trust company or other nominee to determine whether any charges will apply.

Is ACI's financial condition relevant to my decision to tender S1 Shares in the Exchange Offer?

Yes. ACI's financial condition is relevant to your decision to tender your S1 Shares because the consideration you will receive if your S1 Shares are exchanged in the Exchange Offer will consist of a combination of ACI Shares and cash. You should therefore consider ACI's financial condition before you decide to become one of ACI's stockholders through the Exchange Offer. You should also consider the likely effect that ACI's acquisition of S1 will have on ACI's financial condition. This prospectus/offer to exchange contains financial information regarding ACI and S1, as well as pro forma financial information (which does not reflect any of our expected synergies) for the acquisition of all of the issued and outstanding S1 Shares by ACI, all of which we encourage you to review.

Does ACI have the financial resources to complete the Exchange Offer and the Second-Step Merger?

The Exchange Offer consideration will consist of ACI Shares and cash (including, cash paid in lieu of any fractional ACI Shares to which any S1 stockholder may be entitled). The Exchange Offer and the Second-Step Merger are not conditioned upon any financing arrangements or contingencies.

ACI has received a commitment letter from Wells Fargo Securities, LLC ("Wells Fargo") and Wells Fargo Bank, N.A. ("Wells Fargo Bank"), to provide, subject to certain conditions, up to \$450 million for the purpose of financing a portion of the cash component of the consideration to be paid for each S1 Share, as well as for other payments made in connection with the Exchange Offer and refinancing of ACI's existing revolving facility. No other plans or arrangements have been made to finance or repay such financing after the consummation of the Exchange Offer and the Second-Step Merger. No alternative financing arrangements or alternative financing plans have been made in the event such financings fail to materialize. Please see the section of this prospectus/offer to exchange titled "The Exchange Offer — Source and Amount of Funds."

What percentage of ACI Shares will former S1 stockholders own after the Exchange Offer?

Based on ACI's and S1's respective capitalizations as of October 12, 2011 and the exchange ratio of 0.3148, ACI estimates that if all S1 Shares are exchanged pursuant to the Exchange Offer and/or the Second-Step Merger, former S1 stockholders would own, in the aggregate, approximately 14.4% of the aggregate ACI Shares on a fully diluted basis. For a detailed discussion of the assumptions on which this estimate is based, please see the section of this prospectus/offer to exchange titled "The Exchange Offer — Ownership of ACI After the Exchange Offer."

What will happen to my employee stock options, stock appreciation rights, restricted stock units and/or restricted units in the Exchange Offer?

The Transaction Agreement provides that each stock option of S1 (each, an "S1 Stock Option") issued under the S1 Corporation 1997 Stock Option Plan (the "1997 Stock Option Plan") and the Security First Technologies Corporation 1998 Directors' Stock Option Plan (the "1998 Directors' Stock Option Plan") that is

outstanding will, if elected by the holder, be exercised effective as of immediately prior to the Effective Time, with the effect that the S1 Shares issuable upon exercise will be deemed for all purposes to be issued and outstanding immediately prior to the Effective Time and will have the right to receive the Proration Amount of Cash and Stock Consideration. The holders of the S1 Stock Options under the 1997 Stock Option Plan and the 1998 Directors' Stock Option Plan will be notified that such S1 Stock Options may be exercised at any time during the period beginning on October 3, 2011 and ending on the day before the Effective Time, provided that (1) any such exercise, to the extent that it relates to an S1 Stock Option that would become exercisable only at the Effective Time, will be contingent until, and will become effective only upon, the occurrence of the Effective Time and (2) no S1 Stock Option may be exercised after the relevant exercise period.

Each outstanding S1 Stock Option under the 1997 Stock Option Plan and the 1998 Directors' Stock Option Plan that is not exercised before the day prior to the Effective Time, and any other S1 Stock Option that is outstanding as of immediately before the Effective Time, will be terminated and canceled at the Effective Time, and the holder of each S1 Stock Option under the S1 Corporation 2003 Stock Incentive Plan (the "2003 Plan") and each S1 Stock Option that will have vested as of or prior to the Effective Time pursuant to the terms of the 2003 Plan, the 1997 Stock Option Plan and the 1998 Directors' Stock Option Plan (each, an "S1 Stock Plan") as applicable and/or related award agreement will, subject to any required tax withholding, be entitled to receive an amount in cash equal to the product of (1) the excess, if any, of (a) the sum of (i) \$6.62 plus (ii) an amount equal to the product (rounded to the nearest cent) of (x) 0.1064 times (y) the volume weighted average sales price per share of ACI Common Stock for the ten consecutive days that ACI Shares have traded ending on and including the second clear trading day immediately prior to the Effective Time as reported on NASDAQ (the "Blended Value") over (b) the exercise price per S1 Share subject to such S1 Stock Option and (2) the total number of S1 Shares subject to such S1 Stock Option as in effect immediately prior to the Effective Time (the "Option Consideration"); provided, however, that if the Option Consideration is zero or a negative number as of the Effective Time, such S1 Stock Option will be canceled and no amount will be paid in respect thereof. ACI will pay or cause to be paid the Option Consideration to the holders of the S1 Stock Options in a lump sum as soon as practicable after the Effective Time but in no event later than five business days following the Effective Time.

At the Effective Time, each stock appreciation right granted under the applicable S1 Stock Plan (the "SARs") will be canceled at the Effective Time, and the holder of each SAR that has vested as of or prior to the Effective Time pursuant to the applicable S1 Stock Plan will, subject to any required tax withholding, be entitled to receive an amount in cash equal to the product of (1) the excess, if any, of (a) the Blended Value over (b) the exercise price per share of the S1 Shares, if any, subject to such SARs and (2) the total number of S1 Shares, if any, subject to such SARs as in effect immediately prior to the Effective Time (the "SARs Consideration"). ACI will pay or cause to be paid the applicable SARs Consideration to the holders of the SARs in a lump sum as soon as practicable after the Effective Time but in no event later than five business days following the Effective Time.

At the Effective Time, each outstanding restricted S1 Share, restricted stock unit and restricted cash unit (other than certain restricted S1 Shares to be converted into restricted ACI Shares pursuant to the Transaction Agreement) that has vested as of or prior to the Effective Time pursuant to the applicable S1 Stock Plan will be treated as an outstanding S1 Share and will have the right to receive the Proration Amount of Cash and Stock Consideration.

For further information on the treatment of S1 Stock Options, SARs and restricted S1 Shares, please see the section of this prospectus/offer to exchange titled "The Transaction Agreement — Treatment of Stock Options; SARs; Restricted Stock."

When does the Exchange Offer expire:

The Exchange Offer is scheduled to expire at 5:00 p.m., Eastern time, on Monday, October 31, 2011, which is the Expiration Time, unless further extended by Offeror. When we make reference to the "Expiration Time" anywhere in this prospectus/offer to exchange, this is the time to which we are referring, including when applicable, any extension period that may apply.

Can the Exchange Offer be extended and, if so, under what circumstances?

The Transaction Agreement provides that Offeror will extend the Exchange Offer if any of the conditions specified in "The Exchange Offer — Conditions of the Exchange Offer" is not satisfied or waived prior to the scheduled Expiration Time. The Expiration Time may be subject to multiple extensions. For more

information, please see the section of this prospectus/offer to exchange titled "The Exchange Offer — Extension, Termination, Waiver and Amendment."

Any decision by Offeror to extend the Exchange Offer will be made public by a public announcement regarding such extension prior to 9:00 a.m., Eastern time, on the first business day after the previously scheduled Expiration Time.

Offeror may also elect to provide a "subsequent offering period" for the Exchange Offer. A subsequent offering period would not be an extension of the Exchange Offer. Rather, a subsequent offering period would be an additional period of time, beginning after Offeror has accepted for exchange all S1 Shares tendered during the Exchange Offer, during which S1 stockholders who did not tender their S1 Shares in the Exchange Offer may tender their S1 Shares and receive the Proration Amount of Cash and Stock Consideration. Offeror does not currently intend to include a subsequent offering period, although it reserves the right to do so.

How do I tender my S1 Shares?

To tender your S1 Shares represented by physical certificates into the Exchange Offer, you must deliver the certificates representing your S1 Shares, together with a completed letter of election and transmittal and any other documents required by the letter of election and transmittal, to Wells Fargo Bank, the exchange agent for the Exchange Offer, not later than the Expiration Time. The letter of election and transmittal is enclosed with this prospectus/offer to exchange.

If your S1 Shares are held in "street name" (i.e., through a broker, dealer, commercial bank, trust company or other nominee), your S1 Shares can be tendered by your nominee by book-entry transfer through The Depository Trust Company.

If you are unable to deliver any required document or instrument to the exchange agent by the Expiration Time, you may have a limited amount of additional time by having a broker, a bank or other fiduciary that is an eligible guarantor institution guarantee that the missing items will be received by the exchange agent by using the enclosed notice of guaranteed delivery circulated with this prospectus/offer to exchange (the "Notice of Guaranteed Delivery"). For the tender to be valid, however, the exchange agent must receive the missing items within three NASDAQ trading days after the date of execution of such Notice of Guaranteed Delivery. In all cases, an exchange of tendered S1 Shares will be made only after timely receipt by the exchange agent of certificates for such S1 Shares (or of a confirmation of a book-entry transfer of such shares) and a properly completed and duly executed letter of election and transmittal and any other required documents.

For a complete discussion on the procedures for tendering your S1 Shares, please see the section of this prospectus/offer to exchange titled "The Exchange Offer — Procedure for Tendering."

Until what time may I withdraw tendered S1 Shares?

You may withdraw previously tendered S1 Shares any time prior to the Expiration Time. S1 Shares tendered during the subsequent offering period, if one is provided, may not be withdrawn. For a complete discussion on the procedures for withdrawing your S1 Shares, please see the section of this prospectus/offer to exchange titled "The Exchange Offer — Withdrawal Rights."

How do I withdraw previously tendered S1 Shares?

To withdraw previously tendered S1 Shares, you must deliver a written or facsimile notice of withdrawal with the required information to the exchange agent while you still have the right to withdraw. If you tendered S1 Shares by giving instructions to a broker, dealer, commercial bank, trust company or other nominee, you must instruct the broker, dealer, commercial bank, trust company or other nominee to arrange for the withdrawal of your S1 Shares. For a complete discussion on the procedures for withdrawing your S1 Shares, please see the section of this prospectus/offer to exchange titled "The Exchange Offer — Withdrawal Rights."

When and how will I receive the Exchange Offer consideration in exchange for my tendered S1 Shares?

Offeror will exchange all validly tendered and not properly withdrawn S1 Shares promptly after the Expiration Time, subject to the terms thereof and the satisfaction or waiver of the conditions to the Exchange Offer, as set forth in the section of this prospectus/offer to exchange titled "The Exchange Offer — Conditions

of the Exchange Offer." Offeror will deliver the consideration for your validly tendered and not properly withdrawn S1 Shares by depositing the consideration therefore with the exchange agent, which will act as your agent for the purpose of receiving the Exchange Offer consideration from Offeror and transmitting such consideration to you. In all cases, an exchange of tendered S1 Shares will be made only after timely receipt by the exchange agent of certificates for such S1 Shares (or of a confirmation of a book-entry transfer of such S1 Shares as set forth in the section of this prospectus/offer to exchange titled "The Exchange Offer — Procedure for Tendering") and a properly completed and duly executed letter of election and transmittal (or Agent's Message (as defined below)) and any other required documents.

Will S1 continue as a public company following the Exchange Offer?

If the Second-Step Merger occurs, S1 will become a wholly owned subsidiary of ACI and will no longer be publicly owned. Even if the Second-Step Merger does not occur, if Offeror exchanges all S1 Shares which have been tendered, there may be so few remaining stockholders and publicly held shares that S1 Shares will no longer be eligible to be traded on the NASDAQ or any other securities market, there may not be a public trading market for such shares, and S1 may cease making filling with the SEC or otherwise cease being required to comply with applicable law and SEC rules relating to publicly held companies. Please see the sections of this prospectus/offer to exchange titled "The Exchange Offer — Plans for S1" and "The Exchange Offer — Effect of the Exchange Offer on the Market for S1 Shares; NASDAQ Listing; Registration Under the Securities Exchange Act of 1934; Margin Regulations."

Are dissenters' or appraisal rights available in either the Exchange Offer and/or the Second-Step Merger?

No dissenters' or appraisal rights are available in connection with the Exchange Offer. However, upon consummation of the Second-Step Merger, S1 stockholders who have not tendered their S1 Shares in the Exchange Offer and who, if a stockholder vote is required, did not vote in favor of or consent to the approval of the Second-Step Merger will have rights under Delaware law to dissent from the Second-Step Merger and demand appraisal of their S1 Shares. Stockholders at the time of a "short form" merger under Delaware law would also be entitled to exercise dissenters' rights pursuant to such a "short form" merger. Stockholders who perfect dissenters' rights by complying with the procedures set forth in Section 262 of the DGCL will be entitled to receive a cash payment equal to the "fair value" of their S1 Shares, as determined by a Delaware court. Please see the section of this prospectus/offer to exchange titled "The Exchange Offer — Appraisal/Dissenters' Rights."

What is the market value of my S1 Shares as of a recent date?

On October 12, 2011, the last trading day prior to the date of this prospectus/offer to exchange, the closing price of an S1 Share was \$9.61. S1 stockholders are encouraged to obtain a recent quotation for S1 Shares before deciding whether or not to tender such S1 Shares pursuant to the Exchange Offer, whether to exercise withdrawal rights as provided herein and, with respect to the election, whether to receive the Cash Consideration or the Stock Consideration or some combination thereof.

Where can I find more information on ACI and S1?

You can find more information about ACI and S1 from various sources described in the section of this prospectus/offer to exchange titled "Where You Can Find More Information."

Who can I contact with any additional questions about the Exchange Offer?

You can call the information agent for the Exchange Offer.

The information agent for the Exchange Offer is:



501 Madison Avenue, 20th Floor New York, New York 10022 Stockholders May Call Toll Free: (888) 750-5834 Banks and Brokers May Call Collect: (212) 750-5833

SUMMARY OF THE EXCHANGE OFFER

This summary highlights the material information in this prospectus/offer to exchange. To more fully understand the Exchange Offer to holders of S1 Shares, and for a more complete description of the terms of the Transaction Agreement, the Exchange Offer and the Second-Step Merger, you should read carefully this entire document, including the exhibits, schedules and documents incorporated by reference herein, and the other documents referred to herein. For information on how to obtain the documents that are on file with the SEC, please see the section of this prospectus/offer to exchange titled "Where You Can Find More Information."

The Companies (See page 30)

ACI

ACI is a Delaware corporation with its principal executive offices located at 120 Broadway, Suite 3350, New York, New York 10271. The telephone number of ACI is (646) 348-6700. ACI develops, markets, installs and supports a broad line of software products and services primarily focused on facilitating electronic payments. In addition to its own products, ACI distributes, or acts as a sales agent for, software developed by third parties. These products and services are used principally by financial institutions, retailers and electronic payment processors, both in domestic and international markets. Most of ACI's products are sold and supported through distribution networks covering three geographic regions — the Americas, Europe/Middle East/Africa and Asia/Pacific. As of June 30, 2011, ACI had total stockholders' equity of approximately \$280 million and total assets of approximately \$614 million. ACI Shares are listed on the NASDAQ Global Select Market under the ticker symbol "ACIW" and, as of October 12, 2011, the last practicable date prior to the date of this prospectus/offer to exchange, ACI had an equity capital market capitalization of approximately \$963.9 million. As of December 31, 2010, ACI had a total of approximately 2,134 employees, of whom 1,124 were in the Americas reportable segment.

As of the date of this prospectus/offer to exchange with the SEC, ACI was the beneficial owner of 1,107,000 S1 Shares, or 2.0% of the amount outstanding.

Offeror

Offeror, a Delaware limited liability company, is a wholly owned subsidiary of ACI. Offeror is newly formed, and was organized for the purpose of making the Exchange Offer and consummating the Second-Step Merger. Offeror has engaged in no business activities to date and it has no material assets or liabilities of any kind, other than those incident to its formation and those incurred in connection with the Exchange Offer and the Second-Step Merger.

 S_1

S1 is a leading global provider of payments and financial services software solutions. S1 offers payments solutions for ATM and retail point-of-sale driving, card management and merchant acquiring, as well as financial services solutions for consumer, small business and corporate online banking, trade finance, mobile banking, voice banking, branch and call center banking. S1 sells its solutions primarily to banks, credit unions, retailers and transaction processors and also provides software, custom software development, hosting and other services to State Farm Mutual Automobile Insurance Company, a relationship that will conclude by the end of 2011. Founded in 1996, S1 started the world's first Internet bank, Security First Network Bank. In 1998, S1 sold the banking operations and focused on software development, implementation and support services. For several years, S1's core business was primarily providing Internet banking and insurance applications. Then, through a series of strategic acquisitions and product development initiatives, S1 expanded its solution set to include applications that deliver financial services across multiple channels and provide payments and card management functionality.

S1 Shares are listed on the NASDAQ under the ticker symbol "SONE." S1's principal executive offices are located at 705 Westech Drive, Norcross, Georgia 30092 and its telephone number is (404) 923-3500.

The Exchange Offer (See page 55)

Offeror is offering, upon the terms and subject to the conditions set forth in this prospectus/offer to exchange and in the accompanying letter of election and transmittal, to exchange for each issued and outstanding share of common stock of S1, validly tendered pursuant to the Exchange Offer and not properly withdrawn one of the following:

- 0.3148 of an ACI Share (Stock Consideration); or
- \$10.00 in cash, without interest (Cash Consideration),

subject to the proration procedures described in this prospectus/offer to exchange and the related letter of election and transmittal. The blended value of the Cash-Stock Consideration as of the close of trading on October 12, 2011, assuming full proration, was \$9.68 per S1 Share.

The equity capital markets have been highly volatile and market prices for ACI Shares have fluctuated and will fluctuate, and could be higher or lower than the price of ACI Shares at or after the Expiration Time. Accordingly, S1 stockholders are urged to obtain current trading price information for ACI Shares prior to deciding whether to tender shares pursuant to the Exchange Offer, whether to exercise withdrawal rights as provided herein and, with respect to the election, whether to receive the Cash Consideration or the Stock Consideration or some combination thereof.

S1 stockholders electing either the Cash Consideration or the Stock Consideration will be subject to proration so that 66.2% of S1 Shares will be exchanged for the Cash Consideration and 33.8% of S1 Shares will be exchanged for the Stock Consideration in the Exchange Offer. S1 stockholders who do not participate in the Exchange Offer and whose shares are acquired in the Second-Step Merger will receive the Proration Amount of Cash and Stock Consideration. The elections of other S1 stockholders will affect whether a tendering S1 stockholder lecting the Cash Consideration or the Stock Consideration receives solely the type of consideration elected or if a portion of such S1 stockholder's tendered S1 Shares is exchanged for another form of consideration. S1 stockholders who otherwise would be entitled to receive a fractional ACI Share will instead receive cash in lieu of any fractional ACI Share such holder may have otherwise been entitled to receive based on ten-day volume weighted average trading prices. For a complete discussion of the proration procedure and the treatment of fractional ACI Shares, please see the sections of this prospectus/offer to exchange titled "The Exchange Offer — Cash In Lieu of Fractional ACI Shares."

The Second-Step Merger (See page 44)

The Exchange Offer is being made pursuant to the Transaction Agreement. Pursuant to the Transaction Agreement, after the Exchange Offer is completed, subject to the approval of the S1 stockholders if required by applicable law, Offeror will merge with and into S1.

The Transaction Agreement provides that at the effective time of the Second-Step Merger (the "Effective Time"), the separate corporate existence of Offeror will cease and S1 will continue as the surviving corporation in the Second-Step Merger. The directors of Offeror immediately prior to the Effective Time will be the initial directors of the surviving corporation, and the officers of S1 immediately prior to the Effective Time will be the initial officers of the surviving corporation. Please see the section of this prospectus/offer to exchange titled "The Transaction Agreement — Second-Step Merger; Effect on Capital Stock."

Recommendation of the S1 Board

(See page 42)

The S1 Board has unanimously (1) determined that the transactions contemplated by the Transaction Agreement are fair to, and in the best interests of, S1 and the S1 stockholders; (2) approved the transactions contemplated by the Transaction Agreement; and (3) determined to recommend that the S1 stockholders accept the Exchange Offer and tender their S1 Shares to Offeror pursuant to the Exchange Offer. The S1 Board unanimously recommends that S1 stockholders accept the Exchange Offer by tendering their S1 Shares into the Exchange Offer. Information about the recommendation of the S1 Board is more fully described in Amendment No. 2 to the S1's Solicitation/Recommendation Statement on Schedule 14D-9, which is being mailed to S1's stockholders together with this prospectus/offer to exchange and is incorporated herein by reference. Please see the section of this prospectus/offer to exchange titled "The Transaction Agreement — Recommendation of the S1 Board."

Reasons for the Exchange Offer and the Second-Step Merger (See page 39)

ACI believes that the combination of ACI's and S1's businesses will create significant value for both ACI's and S1's current stockholders. We believe the combination of ACI and S1 is a compelling combination with a number of strategic benefits, including the following:

Value

At \$9.68 per S1 Share, the blended value of the Cash-Stock Consideration as of October 12, 2011, assuming full proration, the Exchange Offer represents (1) a 35.8% premium to the closing sales price of S1 Shares on July 25, 2011, the last trading day prior to the public announcement of Original ACI Merger Proposal, (2) a 34.3% premium to the volume weighted average closing price of S1 Shares over the previous 90 days prior to the announcement of the Original ACI Merger Proposal, and (3) a 24.9% premium to the 52-week high of S1 Shares for the 52-Week Period.

S1 stockholders who elect the Cash-Stock Consideration contemplated by the Exchange Offer will be subject to proration. The elections of other S1 stockholders will affect whether S1 stockholders receive solely the type of consideration they elect or whether a portion of the consideration S1 stockholders elect is exchanged for another form of consideration as a result of the pro ration procedures contemplated by the Exchange Offer. Since the value of ACI Shares fluctuates, the per S1 Share Stock Consideration necessarily could have a value that is different than the per S1 Share Cash Consideration. As a consequence, in the Exchange Offer S1 stockholders could receive a combination of Cash-Stock Consideration with a value that is different from the value of such consideration on the date of this prospectus/offer to exchange, the Expiration Time and the date of the consummation of the Exchange Offer.

Solely for purposes of illustration, the following table indicates the value of the Cash Consideration, the Stock Consideration and the blended value of the Cash-Stock Consideration based on different assumed prices for ACI Shares.

	Assuming No	Proration	Assuming Full Proration				
Assumed ACI Share Price	Value of Stock Consideration	Value of Cash Consideration	Value of Stock Consideration	Value of Cash Consideration	Value of Cash-Stock Consideration		
\$37.93(1)	\$11.94	\$10.00	\$4.04	\$6.62	\$10.66		
\$35.70(2)	\$11.24	\$10.00	\$3.80	\$6.62	\$10.42		
\$30.49(3)	\$ 9.60	\$10.00	\$3.24	\$6.62	\$ 9.86		
\$27.54(4)	\$ 8.67	\$10.00	\$2.93	\$6.62	\$ 9.55		
\$28.77(5)	\$ 9.06	\$10.00	\$3.06	\$6.62	\$ 9.68		
\$22.70(6)	\$ 7.15	\$10.00	\$2.42	\$6.62	\$ 9.04		

⁽¹⁾ Represents highest sales price for ACI Shares in the 52-Week Period.

- (2) Represents closing sales price for ACI Shares on July 25, 2011, the last trading day prior to the announcement of the Original ACI Merger Proposal.
- (3) Represents closing sales price for ACI Shares on August 29, 2011, the last trading day prior to the commencement of the Original ACI Exchange Offer.
- (4) Represents closing sales price for ACI Shares on September 30, 2011, the last trading day prior to the announcement of the Transaction Agreement
- (5) Represents closing sales price for ACI Shares on October 12, 2011, the last trading day prior to the date of this prospectus/offer to exchange.
- (6) Represents the lowest sales price for ACI Shares in the 52-Week Period.

The equity capital markets have been highly volatile and market prices for ACI Shares and S1 Shares have fluctuated and can be expected to continue to fluctuate. S1 stockholders are urged to obtain current trading price information prior to deciding how to vote. The premium represented by the Exchange Offer may be larger or smaller depending on market prices on any given date and will fluctuate between the date of this prospectus/offer to purchase, the Expiration Time and the date of the consummation of the Exchange Offer.

Strateaic Rationale:

The Exchange Offer provides immediate cash value to S1 stockholders, as well as the opportunity to participate in the value creation in the Exchange Offer through the receipt of ACI Shares. ACI believes that the complementary nature of ACI and S1 creates a compelling opportunity to establish a full-service global leader of financial and payments software with significant scale and financial strength, including as follows:

- Highly Complementary Product and Customer Bases: Combined, ACI and S1 would provide a rich set of capabilities and a broad portfolio of products to customers across the entire electronic payments spectrum. In particular, ACI believes that the acquisition of S1 would provide breadth and additional capabilities to what ACI does today, including: (1) expand ACI's retailer business beyond North America; (2) increase ACI's retail banking payments business down into lower and mid-tier financial institutions; and (3) add function and global reach to ACI's online business banking offering, including new capabilities around branch banking and trade. The acquisition of S1 would support ACI's position as a leading provider of the most unified payments solution to serve retail banking, wholesale banking, processors and retailers and would enable its customers to lower their operational costs and improve time-to-market.
- Enhanced Scale and Global Position: ACI's and S1's principal competitors are substantially larger companies with greater financial resources than ACI and S1 have. The combined ACI and S1 would have revenue of \$683 million and adjusted EBITDA of \$123 million for the 12 months ended June 30, 2011. This scale advantage would enable the combined ACI and S1 to more effectively serve its combined global customer base and compete against the very large companies which operate in the electronic payments software business.
- Significant Synergy Opportunities: ACI expects the combination of ACI and S1 will generate a significant amount of operational efficiencies and cost savings that will drive margin expansion for
 the acquired S1 business and earnings accretion for the combined company. ACI estimates that the annual pre-tax cost savings related to the Exchange Offer would be approximately \$30 million,
 primarily attributable to elimination of \$1's public company costs and rationalization of duplicate general and administrative functions, sales/marketing functions and costs, occupancy costs,
 product management and R&D functions. In addition, ACI expects to consolidate the combined company's hosting data centers and infrastructure. Further, ACI expects the cost savings will
 improve \$1's margins in line with ACI's margins for adjusted EBITDA. Assuming that the Exchange Offer is closed in the fourth calendar quarter of this year, ACI anticipates the cost savings
 would be fully realizable in 2012.
- Strong Financial Position: ACI would continue to have a strong financial profile driven by a solid balance sheet with substantial liquidity and a recurring revenue model that generates significant free cash flows, allowing for further future investments in the business. In addition, ACI expects the transaction to be accretive to full year earnings in 2012.

The following metrics provide relevant information with respect to ACI's recent financial performance, as of July 26, 2011, the date of the Original ACI Merger Proposal:

- · ACI has produced a stockholder return of approximately 90% over the past three years, significantly outperforming the relevant peer group;
- ACI has increased its 60-month backlog to \$1.6 billion in 2010, up \$350 million since 2006;
- ACI has driven monthly recurring revenue to 68% in 2010, up nearly 29% since 2007; and
- ACI has increased adjusted EBITDA margin to 21% in 2010, from 7% in 2007.

This prospectus/offer to exchange includes summary selected unaudited pro forma combined financial information that is intended to provide S1 stockholders with information relating to ACI's financial results assuming that ACI and S1 had already been combined.

Conditions of the Exchange Offer (See page 69)

The Exchange Offer is conditioned upon, among other things, the following:

- S1 stockholders shall have validly tendered and not properly withdrawn prior to the Expiration Time at least that number of S1 Shares (together with the S1 Shares then owned by ACI, Offeror or any of ACI's other subsidiaries), shall constitute a majority of the S1 Shares issued and outstanding on a fully diluted basis
- The registration statement of which this prospectus/offer to exchange is a part shall have been declared effective under the Securities Act, and no stop order suspending the effectiveness of the registration statement shall have been issued and no proceedings for that purpose shall have been initiated or threatened by the SEC, and ACI shall have received all necessary state securities law or "blue sky" authorizations
- · the HSR Condition shall have been satisfied.
- Any clearance, approval, permit, authorization, waiver, determination, favorable review or consent of any Governmental Authority, other than the HSR Condition, shall have been obtained and such approvals shall be in full force and effect, or any applicable waiting periods for such clearances or approvals shall have expired, except for any failures that would not reasonably be expected to have a material adverse effect on ACI or S1.
- · Any of the following fail to be true
 - (1) the Fundamental S1 Corporate Representations were true and correct as of October 3, 2011 and will be true and correct on and as of the Expiration Time with the same force and effect as if made at the Expiration Time (in either case other than those representations and warranties which address matters only as of a particular date, which representations and warranties shall have been true and correct as of such particular date), except in either case contemplated by this clause (1) for de minimis inaccuracies and (2) the other representations and warranties of S1 set forth in the Transaction Agreement were true and correct as of October 3, 2011 and will be true and correct on and as of the Expiration Time with the same force and effect as if made on the Expiration Time (in either case other than those representations and warranties which address matters only as of a particular date, which representations shall have been true and correct as of such particular date), except in either case contemplated by this clause (2) where the failure of such representations and warranties to be true and correct (disregarding all qualifications or limitations as to materiality, material adverse effect or words of similar import set forth therein) has not had and would not reasonably be expected to have a material adverse effect on S1;

- S1 has performed or complied in all material respects with all agreements and covenants required by the Transaction Agreement to be performed or complied with by it on or prior to the Expiration Time; and
- since the October 3, 2011, there shall not have occurred any material adverse change in the business, financial condition or continuing results of S1 and its subsidiaries, taken as a whole (excluding certain events specified in the Transaction Agreement).

Ownership of ACI After the Exchange Offer (See page 63)

Based on ACI's and S1's respective capitalizations as of October 12, 2011 and the exchange ratio of 0.3148, ACI estimates that if all S1 Shares are exchanged pursuant to the Exchange Offer and/or the Second-Step Merger, former S1 stockholders would own, in the aggregate, approximately 14.4% of the aggregate ACI Shares on a fully diluted basis. For a detailed discussion of the assumptions on which this estimate is based, please see the section of this prospectus/offer to exchange titled "The Exchange Offer — Ownership of ACI After the Exchange Offer."

Comparative Market Price and Dividend Information (See page 23)

ACI Shares are listed on the NASDAQ Global Select Market under the ticker symbol "ACIW." S1 Shares are listed on the NASDAQ under the ticker symbol "SONE."

Based on the \$28.77 closing trading price per ACI Share on October 12, 2011, the last trading day prior to the date of this prospectus/offer to exchange, the relative value of the Cash-Stock Consideration reflected by this Exchange Offer consisted of \$6.62 in cash and \$3.06 in ACI Shares per \$1 Share as of such date, or an aggregate blended value of \$9.68 per \$1 Share as of such date, assuming full proration.

The equity capital markets have been highly volatile and market prices for ACI Shares have fluctuated and will fluctuate prior to the Expiration Time, and could be higher or lower than the ACI Share price at or after the Expiration Time. Accordingly, S1 stockholders are urged to obtain current trading price information for ACI Shares prior to deciding whether to tender shares pursuant to the Exchange Offer, whether to exercise withdrawal rights as provided herein and, with respect to the election, whether to receive the Cash Consideration or the Stock Consideration or some combination thereof.

Solely for purposes of illustration, the following table indicates the value of the Cash Consideration, the Stock Consideration and the blended value of the Cash-Stock Consideration based on different assumed prices for ACI Shares.

				Assuming Full Proration	
Assuming No Proration					Value of
Assumed ACI	Value of	Value of	Value of	Value of	Cash-Stock
Share Price	Stock Consideration	Cash Consideration	Stock Consideration	Cash Consideration	Consideration
\$37.93(1)	\$11.94	\$10.00	\$4.04	\$6.62	\$10.66
\$35.70(2)	\$11.24	\$10.00	\$3.80	\$6.62	\$10.42
\$30.49(3)	\$ 9.60	\$10.00	\$3.24	\$6.62	\$ 9.86
\$27.54(4)	\$ 8.67	\$10.00	\$2.93	\$6.62	\$ 9.55
\$28.77(5)	\$ 9.06	\$10.00	\$3.06	\$6.62	\$ 9.68
\$22.70(6)	\$ 7.15	\$10.00	\$2.42	\$6.62	\$ 9.04

- (1) Represents highest sales price for ACI Shares in the 52-Week Period.
- (2) Represents closing sales price for ACI Shares on July 25, 2011, the last trading day prior to the announcement of the Original ACI Merger Proposal.

- (3) Represents closing sales price for ACI Shares on August 29, 2011, the last trading day prior to the commencement of the Original ACI Exchange Offer.
- (4) Represents closing sales price for ACI Shares on September 30, 2011, the last trading day prior to the announcement of the Transaction Agreement
- (5) Represents closing sales price for ACI Shares on October 12, 2011, the last trading day prior to the date of this prospectus/offer to exchange.
- (6) Represents the lowest sales price for ACI Shares in the 52-Week Period.

The prices of ACI Shares used in the above table, and the assumptions regarding the mix of cash and/or stock a hypothetical S1 stockholder would receive, are for purposes of illustration only. The value of the Stock Consideration will change as the price of ACI Shares fluctuates during the Exchange Offer period and thereafter, and may therefore be higher or lower than the prices set forth in the examples above at the expiration of the Exchange Offer and at the time you receive the ACI Shares. S1'S stockholders are encouraged to obtain current market quotations for the ACI Shares and the S1 Shares prior to making any decision with respect to the Exchange Offer. Please see the section of this prospectus/offer to exchange titled "Risk Factors."

Interest of Executive Officers and Directors of ACI in the Exchange Offer (See page 79)

Except as set forth in this prospectus/offer to exchange, neither we nor, after due inquiry and to the best of our knowledge and belief, any of our directors, executive officers or other affiliates has any contract, arrangement, understanding or relationship with any other person with respect to any securities of S1, including, but not limited to, any contract, arrangement, understanding or relationship concerning the transfer or the voting of any securities, joint ventures, loan or option arrangements, puts or calls, guaranties of loans, guaranties against loss or the giving or withholding of proxies.

ACI does not believe that the Exchange Offer and the Second-Step Merger will result in a change in control under any of ACI's stock option plans or any employment agreement between ACI and any of its employees. As a result, no options or other equity grants held by such persons will vest as a result of the Exchange Offer and the Second-Step Merger. Please see the section of this prospectus/offer to exchange titled "The Exchange Offer — Certain Relationships With S1 and Interests of ACI in the Exchange Offer."

Interest of Executive Officers and Directors of S1 in the Exchange Offer (See page 79)

In considering the recommendation of the S1 Board regarding the Exchange Offer and the Second-Step Merger, S1 stockholders should be aware that certain directors and officers of S1 may be deemed to have interests in the Exchange Offer and the Second-Step Merger that are different from or in addition to the interests of other S1 stockholders, S1 has informed us that the S1 Board was aware of these interests and considered them, among other matters, in approving the Transaction Agreement, the Exchange Offer and the Second-Step Merger and recommending that S1 stockholders accept the Exchange Offer by tendering their S1 Shares into the Exchange Offer and, if required by applicable law, approve the Second-Step Merger.

As a result of these interests, S1 directors and officers may have reasons for tendering their S1 Shares and, if necessary, voting to approve the Second-Step Merger that are not the same as your interests. S1 stockholders should consider whether these interests may have influenced these directors and officers to support or recommend the Exchange Offer and the Second-Step Merger.

Information on the interests of executive officers and directors of S1 in the Exchange Offer and the Second-Step Merger is more fully described in Amendment No 2 to S1's Solicitation/Recommendation Statement on Schedule 14D-9, which is being mailed to S1 stockholders together with this prospectus/offer to exchange and is incorporated herein by reference. Please see the section of this prospectus/offer to exchange titled "The Exchange Offer — Interest of Executive Officers and Directors of S1 in the Exchange Offer."

Source and Amount of Funds; Financing (See page 70)

The Exchange Offer consideration will consist of ACI Shares and cash (including, cash paid in lieu of any fractional ACI Shares to which any S1 stockholder may be entitled). The Exchange Offer and the Second-Step Merger are not conditioned upon any financing arrangements or contingencies.

ACI has received a commitment letter from Wells Fargo, to arrange, and Wells Fargo Bank to provide, subject to certain conditions, up to \$450 million for the purpose of financing a portion of the cash component of the consideration to be paid for each S1 Share, as well as for other payments made in connection with the Exchange Offer and to refinance ACI's existing revolving facility. No other plans or arrangements have been made to finance or repay such financing after the consummation of the Exchange Offer and the Second-Step Merger. No alternative financing arrangements or alternative financing plans have been made in the event such financings fail to materialize. Please see the section of this prospectus/offer to exchange titled "The Exchange Offer — Source and Amount of Funds."

Non-Solicitation by S1 of S1 Acquisition Proposals (See page 48)

The Transaction Agreement provides that, subject to limited exceptions, S1 will not, and will cause its subsidiaries not to and will use reasonable best efforts to cause its representatives not to, directly or indirectly initiate, solicit, knowingly encourage (including by way of furnishing non-public information), or take any other action designed to lead to, any inquiries or the making of any proposal that constitutes, or would reasonably be expected to lead to, the submission of any S1 Acquisition Proposal (as defined below at "The Transaction Agreement — Non-Solicitation by S1 of S1 Acquisition Proposals; Board Recommendation") or engage, enter into, continue or participate in any negotiations or discussions with respect thereto or furnish any non-public information concerning S1 and its subsidiaries to any person in connection with any S1 Acquisition Proposal.

However, prior to the earlier of the Acceptance Time or the receipt of the approval of the merger of S1 and Offeror by the S1 stockholders, if S1 receives a written S1 Acquisition Proposal that the S1 Board believes in good faith is bona fide, and the S1 Board, after consultation with its financial advisors and outside legal counsel, determines in good faith that such S1 Acquisition Proposal constitutes or would reasonably be expected to lead to or result in an S1 Superior Offer (as defined below at "The Transaction Agreement — Non-Solicitation by S1; Board Recommendation"), then the S1 Board may, subject to certain conditions, furnish in information with irrespect to S1 and participate in discussions with respect to such S1 Acquisition Proposal. If S1 receives a written S1 Acquisition Proposal that the S1 Board believes in good faith is bona fide, and the S1 Board, after consultation with its financial advisors and outside legal counsel, determines in good faith that such S1 Acquisition Proposal constitutes an S1 Superior Offer, then the S1 Board may at any time prior to the earlier of the Acceptance Time or the receipt of the approval of the Second-Step Merger by the S1 stockholders, if it determines in good faith, after consultation with outside counsel, that the failure to take such action would be inconsistent with its fiduciary duties, change its recommendation and/or, subject to certain conditions, cause S1 to terminate the Transaction Agreement and concurrently with such termination, upon payment of a termination fee in the amount of \$19.14 million, enter into a definitive agreement with respect to such S1 Acquisition Proposal. The S1 Board may not change its recommendation and terminate the Transaction Agreement unless S1 has provided prior written notice to ACI of the reasons for such action at least five business days in advance of its taking such action, and during such notice period, S1 must negotiate with ACI in good faith and take into account all changes to the terms of the Transaction Agreement

Termination of the Transaction Agreement

(See page 52)

The Transaction Agreement may be terminated and the Exchange Offer and the Second-Step Merger may be abandoned:

- · by mutual written consent of ACI and S1;
- by either ACI or S1 at any time prior to the Effective Time if the Acceptance Time shall not have occurred on or prior to the close of business on July 31, 2012;
- by either ACI or S1 at any time prior to the Effective Time, if a Governmental Authority shall have enacted, issued, promulgated, enforced or entered any law (including an injunction or other order) or taken any other action, in any case having the effect of permanently restraining, enjoining or otherwise prohibiting the Exchange Offer or the Second-Step Merger, which law (including any such injunction or other order) or other action shall have become final and nonappealable;
- by either ACI or S1 at any time prior to the Effective Time if the Exchange Offer shall have expired or been terminated without any S1 Shares being purchased therein as a result of the failure to satisfy the Minimum Tender Condition:
- by ACI at any time prior to the Acceptance Time, if: (1) an S1 Change of Recommendation shall have occurred; (2) S1 shall have delivered a notice to ACI of its intent to effect an S1 Change of Recommendation; or (3) following the request in writing by ACI, the S1 Board shall have failed to reaffirm publicly the S1 recommendation within five business days after ACI requests in writing that such recommendation be reaffirmed publicly;
- by ACI at any time prior to the Acceptance Time if there shall have been a breach by S1 of any of its representations, warranties, covenants or obligations contained in the Transaction Agreement, which breach would result in the failure to satisfy by July 31, 2012 one or more of the conditions to the Exchange Offer, and in any such case such breach shall be incapable of being cured or, if capable of being cured, shall not have been cured within 30 days after written notice thereof will have been received by S1 of such breach;
- by S1 at any time prior to the Acceptance Time if (1) Offeror fails to amend the Exchange Offer to give effect to the terms of the Transaction Agreement or (2) there shall have been a breach by ACI or Offeror of any of its representations, warranties, covenants or obligations contained in the Transaction Agreement, which breach would result in the failure to satisfy by July 31, 2012 one or more of the conditions to the Exchange Offer, and in any such case such breach shall be incapable of being cured or, if capable of being cured, shall not have been cured within 30 days after written notice thereof will have been received by ACI or Offeror of such breach; or
- by S1 if S1 effects an S1 Change of Recommendation to accept an S1 Acquisition Proposal.

Alternatives to the Exchange Offer

(See page 44)

ACI may only complete the Second-Step Merger if it purchases the S1 Shares pursuant to the Exchange Offer. However, the Transaction Agreement gives ACI the right to require that S1 convene a stockholders' meeting to approve a merger in which the S1 stockholders would have the right to receive the Proration Amount of Cash and Stock Consideration as a result of a merger of Offeror and S1 instead of the Exchange Offer. The terms and conditions of such a transaction would be substantially the same as the terms and conditions of the Exchange Offer. ACI had not determined whether to exercise this right as of the date of this prospectus/offer to exchange. Please see the section of this prospectus/offer to exchange titled "The Transaction Agreement — Merger Without Meeting of Stockholders; Special Meeting

Termination Fees

(See page 53)

The Transaction Agreement provides that upon the termination of the Transaction Agreement under specified circumstances, S1 will owe ACI a cash termination fee of \$19.14 million. Please see the section of this prospectus/offer to exchange titled "The Transaction Agreement — Termination Fees."

Appraisal/Dissenters' Rights

(See page 67)

No dissenters' or appraisal rights are available in connection with the Exchange Offer. However, upon consummation of the Second-Step Merger, S1 stockholders who have not tendered their S1 Shares in the Exchange Offer and who, if a stockholder vote is required, do not vote for, or otherwise consent to, the approval of the Second-Step Merger will have rights under Delaware law to dissent from the Second-Step Merger and demand appraisal of their S1 Shares. Stockholders at the time of a "short form" merger under Delaware law would also be entitled to exercise dissenters' rights pursuant to such a "short form" merger. Stockholders who perfect dissenters' rights by complying with the procedures set forth in Section 262 of the DGCL will be entitled to receive a cash payment equal to the "fair value" of their S1 Shares, as determined by a Delaware court. Please see the section of this prospectus/offer to exchange titled "The Transaction Agreement — Appraisal Rights/Dissenting Shares."

Material Federal Income Tax Consequences (See page 63)

Based on closing trading prices of ACI Shares as of October 12, 2011, the Exchange Offer would be taxable to you because the integrated transaction would not qualify as a reorganization. If the integrated transaction does not qualify as a reorganization, you may be taxed on your exchange of \$1 Shares for the Stock Consideration in the Exchange Offer or the Second-Step Merger depending on the surrounding facts. In general in this case, you will recognize a capital gain or a capital loss to the extent of the difference between your adjusted tax basis in your shares and the sum of the Cash Considerations and the fair market value of the Stock Consideration you receive.

If the Exchange Offer and the Second-Step Merger qualified as component parts of an integrated transaction that constitutes a reorganization within the meaning of Section 368(a) of the Internal Revenue Code of 1986, as amended (the "Internal Revenue Code"), your exchange of S1 Shares for the Stock Consideration should be tax free, except to the extent that you also receive cash, as discussed below. Whether or not such transactions will so qualify is dependent on whether certain factual requirements are met, including that the Exchange Offer and Second-Step Merger are "interdependent" (that is, ACI would not undertake the Exchange Offer without the intention and expectation of completing the Second-Step Merger). In addition, there must be a "continuity of interest" of holders of S1 Shares in the combined company. ACI believes that this test should be satisfied if the total value of the Stock Consideration represents at least 40% of the total value of the consideration received by holders of S1 Shares, and may be satisfied at a slightly lower percentage. If market prices for ACI Shares upon consummation of the Exchange Offer are less than \$41.48, the Stock Consideration would represent less than 40% of the total value of the Exchange Offer consideration. You are urged to obtain current trading price information prior to making any decision with respect to the Exchange Offer. We cannot provide any assurance as to whether these conditions will be satisfied at this time, since it may be affected, among other things, by the total value of the Stock Consideration at the time of the consummation of the Exchange Offer and the Second-Step Merger.

If the integrated transaction constitutes a reorganization within the meaning of Section 368(a) of the Internal Revenue Code, any gain (but not loss) you realize on the transaction will be treated as a taxable capital gain or dividend in an amount equal to the lesser of (1) the excess of the sum of the Cash Consideration and the fair market value of the Stock Consideration you receive in the transaction over your basis in your shares and (2) the amount of cash you receive in the transaction, including any cash you receive in lieu of a fractional ACI Share,

depending on your circumstances. For more information, please see the section of this prospectus/offer to exchange titled "The Exchange Offer — Material Federal Income Tax Consequences."

THIS PROSPECTUS/OFFER TO EXCHANGE CONTAINS A GENERAL DESCRIPTION OF CERTAIN MATERIAL FEDERAL INCOME TAX CONSEQUENCES OF THE OFFER AND THE SECOND-STEP MERGER. THIS DESCRIPTION DOES NOT ADDRESS ANY NON-U.S. TAX CONSEQUENCES, NOR DOES IT PERTAIN TO STATE OR OTHER TAX CONSEQUENCES. CONSEQUENTLY, ACI URGES YOU TO CONTACT YOUR OWN TAX ADVISOR TO DETERMINE THE PARTICULAR TAX CONSEQUENCES TO YOU OF THE OFFER.

Accounting Treatment (See page 80)

ACI will account for the acquisition of S1 Shares under the acquisition method of accounting for business transactions. ACI will be considered the acquirer of S1 for accounting purposes. In determining the acquirer for accounting purposes, ACI considered the factors required under the accounting principles generally accepted in the U.S., which is referred to as "U.S. GAAP."

Regulatory Approval and Status (See page 77)

U.S. Antitrust Clearance

The Exchange Offer is subject to review by the FTC and the Antitrust Division. Under the HSR Act, the Exchange Offer may not be completed until certain information has been provided to the FTC and the Antitrust Division and a required waiting period has expired or has been terminated.

ACI filed the required Notification and Report Form under the HSR Act with the Antitrust Division and the FTC on July 27, 2011. Thereafter, the Antitrust Division informed ACI that, as between the FTC and the Antitrust Division, the Antitrust Division would review ACI's filing, ACI withdrew its initial filing on August 26, 2011, and refiled it on August 29, 2011 in order to permit the Antitrust Division to have additional time to review the filing. On September 27, 2011, ACI withdrew its initial HSR filing and refiled it on September 28, 2011 in order to permit the Antitrust Division to have additional time to review the filing. The 30-calendar day waiting period recommenced in connection with such refiling so that it now expires, unless terminated earlier or extended, at 11:59 p.m., Eastern time, on October 28, 2011. The Antitrust Division may extend its review beyond the 30-calendar day waiting period by requesting additional information and documentary material. In the event of such a request, the waiting period would be extended until 11:59 p.m., Eastern time, on the 30th calendar day after ACI has made a proper response to that request as specified by the HSR Act and the implementing rules.

We believe that the combination with S1 would provide ACI with enhanced scale, breadth and additional capabilities to compete more effectively in the highly competitive payment systems marketplace. If ACI were to acquire S1, we believe that the combined company would continue to face intense competition from third-party software vendors, in house solutions, processors, IT service organizations and credit card associations, including from companies which are substantially larger and have substantially greater market shares than the combined company would have. Moreover, we believe that the dynamic worldwide nature of the industry means that competitive alternatives can and do regularly emerge. Thus, ACI does not believe the transaction would enable it to obtain market power in, or even a significant share of, any relevant market. However, ACI has twice withdrawn and refiled its HSR Act filing prior to the date of this prospectus/offer to exchange in an effort to convince the DOJ staff of ACI's view as to the competitive nature of payment systems marketplace, and there can be no assurance that the DOJ will concur with our belief. If ACI again withdraws and refiles its HSR Act filing, the DOJ issues a request for additional information or documentary material or the DOJ institutes an action challenging the transaction, the Expiration Time would be extended and the completion of the Exchange Offer could be prevented.

In the Transaction Agreement, each of ACI and S1 has agreed to, among other things, use its reasonable best efforts to obtain promptly, and no later than July 31, 2012, obtain any clearance required under the antitrust laws and to avoid any impediment under the antitrust laws. If necessary to avoid the entry of an injunction sought or issued by the Antitrust Division (a "DOJ Impediment"), ACI's covenants under the Transaction Agreement include ACI being required under the Transaction Agreement to offer to the DOJ that it or its subsidiaries take, and, if such offer is accepted by the DOJ, use its best efforts to eliminate any DOJ Impediment. For this purpose, ACI's "best efforts to eliminate any DOJ Impediment" as set forth in the immediately preceding sentence will require that ACI use its best efforts to eliminate any DOJ Impediment. (1) the sale, holding separate, licensing, modifying or otherwise disposing of all or any portion of the business, assets or properties of S1 or its subsidiaries, whether located in or outside the United States; (2) ACI conducting or limiting the conduct of the business, assets or properties of S1 or its subsidiaries, whether located in or outside the United States, in a specified manner; or (3) S1 or its subsidiaries' entry with the DOJ into any agreement, settlement, order, other relief or action of a type referred to in clause (2).

The Exchange Offer and the Second-Step Merger will also be subject to review by antitrust and other authorities in jurisdictions outside the U.S. ACI is in the process of filing as soon as practicable all applications and notifications determined by ACI to be necessary or advisable under the laws of the respective jurisdictions for the consummation of the Exchange Offer and the Second-Step Merger.

For more information, please see the section of this prospectus/offer to exchange titled "The Exchange Offer — Certain Legal Matters; Regulatory Approvals."

Listing of ACI Shares to be Issued Pursuant to the Exchange Offer and the Second-Step Merger (See page 51)

ACI will submit the necessary applications to cause the ACI Shares to be issued as the Stock Consideration of the Exchange Offer and the Second-Step Merger to be authorized for listing on the NASDAQ Global Select Market.

Comparison of Stockholders' Rights (See page 83)

You may receive ACI Shares as a portion of the Exchange Offer consideration, subject to your election and proration. Because there are a number of differences between the rights of a stockholder of S1 and the rights of a stockholder of ACI, ACI urges you to review the discussion in the section of this prospectus/offer to exchange titled "Comparison of Stockholders' Rights."

Expiration Time of the Exchange Offer

(See page 55)

The Exchange Offer is scheduled to expire at 5:00 p.m., Eastern time, on Monday, October 31, 2011, which is the Expiration Time, unless further extended by Offeror. For more information, you should read the discussion in the section of this prospectus/offer to exchange titled "The Exchange Offer — Extension, Termination, Waiver and Amendment.

Extension, Termination, Waiver and Amendment (See page 56)

Transaction Agreement provides that Offeror will extend the Expiration Time from time to time, including as follows:

- for one or more periods of not more than 20 business days each if at any otherwise scheduled expiration date any of the Exchange Offer conditions is not satisfied or waived by Offeror; or
- for any period required by any rule, regulation, interpretation or position of the SEC or the staff thereof or the NASDAQ applicable to the Exchange Offer,

in each case, by making public announcement thereof.

The Expiration Time may be subject to multiple extensions and any decision to extend the Exchange Offer will be made prior to the Expiration Time. Additionally, Offeror may elect to provide a subsequent offering period of at least three business days following the Expiration Time.

Subject to the applicable rules of the SEC and the terms and conditions of the Exchange Offer, Offeror expressly reserves the right (but shall not be obligated) at any time or from time to time in its sole discretion to waive any Exchange Offer condition or modify or amend the terms of the Exchange Offer, except that, without the prior written consent of S1,

- · the Minimum Tender Condition may not be amended or waived; and
- · no change may be made to the Exchange Offer that:
 - · decreases the offer price or changes the form of consideration;
 - decreases the number of S1 Shares to be purchased by Offeror in the Exchange Offer;
 - modifies the Exchange Offer or the Exchange Offer conditions in a manner that adversely affects or reasonably could adversely affect the S1 stockholders;
 - · adds to the Exchange Offer conditions; or
- extends the Expiration Time of the Exchange Offer except as required or permitted by the Transaction Agreement.

For more information, please see the section of this prospectus/offer to exchange titled "The Exchange Offer — Extension, Termination, Waiver and Amendment."

Procedure for Tendering Shares

(See page 59)

The procedure for tendering S1 Shares varies depending on whether you possess physical certificates, a nominee holds your certificates for you, or whether you or a nominee hold your S1 Shares in book-entry form. ACI urges you to read the section of this prospectus/offer to exchange titled "The Exchange Offer — Procedure for Tendering" as well as the transmittal materials, including the letter of election and transmittal.

Withdrawal Rights (See page 62)

You can withdraw tendered S1 Shares at any time until the Exchange Offer has expired. If Offeror decides to provide a subsequent offering period, it will accept S1 Shares validly tendered during that period immediately and you will not be able to withdraw shares tendered in the Exchange Offer during any subsequent offering period. Please see the section of this prospectus/offer to exchange titled "The Exchange Offer — Withdrawal Rights.'

Acceptance for Exchange and Exchange of S1 Shares; Delivery of Exchange Offer Consideration

Upon the terms and subject to the conditions of the Exchange Offer (including, if the Exchange Offer is extended or amended, the terms and conditions of any such extension or amendment), Offeror will accept for exchange, and will exchange for ACI Shares and cash promptly after the Expiration Time, all S1 Shares validly tendered and not properly withdrawn. If Offeror elects to provide a subsequent offering period following the Expiration Time, S1 Shares validly tendered during such subsequent offering period will be accepted for exchange immediately upon tender and will be promptly exchanged for the Exchange Offer consideration. For more information, please see the section of this prospectus/offer to exchange under the caption titled "The Exchange Offer — Acceptance for Exchange and Exchange and Exchange of the Ex of S1 Shares; Delivery of Exchange Offer Consideration.

Cash in Lieu of Fractional ACI Shares (See page 58)

Certificates representing fractional ACI Shares will not be distributed in the Exchange Offer or the Second-Step Merger. Instead, each tendering S1 stockholder who would otherwise be entitled to a fractional ACI Share will receive cash (rounded to the nearest whole cent) in an amount (without interest) equal to the product of (1) such fraction, multiplied by (2) the volume weighted average sales price per share of ACI Shares for the ten consecutive days that ACI Shares have traded ending on and including the second clear trading day immediately prior to the Acceptance Time or Effective Time, as applicable, as reported on the NASDAQ.

Elections and Proration (See page 58)

S1 stockholders may elect to receive the Stock Consideration or the Cash Consideration in exchange for each S1 Share validly tendered and not withdrawn pursuant to the Exchange Offer, subject, in the case of elections of the Cash Consideration or the Stock Consideration, to the proration procedures described in this prospectus/offer to exchange and the related letter of election and transmittal, by indicating their elections in the applicable section of the letter of election and transmittal. If an S1 stockholder decides to change its election after tendering its S1 Shares, such S1 stockholder must first properly withdraw the tendered \$1 Shares and then retender the \$1 Shares prior to the Expiration Time, with a new letter of election and transmittal that indicates the revised election. \$1 stockholders who do not make an election will be deemed to have elected the Cash Consideration.

S1 stockholders electing either the Cash Consideration or the Stock Consideration will be subject to proration so that 66.2% of S1 Shares will be exchanged for the Cash Consideration and 33.8% of S1 Shares will be exchanged for the Stock Consideration in the Exchange Offer. S1 stockholders who do not participate in the Exchange Offer and whose shares are acquired in the Second-Step Merger will receive the Proration Amount of Cash and Stock Consideration. The elections of other S1 stockholders will affect whether a tendering S1 stockholder electing the Cash Consideration or the Stock Consideration receives solely the type of consideration elected or if a portion of such S1 stockholder's tendered S1 Shares is exchanged for another form of consideration. S1 stockholders who otherwise would be entitled to receive a fractional ACI Share will instead receive cash in lieu of any fractional ACI Share such holder may have otherwise been entitled to receive.

Risk Factors (See page 25)

The Exchange Offer and the Second-Step Merger are, and if the Exchange Offer and the Second-Step Merger are consummated, the combined company will be, subject to several risks which you should carefully consider prior to participating in the Exchange Offer

SELECTED HISTORICAL CONSOLIDATED FINANCIAL DATA OF ACI

Set forth below is certain selected historical consolidated financial data relating to ACI. The financial data has been derived from ACI's Quarterly Report on Form 10-Q for the six months ended June 30, 2011, which is incorporated by reference into this prospectus/offer to exchange (the "ACI 10-Q") and ACI's Annual Report on Form 10-K for the year ended December 31, 2010, which is incorporated by reference into this prospectus/offer to exchange (the "ACI 10-K"). You should not take historical results as necessarily indicative of the results that may be expected for the remainder of this fiscal year or any other future period. This financial data should be read in conjunction with the financial statements and the related notes and other financial information contained in the ACI 10-Q and the ACI 10-K. More comprehensive financial information, including "Management's Discussion and Analysis of Financial Condition and Results of Operations," is contained in the ACI 10-Q and ACI 10-K and all of the financial information and notes contained therein. Please see the section of the prospectus/offer to exchange titled "Where You Can Find More Information."

The following table sets forth selected historical consolidated financial data for the years ended December 31, 2010, 2009 and 2008, the three months ended December 31, 2007 and the years ended September 30, 2007 and 2006 and the six months ended June 30, 2011 and June 30, 2010:

						Three Months Ended		
	Six Months En	ded June 30,	Year	Years Ended December 31,(3)		December 31,	Years Ended September 30	
	2011	2010	2010	2009	2008	2007	2007	2006
				(In thousands, e	xcept per share data)			
Income Statement Data								
Total revenues	\$217,909	\$180,166	\$418,424	\$405,755	\$417,653	\$101,282	\$366,218	\$347,902
Net income (loss)	\$ 11,422	\$ (2,239)	\$ 27,195	\$ 19,626	\$ 10,582	\$ (2,016)	\$ (9,131)	\$ 55,365
Earnings (loss) per share:								
Basic	\$ 0.34	\$ (0.07)	\$ 0.81	\$ 0.57	\$ 0.31	\$ (0.06)	\$ (0.25)	\$ 1.48
Diluted	\$ 0.33	\$ (0.07)	\$ 0.80	\$ 0.57	\$ 0.30	\$ (0.06)	\$ (0.25)	\$ 1.45
Shares used in computing earnings (loss) per								
share:								
Basic	33,383	33,612	33,560	34,368	34,498	35,700	36,933	37,369
Diluted	34,120	33,612	33,870	34,554	34,795	35,700	36,933	38,237
		of June 30,			December 31,(3)			tember 30,
	2011	2010	2010	2009	2008	2007	2007	2006
				(Ir	thousands)			
Balance Sheet Data								
Working capital(2)	\$ 22,509	\$ 76,409	\$ 24,045	\$ 78,662	\$ 80,260	\$ 39,585	\$ 17,358	\$ 67,932
Total assets	613,647	552,516	601,529	590,043	552,842	570,458	506,741	539,365
Current portion of debt(2)	75,000	_	75,000	_	_	_	_	_
Debt (long-term portion)(1)(2)	1,745	78,126	2,790	77,408	76,014	75,911	76,546	78,093
Stockholders' equity	\$279,540	\$217,267	\$255,623	\$236,063	\$213,841	\$241,039	\$225,012	\$267,212

(1)	Debt (long-term portion) includes long-term capital lease obligations of \$1.3 million, \$2.4 million, \$1.8 million, \$1.5 million, \$1.0 million, \$0.9 million, \$1.5 million, and \$3.1 million as of June 30,	
	2011 and 2010, December 31, 2010, 2009, 2008 and 2007, and September 30, 2007 and 2006, respectively, which is included in other noncurrent liabilities in the consolidated balance sheets.	
	ACI's revolving credit facility has a maturity date of September 29, 2016. This revolving credit facility was refinanced on September 29, 2011.	
	On February 27, 2007, ACI's Board of Directors approved a change in ACI's fiscal year from a September 30 fiscal year-end to a December 31 fiscal year-end, effective as of January 1, 2008 for the year ended December 31, 2008.	
		_

SELECTED HISTORICAL CONSOLIDATED FINANCIAL DATA OF S1

Set forth below is certain selected historical consolidated financial data relating to S1. The financial data has been derived from S1's Quarterly Report on Form 10-Q for the six months ended June 30, 2011 (the "S1 10-Q"), which is incorporated by reference into this prospectus/offer to exchange, and S1's Annual Report on Form 10-K for the year ended December 31, 2010 (the "S1 10-K"), which is incorporated by reference into this prospectus/offer to exchange. You should not take historical results as necessarily indicative of the results that may be expected for any future period. This financial data should be read in conjunction with the financial statements and the related notes and other financial information contained in the S1 10-Q and the S1 10-K. More comprehensive financial information, including "Management's Discussion and Analysis of Financial Condition and Results of Operations," is contained in other documents filed by S1 with the SEC, and the following summary is qualified in its entirety by reference to such other documents and all of the financial information and notes contained in those documents. Please see the section of this prospectus/offer to exchange titled "Where You Can Find More Information."

The following table sets forth selected historical consolidated financial data for the years ended December 31, 2010, 2009, 2008, 2007 and 2006 and the six months ended June 30, 2011 and June 30, 2010:

	Six Months E	Six Months Ended June 30,		Years Ended December 31,			
_	2011	2010	2010(3)	2009	2008	2007	2006(4)
			(In th	ousands, except per sha	re data)		
Statement of Operations Data:							
Total revenue	\$121,165	\$102,933	\$209,086	\$238,927	\$228,435	\$204,925	\$192,310
(Loss) income from continuing operations	2,189	(2,830)	(6,283)	30,423	21,850	19,495	(12,239)
Income from discontinued operations	_	_	_	_	_	_	30,141
Net (loss) income	2,189	(2,830)	(6,283)	30,423	21,850	19,495	17,902
Revenue from significant customer(1)	10,636	14,698	25,168	38,402	42,084	43,425	47,898
Stock-based compensation expense	2,485	1,182	3,700	1,602	8,092	8,522	5,663
Basic (loss) income per share:							
Continuing operations	\$ 0.04	\$ (0.05)	\$ (0.12)	\$ 0.56	\$ 0.38	\$ 0.32	\$ (0.17)
Discontinued operations	_	_	_	_	_	_	\$ 0.42
Net (loss) income	\$ 0.04	\$ (0.05)	\$ (0.12)	\$ 0.56	\$ 0.38	\$ 0.32	\$ 0.25
Diluted (loss) income per share:							
Continuing operations	\$ 0.04	\$ (0.05)	\$ (0.12)	\$ 0.55	\$ 0.38	\$ 0.32	\$ (0.17)
Discontinued operations	_	_	_	_	_	_	\$ 0.42
Net (loss) income	\$ 0.04	\$ (0.05)	\$ (0.12)	\$ 0.55	\$ 0.38	\$ 0.32	\$ 0.25

	As of J	June 30,			As of December 31,		
	2011	2010	2010(3)	(In thousands)	2008	2007	2006(4)
Balance Sheet Data:							
Cash and cash equivalents	\$ 71,720	\$ 51,707	\$ 61,917	\$ 61,784	\$ 63,840	\$ 45,011	\$ 69,612
Working capital(5)(6)	59,094	50,300	48,843	82,942	55,804	64,318	83,227
Goodwill	148,236	145,325	147,544	126,605	124,362	125,281	125,300
Total assets	327,113	305,767	309,653	300,066	278,686	281,844	307,805
Debt obligations, excluding current portion	27	14	35	5,026	6,126	8,805	4,119
Total liabilities	83,430	70,151	72,040	61,425	69,946	71,939	83,576
Stockholders' equity	243.683	235,616	237.613	238.641	208.740	209.905	224.229

	Six Months Ended June 30, 2011 2010		_	2010(3)	Years Ended December			er 31,		2	2006(4)		
Other Selected Data:						(In	thousands)						
Cash provided by operating activities	\$ 16,	,938	\$ 23,311	\$	37,249	\$	16,035	\$	34,147	\$	31,332	\$	3,460
Cash (used in) provided by investing activities	(3,	,039)	(32,371)		(37,704)		(7,688)		15,765		(13,893)		31,626
Cash used in financing activities(2)	(4,	,176)	(815)		(364)		(12,172)		(27,488)		(42,490)		(50,671)
Weighted average common shares outstanding — basic	53,	,475	51,791		52,495		52,584		55,734		59,746		70,780
Weighted average common shares outstanding — diluted	54,	,277	51,791		52,495		53,291		56,449		60,596		70,780

- Revenue from State Farm.
- (2) Cash used in financing activities included the repurchase of common stock of \$9.6 million in 2009, \$25.1 million in 2008, \$51.0 million in 2007 and \$55.8 million in 2006 pursuant to authorized stock repurchase programs.
- (3) S1's 2010 selected financial data reflects, as of their respective dates of acquisition, S1's purchase of PM Systems Corporation for approximately \$29.2 million, net of cash acquired, in March 2010 and certain assets from a reseller in Latin America for approximately \$1.9 million, net of cash acquired, in August 2010.
- (4) In 2004, S1 acquired Mosaic Software Holdings Limited and S1 paid an additional acquisition cost of \$14.0 million as earn-out consideration in 2006. Discontinued operations included S1's Risk and Compliance business sold in 2006 for approximately \$32.6 million.
- (5) Working capital includes deferred revenue of \$50.0 million and \$38.0 million as of June 30, 2011 and December 31, 2010, respectively.
- (6) Working capital includes deferred revenue of \$36.8 million and \$26.8 million as of June 30, 2010 and December 31, 2009, respectively.

SUMMARY SELECTED UNAUDITED PRO FORMA COMBINED FINANCIAL INFORMATION

The following summary selected unaudited pro forma combined financial information has been prepared to illustrate the effect of the combination of ACI and S1 and has been prepared for informational purposes only. The unaudited pro forma combined balance sheet information combines information from the historical consolidated balance sheets of ACI and of S1 as of June 30, 2011, giving effect to the acquisition of S1 by ACI as if it had occurred on June 30, 2011. The unaudited pro forma combined statements of operations information combines information from the historical consolidated statements of operations of ACI and of S1 for the year ended December 31, 2010 and the six months ended June 30, 2011, giving effect to the acquisition of S1 by ACI as if it had occurred on January 1, 2010. The summary selected unaudited pro forma combined financial information has been prepared using the acquisition method of accounting under U.S. GAAP. ACI has been treated as the acquirer for accounting purposes.

The summary selected unaudited pro forma combined financial information has been presented for informational purposes only. The pro forma information is not necessarily indicative of what the combined company's financial position or results of operations actually would have been had the acquisition been completed as of the dates indicated. In addition, the summary selected unaudited pro forma combined financial information does not purport to project the future financial position or operating results of the combined company. The following information has been derived from, and should be read in conjunction with, the unaudited pro forma condensed combined financial information and related notes included in this prospectus/offer to exchange. See "Unaudited Condensed Combined Pro Forma Financial Information."

conjunction with, the unaudited pro forma condensed combined financial information and related notes included in this prospectus/offer to exchange. See "Unaudited Condensed Combined Pro Forma Financial Information".

This pro forma information is subject to risks and uncertainties, including those discussed in the section of this prospectus/offer to exchange titled "Risk Factors."

The following sets forth unaudited summarized pro forma statement of operations data for the six months ended June 30, 2011 and the year ended December 31, 2010 (in thousands of dollars):

	Six M 	Year Ended December 31, 2010		
Revenues:				
Software license fees	\$	107,768	\$	190,796
Maintenance fees		105,373		198,557
Services		75,870		139,169
Software hosting fees		50,063		98,988
Total revenues	-	339,074		627,510
Expenses:				
Cost of software license fees		8,702		14,833
Cost of maintenance, services, and hosting fees		123,857		227,505
Research and development		64,234		109,584
Selling and marketing		55,574		98,725
General and administrative		48,478		97,230
Depreciation and amortization		15,929		30,489
Total expenses		316,774		578,366
Operating income		22,300		49,144
Other income (expense):				
Interest income		547		879
Interest expense		(5,734)		(11,784)
Other, net		(970)		(4,982)
Total other income (expense)		(6,157)		(15,887)
Income before income taxes		16,143		33,257
Income tax expense		5,464		18,411
Net income	\$	10,679	\$	14,846

The following sets forth unaudited summarized pro forma balance sheet data as of June 30, 2011 (in thousands of dollars): ASSETS Cash and cash equivalents Billed receivables, net Accrued receivables Income taxes receivable Deferred income taxes, net Prepaid expenses	\$ June 30, 2011
Cash and cash equivalents Billed receivables, net Accrued receivables Income taxes receivable Deferred income taxes, net Prepaid expenses	\$ 142,527
Cash and cash equivalents Billed receivables, net Accrued receivables Income taxes receivable Deferred income taxes, net Prepaid expenses	\$
Billed receivables, net Accrued receivables Income taxes receivable Deferred income taxes, net Prepaid expenses	\$
Accrued receivables Income taxes receivable Deferred income taxes, net Prepaid expenses	
Income taxes receivable Deferred income taxes, net Prepaid expenses	116,348
Deferred income taxes, net Prepaid expenses	19,081
Prepaid expenses	1,953
	13,931
	19,143
Other current assets	 14,637
Total current assets	327,620
Property and equipment, net	43,488
Software, net	28,455
Goodwill	658,265
Other intangible assets, net	29,075
Deferred income taxes, net	28,776
Other noncurrent assets	 27,483
TOTAL ASSETS	\$ 1,143,162
LIABILITIES AND STOCKHOLDERS' EQUITY	
Current liabilities	
Accounts payable	\$ 24,678
Accrued employee compensation	39,470
Deferred revenue	181,753
Income taxes payable	2,159
Alliance agreement liability	1,600
Current portion of note payable	8,750
Accrued and other current liabilities	23,415
Total current liabilities	 281,825
Deferred revenue	30,035
Long term note payable	356,733
Alliance agreement noncurrent liability	20,667
Other noncurrent liabilities	20,818
Total liabilities	710,078
Stockholders' equity	
Preferred stock	_
Common stock	263
Common stock warrants	24,003
Treasury stock	(167,286)
Additional paid-in capital	484,948
Retained earnings	101,943
Accumulated other comprehensive loss	(10,787)
Total stockholders' equity	433,084
TOTAL LIABILITIES AND STOCKHOLDERS' EQUITY	\$ 1,143,162

HISTORICAL AND PRO FORMA PER SHARE INFORMATION

The historical per share earnings, dividends, and book value of ACI and S1 shown in the tables below are derived from their respective audited consolidated financial statements for the year ended December 31, 2010 and their respective unaudited consolidated financial statements for the six months ended June 30, 2011. The pro forma comparative basic and diluted earnings per share data give effect to the acquisition using the acquisition method of accounting as if it had been completed on January 1, 2010. The pro forma book value per share information was computed as if the acquisition had been completed on June 30, 2011. You should read this information in conjunction with the historical financial information of ACI and of \$1 included elsewhere or incorporated in this prospectus/offer to exchange, including ACI's and S1's financial statements and related notes. The per share proforma information assumes that all S1 Shares are converted into ACI Shares at the exchange ratio of 0.1064. The equivalent pro forma per share information was derived by multiplying the combined company pro forma per share information by the exchange ratio of 0.1064.

The pro forma data shown in the tables below is unaudited and for illustrative purposes only. You should not rely on this data as being indicative of the historical results that would have been achieved had ACI and S1 always been combined or the future results that the combined company will achieve after the consummation of the acquisition. This pro forma information is subject to risks and uncertainties, including those discussed in the section entitled "Risk Factors."

Historical ACI	Historical S1	Company Pro Forma	Equivalent Pro Forma
\$0.34	\$0.04	\$ 0.27	\$0.03
0.33	0.04	0.27	0.03
_	_	_	_
8.19	4.49	10.84	n/a
	Year Ended D	ecember 31, 2010	
Historical ACI	Year Ended Do Historical S1	ecember 31, 2010 Combined Company Pro Forma	Equivalent Pro Forma
	Historical	Combined Company	
ACI	Historical S1	Combined Company Pro Forma	Pro Forma
\$0.81	Historical S1 \$(0.12)	Combined Company Pro Forma \$0.38	Pro Forma \$0.04
	\$0.34 0.33	*** S1 **** \$0.04 **** 0.03 **** 0.04 **** 0.04 **** 0.04 **** 0.04 **** 0.04 **** 0.04 **** 0.04 **** 0.04 **** 0.04 **** 0.04 **** 0.04 **** 0.04 **** 0.04 **** 0.04 **** 0.04 **** 0.04 *** 0.04 **** 0.04	Historical

COMPARATIVE MARKET PRICE AND DIVIDEND INFORMATION

The following table sets forth the high and low sales prices per share of ACI Shares and S1 Shares for the periods indicated as reported on the consolidated tape of the NASDAQ Global Select Market and the NASDAQ, as reported in the ACI 10-K and the S1 10-K, respectively, with respect to the years 2009 and 2010, and thereafter as reported in publicly available sources. As reported in the ACI 10-K, ACI has never declared or paid cash dividends on its capital stock and does not anticipate paying any cash dividends in the foreseeable future. Loan covenants contained in ACI's current credit facility limit its ability to pay dividends on ACI's capital stock. As reported in the S1 10-K, S1 has never declared or paid cash dividends on its capital stock and does not anticipate paying any cash dividends in the foreseeable future, although there are no restrictions on S1's ability to do so. Please see the section of this prospectus/offer to exchange titled "Note on S1 Information."

		ACI		81
	High	Low	High	Low
Year Ended December 31, 2011				
Fourth Quarter (through October 12, 2011)	\$29.06	\$24.23	\$9.65	\$9.00
Third Quarter	\$37.93	\$25.76	\$9.47	\$6.84
Second Quarter	\$34.65	\$28.70	\$7.75	\$6.50
First Quarter	\$33.03	\$24.96	\$7.33	\$5.90
Year Ended December 31, 2010				
Fourth Quarter	\$28.15	\$22.28	\$7.24	\$5.16
Third Quarter	\$22.39	\$18.31	\$6.18	\$4.73
Second Quarter	\$21.03	\$17.79	\$6.80	\$5.45
First Quarter	\$21.59	\$15.32	\$6.84	\$5.80
Year Ended December 31, 2009				
Fourth Quarter	\$17.97	\$14.39	\$6.60	\$5.65
Third Quarter	\$15.98	\$13.20	\$7.43	\$5.87
Second Quarter	\$20.32	\$13.28	\$7.42	\$5.04
First Quarter	\$19.14	\$15.90	\$8.00	\$4.75

Based on the \$28.77 closing trading price per ACI Share on October 12, 2011, the last trading day prior to the date of this prospectus/offer to exchange, the relative value of the Cash-Stock Consideration reflected by this Exchange Offer consisted of \$6.62 in cash and \$3.06 in ACI Shares per S1 Share as of such date, or an aggregate blended value of \$9.68 per S1 Share as of such date, assuming full proration. The value of the Stock Consideration will change as the price of ACI Shares fluctuates during the Exchange Offer period and thereafter may be higher or lower than the prices set forth in the examples above at the expiration of the Exchange Offer and at the time you receive the ACI Shares. You are encouraged to obtain current market quotations for the ACI Shares and the S1 Shares prior to making any decision with respect to the Exchange Offer.

Solely for purposes of illustration, the following table indicates the value of the Cash Consideration, the Stock Consideration and the blended value of the Cash-Stock Consideration based on different assumed prices for ACI Shares.

			Assuming Full Proration		
Assumed ACI Share Price	Assuming No Value of Stock Consideration	Proration Value of Cash Consideration	Value of Stock Consideration	Value of Cash Consideration	Value of Cash-Stock Consideration
\$37.93(1)	\$11.94	\$10.00	\$4.04	\$6.62	\$10.66
\$35.70(2)	\$11.24	\$10.00	\$3.80	\$6.62	\$10.42
\$30.49(3)	\$ 9.60	\$10.00	\$3.24	\$6.62	\$ 9.86
\$27.54(4)	\$ 8.67	\$10.00	\$2.93	\$6.62	\$ 9.55
\$28.77(5)	\$ 9.06	\$10.00	\$3.06	\$6.62	\$ 9.68
\$22.70(6)	\$ 7.15	\$10.00	\$2.42	\$6.62	\$ 9.04

- (1) Represents highest sales price for ACI Shares in the 52-Week Period.
- (2) Represents closing sales price for ACI Shares on July 25, 2011, the last trading day prior to the announcement of the Original ACI Merger Proposal.
- (3) Represents closing sales price for ACI Shares on August 29, 2011, the last trading day prior to the commencement of the Original ACI Exchange Offer.
- (4) Represents closing sales price for ACI Shares on September 30, 2011, the last trading day prior to the announcement of the Transaction Agreement.
- (5) Represents closing sales price for ACI Shares on October 12, 2011, the last trading day prior to the date of this prospectus/offer to exchange.
- (6) Represents the lowest sales price for ACI Shares in the 52-Week Period.

The prices of ACI Shares used in the above table, and the assumptions regarding the mix of cash and/or stock a hypothetical S1 stockholder would receive, are for purposes of illustration only. The value of the Stock Consideration will change as the price of ACI Shares fluctuates during the Exchange Offer period and thereafter, and may therefore be higher or lower than the prices set forth in the examples above at the expiration of the Exchange Offer and at the time you receive the ACI Shares. S1's stockholders are encouraged to obtain current market quotations for the ACI Shares and the S1 Shares prior to deciding whether to tender shares pursuant to the Exchange Offer, whether to exercise withdrawal rights as provided herein and, with respect to the election, whether to receive the Cash Consideration or the Stock Consideration or some combination thereof. Please see the section of this prospectus/offer to exchange titled "Risk Factors."

Please also see the section of this prospectus/offer to exchange titled "The Exchange Offer — Effect of the Exchange Offer on the Market for S1 Shares; NASDAQ Listing; Registration Under the Securities Exchange Act of 1934; Margin Regulations" for a discussion of the possibility that S1 Shares will cease to be listed on the NASDAQ.

RISK FACTORS

In addition to the risk factors set forth below, you should read and consider other risk factors specific to each of the ACI and S1 businesses that will also affect ACI after consummation of the Exchange Offer and the Second-Step Merger, described in Part I, Item 1A of each company's annual report on Form 10-K for the year ended December 31, 2010 and other documents that have been filed with the SEC, all of which are incorporated by reference into this prospectus/offer to exchange. If any of the risks described below or in the reports incorporated by reference into this prospectus/offer to exchange actually occurs, the respective businesses, financial results, financial conditions, operating results or share prices of ACI or S1 could be materially adversely affected.

Risk Factors Relating to the Exchange Offer and the Second-Step Merger

The value of the ACI Shares that the S1 stockholders could receive in the Exchange Offer as Stock Consideration will vary as a result of the fixed exchange ratio and possible fluctuations in the price of ACI Shares.

Upon consummation of the Exchange Offer, each S1 Share validly tendered into the Exchange Offer and accepted by Offeror for exchange that is exchanged for ACI Shares will be exchanged at a fixed exchange ratio of 0.3148 of an ACI Share for each S1 Share, subject to proration. The market value of the ACI Shares issued in exchange for S1 Shares in the Exchange Offer will depend upon the market price of an ACI Share on the date the Exchange Offer is consummated. If the price of ACI Shares declines, S1 stockholders who receive the Stock Consideration could receive less value for their S1 Shares upon the consummation of the Exchange Offer than the value calculated pursuant to the exchange ratio as of the date of this prospectus/offer to exchange. Stock price changes may result from a variety of factors that are beyond the companies' control, including general market conditions, changes in business prospects, catastrophic events, both natural and man-made, and regulatory considerations. In addition, the ongoing business of ACI may be adversely affected by actions taken by ACI in connection with the Exchange Offer, including as a result of (1) the attention of management of ACI having been diverted to the Exchange Offer instead of being directed solely to ACI's own operations and pursuit of other opportunities that could have been beneficial to ACI and the combined entity and (2) payment by ACI of certain costs relating to the Exchange Offer, including certain legal, accounting and financial and capital markets advisory fees.

Because the Exchange Offer and the Second-Step Merger will not be completed until certain conditions have been satisfied or, where relevant, waived (please see the section of this prospectus/offer to exchange titled "The Exchange Offer — Conditions of the Exchange Offer"), a period of time, which may be significant, may pass between the commencement of the Exchange Offer and the time that Offeror accepts S1 Shares for exchange. Therefore, at the time when you tender your S1 Shares pursuant to the Exchange Offer, you will not know the exact market value of the ACI Shares that will be issued if Offeror accepts such S1 Shares for exchange. However, tendered S1 Shares may be withdrawn at any time prior to the Expiration Time. Please see the sections of this prospectus/offer to exchange titled "Comparative Market Price and Dividend Information" for the historical high and low closing prices of ACI Shares and S1 Shares for each quarter of the period 2009 through the date of this prospectus/offer to exchange and "The Exchange Offer — Withdrawal Rights."

Furthermore, in connection with the Exchange Offer and the Second-Step Merger, ACI will need to issue approximately 5.9 million ACI Shares. The increase in the number of ACI Shares may lead to sales of such ACI Shares or the perception that such sales may occur, either of which may adversely affect the market for, and the market price of, ACI Shares.

S1 stockholders are urged to obtain market quotations for ACI Shares and S1 Shares when they consider whether to tender their S1 Shares pursuant to the Exchange Offer. Please see the section of this prospectus/offer to exchange titled "Comparative Market Price and Dividend Information."

The Exchange Offer may adversely affect the liquidity and value of non-tendered S1 Shares.

In the event that not all S1 Shares are tendered in the Exchange Offer and we accept for exchange those S1 Shares tendered into the Exchange Offer, the number of stockholders and the number of S1 Shares held by individual holders will be greatly reduced. As a result, Offeror's acceptance of S1 Shares for exchange in the Exchange Offer could adversely affect the liquidity and could also adversely affect the market value of the remaining S1 Shares held by the public. Additionally, subject to the rules of the NASDAQ, ACI may delist the S1 Shares on the NASDAQ. As a result of such delisting, each issued and outstanding S1 Share not tendered pursuant to the Exchange Offer may become illiquid and may be of reduced value. Please see the section of this prospectus/offer to exchange titled "The Exchange Offer — Plans for S1."

The receipt of ACI Shares pursuant to the Exchange Offer and the Second-Step Merger would be taxable based on the price of ACI Shares as of October 12, 2011 and could be taxable to S1 stockholders depending on facts surrounding the Exchange Offer and the Second-Step Merger.

Based on closing trading prices of ACI Shares as of October 12, 2011, the Exchange Offer would be taxable to you because the integrated transaction would not qualify as a reorganization. If the integrated transaction does not qualify as a reorganization, you may be taxed on your exchange of \$1 Shares for the Stock Consideration in the Exchange Offer or the Second-Step Merger, depending on the surrounding facts. In general in this case, you will recognize a capital gain or a capital loss to the extent of the difference between your adjusted tax basis in your shares and the sum of the Cash Considerations and the fair market value of the Stock Consideration vou receive.

If the Exchange Offer and the Second-Step Merger qualified as component parts of an integrated transaction that constitute a reorganization within the meaning of Section 368(a) of the Internal Revenue Code of 1986, as amended (the "Internal Revenue Code"), your exchange of S1 Shares for the Stock Consideration should be tax free, except to the extent that you also receive cash, as discussed below. Whether or not they will so qualify is dependent on whether certain factual requirements are met, including that the Exchange Offer and Second-Step Merger are "interdependent" (that is, ACI would not undertake the Exchange Offer without the intention and expectation of completing the Second-Step Merger). In addition, there must be a "continuity of interest" of holders of S1 Shares in the combined company. ACI believes that this test should be satisfied if the total value of the Stock Consideration represents at least 40% of the total value of the consideration received by holders of S1 Shares, and may be satisfied at a slightly lower percentage. If market prices for ACI Shares upon consummation of the Exchange Offer are less than \$41.48, the Stock Consideration would represent less than 40% of the total value of the Exchange Offer consideration. You are urged to obtain current trading price information prior to making any decision with respect to the Exchange Offer. We cannot provide any assurance as to whether these conditions will be satisfied at this time, since it may be affected, among other things, by the total value of the Stock Consideration at the time of the consummation of the Exchange Offer and the Second-Step Merger.

If the integrated transaction constitutes a reorganization within the meaning of Section 368(a) of the Internal Revenue Code, any gain (but not loss) you realize on the transaction will be treated as a taxable capital gain or dividend in an amount equal to the lesser of (1) the excess of the sum of the Cash Consideration and the fair market value of the Stock Consideration you receive in the transaction over your basis in your shares and (2) the amount of cash you receive in the transaction, including any cash you receive in lieu of a fractional ACI Share, depending on your circumstances. For more information, please see the section of this prospectus/offer to exchange titled "The Exchange Offer — Material Federal Income Tax Consequences."

ACI must obtain governmental and regulatory approvals to consummate the Exchange Offer, which, if delayed or not granted, may delay or jeopardize the Exchange Offer, result in additional expenditures of money and resources and/or reduce the anticipated benefits of the combination contemplated by the Exchange Offer and the Second-Step Merger.

The Exchange Offer is conditioned on the expiration or termination of the applicable waiting period under the HSR Act. You should be aware that all required regulatory approvals may not be obtained in a timely manner, and this could result in a delay in the completion of the Exchange Offer. ACI has twice withdrawn and refiled its HSR Act filing prior to the date of this prospectus/offer to exchange in an effort to convince the DOJ staff of ACI's view as to the competitive nature of payment systems marketplace, and there can be no assurance that the DOJ will concur with its belief that the transaction should be permitted to close. If ACI again withdraws and refiles its HSR Act filing, the DOJ issues a request for additional information or documentary material or the DOJ institutes an action challenging the transaction, the Expiration Time would be extended and the completion of the Exchange Offer could be prevented.

The governmental and regulatory agencies from which ACI will seek these approvals have broad discretion in administering the applicable governing regulations. As a condition to their approval of the transactions contemplated by this prospectus/offer to exchange, agencies may impose requirements, limitations or costs or require divestitures or place restrictions on the conduct of the combined company's business. These requirements, limitations, costs, divestitures or restrictions could jeopardize or delay the consummation of the Exchange Offer or may reduce the anticipated benefits of the combination contemplated by the Exchange Offer. Further, no assurance can be given that the required consents and approvals will be obtained or that the required conditions to the Exchange Offer will be satisfied, and, if all required consents and approvals are obtained and the conditions to the consummation of the Exchange Offer are satisfied, no assurance can be given as to the terms, conditions and timing of the consents and approvals.

If ACI agrees to any material requirements, limitations, costs, divestitures or restrictions in order to obtain any consents or approvals required to consummate the Exchange Offer, these requirements, limitations, additional costs or restrictions could adversely affect ACI's ability to integrate the operations of ACI and S1 or reduce the anticipated benefits of the combination contemplated by the Exchange Offer. This could have a material adverse effect on the business, financial condition and results of operations of the combined company and the market value of ACI Shares after the acquisition. In addition, a third party could attempt to intervene in any governmental or regulatory filings to be made by ACI or otherwise object to the granting to ACI of any such governmental or regulatory authorizations, consents, orders or approvals, which may cause a delay in obtaining, or the imposition of material requirements, limitations, costs, divestitures or restrictions on, or the failure to obtain, any such authorizations, consents, orders or approvals. Please see the section titled "The Exchange Offer — Conditions of the Exchange Offer and the Second-Step Merger.

The Exchange Offer remains subject to other conditions that ACI cannot control.

The Exchange Offer is subject to other conditions, including the HSR Condition; the tender without withdrawal of a sufficient number of S1 Shares to satisfy the Minimum Tender Condition; no material adverse change in the business, financial condition or continuing results of S1 and its subsidiaries, taken as a whole; S1's compliance with its covenants in the Transaction Agreement in all material respects; and the continued accuracy of S1's representations and warranties in the Transaction Agreement, subject to certain exceptions. There are no assurances that all of the conditions to the Exchange Offer will be satisfied. If the conditions to the Exchange Offer are not met, then Offeror is required under the Transaction Agreement to extend the Exchange Offer.

Please see the section of this prospectus/offer to exchange titled "The Exchange Offer — Conditions of the Exchange Offer" for a discussion of the conditions to the Exchange Offer.

Risk Factors Relating to S1's Businesses

You should read and consider other risk factors specific to S1's businesses that will also affect ACI after the acquisition contemplated by this prospectus/offer to exchange, described in Part I, Item 1A of the S1 10-K and other documents that have been filed by S1 with the SEC and which are incorporated by reference into this document. See the section of this prospectus/offer to exchange titled "Where You Can Find More Information."

Risk Factors Relating to ACI's Businesses

You should read and consider other risk factors specific to ACI's businesses that will also affect ACI after the acquisition contemplated by this prospectus/offer to exchange, described in Part I, Item 1A of the ACI 10-K and other documents that have been filed by ACI with the SEC and which are incorporated by reference into this prospectus/offer to exchange. See the section of this prospectus/offer to exchange titled "Where You Can Find More Information."

Risk Factors Relating to ACI Following the Exchange Offer

ACI may experience difficulties integrating S1's businesses, which could cause ACI to fail to realize the anticipated benefits of the acquisition.

If ACI's acquisition of S1 is consummated, achieving the anticipated benefits of the acquisition will depend in part upon whether the two companies integrate their businesses in an effective and efficient manner. The companies may not be able to accomplish this integration process smoothly or successfully. The integration of certain operations following the acquisition will take time and will require the dedication of significant management resources, which may temporarily distract management's attention from the routine business of the combined entity.

Any delay or inability of management to successfully integrate the operations of the two companies could compromise the combined entity's potential to achieve the anticipated long-term strategic benefits of the acquisition and could have a material adverse effect on the business, financial condition and results of operations of the combined company and the market value of ACI Shares after the acquisition.

Future results of the combined company may differ materially from the Selected Unaudited Condensed Consolidated Pro Forma Financial Information of ACI and S1 presented in this prospectus/offer to exchange.

The future results of ACI following the consummation of the Exchange Offer may be materially different from those shown in the Unaudited Condensed Combined Pro Forma Financial Information presented in this prospectus/offer to exchange, which show only a combination of ACI's and S1's historical results after giving effect to the Exchange Offer. ACI has estimated that it will record approximately \$26.5 million in transaction expenses, as described in the notes to the Unaudited Condensed Combined Pro Forma Financial Information included in this prospectus/offer to exchange. In addition, the final amount of any charges relating to acquisition accounting adjustments that ACI may be required to record will not be known until following the consummation of Exchange Offer and Second-Step Merger. These and other expenses and charges may be higher or lower than estimated.

Our business, which depends heavily on revenue from customers in the banking and insurance industries and other financial services firms, may be materially adversely impacted by volatile U.S. and global market and economic conditions, which could adversely affect the value of ACI Shares received as part of the Exchange Offer.

For the foreseeable future, we expect to continue to derive most of our revenue from products and services we provide to the banking and insurance industries and other financial services firms and retailers. Given the concentration of our business activities in financial industries, we may be particularly exposed to economic downturns in those industries. U.S. and global market and economic conditions have been disjurited.

and volatile over the past several years. General business and economic conditions that could affect us and our customers include fluctuations in debt and equity capital markets, liquidity of the global financial markets, the availability and cost of credit, investor and consumer confidence, and the strength of the economies in which our customers operate. A poor economic environment could result in significant decreases in demand for our products and services, including the delay or cancellation of current or anticipated projects, and adversely affect our operating results. In addition to mergers and acquisitions in the banking industry, we have seen an increased level of bank closures and government supervised consolidation transactions. Our existing customers may be acquired by or merged into other financial institutions that have their own financial software solutions, be closed by regulators, or decide to terminate their relationships with us for other reasons. As a result, our sales could decline if an existing customer is merged with or acquired by another company or closed. Additionally, our investment portfolio is generally subject to credit, market, liquidity and interest rate risks and the value and liquidity of our investments may be adversely impacted by U.S. and global market and economic conditions including bank closures.

THE COMPANIES

ΔC

ACI is a Delaware corporation with its principal executive offices located at 120 Broadway, Suite 3350, New York, New York 10271. The telephone number of ACI is (646) 348-6700. ACI develops, markets, installs and supports a broad line of software products and services primarily focused on facilitating electronic payments. In addition to its own products, ACI distributes, or acts as a sales agent for, ACI's products are sold and supported through distribution networks covering three geographic regions — the Americas, Europe/Middle East/Africa and Asia/Pacific. As of June 30, 2011, ACI had total stockholders' equity of approximately \$280 million and total assets of approximately \$614 million. ACI Shares are listed on the NASDAQ Global Select Market under the ticker symbol "ACIW" and, as of October 12, 2011, the last practicable date prior to the date of this prospectus/offer to exchange, ACI had an equity capital market capitalization of approximately \$963.9 million. As of December 31, 2010, ACI had a total of approximately 2,134 employees, of whom 1,124 were in the Americas reportable segment.

As of the date of this prospectus/offer to exchange, ACI was the beneficial owner of 1,107,000 S1 Shares, or 2.0% of the amount outstanding.

Offeror

Offeror, a Delaware limited liability company, is a wholly owned subsidiary of ACI. Offeror is newly formed, and was organized for the purpose of making the Exchange Offer and consummating the Second-Step Merger. Offeror has engaged in no business activities to date and it has no material assets or liabilities of any kind, other than those incident to its formation and those incurred in connection with the Exchange Offer and the Second-Step Merger.

S

S1 is a leading global provider of payments and financial services software solutions. S1 offers payments solutions for ATM and retail point-of-sale driving, card management and merchant acquiring, as well as financial services solutions for consumer, small business and corporate online banking, trade finance, mobile banking, voice banking, branch and call center banking. S1 sells its solutions primarily to banks, credit unions, retailers and transaction processors and also provides software, custom software development, hosting and other services to State Farm Mutual Automobile Insurance Company, a relationship that will conclude by the end of 2011. Founded in 1996, S1 started the world's first Internet bank, Security First Network Bank. In 1998, S1 sold the banking operations and focused on software development, implementation and support services. For several years, S1's core business was primarily providing Internet banking and insurance applications. Then, through a series of strategic acquisitions and product development initiatives, S1 expanded its solution set to include applications that deliver financial services across multiple channels and provide payments and card management functionality.

S1 Shares are listed on the NASDAQ under the ticker symbol "SONE." S1's principal executive offices are located at 705 Westech Drive, Norcross, Georgia 30092 and its telephone number is (404) 923-3500.

BACKGROUND AND REASONS FOR THE EXCHANGE OFFER

Background of the Exchange Offer

As part of the ongoing evaluation of its businesses, ACI regularly considers strategic acquisitions, capital investments, divestitures and other possible transactions. In connection with such strategic evaluation, ACI has in the past considered a potential business combination transaction involving S1 and in connection therewith engaged in discussions with representatives of S1 over an approximately one-year period beginning in the Summer of 2010.

On August 30, 2010, Philip G. Heasley, ACI's Chief Executive Officer, met in Atlanta, Georgia, with Johann Dreyer, S1's Chief Executive Officer. During that meeting, Mr. Heasley expressed an interest in pursuing a possible acquisition of S1 by ACI.

On September 30, 2010, members of ACI's senior management met in Atlanta, Georgia with members of S1's senior management to discuss a possible acquisition of S1 by ACI. In that meeting, the representatives of ACI indicated a possible price range of \$7.50 to \$8.00 per S1 Share. The closing sales price for S1 Shares as reported on the NASDAQ Market was \$5.25 per share on September 29, 2010, the last trading day prior to that meeting. At the meeting, Mr. Dreyer indicated that he did not believe that it was opportune timing for S1 to be sold, but S1 might consider an enhanced proposal.

On October 6, 2010, representatives of S1 and ACI had a follow-up conversation in which the representatives of S1 informed the representatives of ACI that, after reviewing the matter with the S1 Board, S1 was not for sale and the S1 Board did not desire to initiate a sale process. They also mentioned that they believed that the price range that ACI had indicated was too low, but indicated that the S1 Board might be willing to consider a transaction at an increased valuation. ACI interpreted that communication as meaning that S1 would consider a transaction at a higher price other than the \$7.50-\$8.00 per share range that ACI had indicated, although there can be no assurance that this was intended. In the October 6, 2010 call, the representatives of S1 also said that the S1 Board acknowledged the rationale for a possible combination of S1 and ACI, but indicated that S1 would be willing to continue discussions only if the parties signed a standstill agreement.

On October 22, 2010, S1 and ACI signed an agreement that restricted ACI's ability to acquire S1 Shares or make any tender offer or other proposal to acquire S1. These restrictions expired prior to July 26, 2011. During the standstill period, ACI did not buy any S1 Shares and made proposals to acquire S1 confidentially.

On October 25, 2010, representatives of ACI's and S1's managements and financial advisors met in Atlanta, Georgia to discuss the S1 business and a possible transaction. From time to time thereafter, certain of S1's senior managers, representatives of S1's financial advisor and S1's counsel held additional discussions with members of ACI's senior management team and legal and financial advisors concerning a possible transaction.

On November 19, 2010, ACI submitted a written proposal to S1 to acquire S1 in an all-cash transaction at a price of \$8.40 per S1 Share, subject to confirmatory due diligence. ACI included a letter from a major financial institution stating that such institution was highly confident that ACI could raise the funds necessary to acquire S1 in an all-cash transaction at \$8.40 per share. In the November 19, 2010 proposal, ACI noted, among other things, "[w]e believe our proposal constitutes an extremely attractive opportunity for your stockholders. Our price represents a premium of 38% over the current market price of S1's common stock and a premium of 42% over the average market price over the past year." After ACI submitted the proposal letter, S1 representatives raised concerns about ACI's ability to finance an all-cash acquisition of S1 and regulatory considerations. ACI representatives indicated that ACI believed that it could satisfy any such concerns, and undertook to do so.

On December 9, 2010, Mr. Heasley spoke with Messrs. Dreyer and John W. Spiegel, Chairman of the S1 Board, regarding ACI's November 19th proposal. The parties also discussed ACI and S1's overlapping stockholder base and the potential for a mix of stock and cash consideration in an ACI-S1 transaction. On

December 20, 2010, ACI delivered a draft merger agreement to S1. The draft merger agreement contemplated the payment of the purchase price in cash or stock, as elected by S1 stockholders.

From time to time between December 2010 and February 2011, representatives of ACI's management and ACI's legal and financial advisors held additional discussions with representatives of S1's management and S1's legal and financial advisors concerning a possible transaction. On January 13, 2011, ACI sent a follow-up letter to S1 in an effort to progress the dialogue between the parties and to commence due diligence. During January 2011, S1's financial advisor on several times rescheduled a lender due diligence session, which was finally scheduled for March 3, 2011 but cancelled after S1 sent a letter to ACI on February 18, 2011 stating among other things that S1 was terminating discussions with ACI as the S1 Board had "determined that it is in the best interests of S1 and its stockholders to focus our efforts on executing our long-term business plan."

On March 10, 2011, S1 published its 2010 earnings release and provided public guidance with respect to its 2011 outlook. In late March 2011, Mr. Heasley initiated contact with S1 in an effort to continue discussion regarding a possible transaction. On April 5, 2011, Mr. Heasley met in person with Messrs. Dreyer and Spiegel in Atlanta, Georgia. On April 12, 2011, ACI submitted an acquisition proposal (including a revised draft of a definitive merger agreement) at a price of \$8.40 per S1 Share, 55% of which was to be paid in cash and 45% in ACI Shares. In its proposal, ACI noted, among other things, "[t]his proposal represents a premium of 26.1% over the current market price of S1's common stock and a premium of 37.4% over the average market price over the past year. We believe that this price is at a level at which your stockholders would enthusiastically support such a transaction."

On April 15, 2011, representatives of ACI's financial advisor held a discussion with representatives of S1's financial advisor regarding ACI's proposal. The financial advisors had additional contacts from time to time concerning the proposal between April 15, 2011 and June 14, 2011.

On June 14, 2011, Mr. Heasley spoke with Messrs. Dreyer and Spiegel regarding ACI's proposal. During the call, Mr. Spiegel informed Mr. Heasley that S1 was not interested in pursuing a possible transaction with ACI. No mention was made that S1 was simultaneously pursuing discussions with Fundtech Ltd., a company organized under the laws of Israel ("Fundtech"), relating to a possible merger transaction. Later that day, Mr. Heasley sent a follow-up letter to Mr. Spiegel requesting a response from the S1 Board regarding ACI's proposed valuation and other key terms. The June 14, 2011 letter, in relevant part, is as follows:

"June 14, 2011

Mr. John W. Spiegel Chairman of the Board of Directors S1 Corporation 705 Westech Drive Norcross, Georgia 30092

Dear John,

I appreciated your feedback during our call this morning. I was surprised by your Board's lack of response to our April 12th proposal.

ACI and our advisors have complied with all of the process requirements that S1 management and your advisors have communicated to us since last Fall. First, our financing advisors, Goldman Sachs and Wells Fargo, have had multiple interactions with S1 management and your advisor providing you with certainty of the financial structure we proposed. Second, our legal advisor, Jones Day, has had several conversations with your external counsel to address any regulatory concerns around the proposed transaction. Also, Jones Day submitted on December 20, 2010, a fair and balanced merger agreement and a revised version on April 12, 2011, to which we have still not received any feedback.

We have studied the regulatory backdrop applicable to the proposed transaction. As reflected in the April 12th merger agreement, we believe the regulatory review process will not impact the certainty of closing and we have outlined measures in the agreement that demonstrate our confidence in this view.

To date, your Board has not provided any response to our proposed valuation or other key terms. We would have liked to have had a discussion on value, but are now left to determine valuation based on publicly available information. With the nine-month standstill period expiring on July 22nd, we still believe it would be in the best interests of S1 and your Board to engage with ACI to maximize value for S1's shareholders.

The combination of ACI and S1 would create a leading global player in the enterprise payments software industry. As I have indicated, the combination of our companies would not only benefit your shareholders, but would also offer more and better options to customers within our marketplace. We sincerely hope that we will move forward in a negotiated transaction which can be presented to your stockholders as the joint effort of ACI and S1 Boards of Directors and management teams.

This opportunity has the highest priority for us and we are committed to work with S1 and your Board in any way we can to expeditiously move this forward.

Sincerely

/s/ Philip G. Heasley President and CEO ACI Worldwide, Inc.

cc: Mr. Johann Dreyer, Chief Executive Officer, S1 Corporation"

On June 27, 2011, S1 and Fundtech announced that they had entered into the Plan of Merger and Reorganization, dated as of June 26, 2011, by and among S1, Finland Holdings (2011) Ltd., a company organized under the laws of Israel and a wholly owned subsidiary of S1, and Fundtech (the "Fundtech Merger Agreement").

On July 26, 2011, ACI delivered a proposal letter containing the Original ACI Merger Proposal to the S1 Board and issued a press release announcing the Original ACI Merger Proposal. The letter read as follows:

"July 26, 2011

Board of Directors \$1 Corporation 70 Westech Drive Norcross, Georgia 30092 Attn: Mr. John W. Spiegel, Chairman of the Board

Gentlemen:

We are pleased to submit the following proposal by which ACI Worldwide and S1 Corporation would combine to create a leading global enterprise payments company. We propose to acquire 100% of the issued and outstanding common stock of S1 in a cash and stock transaction valued at \$9.50 per share. This equates to a 33% premium to S1's closing market price on July 25, 2011, a 32% premium to S1's 90-day volume weighted average price and a 23% premium to S1's 52-week high. Our proposal is being made pursuant to and in accordance with the superior offer provisions you provided for in your June 26, 2011 merger agreement with Fundtech.

Given the overlapping shareholder base of our companies, we believe that a cash and stock transaction is ideal for all stakeholders, as it provides a mix of immediate value, tax efficiency and the ability to benefit from significant synergies. Accordingly, the form of consideration in our proposal consists of 40% in ACI stock and 60% in cash. In addition, our proposal includes a cash election feature, subject to proration, designed to provide your shareholders with the optimal consideration of cash and/or stock for their individual circumstances and preferences. Upon completion of our proposed transaction and based on the most recent closing price of ACI's common stock, S1 shareholders would own approximately 15% of the combined company on a fully diluted basis.

We believe the combination of ACI and S1 provides specific tangible benefits to the combined shareholders, including, among others:

- Combination of complementary products and expanded customer bases, providing a rich set of capabilities and a broad portfolio of products to serve customers across the entire electronic navments spectrum:
- · The creation of an approximate \$100 million in revenue hosting business serving our collective customer base with enhanced margins due to the consolidation of fixed infrastructure;
- · Expanded presence in high-growth international markets and additional capabilities with respect to ACI's retailer payments and online banking solutions;
- · Substantial synergy opportunities by leveraging ACI's established global cost structure, eliminating redundant operating expenses and consolidating our on-demand operations and facilities; and
- Strong financial profile with full year earnings accretion in 2012.

We believe that our premium stock and cash proposal is both financially and strategically superior to your proposed transaction with Fundtech. Our proposal offers substantially greater current financial value to S1 shareholders in the form of a meaningful premium to the current stock price and a clearer, more expedient path to value creation over the long-term through the realization of significant synergies, with less risk and uncertainty than the Fundtech transaction. Additionally, our proposed combination creates a more diverse, long-term shareholder base for the pro forma company.

Our proposal contemplates that, following the completion of the transaction, S1 shareholders would have a meaningful ownership stake in ACI, which has:

- · Produced a shareholder return of approximately 91% over the past three years, significantly outperforming the relevant peer group;
- Increased 60-month backlog to \$1.6 billion in 2010, up \$350 million since 2006;
- · Driven monthly recurring revenue to 68% in 2010, up nearly 29% since 2007; and
- Increased Adjusted EBITDA margin to 21% in 2010, from 7% in 2007.

Not only have we executed our historical business plan, as evidenced by our strong second quarter earnings, we have raised our 2011 guidance and are firmly committed to achieving our five-year strategy.

Our proposal includes committed financing from Wells Fargo Bank for the cash portion of the transaction. As such, the proposed transaction is not subject to any financing condition. In addition, we have completed a review of applicable regulatory requirements and, while we do not expect any issues to delay closing, our merger agreement contains appropriate undertakings by us to assure HSR clearance.

Our proposal is subject to the negotiation of a mutually acceptable definitive merger agreement, a draft of which we are including as part of our proposal. Consummation of the transaction is subject to satisfaction of customary closing conditions, including expiration of the waiting period under the Hart-Scott-Rodino Antitrust Improvements Act. You will see that our draft is the same as the Fundtech Merger Agreement except for changes required in order to effect our transaction. We are prepared to promptly conclude our confirmatory due diligence and to give you and your representatives immediate due diligence access to us.

We believe that our proposal represents a Parent Superior Offer that clearly meets the standards set forth in Section 6.7(a) of your Fundtech Merger Agreement as it is more favorable to S1 shareholders from a financial point of view than the Fundtech transaction, and it is likely to be completed, taking into account all financial, regulatory, legal and other aspects of our proposal. Accordingly, we believe that you must, consistent with the Fundtech Merger Agreement, provide us with confidential information and participate in discussions and negotiations with us to finalize a transaction.

We stand ready and willing to promptly engage with S1 on this transaction, so that together we can effect a transaction that benefits both companies' shareholders. That said, we are committed to making this transaction a reality.

Our Board of Directors has unanimously approved the submission of this proposal. We and our financial and legal advisors are prepared to move forward immediately with you and your advisors to finalize a mutually beneficial agreement, and make the combination of S1 and ACI a reality, for the benefit of both companies' shareholders.

We look forward to hearing from you.

Sincerely,

/s/ Philip G. Heasley President and CEO ACI Worldwide, Inc.

Enclosures'

On July 27, 2011, ACI filed a Notification and Report Form with the FTC and Antitrust Department under the HSR Act relating to the Original ACI Merger Proposal.

On August 2, 2011, S1 announced that the S1 Board had rejected the Original ACI Merger Proposal based on the S1 Board's determination that pursuing discussions with ACI at this time "is not in the best financial or strategic interests of S1 and its stockholders." According to S1's August 2, 2011 press release, Mr. Spiegel said:

"The S1 Board gave careful consideration to each of the proposed terms and conditions of ACI's proposal. In the end, the Board determined that ACI's proposal is not in the best interests of S1 and its stockholders. We believe that continuing to execute on our long-term business plan, which includes the business combination with Fundtech, will best help us maximize stockholder value and achieve our strategic goals."

On August 11, 2011, S1 announced that it had set August 18, 2011 as the record date and September 22, 2011 as the date of the S1 special stockholder meeting. On August 22, 2011, S1 filed its definitive proxy statement with the SEC and reported that it had commenced mailing its proxy statement to S1 stockholders on or about August 22, 2011.

On August 25, 2011, ACI delivered a proposal letter to S1's Board containing a revised merger proposal, increasing the cash consideration by \$0.50 per S1 Share, assuming full proration, and issued a press release announcing the revised merger proposal. The letter read as follows:

"August 25, 2011

PERSONAL AND CONFIDENTIAL ELECTRONIC DELIVERY

John W. Spiegel Chairman of the Board of Directors S1 Corporation 705 Westech Drive Norcross, Georgia 30092

Dear John

We remain committed to acquiring S1 Corporation and are pleased to inform you that we have enhanced our proposal in order to provide S1 shareholders with additional value certainty for their investment. Given the recent significant market volatility, ACI Worldwide, Inc. has increased its cash and stock proposal from \$5.70 per share plus 0.1064 ACI shares to \$6.20 per share, plus 0.1064 ACI shares, assuming full proration.

We are confident that your shareholders will find our enhanced proposal to be superior to the Fundtech Ltd. transaction, and we stand ready and willing to promptly engage with S1 to consummate a transaction that benefits both companies' shareholders. Based on the closing price of ACI stock on July 25, 2011, the day

prior to our initial proposal, our enhanced proposal provides a per share consideration of \$10.00 to each S1 shareholder. Based on the closing price of ACI stock on August 24, 2011, our enhanced proposal provides a per share consideration of \$9.29 to each S1 shareholder. ACI's enhanced proposal also equates to a:

- 30% premium to S1's unaffected closing market price on July 25, 2011;
- 29% premium to the volume weighted average price of S1 shares over the previous 90 days prior to July 25, 2011; and
- · 20% premium to the 52-week high of S1 shares, for the 52-week period ending July 25, 2011.

When evaluating our enhanced proposal, we strongly encourage you to consider at what price levels S1 would be trading absent the ACI proposal. Since we made our proposal on July 26, 2011, the NASDAQ Index has declined by 13% while S1's stock price, affected by the value of the ACI proposal, has generally avoided the declines experienced in the overall market. Furthermore, we believe that your shareholders know that, had ACI not made its proposal, S1's share price would have been affected by the overall decline in stock market valuations. We also believe that the S1 shareholder reaction to our proposal, despite the significant ensuing market volatility, underscores its strength.

Your August 22, 2011, shareholder letter questioned whether we had the financing for the cash portion of our merger proposal as well as our commitment to obtain clearance under the Hart-Scott-Rodino (HSR) Act. To resolve these issues, we have a fully executed commitment letter from Wells Fargo Bank, N.A. sufficient to fund the cash required by our proposal and to finance our ongoing operations, and we would be pleased to provide a copy of such commitment letter upon request. In addition, we reiterate that we are willing to provide appropriate assurance of satisfaction of the HSR Act condition, including a divestiture commitment (if required) and substantial break-up compensation. However, it does not withstand scrutiny for S1 to, on the one hand, refuse to engage with us on these issues and, on the other hand, point to these issues as a reason for not engaging in the first place.

As S1 has been unwilling to engage, we are taking the actions we believe necessary to consummate our proposed transaction. We are filing our definitive proxy statement to begin solicitation of votes against the proposed Fundtech transaction and, rest assured, we will take all actions necessary to advance our proposal. We would, however, strongly prefer to begin a direct dialogue with S1's management and advisors

We believe that our proposal represents a Parent Superior Offer that clearly meets the standards set forth in Section 6.7(a) of the Fundtech merger agreement as it is more favorable to S1 shareholders from a financial point of view than the Fundtech transaction and it is likely to be completed, taking into account all financial, regulatory, legal and other aspects of our proposal.

We remain convinced of the strategic benefits of this transaction and strongly believe that it is in the best interests of both ACI's and S1's shareholders. We look forward to your prompt reply.

Sincerely

/s/ Philip G. Heasley

cc: Johann Dreyer, Chief Executive Officer, S1 Corporation"

On August 25, 2011, ACI filed with the SEC and began mailing its proxy statement soliciting votes "against" the Fundtech Merger Agreement and related proposals.

On August 26, 2011, ACI withdrew its initial HSR filing and refiled it on August 29, 2011 in order to permit the Antitrust Division to have additional time to review the filing. On September 27, 2011, ACI withdrew its filing under the HSR Act and refiled it on September 28, 2011. The 30-calender day waiting period recommenced in connection with such refiling so that the waiting period now expires, unless earlier terminated or extended, at 11:59 p.m., Eastern time, on October 28, 2011.

On August 29, 2011, after the S1 Board determined that the conditions to the Fundtech Merger Agreement that would permit discussions with a third party had been satisfied, an authorized representative of S1 contacted a representative of ACI to discuss the value and certainty of closure of ACI's revised acquisition proposal. On August 30, 2011, ACI commenced the Original ACI Exchange Offer. The Original ACI Exchange Offer disclosed, among other things, that a representative of S1 had contacted a representative of ACI with respect to the value and certainty of closure of ACI's revised acquisition proposal.

On August 30, 2011 and at a meeting on September 7, 2011, the S1 Board met with its advisors and certain members of S1's senior management to consider, among other things, the terms of the Original ACI Exchange Offer.

Between August 30, 2011 and October 3, 2011, senior managers and representatives of S1 and ACI had additional discussions regarding ACI's revised acquisition proposal and conducted diligence of the companies' respective businesses and operations.

On August 30, 2011, ACI filed this prospectus/offer to exchange with the SEC with respect to the Original ACI Exchange Offer.

On September 13, 2011, S1 filed a Solicitation/Recommendation Statement on Schedule 14D-9 with the SEC and released a letter to its stockholders in which it responded to the Original ACI Exchange Offer. The S1 Board recommended that S1 stockholders not tender their S1 Shares pursuant to the Original ACI Exchange Offer and described certain concerns regarding the Original ACI Exchange Offer, including with respect to the timing of the launch of the Original ACI Exchange Offer, the conditions to the Original ACI Exchange Offer, the consideration and value offered by ACI, the use of debt financing, the tax consequences of the Original ACI Exchange Offer and certain assertions by ACI in the Original ACI Exchange Offer, including with respect to the terms and conditions of ACI's revised acquisition proposal and S1's track record of creating stockholder value. The filing also disclosed that a representative of S1 contacted a representative of ACI to discuss the value and certainty of closure of ACI's revised acquisition proposal, and that, between August 30 and September 12, 2011, senior managers and representatives of S1 and ACI had additional discussions regarding ACI's revised acquisition proposal.

On September 15, 2011, ACI filed with the SEC an amendment to its Registration Statement on Form S-4 of which this prospectus/offer to exchange forms a part.

Also, on September 15, 2011, S1 announced that Fundtech had delivered to S1 a notice of its intent to change its recommendation with respect to the pending merger with S1, to terminate the Fundtech Merger Agreement and to enter into a written definitive agreement with entities formed by GTCR Fund X/A LP and its affiliated entities. The S1 Board determined not to revise its proposal to acquire Fundtech and instead to terminate the Fundtech Merger Agreement. S1 announced on September 16, 2011 that it had terminated the Fundtech Merger Agreement and received an \$11.9 million termination fee from Fundtech. S1 also announced that (i) its special meeting of S1 stockholders scheduled for October 13, 2011 was canceled and (ii) despite its determination to terminate the Fundtech Merger Agreement, the S1 Board had not changed its recommendation with respect to the Original ACI Exchange Offer.

On September 16, 2011, S1 filed Amendment No. 1 to its Solicitation/Recommendation Statement on Schedule 14D-9, reporting that the S1 Board has not changed its recommendation with respect to the Original ACI Exchange Offer.

On September 18, 2011, S1 received from ACI proposed language relating to antitrust matters and certainty of closure that ACI proposed to include in a transaction agreement between S1 and ACI. Following receipt of such language, senior managers and representatives of S1 and ACI held additional discussions regarding the terms of ACI's revised acquisition proposal.

On September 21, 2011, ACI filed with the SEC an amendment to its Registration Statement on Form S-4 relating to the Original ACI Exchange Offer. The amendment disclosed, among other things, that between

August 30, 2011 and September 20, 2011, senior managers and representatives of ACI and S1 had additional discussions regarding ACI's revised acquisition proposal.

On September 22, 2011, members of S1's and ACI's senior management held a conference call to discuss certain diligence matters

On September 25, 2011, ACI's chief executive officer and the chairman of the S1 Board and S1's chief executive officer met in person to further discuss specific elements of a potential transaction between the companies. Among other things, the parties discussed (1) the price per S1 Share offered by ACI, (2) ACI's willingness to strengthen its previously proposed undertaking to obtain clearance under the HSR Act or pay a reverse break-up fee in the event such clearance was not obtained, (3) employee retention matters and (4) integration and transition concerns. ACI's chief executive officer indicated that ACI might be willing to increase the cash component of its proposal from \$6.20 to \$6.42 per S1 Share, assuming full proration, and agree that ACI would use its best efforts to obtain clearance of the transaction under the HSR Act, which could include holding separate or divesting S1 assets, but that ACI would not agree to a reverse break-up fee if HSR Act clearance was not obtained and would only agree to an increase in the value offered if all other issues were resolved to its satisfaction. S1's chairman indicated that he did not believe that the proposed increase in value was sufficient.

On September 27, 2011, the S1 Board met with its advisors and certain members of senior management present to review the status of discussions with ACI and to consider the relative merits of the adoption of a stockholder rights plan to give S1 greater leverage to negotiate against ACI or consider its strategic alternatives in the event that the Original ACI Exchange Offer was not extended beyond its original expiration date of September 28, 2011.

On September 28, 2011, ACI's chief executive officer and the chairman of the S1 Board discussed, among other things, the value of ACI's acquisition proposal. During the discussion, ACI's chief executive officer offered to increase the value of the cash component of ACI's acquisition proposal from \$6.20 to \$6.62. Additionally, later that day, ACI announced that Offeror had extended the Exchange Offer until October 31, 2011, unless further extended.

On September 29, 2011, the S1 Board met with its advisors and certain members of senior management to review the status of discussions with ACI. The board received an update as to the matters discussed between ACI's chief executive officer and S1's chairman on September 28, 2011. Following the meeting, senior managers and representatives of S1 and ACI had additional discussions and exchanged drafts of the Transaction Agreement.

On October 2, 2011, the S1 Board met with representatives of Hogan Lovells, Raymond James, PricewaterhouseCoopers LLP and certain members of S1's senior management to consider the possible transaction with ACI. Following a report to the S1 Board concerning the results of S1's due diligence review of ACI, a representative of Hogan Lovells advised the S1 Board of its fiduciary duties and, along with members of S1's senior management, reviewed the terms of the Transaction Agreement with the S1 Board, and answered questions from the S1 Board members about the transaction. A representative of Raymond James' inancial analysis of the proposed transaction and orally expressed Raymond James' opinion (subsequently confirmed in writing) that as of such date, based upon and subject to the considerations, assumptions, qualifications and limitations set forth in the opinion, the Exchange Offer price and the Proration Amount of Cash and Stock Consideration, were fair, from a financial point of view, to S1's stockholders. (See Amendment No. 2 to S1's Solicitation/Recommendation Statement on Schedule 14D-9 for a discussion of Raymond James of Paymond James and the other information presented at that and prior S1 Board meetings, unanimously determined that the transaction with ACI as proposed was advisable, fair to and in the best interests of S1 and the S1 stockholders and approved the Transaction Agreement and the transactions contemplated thereby. Raymond James later delivered its written fairness opinion, dated October 3, 2011, a copy of which is attached as Annex 1 to Amendment No. 2 to S1's Solicitation/Recommendation Schedule 14D-9.

Following the approval of the boards of directors of S1 and ACI, the parties executed the Transaction Agreement on October 3, 2011 and issued a joint press release announcing the transaction on the morning of that day. ACI also filed with the SEC on that date an amendment to its Schedule TO which reported the transaction and communicated the terms of the Exchange Offer.

On October 13, 2011, ACI filed with the SEC a post-effective amendment to its Registration Statement on Form S-4 of which this prospectus/offer to exchange forms a part.

Reasons for the Exchange Offer

ACI believes that the combination of ACI's and S1's businesses will create significant value for both ACI's and S1's current stockholders. We believe the combination of ACI and S1 is a compelling combination with a number of strategic benefits, including the following:

Value

At the \$9.68 per S1 Share value of the Cash-Stock Consideration as of October 12, 2011, assuming full proration, the Exchange Offer represents (1) a 35.8% premium to the closing sales price of S1 Shares on July 25, 2011, the last trading day prior to the public announcement of the Original ACI Merger Proposal, (2) a 34.3% premium to the volume weighted average closing price of S1 Shares over the previous 90 days prior to the announcement of the Original ACI Merger Proposal, and (3) a 24.9% premium to the 52-week high of S1 Shares for the 52-Week Period.

S1 stockholders who elect the Cash-Stock Consideration contemplated by the Exchange Offer will be subject to proration. Since the value of ACI Shares fluctuates, the per S1 Share Stock Consideration necessarily could have a value that is different than the per S1 Share Cash Consideration. As a consequence, in the Exchange Offer, S1 stockholders could receive a combination of Cash-Stock Consideration with a value that is different from the value of such consideration on the date of the Exchange Offer and the date of the consummation of a transaction with ACI.

The elections of other S1 stockholders would affect whether S1 stockholders received solely the type of consideration they had elected or whether a portion of the consideration S1 stockholders elected were exchanged for another form of consideration as a result of the pro ration procedures contemplated by the Exchange Offer.

Solely for purposes of illustration, the following table indicates the value of the Cash Consideration, the Stock Consideration and the blended value of the Cash-Stock Consideration based on different assumed prices for ACI Shares.

	Assuming N	o Proration	Assuming Full Proration				
Assumed ACI Share Price	Value of Stock Consideration	Value of Cash Consideration	Value of Stock Consideration	Value of Cash Consideration	Value of Cash-Stock Consideration		
\$37.93(1)	\$11.94	\$10.00	\$4.04	\$6.62	\$10.66		
\$35.70(2)	\$11.24	\$10.00	\$3.80	\$6.62	\$10.42		
\$30.49(3)	\$ 9.60	\$10.00	\$3.24	\$6.62	\$ 9.86		
\$27.54(4)	\$ 8.67	\$10.00	\$2.93	\$6.62	\$ 9.55		
\$28.77(5)	\$ 9.06	\$10.00	\$3.06	\$6.62	\$ 9.68		
\$22.70(6)	\$ 7.15	\$10.00	\$2.42	\$6.62	\$ 9.04		

⁽¹⁾ Represents highest sales price for ACI Shares in the 52-Week Period.

⁽²⁾ Represents closing sales price for ACI Shares on July 25, 2011, the last trading day prior to the announcement of the Original ACI Merger Proposal.

- (3) Represents closing sales price for ACI Shares on August 29, 2011, the last trading day prior to the commencement of the Original ACI Exchange Offer.
- (4) Represents closing sales price for ACI Shares on September 30, 2011, the last trading day prior to the announcement of the Transaction Agreement.
- (5) Represents closing sales price for ACI Shares on October 12, 2011, the last trading day prior to the date of this prospectus/offer to exchange.
- (6) Represents the lowest sales price for ACI Shares in the 52-Week Period.

The equity capital markets have been highly volatile and market prices for ACI Shares and S1 Shares have fluctuated and can be expected to continue to fluctuate. S1 stockholders are urged to obtain current trading price information prior to deciding whether to tender shares pursuant to the Exchange Offer, whether to exercise withdrawal rights as provided herein and, with respect to the election, whether to receive the Cash Consideration or the Stock Consideration or some combination thereof. The premium represented by the Exchange Offer may be larger or smaller depending on market prices on any given date and will fluctuate between the date of this prospectus/offer to purchase, the Expiration Time and the date of the consummation of the Exchange Offer.

Strategic Rationale:

The Exchange Offer provides immediate cash value to S1 stockholders, as well as the opportunity to participate in the value creation in the Exchange Offer through the receipt of ACI Shares. ACI believes that the complementary nature of ACI and S1 creates a compelling opportunity to establish a full-service global leader of financial and payments software with significant scale and financial strength, including as follows:

- Highly Complementary Product and Customer Bases: Combined, ACI and S1 would provide a rich set of capabilities and a broad portfolio of products to customers across the entire electronic payments spectrum. In particular, ACI believes that the acquisition of S1 would provide breadth and additional capabilities to what ACI does today, including: (1) expand ACI's retailer business beyond North America; (2) increase ACI's retail banking payments business down into lower and mid-tier financial institutions; and (3) add function and global reach to ACI's online business banking offering, including new capabilities around branch banking and trade. The acquisition of S1 would support ACI's position as a leading provider of the most unified payments solution to serve retail banking, wholesale banking, processors and retailers and would enable its customers to lower their operational costs and improve time-to-market.
- Enhanced Scale and Global Position: ACI's and S1's principal competitors are substantially larger companies with greater financial resources than ACI and S1 have. The combined ACI and S1 would have revenue of \$683 million and adjusted EBITDA of \$123 million for the 12 months ended June 30, 2011. This scale advantage would enable the combined ACI and S1 to more effectively serve its combined global customer base and compete against the very large companies which operate in the electronic payments software business.
- Significant Synergy Opportunities: ACI expects the combination of ACI and S1 will generate a significant amount of operational efficiencies and cost savings that will drive margin expansion for the acquired S1 business and earnings accretion for the combined company. ACI estimates that the annual pre-tax cost savings related to the Exchange Offer would be approximately \$30 million, primarily attributable to elimination of \$1's public company costs and rationalization of duplicate general and administrative functions, sales/marketing functions and costs, occupancy costs, product management and R&D functions. In addition, ACI expects to consolidate the combined company's hosting data centers and infrastructure. Further, ACI expects the cost savings will improve \$1's margins in line with ACI's margins for adjusted EBITDA. Assuming that the Exchange Offer is closed in the fourth calendar quarter of this year, ACI anticipates the cost savings would be fully realizable in 2012.

• Strong Financial Position: ACI would continue to have a strong financial profile driven by a solid balance sheet with substantial liquidity and a recurring revenue model that generates significant free cash flows, allowing for further future investments in the business. In addition, ACI expects the transaction to be accretive to full year earnings in 2012.

The following metrics provide relevant information with respect to ACI's recent financial performance, as of July 26, 2011, the date of the Original ACI Merger Proposal:

- · ACI has produced a stockholder return of approximately 90% over the past three years, significantly outperforming the relevant peer group;
- ACI has increased its 60-month backlog to \$1.6 billion in 2010, up \$350 million since 2006;
- ACI has driven monthly recurring revenue to 68% in 2010, up nearly 29% since 2007; and
- ACI has increased adjusted EBITDA margin to 21% in 2010, from 7% in 2007.

This prospectus/offer to exchange includes summary selected unaudited pro forma combined financial information that is intended to provide S1 stockholders with information relating to ACI's financial results assuming that ACI and S1 had already been combined.

THE TRANSACTION AGREEMENT

Overview

On October 3, 2011, ACI and Offeror entered into the Transaction Agreement with S1 as a means to acquire all of the outstanding S1 Shares. The S1 Board has unanimously (1) determined that the transactions contemplated by the Transaction Agreement; and in the best interests of, S1 and the S1 stockholders; (2) approved the transactions contemplated by the Transaction Agreement; and (3) determined to recommend that the S1 stockholders accept the Exchange Offer and tender their S1 Shares to Offeror pursuant to the Exchange Offer. The S1 Board unanimously recommends that S1 stockholders accept the Exchange Offer by tendering their S1 Shares into the Exchange Offer. Information about the recommendation of the S1 Board is more fully described in Amendment No. 2 to S1's Solicitation/Recommendation Statement on Schedule 14D-9, which is being mailed to S1 stockholders together with this prospectus/offer to exchange and is incorporated herein by reference.

The following is a summary of selected material provisions of the Transaction Agreement. This summary is qualified in its entirety by reference to the Transaction Agreement, which is incorporated by reference in its entirety and attached to this prospectus/offer to exchange as Appendix D. This summary does not purport to be complete and may not contain all of the information about the Transaction Agreement that may be important to you. We encourage you to read the Transaction Agreement carefully and in its entirety, as it is the legal document governing the Exchange Offer and the Second-Step Merger.

The Transaction Agreement has been provided solely to inform investors of its terms. The representations, warranties and covenants contained in the Transaction Agreement were made only for the purposes of such agreement and as of specific dates, were made solely for the benefit of the parties to the Transaction Agreement and may be intended not as statements of fact, but rather as a way of allocating risk to one of the parties if those statements prove to be inaccurate. In addition, such representations, warranties and covenants may have been qualified by certain disclosures not reflected in the text of the Transaction Agreement, and may apply standards of materiality in a way that is different from what may be viewed as material by stockholders of, or other investors in, S.I. SI stockholders and other investors are not third-party beneficiaries under the Transaction Agreement and should not rely on the representations, warranties and covenants or any descriptions thereof as characterizations of the actual state of facts or conditions of SI, ACI, Offeror or any of their respective subsidiaries or affiliates. ACI acknowledges that, notwithstanding the inclusion of the foregoing cautionary statements, it is responsible for considering whether additional specific disclosures of material information regarding material contractual provisions are required to make the statements in this prospectus/offer to exchange not misleading.

The Exchange Offer

The Transaction Agreement provides that the Exchange Offer will be conducted on the terms and subject to the conditions set forth in the section of this prospectus/offer to exchange titled "The Exchange Offer."

Recommendation of the S1 Board

The S1 Board has unanimously (1) determined that the transactions contemplated by the Transaction Agreement are fair to, and in the best interests of, S1 and the S1 stockholders; (2) approved the transactions contemplated by the Transaction Agreement; and (3) determined to recommend that the S1 stockholders accept the Exchange Offer and tender their S1 Shares to Offeror pursuant to the Exchange Offer.

S1 Stockholder Meeting

If the S1 stockholder approval is required by applicable law, S1 has agreed to take all necessary actions to duly call, give notice of, convene and hold a meeting of the S1 stockholders or, if possible, arrange for action by written consent of the S1 stockholders for the purpose of obtaining such stockholder approval and to use reasonable best efforts to solicit its stockholders to obtain such stockholder approval. At such S1 stockholder meeting or any postponement or adjournment thereof, ACI has agreed to vote, or cause to be voted, all of the S1 Shares then owned by it, Offeror or any of their respective subsidiaries in favor of the

adoption of the Transaction Agreement and to deliver or provide, in its capacity as an S1 stockholder, any other approvals that are required pursuant to applicable law to effect the transactions contemplated by the Transaction Agreement.

Top-Up Option

S1 has granted to Offeror the Top-Up Option, for so long as the Transaction Agreement has not been terminated, to purchase from S1 up to the number (but not less than that number) of authorized and unissued S1 Shares equal to the lowest number of S1 Shares that, when added to the number of S1 Shares owned by ACI, Offeror or any subsidiary of ACI at the time of the exercise of the Top-Up Option, constitutes at least one S1 Share more than 90% of the S1 Shares (after giving effect to the issuance of S1 Shares to be issued upon exercise of the Top-Up Option (the Top-Up Shares)).

The Top-Up Option may be exercised by Offeror only once, in whole but not in part, at any time during the two-business day period following the Acceptance Time, or if the Exchange Offer is extended, during the two-business day period following the expiration date of such Subsequent Offering Period, and only if Offeror owns as of such time more than 50% but less than 90% of S1 Shares outstanding.

The Top-Up Option is not exercisable (1) to the extent the number of S1 Shares issuable upon exercise of the Top-Up Option would exceed the number of authorized but unissued and unreserved S1 Shares, (2) if any law, injunction or other order (whether temporary, preliminary or permanent), judgment, decree, executive order or award then in effect would prohibit the exercise of the Top-Up Option or the delivery of the Top-Up Shares, (3) unless ACI or Offeror has accepted for payment all S1 Shares validly tendered in the Exchange Offer and not withdrawn, and (4) unless immediately after such exercise and the issuance of S1 Shares pursuant thereto, Offeror would own at least 90% of the outstanding S1 Shares (assuming the issuance of the Top-Up Shares).

The aggregate purchase price payable for the Top-Up Shares being purchased by Offeror pursuant to the Top-Up Option will be determined by multiplying the number of such Top-Up Shares by the greater of (1) the closing price of a Company Share on NASDAQ the last trading day prior to the exercise of the Top-Up Option or (2) the Cash Consideration, without interest. Such purchase price may be paid by Offeror, at its election, either (a) in cash or (b) by paying in cash an amount equal to not less than the aggregate par value of such Top-Up Shares and by executing and delivering to S1 a promissory note having a principal amount equal to the balance of such purchase price.

Appraisal Rights/Dissenting Shares

The Transaction Agreement provides that any S1 Shares that are outstanding immediately prior to the Effective Time and that are held by S1 stockholders who have neither voted in favor of the Second-Step Merger nor consented thereto in writing and who have demanded properly in writing appraisal for such S1 Shares in accordance with Section 262 of the DGCL not be converted into, or represent the right to receive, the Proration Amount of Cash and Stock Consideration. Such stockholders will be entitled to receive payment of the appraised value of such dissenting shares held by them in accordance with the provisions of Section 262 of the DGCL, except that all such dissenting shares held by S1 stockholders who have failed to perfect or who effectively have withdrawn or lost their rights to appraisal of such dissenting shares under Section 262 of the DGCL will thereupon be deemed to have been converted into, and to have become exchangeable for, as of the Effective Time, the right to receive the Proration Amount of Cash and Stock Consideration, without any interest thereon, upon surrender, in the manner provided in the Transaction Agreement.

Directors After the Acceptance Time

The Transaction Agreement provides that, upon the payment by Offeror for S1 Shares tendered pursuant to the Exchange Offer representing at least a majority of the outstanding S1 Shares on a fully diluted basis, Offeror will be entitled to designate such number of directors, rounded up to the next whole number, on the S1 Board as is equal to the product of (1) the total number of directors on the S1 Board (after giving effect to any increase in the number of directors described in this paragraph) and (2) the percentage that such number

of S1 Shares so purchased bears to the total number of then-outstanding S1 Shares on a fully-diluted basis. S1 will, upon request by Offeror, promptly increase the size of the S1 Board or use commercially reasonable efforts to seek the resignations of such number of directors as is necessary to provide Offeror with such level of representation and will use commercially reasonable efforts to cause Offeror's designees to be so elected or appointed. Offeror's director designees will be one or more persons identified on Appendix A to this document. S1 will use commercially reasonable efforts to cause each committee of the S1 Board to include persons designated by Offeror constituting the same percentage of each such committee as Offeror's designees constitute on the S1 Board. Prior to the Effective Time, the S1 Board will have at least three members who were directors as the date of the Transaction Agreement, who are independent directors for purposes of the continued listing requirements of NASDAQ and who are eligible to serve on S1's audit committee.

Second-Step Merger; Effect on Capital Stock

The Transaction Agreement provides that at the Effective Time, Offeror will be merged with and into S1. As a result of the Second-Step Merger, the separate corporate existence of Offeror will cease, and S1 will continue as the surviving corporation. The directors of Offeror immediately prior to the Effective Time will be the initial directors of the surviving corporation, and the officers of S1 immediately prior to the Effective Time will be the initial officers of the surviving corporation.

Pursuant to the Transaction Agreement, each S1 Share held in the treasury of S1 or owned by ACI or Offeror immediately prior to the Effective Time will be canceled and retired without any conversion thereof. At the Effective Time, each S1 Share (other than S1 Shares held in treasury by S1 or owned by ACI or its wholly owned subsidiaries, certain restricted S1 Shares converted into restricted ACI Shares pursuant to the Transaction Agreement and any dissenting S1 Shares) by virtue of the Second-Step Merger and without any action on the part of the holder thereof, will be converted into the right to receive the Proration Amount of Cash and Stock Consideration, and all such S1 Shares will no longer be outstanding and will automatically be canceled and cease to exist, and each holder of an S1 Share certificate or evidence of S1 Shares in book-entry form which immediately prior to the Effective Time represented any such S1 Shares will cease to have any rights with respect thereto, except the right to receive the Proration Amount of Cash and Stock Consideration, without interest. At the Effective Time, each share of common stock of Offeror issued and outstanding immediately prior to the Effective Time will be converted into one share of common stock of the surviving corporation.

Merger Without Meeting of Stockholders; Special Meeting

If Offeror owns at least 90% of the outstanding S1 Shares following the Exchange Offer and any subsequent offer period (including through exercise of the Top-Up Option), Offeror will cause the Second-Step Merger to become effective promptly after the consummation of the Exchange Offer, without a meeting of S1 stockholders.

ACI may only complete the Second-Step Merger if it purchases the S1 Shares pursuant to the Exchange Offer. However, the Transaction Agreement gives ACI the right to require that S1 convene a stockholders' meeting to approve a merger in which the S1 stockholders would have the right to receive the Proration Amount of Cash and Stock Consideration as a result of a merger of Offeror and S1 instead of the Exchange Offer. The terms and conditions of such a transaction would be substantially the same as the terms and conditions of the Exchange Offer. ACI had not determined whether to exercise this right as of the date of this prospectus/offer to exchange.

Treatment of Stock Options; SARs; Restricted Stock

The Transaction Agreement provides that each S1 Stock Option issued under S1's 1997 Stock Option Plan and 1998 Directors' Stock Option Plan that is outstanding will, if elected by the holder, be exercised effective as of immediately prior to the Effective Time, with the effect that the S1 Shares issuable upon exercise will be deemed for all purposes to be issued and outstanding immediately prior to the Effective Time and will have the right to receive the Proration Amount of Cash and Stock Consideration.

The holders of S1 Stock Options under the 1997 Stock Option Plan and the 1998 Directors' Stock Option Plan will be notified that such S1 Stock Options may be exercised at any time during the relevant exercise

period beginning on October 3, 2011 and ending on the day before the Effective Time, provided that (1) any such exercise, to the extent that it relates to an S1 Stock Option that would become exercisable only at the Effective Time, will be contingent until, and will become effective only upon, the occurrence of the Effective Time and (2) no S1 Stock Option may be exercised after the relevant exercise period.

Each outstanding S1 Stock Option under the 1997 Stock Option Plan and the 1998 Directors Stock Option Plan that is not exercised before the day prior to the Effective Time, and any other S1 Stock Option that is outstanding as of immediately before the Effective Time will be terminated and canceled at the Effective Time, and the holder of each S1 Stock Option under the 2003 Plan and each S1 Stock Option that will have vested as of or prior to the Effective Time pursuant to the terms of the applicable S1 Stock Plan and/or related award agreement will, subject to any required tax withholding, be entitled to receive an amount in cash equal to the Option Consideration; provided, however, that if the Option Consideration is zero or a negative number as of the Effective Time, such S1 Stock Option will be canceled and no amount will be paid in respect thereof. ACI will pay or cause to be paid the Option Consideration to the holders of the S1 Stock Options in a lump sum as soon as practicable after the Effective Time but in no event later than five business days following the Effective Time.

At the Effective Time, each stock appreciation right granted under the applicable S1 Stock Plan (the "SARs") will be canceled at the Effective Time, and the holder of each SAR that will have vested as of or prior to the Effective Time pursuant to the applicable S1 Stock Plan will, subject to any required tax withholding, be entitled to receive an amount in cash equal to the SARS Consideration. ACI will pay or cause to be paid the applicable SARs Consideration to the holders of the SARs in a lump sum as soon as practicable after the Effective Time but in no event later than five business days following the Effective Time.

At the Effective Time, each outstanding restricted S1 Share, (other than certain restricted S1 Shares to be converted into restricted ACI Shares pursuant to the Transaction Agreement), restricted stock unit and restricted cash unit that will have vested as of or prior to the Effective Time pursuant to the applicable S1 Stock Plan will be treated as an outstanding fully vested S1 Share and will have the right to receive the Proration Amount of Cash and Stock Consideration.

Representations and Warranties of the Parties in the Transaction Agreement

In the Transaction Agreement, S1 has made customary representations and warranties to ACI and Offeror, including representations relating to, among other things:

- · S1's corporate organization, standing and power;
- · S1 subsidiaries;
- · corporate authority;
- S1's articles of organization and bylaws;
- capital structure;
- absence of conflicts with or consents required under third-party contracts in connection with the Transaction Agreement;
- · required government approvals, filings and consents;
- · SEC reports;
- · financial statements;
- · absence of undisclosed liabilities;
- · absence of certain changes or events;
- · material contracts:
- · intellectual property;
- · certain business practices;

- applicability of anti-takeover laws;
- tax matters;
- employee benefits plans;
- · labor and employment matters;
- absence of litigation;
- · compliance with laws;
- brokers;
- opinion of S1's financial advisor;
- stockholder vote required;
- · related party transactions;
- insurance; and
- · environmental matters.

In the Transaction Agreement, ACI and Offeror have made customary representations and warranties to S1, including representations relating to, among other things:

- corporate organization, standing and power;
- ACI subsidiaries;
- corporate authority;
- capital structure;
- $\bullet \quad \text{absence of conflicts with or consents required under third-party contracts in connection with the Transaction Agreement;}\\$
- required government approvals, filings and consents;
- SEC reports;
- · financial statements;
- absence of undisclosed liabilities;
- absence of certain changes or events;
- material contracts;
- intellectual property;
- · certain business practices;
- applicability of anti-takeover laws;
- tax matters;
- benefits matters;
- · labor and employment matters;
- absence of litigation;
- · compliance with laws;
- brokers;
- issuance of ACI Shares;

- funds;
- related party transactions;
- insurance;
- · environmental matters; and
- · the Exchange Offer documents.

Conduct of Business Pending the Closing of the Second-Step Merger

The Transaction Agreement provides that from the date of the Transaction Agreement until completion of the Second-Step Merger, S1 will, and cause its subsidiaries to, conduct its business in the ordinary course consistent with past practice. Subject to certain exceptions, the Transaction Agreement provides that during the same time period, S1 and/or its subsidiaries are subject to customary operating covenants and restrictions, including restrictions on:

- · declaring, setting aside or paying any dividends;
- making any issuance, grant, conferral, award, delivery, sale, pledge, disposal, or encumbrance of S1 securities;
- amending of the articles of organization and by-laws of S1;
- making or offering to make, by merger or otherwise, acquisitions of any business, assets, or securities in excess of \$2 million;
- granting or announcing any incentive awards or any increase in compensation, severance or termination pay to any employee, officer, director of other S1 service provider, other than persons with an annual base salary less than \$225,000;
- · hiring new employees or officers;
- establishing, adopting, entering into, amending, modifying or terminating any collective bargaining agreement of S1 employee benefit plan;
- taking any action to accelerate any rights or benefits of any person;
- making any change in accounting methods;
- selling, leasing, licensing or otherwise disposing of or subjecting to any Lien any material properties or assets;
- incurring any indebtedness for borrowed money or guaranteeing any such indebtedness of another person;
- making or agreeing to make any new capital expenditure;
- encumbering, impairing, abandoning, failing to maintain, transferring, licensing or otherwise disposing of any right, title or interest in S1 intellectual property;
- divulging, furnishing to or making accessible any material confidential information in which S1 has trade secret or equivalent rights;
- making or changing any material tax election;
- waiving, releasing, assigning, settling or compromising any action, action or proceeding involving the payment of monetary damages in excess of \$1 million in the aggregate;
- entering into any new line of business outside of S1's existing business;

- taking any action that is intended to result in any of its representations or warranties set forth in the Transaction Agreement being or becoming untrue in any material respect such that the closing conditions cannot be fulfilled; or
- · authorizing any of, or committing or agreeing to take any of, the foregoing actions.

Non-Solicitation by S1 of S1 Acquisition Proposals; Board Recommendation

The Transaction Agreement provides that until the earlier of the Acceptance Time or the receipt of approval of the Second-Step Merger by the S1 stockholders, S1 will not, and will cause its subsidiaries not to and will use reasonable best efforts to cause its representatives not to, directly or indirectly:

- initiate, solicit, knowingly encourage (including by way of furnishing non-public information), or take any other action designed to lead to, any inquiries or the making of any proposal that constitutes, or would reasonably be expected to lead to, the submission of any S1 Acquisition Proposal (as defined below) or engage, enter into, continue or participate in any negotiations or discussions with respect thereto or furnish any non-public information concerning S1 and its subsidiaries to any person in connection with any S1 Acquisition Proposal;
- except for a confidentiality agreement contemplated by the Transaction Agreement, enter into any Transaction Agreement, letter of intent, agreement in principle, share purchase agreement, asset purchase agreement or share exchange agreement, option agreement or other similar agreement relating to an S1 Acquisition Proposal (an "S1 Acquisition Agreement"); or
- withdraw, modify or qualify, or propose publicly to withdraw, modify or qualify, in a manner adverse to ACI or Offeror, the recommendation of the S1 Board that the S1 stockholders tender their S1 Shares in the Exchange Offer (the "S1 Board Recommendation") or publicly recommend the approval or adoption of, or publicly approve or adopt, or propose to publicly recommend, approve or adopt, any S1 Acquisition Proposal (any action described in this bullet being referred to as a "S1 Change of Recommendation").

In the Transaction Agreement, S1 agreed to cease and cause to be terminated any solicitation, discussion or negotiation with any person conducted prior to the date of the Transaction Agreement by S1, its subsidiaries or any of its representatives with respect to any S1 Acquisition Proposal and to request the return or destruction of all confidential information provided by or on behalf of S1 or its subsidiaries to any such person.

Notwithstanding the foregoing restrictions, the Transaction Agreement provides that at any time following the date of the Transaction Agreement and prior to the earlier of the Acceptance Time or the receipt of approval of the Second-Step Merger by the S1 stockholders, if S1 receives a written S1 Acquisition Proposal that the S1 Board believes in good faith is bona fide, and the S1 Board, after consultation with its financial advisors and outside legal counsel, determines in good faith that such S1 Acquisition Proposal constitutes or would reasonably be expected to lead to or result in an S1 Superior Offer (as defined below) and such S1 Acquisition Proposal did not result from a material breach of S1's obligations with respect to S1 Acquisition Proposals, then S1 may, subject to certain restrictions, (1) furnish information with respect to S1 and its subsidiaries to the person making such S1 Acquisition Proposal and (2) participate in discussions or negotiations regarding such S1 Acquisition Proposal.

If S1 receives a written S1 Acquisition Proposal that the S1 Board believes in good faith is bona fide, and the S1 Board, after consultation with its financial advisors and outside legal counsel, concludes in good faith such S1 Acquisition Proposal constitutes an S1 Superior Offer, the S1 Board may, if it determines in good faith, after consultation with outside counsel, that the failure to take such action would be inconsistent with its fiduciary duties under applicable law and, subject to payment of a \$19.14 million termination fee to ACI and to certain restrictions, effect a Change of Board Recommendation

The S1 Board may not effect an S1 Change of Recommendation or terminate the Transaction Agreement unless S1 has provided prior written notice to ACI specifying in reasonable detail the reasons for such action (including a description of the material terms of such S1 Acquisition Proposal and delivering to ACI a copy of

the S1 Acquisition Agreement and other relevant documents for such S1 Superior Offer in the form to be entered into), at least five business days in advance of its taking such action with respect to such S1 Superior Offer (or if there are less than three business days prior to the then scheduled expiration date, such written notice will be provided as far in advance of the then scheduled expiration date as practicable) (the "Notice Period").

During the Notice Period, S1 must negotiate with ACI in good faith and take into account all changes to the terms of the Transaction Agreement proposed by ACI in determining whether such S1 Acquisition Proposal continues to constitute an S1 Superior Offer. Any amendment, in any material respect, to the terms of such S1 Superior Offer will require a new notice to ACI and a two business day extension of the Notice Period. After delivery of such written notice, S1 must keep ACI informed of all material developments affecting the material terms of any such S1 Superior Offer (and S1 will provide ACI with copies of any additional written materials received that relate to such S1 Superior Offers).

As used in the Transaction Agreement, "S1 Acquisition Proposal" means any offer, proposal or indication of interest received from a third party (other than a party to the Transaction Agreement) providing for any S1 Acquisition Transaction, including any renewal or revision to such a previously made offer, proposal or indication of interest.

As used in the Transaction Agreement, "S1 Acquisition Transaction" means any transaction or series of transactions involving: (1) any merger, consolidation, share exchange, recapitalization, business combination or similar transaction involving S1 or any of its Subsidiaries; (2) any direct or indirect acquisition of securities, tender offer, exchange offer or other similar transaction in which a Person or "group" (as defined in the Exchange Act) of persons directly or indirectly acquires beneficial or record ownership of securities representing 20% or more of the outstanding S1 Shares; (3) any direct or indirect acquisition of any business or businesses or of assets that constitute or account for 20% or more of the consolidated net revenues, net income or assets of S1 and its subsidiaries, taken as a whole (based on the fair market value thereof); (4) any liquidation or dissolution of S1 or any material subsidiary of S1; or (5) any combination of the foregoing (in each case, other than any of the transactions contemplated by the Transaction Agreement).

As used in the Transaction Agreement, "S1 Superior Offer" means a bona fide written S1 Acquisition Proposal (for purposes of this definition, replacing all references in such definition to 20% with 50%) that the S1 Board or any committee thereof determines, in good faith, after consultation with outside legal counsel and a financial advisor (1) is on terms that are more favorable from a financial point of view to the S1 stockholders than the transactions contemplated by the Transaction Agreement (including any proposal by ACI to amend the terms of the Transaction Agreement) after taking into account all of the terms and conditions of such proposal and (2) is likely to be completed (without material modification of its terms), in each of the cases of clause (1) and (2) taking into account all financial, regulatory, legal and other aspects of such S1 Acquisition Proposal (including the timing and likelihood of consummation thereof) and the payment, if any, of the S1 termination fee.

Employee Matters

The Transaction Agreement provides that for the period of one year following the Effective Time, individuals employed by S1 immediately prior to the Effective Time and that continue to be employed by ACI or any of its subsidiaries, including S1 (the "Post-Merger Employees"), will be (1) provided salaries and benefits that are in the aggregate approximately equal to the salaries and benefits (other than equity compensation) they received prior to the Effective Time and (2) given credit for all service with S1 and its subsidiaries and their respective predecessors under any employee benefit plan of ACI, the surviving corporation or any of their subsidiaries, including any such plans providing vacation, sick pay, severance and retirement benefits maintained by ACI or its subsidiaries in which such Post-Merger Employees of eligibility, vesting and entitlement to benefits, including for severance benefits and vacation entitlement (but not for accrual of pension benefits), to the extent past service was recognized for such Post-Merger Employees under the comparable S1 employee plans immediately prior to the Effective Time.

In the event of any change in the welfare benefits provided to Post-Merger Employees following the Effective Time, it is ACI's and S1's intention that ACI will use its reasonable best efforts to cause (1) the

waiver of all limitations as to pre-existing conditions, exclusions and waiting periods with respect to participation and coverage requirements applicable to the Post-Merger Employees (and their eligible dependents) under any welfare benefit plans in which Post-Merger Employees participate following the Effective Time, to the extent that such conditions, exclusions or waiting periods would not apply in the absence of such change, and (2) for the plan year in which the Effective Time occurs, the crediting of each Post-Merger Employee (or his or her eligible dependents) with any co-payments and deductibles paid prior to any such change in satisfying any applicable deductible or out-of-pocket requirements after such change.

Director's and Officer's Indemnification and Insurance

The Transaction Agreement provides that ACI and the surviving corporation will indemnify and hold harmless all current and former directors, officers and employees, as the case may be, of S1 and its subsidiaries to the fullest extent permitted by law for acts or omissions occurring prior to the Effective Time (including acts or omissions occurring in connection with the approval of the Transaction Agreement and the consummation of the transactions contemplated thereto) in their capacities as such. ACI will, and will cause the surviving corporation to, fulfill and honor in all respects the obligations pursuant to any indemnification agreements between S1 or its subsidiaries, on the one hand, and any current or former directors, officers and employees, as the case may be, of S1 and its subsidiaries, on the other hand, in effect immediately prior to October 3, 2011 or, subject to the prior approval of ACI, after October 3, 2011, and any indemnification provisions under the S1 certificate of incorporation or by-laws or the comparable charter or organizational documents of any of S1's subsidiaries as in effect on October 3, 2011, in each case to the maximum extent permitted by law, and will not amend, repeal or otherwise modify any such provision in any manner that would adversely affect the rights of such indemnitee thereunder for any acts or omissions occurring prior to the Effective Time.

Prior to the Effective Time, S1 will endeavor to enter into a directors' and officers' liability insurance policy covering those persons who, as of immediately prior to the Effective Time, are covered by S1's directors' and officers' liability insurance policy (the "Insured Parties") on terms no less favorable to the Insured Parties than those of S1's present directors' and officers' liability insurance policy (such policy, an "S1 D&O Policy"), for a period of seven years after the Effective Time. If S1 is unable to obtain such an S1 D&O Policy prior to the Effective Time, ACI will cause the surviving corporation to maintain in effect, for a period of seven years after the Closing Date, an S1 D&O Policy with a creditworthy issuer; provided, however, that in no event will ACI be required to expend annually more than 250% of the annual premium currently paid by S1 for such coverage and, if the cost for such coverage is in excess of such amount, ACI will be required only to maintain the maximum amount of coverage as is reasonably available for 250% of such annual premium.

Regulatory Filings; Efforts to Close

The Transaction Agreement provides that prior to the Effective Time, each of ACI, Offeror and S1 will use reasonable best efforts to take, or cause to be taken, all actions and to do, or cause to be done, all things necessary, proper or advisable under applicable laws to consummate the Exchange Offer, the Second-Step Merger and the other transactions contemplated by the Transaction Agreement, including to obtain any clearance required under the antitrust laws.

Each of ACI and S1 agree to:

- use its reasonable best efforts to obtain promptly (and in any event no later than July 31, 2012) any clearance required under the HSR Act and any other antitrust laws for the consummation of the transactions pursuant to the Transaction Agreement;
- use its reasonable best efforts to avoid or eliminate any impediment under any antitrust law, or regulation or rule, that may be asserted by any Governmental Authority, or any other person, with respect to the transactions pursuant to the Transaction Agreement so as to enable the Effective Time to occur expeditiously (and in any event no later than July 31, 2012);

- use its reasonable best efforts to defend through any legal or other proceeding by or before any Governmental Authority with respect to the transactions contemplated by the Transaction Agreement ("Agency Litigation") or, if applicable, other litigation on the merits any claim asserted in any court, administrative tribunal or hearing that the Transactions would violate any Law, or any regulation or rule of any Governmental Authority, in order to avoid entry of, or to have vacated or terminated, any Injunction;
- cause its respective inside and outside counsel to cooperate in good faith with counsel and other representatives of each other party and use its reasonable best efforts to facilitate and expedite the identification and resolution of any such issues and, consequently, the expiration of the applicable HSR Act waiting period and the waiting periods under any other antitrust laws at the earliest practicable dates (and in any event no later than July 31, 2012), such reasonable best efforts and cooperation to include causing their respective inside and outside counsel (1) to keep each other appropriately informed on a current basis of communications from and to personnel of the reviewing antitrust authority and (2) to confer on a current basis with each other regarding appropriate contacts with and response to personnel of such antitrust authority;
- use its reasonable best efforts to cause the conditions set forth in the Transaction Agreement to be satisfied on a timely basis; and
- prior to the Acceptance Time, ACI will not acquire any business involving annual revenues in excess of \$25.0 million unless advised by counsel that in such counsel's opinion so doing would not significantly increase the risk of an Injunction or materially delay any antitrust or governmental approval.

ACI's covenants under the Transaction Agreement include ACI being required under the Transaction Agreement to offer to the DOJ that it or its subsidiaries take, and, if such offer is accepted by the DOJ, use its best efforts to eliminate any DOJ Impediment. For this purpose, ACI's "best efforts to eliminate any DOJ Impediment" as set forth in the immediately preceding sentence will require that ACI use its best efforts to effect such of the following as may be necessary to avoid a DOJ Impediment: (1) the sale, holding separate, licensing, modifying or otherwise disposing of all or any portion of the business, assets or properties of S1 or its subsidiaries, whether located in or outside the United States; (2) ACI conducting or limiting the conduct of the business, assets or properties of S1 or its subsidiaries, whether located in or outside the United States, in a specified manner; or (3) S1 or its subsidiaries' entry with the DOJ into any agreement, settlement, order, other relief or action of a type referred to in clause (2).

You should be aware that all required regulatory approvals may not be obtained in a timely manner, and this could result in a delay in the completion of the Exchange Offer. ACI has twice withdrawn and refiled its HSR Act filing prior to the date of this prospectus/offer to exchange in an effort to convince the DOJ staff of ACI's view as to the competitive nature of payment systems marketplace, and there can be no assurance that the DOJ will concur with its belief that the transaction should be permitted to close. If ACI again withdraws and refiles its HSR Act filing, the DOJ issues a request for additional information or documentary material or the DOJ institutes an action challenging the transaction, the Expiration Time would be extended and the completion of the Exchange Offer could be prevented.

Defense of Litigation

In the event that any administrative or judicial action or proceeding (including Agency Litigation) is instituted (or threatened to be instituted) by a Governmental Authority or private party challenging the transactions contemplated by the Transaction Agreement, ACI and S1 agree that they will cooperate in all respects with each other and use their respective commercially reasonable efforts to contest and resist any such action or proceeding and to have vacated, lifted, reversed or overturned any decree, judgment, injunction or other order, whether temporary, preliminary or permanent, that is in effect and that prohibits, prevents or restricts consummation of the transactions contemplated by the Transaction Agreement or delays the Effective Time past July 31, 2012 (collectively, an "Injunction"); provided, however, that (1) ACI will be entitled to direct the defense of any legal, administrative or judicial action or proceeding in respect of the transactions contemplated by the Transaction Agreement, or negotiations with, any Governmental Authority or other person

relating thereto, or regulatory filings under applicable antitrust laws and (2) S1 will not make any offer, acceptance or counter-offer to or otherwise engage in negotiations or discussions with any Governmental Authority with respect to any proposed settlement, stay, toll, extension of any waiting period, consent decree, commitment or remedy, or, in the event of litigation, discovery, admissibility of evidence, timing or scheduling, except as specifically requested by or agreed with ACI or its counsel.

Other Agreements of the Parties

NASDAO Listino

ACI will use its reasonable best efforts to cause the ACI Shares to be issued to S1 stockholders in the Exchange Offer and the Second-Step Merger to be authorized for listing on the NASDAQ Global Select Market, subject to official notice of issuance.

Stockholder Litiaation

Each of ACI and S1 has agreed to promptly advise the other party orally and in writing of any litigation commenced or threatened in writing to be commenced by any stockholder of such party against such party and/or any of its directors relating to the transactions contemplated by the Transaction Agreement, the transactions with Fundtech and/or the Fundtech Merger Agreement, and will keep the other party fully informed regarding any such litigation. Each of ACI and S1 has agreed to give the other party the opportunity to participate in, subject to a customary joint defense agreement, the defense or settlement of any such litigation, will give due consideration to the other party's advice with respect to such litigation and will not settle any such litigation without the prior written consent of the other party (not to be unreasonably withheld, conditioned or delayed).

Financing

The Transaction Agreement provides that ACI will take, or cause to be taken, all actions and to do, or cause to be done, all things reasonably necessary to obtain the financing pursuant to the Facility (as defined below at "The Exchange Offer — Source and Amount of Funds") (the "Financing"). If any portion of the Financing becomes unavailable on the terms and conditions (including the flex provisions) contemplated in the Facility Commitment letter, ACI will use its reasonable best efforts to arrange and obtain alternative financing from alternative sources in an amount sufficient to consummate the transactions contemplated by the Transaction Agreement as promptly as practicable following the occurrence of such event but no later than the business day immediately prior to the closing date of the Transaction Agreement.

Conditions to Closing of the Second-Step Merger

The Transaction Agreement provides that the respective obligations of ACI and Offeror and S1 are subject to the satisfaction at or prior to the Effective Time each of the following:

- this prospectus/offer to exchange has been declared effective by the SEC;
- if required by applicable law, the S1 stockholder approval has been obtained;
- · if the Acceptance Time has occurred, Offeror has purchased all S1 Shares validly tendered and not withdrawn pursuant to the Exchange Offer;
- no order, injunction or decree issued by any Governmental Authority of competent jurisdiction preventing the consummation of the Second-Step Merger will be in effect, and no statute, rule, regulation, order, injunction or decree will have been enacted, entered, promulgated or enforced (and still be in effect) by any governmental entity that prohibits or makes illegal consummation of the Second-Step Merger; and

· the HSR Condition has been satisfied.

Termination of the Transaction Agreement

The Transaction Agreement may be terminated and the Exchange Offer and the Second-Step Merger may be abandoned:

- · by mutual written consent of ACI and S1;
- by either ACI or S1 at any time prior to the Effective Time if the Acceptance Time will not have occurred on or prior to the close of business on July 31, 2012;
- by either ACI or S1 at any time prior to the Effective Time if a Governmental Authority will have enacted, issued, promulgated, enforced or entered any law (including an injunction or other order) or taken any other action, in any case having the effect of permanently restraining, enjoining or otherwise prohibiting the Exchange Offer or the Second-Step Merger, which law (including any such injunction or other order) or other action will have become final and nonappealable;
- by either ACI or S1 at any time prior to the Effective Time if the Exchange Offer will have expired or been terminated without any S1 Shares being purchased therein as a result of the failure to satisfy the Minimum Tender Condition;
- by ACI at any time prior to the Acceptance Time, if: (1) an S1 Change of Recommendation will have occurred; (2) S1 will have delivered a notice to ACI of its intent to effect an S1 Change of Recommendation; or (3) following the request in writing by ACI, the S1 Board will have failed to reaffirm publicly S1 Recommendation within five business days after ACI requests in writing that such recommendation be reaffirmed publicly;
- by ACI at any time prior to the Acceptance Time if there will have been a breach by S1 of any of its representations, warranties, covenants or obligations contained in the Transaction Agreement, which breach would result in the failure to satisfy by July 31, 2012 one or more of the conditions to the Exchange Offer, and in any such case such breach will be incapable of being cured or, if capable of being cured, will not have been cured within 30 days after written notice thereof will have been received by S1 of such breach;
- by S1 at any time prior to the Acceptance Time if (1) Offeror fails to amend the Exchange Offer to give effect to the terms of the Transaction Agreement or (2) there will have been a breach by
 ACI or Offeror of any of its representations, warranties, covenants or obligations contained in the Transaction Agreement, which breach would result in the failure to satisfy by July 31, 2012 one
 or more of the conditions to the Exchange Offer, and in any such case such breach will be incapable of being cured or, if capable of being cured, will not have been cured within 30 days after
 written notice thereof will have been received by ACI or Offeror of such breach; or
- S1 effects an S1 Change of Recommendation to accept an S1 Acquisition Proposal.

Termination Fees

The Transaction Agreement contemplates that a termination fee of \$19.14 million will be payable by S1 to ACI if:

• the Transaction Agreement is terminated by ACI within five business days of an S1 Change of Recommendation pursuant to the fifth bullet under "The Transaction Agreement — Termination of the Transaction Agreement" above;

- the Transaction Agreement is terminated by S1 pursuant to the eighth bullet under "The Transaction Agreement Termination of the Transaction Agreement" above by effecting an S1 Change of Recommendation to accept an S1 Acquisition Proposal; and
- the Transaction Agreement is terminated by (1) ACI pursuant to the sixth bullet under "The Transaction Agreement Termination of the Transaction Agreement" above, (2) S1 or ACI pursuant to the fourth bullet under "The Transaction Agreement Termination of the Transaction Agreement" above, or (3) by S1 or ACI pursuant to the second bullet under "The Transaction Agreement Termination of the Transaction Agreement" above and (a) an S1 Acquisition Proposal is publicly announced or otherwise communicated to the S1 Board after October 3, 2011 and prior to the Acceptance Time, in the case of clause (2), or the date of termination, in the case of clauses (1) or (3), and (b) if within 12 months after the date of such termination, S1 enters into a definitive S1 Acquisition Agreement to consummate, or consummates, any S1 Acquisition Transaction.

Amendment

The Transaction Agreement may be amended and any provision thereto may be waived by the parties at any time prior to the Effective Time, provided that after the S1 stockholder approval, if such amendment or waiver requires the further approval of the S1 stockholders pursuant to applicable law or in accordance with the rules and regulations of NASDAQ, the effectiveness of such amendment or waiver will be subject to the approval of the S1 stockholders.

Remedies; Specific Performance

The parties have agreed that irreparable damage may occur for which monetary damages would not be an adequate remedy in the event that any of the provisions of the Transaction Agreement is not performed in accordance with their specific terms or are otherwise breached. Therefore, the parties have agreed that, if for any reason ACI, Offeror or S1 has failed to perform its obligations under the Transaction Agreement, then the party seeking to enforce the Transaction Agreement against such nonperforming party under the Transaction Agreement will be entitled to specific performance and the issuance of injunctive and other equitable relief.

THE EXCHANGE OFFER

Overview

Offeror is offering to exchange for each outstanding S1 Share that is validly tendered and not properly withdrawn prior to the Expiration Time, either of the following:

- · 0.3148 of an ACI Share (Stock Consideration); or
- \$10.00 in cash, without interest (Cash Consideration),

subject to the proration procedures described in this prospectus/offer to exchange and the related letter of election and transmittal, upon the terms and subject to the conditions contained in this prospectus/offer to exchange and the accompanying letter of election and transmittal. In addition, you will receive cash in lieu of any fractional ACI Share to which you may be entitled.

The term "Expiration Time" means 5:00 p.m., Eastern time, on Monday, October 31, 2011, unless Offeror extends the period of time for which the Exchange Offer is open, in which case the term "Expiration Time" means the latest time and date on which the Exchange Offer, as so extended, expires.

The Exchange Offer is subject to conditions which are described in the section of this prospectus/offer to exchange titled "The Exchange Offer — Conditions of the Exchange Offer." ACI expressly reserves the right, subject to the applicable rules and regulations of the SEC, to waive any condition of the Exchange Offer described herein in its discretion, except that the Minimum Tender Condition cannot be waived without S1's consent. Offeror may not make any changes to the conditions of the Exchange Offer or the offer price without S1's consent (subject to any obligation to extend the Exchange Offer pursuant to the applicable rules and regulations of the SEC).

If you are the record owner of your S1 Shares and you tender your S1 Shares in the Exchange Offer, you will not have to pay any brokerage fees or similar expenses. If you own your S1 Shares through a broker, dealer, commercial bank, trust company or other nominee tenders your S1 Shares on your behalf, your broker or such other nominee may charge a fee for doing so. You should consult your broker, dealer, commercial bank, trust company or other nominee to determine whether any charges will apply.

ACI, through Offeror, intends, promptly following acceptance for exchange and exchange of \$1\$ Shares in the Exchange Offer, to effect the Second-Step Merger in accordance with Delaware law pursuant to which Offeror will acquire all \$1\$ Shares of those \$1\$ stockholders who choose not to tender their \$1\$ Shares pursuant to the Exchange Offer. After the Second-Step Merger, former remaining \$1\$ stockholders will no longer have any ownership interest in \$1\$ and will be stockholders of ACI to the extent they receive any Stock Consideration in this Exchange Offer, and ACI, through Offeror, will own all of the issued and outstanding \$1\$ Shares.

Please see the sections of this prospectus/offer to exchange titled "The Exchange Offer — Purpose and Structure of the Exchange Offer"; "The Exchange Offer — Second-Step Merger"; and "The Exchange Offer — Plans for \$1."

Based on ACI's and S1's respective capitalizations as of October 12, 2011 and the estimated 5.9 million ACI Shares estimated to be issued in the Exchange Offer and the Second-Step Merger, former S1 stockholders would own, in the aggregate, approximately 14.4% of the aggregate ACI Shares on a fully diluted basis. For a detailed discussion of the assumptions on which this estimate is based, please see the section of this prospectus/offer to exchange titled "The Exchange Offer — Ownership of ACI After the Exchange Offer."

Expiration Time of the Exchange Offer

The Exchange Offer is scheduled to expire at 5:00 p.m., Eastern time, on Monday, October 31, 2011, which is the Expiration Time, unless further extended by Offeror. For more information, you should read the discussion below in the section of this prospectus/offer to exchange titled "The Exchange Offer — Extension, Termination, Waiver and Amendment."

Extension, Termination and Amendment

Subject to the applicable rules of the SEC and the terms and conditions of the Exchange Offer, Offeror expressly reserves the right (but shall not be obligated) at any time or from time to time in its sole discretion to waive any Exchange Offer condition or modify or amend the terms of the Exchange Offer, except that, without the prior written consent of \$1, (1) the Minimum Tender Condition may not be amended or waived and (2) no change may be made to the Exchange Offer offer price or changes the form of consideration, (b) decreases the number of \$1 Shares to be purchased by Offeror in the Exchange Offer, (c) modifies the Exchange Offer or the Exchange Offer or annear that adversely affects or reasonably could adversely affect the \$1 stockholders, (d) adds to the Exchange Offer conditions, or (e) extends the Expiration Time of the Exchange Offer except as required or permitted by the Transaction Agreement.

The Expiration Time may also be subject to multiple extensions and any decision to extend the Exchange Offer, and if so, for how long, will be made prior to the Expiration Time.

Any such extension, delay, termination, waiver or amendment will be followed as promptly as practicable by public announcement thereof, which, in the case of an extension, will be made no later than 9:00 a.m., Eastern time, on the next business day after the previously scheduled Expiration Time, Subject to applicable law (including Rules 14d-4(d)(i), 14d-6(c) and 14e-1 under Section 14A of the Securities Exchange Act of 1934, as amended (the "Exchange Act"), which requires that material changes be promptly disseminated to S1 stockholders in a manner reasonably designed to inform them of such changes), and without limiting the manner in which Offeror may choose to make any public announcement, Offeror will have no obligation to publish, advertise or otherwise communicate any such public announcement there than by issuing a press release or other announcement.

Rule 14e-1(c) under the Exchange Act requires Offeror to pay the consideration offered or return the S1 Shares tendered promptly after the termination or withdrawal of the Exchange Offer.

If ACI increases or decreases the percentage of S1 Shares being sought or the consideration offered in the Exchange Offer and the Exchange Offer is scheduled to expire at any time before the expiration of 10 business days from, and including, the date that notice of such increase or decrease is first published, sent or given in the manner specified below, the Exchange Offer will be extended until at least the expiration of 10 business days from, and including, the date of such notice. If Offeror makes a material change in the terms of the Exchange Offer (other than a change in the consideration offered in the Exchange Offer or the percentage of securities sought) or in the information concerning the Exchange Offer, or waives a material condition of the Exchange Offer, Offeror will extend the Exchange Offer, if required by applicable law, for a period sufficient to allow S1 stockholders to consider the amended terms of the Exchange Offer. In a published release, the SEC has stated its view that an offer must remain open for a minimum period of time following a material change in the terms of such offer, and that the waiver of a condition such as the condition described in the section of this prospectus/offer to exchange titled "The Exchange Offer." Conditions of the Exchange Offer." under the subheading "Minimum Tender Condition" is a material change in the terms of an offer. The release states that an offer should remain open for a minimum of five business days from the date that the material change is first published, sent or given to S1 stockholders, and that if material changes are made with respect to information that approaches the significance of the price to be paid in the Exchange Offer or the percentage of shares sought in the Exchange Offer, a minimum of 10 business days may be required to allow adequate dissemination and investor response.

As used in this prospectus/offer to exchange, a "business day" means any day, other than a Saturday, Sunday or a federal holiday, and shall consist of the time period from 12:01 a.m. through 12:00 midnight, Eastern time. If, prior to the Expiration Time, ACI increases the consideration being paid for S1 Shares accepted for exchange pursuant to the Exchange Offer, such increased consideration will be received by all S1 stockholders whose S1 Shares are exchanged pursuant to the Exchange Offer, whether or not such S1 Shares were tendered prior to the announcement of the increase of such consideration.

Pursuant to Rule 14d-11 under the Exchange Act, Offeror may, subject to certain conditions, elect to provide a subsequent offering period of at least three business days following the Expiration Time on the date

of the Expiration Time and acceptance for exchange of the S1 Shares tendered in the Exchange Offer. A subsequent offering period would be an additional period of time, following the first exchange of S1 Shares in the Exchange Offer, during which stockholders could tender S1 Shares not tendered in the Exchange Offer.

During a subsequent offering period, tendering S1 stockholders would not have withdrawal rights and Offeror would promptly exchange and pay for any S1 Shares tendered at the same price paid in the Exchange Offer. Rule 14d-11 under the Exchange Act provides that Offeror may provide a subsequent offering period so long as, among other things, (1) the initial period of a least 20 business days of the Exchange Offer has expired, (2) Offeror offers the same form and amount of consideration for S1 Shares in the subsequent offering period as in the initial offer, (3) Offeror immediately accepts and promptly pays for all S1 Shares tendered prior to the Expiration Time, (4) ACI announces the results of the Exchange Offer, including the approximate number and percentage of S1 Shares deposited in the Exchange Offer, no later than 9:00 a.m., Eastern time, on the next business day after the Expiration Time and immediately begins the subsequent offering period, and (5) Offeror immediately accepts and promptly pays for S1 Shares as they are tendered during the subsequent offering period. If Offeror elects to include a subsequent offering period, it will notify S1 stockholders by making a public announcement on the next business day after the Expiration Time consistent with the requirements of Rule 14d-11 under the Exchange Act.

Pursuant to Rule 14d-7(a)(2) under the Exchange Act, no withdrawal rights apply to S1 Shares tendered during a subsequent offering period and no withdrawal rights apply during a subsequent offering period with respect to S1 Shares tendered in the Exchange Offer and accepted for exchange. The same consideration will be received by S1 stockholders tendering S1 Shares in the Exchange Offer or in a subsequent offering period, if one is included. Please see the section of this prospectus/offer to exchange titled "The Exchange Offer — Withdrawal Rights."

This prospectus/offer to exchange, the letter of election and transmittal and all other relevant materials will be mailed by ACI to record holders of S1 Shares and will be furnished to brokers, dealers, banks, trust companies and similar persons whose names, or the names of whose nominees, appear on S1's stockholders lists, or, if applicable, who are listed as participants in a clearing agency's security position listing for subsequent transmittal to beneficial owners of S1 Shares on or about October 12, 2011.

Acceptance for Exchange and Exchange of S1 Shares; Delivery of Exchange Offer Consideration

Upon the terms and subject to the conditions of the Exchange Offer (including, if the Exchange Offer is extended or amended, the terms and conditions of any such extension or amendment), Offeror will accept for exchange promptly after the Expiration Time all S1 Shares validly tendered (and not withdrawn in accordance with the procedure set out in the section of this prospectus/offer to exchange titled "The Exchange Offer — Withdrawal Rights") prior to the Expiration Time. Offeror will exchange all S1 Shares validly tendered and not withdrawn promptly following the acceptance of S1 Shares for exchange pursuant to the Exchange Offer. Offeror expressly reserves the right, in its discretion, but subject to the applicable rules of the SEC, to delay acceptance for and thereby delay exchange of S1 Shares in order to comply in whole or in part with applicable laws or if any of the conditions referred to in the section of this prospectus/offer to exchange itled "The Exchange Offer — Conditions of the Exchange Offer" have not been satisfied or if any event specified in the section of the prospectus/offer to exchange titled "The Exchange Offer" under the subheading "Other Conditions" has occurred. If Offeror decides to include a subsequent offering period, Offeror will accept for exchange, all validly tendered S1 Shares as they are received during the subsequent offering period. Please see the section of this prospectus/offer to exchange titled "The Exchange Offer — Withdrawal Rights."

In all cases (including during any subsequent offering period), Offeror will exchange all S1 Shares tendered and accepted for exchange pursuant to the Exchange Offer only after timely receipt by the exchange agent of (1) the certificates evidencing such S1 Shares or timely confirmation (a "Book-Entry Confirmation") of a book-entry transfer of such S1 Shares into the exchange agent's account at The Depository Trust Company pursuant to the procedures set forth in the section of this prospectus/offer to exchange titled "The Exchange Offer — Procedure for Tendering," (2) the letter of election and transmittal (or a manually signed facsimile

thereof), properly completed and duly executed, with any required signature guarantees, or, in the case of a book-entry transfer, an Agent's Message (as defined below), and (3) any other documents required under the letter of election and transmittal. This prospectus/offer to exchange refers to The Depository Trust Company as the "Book-Entry Transfer Facility." As used in this prospectus/offer to exchange, the term "Agent's Message" means a message, transmitted by the Book-Entry Transfer Facility to, and received by, the exchange agent and forming a part of the Book-Entry Confirmation which states that the Book-Entry Transfer Facility has received an express acknowledgment from the participant in the Book-Entry Transfer Facility tendering the \$1\$ Shares that are the subject of such Book-Entry Confirmation, that such participant has received and agrees to be bound by the letter of election and transmittal and that ACI may enforce such agreement against such participant.

For purposes of the Exchange Offer (including during any subsequent offering period), Offeror will be deemed to have accepted for exchange, and thereby exchanged, S1 Shares validly tendered and not properly withdrawn as, if and when Offeror gives oral or written notice to the exchange agent of Offeror's acceptance for exchange of such S1 Shares pursuant to the Exchange Offer. Upon the terms and subject to the conditions of the Exchange Offer, exchange of S1 Shares accepted for exchange pursuant to the Exchange Offer will be made by deposit of the Exchange Offer consideration being exchanged therefor with the exchange agent, which will act as agent for tendering S1 stockholders for the purpose of receiving the Exchange Offer consideration from Offeror and transmitting such consideration to tendering S1 stockholders whose S1 Shares have been accepted for exchange.

Under no circumstances will Offeror pay interest on the Exchange Offer consideration for S1 Shares, regardless of any extension of the Exchange Offer or other delay in making such exchange or distributing the Exchange Offer consideration.

If any tendered S1 Shares are not accepted for exchange for any reason pursuant to the terms and conditions of the Exchange Offer, or if certificates representing such S1 Shares are submitted evidencing more S1 Shares than are tendered, certificates evidencing unexchanged or untendered S1 Shares will be returned, without expense to the tendering S1 stockholder (or, in the case of S1 Shares tendered by book-entry transfer into the exchange agent's account at a Book-Entry Transfer Facility pursuant to the procedure set forth in the section of this prospectus/offer to exchange offer — Procedure for Tendering," such S1 Shares will be credited to an account maintained at such Book-Entry Transfer Facility), as promptly as practicable following the expiration or termination of the Exchange Offer. ACI reserves the right to transfer or assign, in whole or from time to time in part, to one or more of its affiliates, the right to exchange all or any portion of the S1 Shares tendered pursuant to the Exchange Offer, but any such transfer or assignment will not relieve Offeror of its obligations under the Exchange Offer or prejudice the rights of tendering stockholders to exchange S1 Shares validly tendered and accepted for exchange pursuant to the Exchange Offer.

Cash In Lieu of Fractional ACI Shares

ACI will not issue certificates representing fractional ACI Shares pursuant to the Exchange Offer. Instead, each tendering S1 stockholder who would otherwise be entitled to a fractional ACI Share will receive cash (rounded to the nearest whole cent) in an amount (without interest) equal to the product of (1) such fraction, multiplied by (2) the volume weighted average sales price per share of ACI Shares for the ten consecutive days that ACI Shares have traded ending on and including the second clear trading day immediately prior to the Acceptance Time as reported on the NASDAQ.

Flections and Proration

S1 stockholders electing either the Cash Consideration or the Stock Consideration will be subject to proration so that not more than 66.2% of S1 Shares will be exchanged for the Cash Consideration and 33.8% of S1 Shares will be exchanged for the Stock Consideration in the Exchange Offer. S1 stockholders who do not participate in the Exchange Offer and whose shares are acquired in the Second-Step Merger will receive the Proration Amount of Cash and Stock Consideration. The elections of other S1 stockholders will affect whether a tendering S1 stockholder electing the Cash Consideration or the Stock Consideration receives solely

the type of consideration elected or if a portion of such S1 stockholder's tendered S1 Shares is exchanged for another form of consideration. S1 stockholders who otherwise would be entitled to receive a fractional ACI Share will instead receive cash in lieu of any fractional ACI Share such holder may have otherwise been entitled to receive.

Over-Subscription of Stock Election Shares

If more than 33.8% of the S1 Shares tendered in the Exchange Offer (the "Stock Election Number") elect to receive the Stock Consideration (each, a "Stock Election Share"), then:

- each S1 Share that is not a Stock Election Share (each, a "Non-Stock Share") will be exchanged for \$10.00 in cash, without interest;
- a number of Stock Election Shares of each stockholder making a stock election equal to the product of (x) the Cash Proration Factor and (y) the total number of Stock Election Shares held by such stockholder, will be exchanged for \$10.00 in cash, without interest; and
- each Stock Election Share that has not been exchanged for \$10.00 in cash, without interest in accordance with the preceding bullet will be exchanged for 0.3148 of an ACI Share.

Subscription of Stock Election Shares Equals Stock Election Number

If the aggregate number of Stock Election Shares is equal to the Stock Election Number, then each Stock Election Share will be exchanged for 0.3148 of an ACI Share, and each Non-Stock Share will be exchanged for \$10.00 in cash, without interest.

Under-Subscription of Stock Election Shares

If the aggregate number of Stock Election Shares is less than 33.8% of the S1 Shares tendered in the Exchange Offer, then:

- each Stock Election Share will be exchanged for 0.3148 of an ACI Share;
- a number of Non-Stock Shares of each stockholder equal to the product of (x) the Stock Proration Factor and (y) the total number of Non-Stock Shares of such stockholder, will be exchanged for 0.3148 of an ACI Share; and
- · each Non-Stock Share that has not been exchanged for 0.3148 of an ACI Share pursuant to the preceding bullet will be exchanged for \$10.00 in cash, without interest.

For purposes of these calculations:

- "Cash Proration Factor" means the quotient of (x) the excess of the total number of Stock Election Shares over the Stock Election Number divided by (y) the total number of Stock Election Shares.
- "Stock Proration Factor" means the quotient of (x) the excess of the Stock Election Number over the total number of Stock Election Shares divided by (y) the total number of Non-Stock Shares.

Consequences of Tendering with No Election

S1 stockholders who do not make an election will be deemed to have elected the Cash Consideration.

Procedure for Tendering

In order for an S1 stockholder to tender S1 Shares pursuant to the Exchange Offer, the exchange agent must receive, prior to the Expiration Time, the letter of election and transmittal (or a manually signed facsimile thereof), properly completed and duly executed, together with any required signature guarantees or, in the case of a book-entry transfer, an Agent's Message, and any other documents required by such letter of election and transmittal, at one of its addresses set forth on the back cover of this prospectus/offer to exchange and either (1) the certificates evidencing tendered S1 Shares must be received by the exchange agent at such

address or such S1 Shares must be tendered pursuant to the procedure for book-entry transfer described below and a Book-Entry Confirmation must be received by the exchange agent (including an Agent's Message), in each case prior to the Expiration Time or the expiration of the subsequent offering period, if one is provided, or (2) the tendering S1 stockholder must comply with the guaranteed delivery procedures described below.

The method of delivery of share certificates and all other required documents, including delivery through the Book-Entry Transfer Facility, is at the option and risk of the tendering S1 stockholder, and the delivery will be deemed made only when actually received by the exchange agent. If delivery is by mail, registered mail with return receipt requested, properly insured, is recommended. In all cases, sufficient time should be allowed to ensure timely delivery.

Book-Entry Transfer. The exchange agent will establish accounts with respect to the S1 Shares at the Book-Entry Transfer Facility for purposes of the Exchange Offer within two business days after the date of this prospectus/offer to exchange. Any financial institution that is a participant in the system of the Book-Entry Transfer Facility may make a book-entry delivery of S1 Shares by causing the Book-Entry Transfer Facility to transfer such S1 Shares into the exchange agent's account at the Book-Entry Transfer Facility in accordance with the Book-Entry Transfer Facility procedures for such transfer. However, although delivery of S1 Shares may be effected through book-entry transfer at the Book-Entry Transfer Facility, an Agent's Message and any other required documents must, in any case, be received by the exchange agent at one of its addresses set forth on the back cover of this prospectus/offer to exchange prior to the Expiration Time or the expiration of the subsequent offering period, if one is provided, or the tendering S1 stockholder must comply with the guaranteed delivery procedures described below. Delivery of documents to the Book-Entry Transfer Facility does not constitute delivery to the exchange agent.

Signature Guarantees. No signature guarantee is required on a letter of election and transmittal (1) if a letter of election and transmittal is signed by a registered holder of S1 Shares who has not completed the box titled "Special Issuance Instructions" on the letter of election and transmittal or (2) if S1 Shares are tendered for the account of a financial institution that is a member of the Securities Transfer Agents Medallion Signature Program, or by any other "Eligible Guarantor Institution," as such term is defined in Rule 17Ad-15 under the Exchange Act (each of the foregoing being referred to as an "Eligible Institution"). In all other cases, all signatures on letters of transmittal must be guaranteed by an Eligible Institution.

If a certificate evidencing S1 Shares is registered in the name of a person other than the signer of a letter of election and transmittal, then such certificate must be endorsed or accompanied by appropriate stock powers, in either case signed exactly as the name(s) of the registered holder(s) appear on the share certificate, with the signature(s) on such certificate or stock powers guaranteed by an Eligible Institution. See Instructions 1 and 5 of the letter of election and transmittal.

Guaranteed Delivery. If an S1 stockholder desires to tender S1 Shares pursuant to the Exchange Offer and such S1 stockholder's certificate(s) evidencing such S1 Shares are not immediately available, such S1 stockholder cannot deliver such certificates and all other required documents to the exchange agent prior to the Expiration Time, or such S1 stockholder cannot complete the procedure for delivery by book-entry transfer on a timely basis, such S1 Shares may nevertheless be tendered, provided that all the following conditions are satisfied:

- (1) such tender is made by or through an Eligible Institution;
- (2) a properly completed and duly executed Notice of Guaranteed Delivery, substantially in the form made available by Offeror, is received prior to the Expiration Time by the exchange agent as provided below; and
- (3) the share certificates (or a Book-Entry Confirmation) evidencing all tendered S1 Shares, in proper form for transfer, in each case together with the letter of election and transmittal (or a manually signed facsimile thereof), properly completed and duly executed, with any required signature guarantees or, in the case of a book-entry transfer, an Agent's Message, and any other documents required by the letter of election and transmittal are received by the exchange agent within three NASDAQ trading days after the date of execution of such Notice of Guaranteed Delivery.

The Notice of Guaranteed Delivery may be delivered by hand or mail or by facsimile transmission to the exchange agent and must include a guarantee by an Eligible Institution in the form set forth in such Notice of Guaranteed Delivery. The procedures for guaranteed delivery above may not be used during any subsequent offering period.

In all cases (including during any subsequent offering period), exchanges of S1 Shares tendered and accepted for exchange pursuant to the Exchange Offer will be made only after timely receipt by the exchange agent of the certificates evidencing such S1 Shares, or a Book-Entry Confirmation of the delivery of such S1 Shares (except during any subsequent offering period), and the letter of election and transmittal (or a manually signed facsimile thereof), properly completed and duly executed, with any required signature guarantees or, in the case of a book-entry transfer, an Agent's Message, and any other documents required by the letter of election and transmittal.

Determination of Validity. Offeror's interpretation of the terms and conditions of the Exchange Offer (including the letter of election and transmittal and the instructions thereto) will be final and binding to the fullest extent permitted by law. All questions as to the form of documents and the validity, form, eligibility (including time of receipt) and acceptance for exchange of any tender of S1 Shares will be determined by offeror, in its discretion, which determination shall be final and binding to the fullest extent permitted by law. Offeror reserves the absolute right to reject any and all tenders determined by it not to be in proper form or the acceptance of or exchange for which may, in the opinion of its counsel, be unlawful. Offeror also reserves the absolute right to waive any condition of the Exchange Offer to the extent permitted by applicable law or any defect or irregularity in the tender of any S1 Shares of any particular S1 stockholder, whether or not similar defects or irregularities are waived in the case of other S1 stockholders. No tender of S1 Shares will be deemed to have been validly made until all defects and irregularities have been cured or waived. None of ACI, Offeror or any of their affiliates or assigns, the exchange agent, the information agent or any other person will be under any duty to give any notification of any defect or irregularity in tenders or to waive any such defect or irregularity or incur any liability for failure to give any such notification or waiver.

A tender of S1 Shares pursuant to any of the procedures described above will constitute the tendering S1 stockholder's acceptance of the terms and conditions of the Exchange Offer, as well as the tendering S1 stockholder's representation and warranty to Offeror that (1) such S1 stockholder owns the tendered S1 Shares (and any and all other S1 Shares or other securities issued or issuable in respect of such S1 Shares), (2) such S1 stockholder has the full power and authority to tender, sell, assign and transfer the tendered S1 Shares (and any and all other S1 Shares or other securities issued or issuable in respect of such S1 Shares) and (3) when the same are accepted for exchange, Offeror will acquire good, marketable and unencumbered title thereto, free and clear of all liens, restrictions, charges and encumbrances and not subject to any adverse claims.

The acceptance for exchange by Offeror of S1 Shares pursuant to any of the procedures described above will constitute a binding agreement between the tendering S1 stockholder and Offeror upon the terms and subject to the conditions of the Exchange Offer, including with respect to the release and discharge from certain claims as described in the letter of election and transmittal.

Appointment as Proxy; Other Agreements. By executing the letter of election and transmittal, or through delivery of an Agent's Message, as set forth above, a tendering S1 stockholder irrevocably appoints designees of Offeror as such S1 stockholder's agents, attorneys-in-fact and proxies, each with full power of substitution, in the manner set forth in such letter of election and transmittal, to the full extent of such S1 stockholder's rights with respect to the S1 Shares tendered by such S1 stockholder and accepted for exchange by Offeror (and with respect to any and all other S1 Shares or other securities issued or issuable in respect of such S1 Shares on or after the date of this prospectus/offer to exchange). All such powers of attorney and proxies shall be considered irrevocable and coupled with an interest in the tendered S1 Shares (and such other S1 Shares and securities). Such appointment will be effective when, and only to the extent that, Offeror accepts such S1 Shares for exchange. Upon appointment, all prior powers of attorney and proxies given by such S1 stockholder with respect to such S1 Shares (and such other S1 Shares and securities) will be revoked,

without further action, and no subsequent powers of attorney or proxies may be given nor any subsequent written consent executed by such S1 stockholder (and, if given or executed, will not be deemed to be effective) with respect thereto. The designees of Offeror will, with respect to the S1 Shares (and such other S1 Shares and securities) for which the appointment is effective, be empowered to exercise all voting, consent and other rights of such S1 stockholder as they in their discretion may deem proper at any annual or special meeting of S1 stockholders or any adjournment or postponement thereof, by written consent in lieu of any such meeting or otherwise. Offeror reserves the right to require that, in order for S1 Shares to be deemed validly tendered, immediately upon Offeror's acceptance of S1 Shares for exchange, ACI must be able to exercise full voting, consent and other rights with respect to such S1 Shares (and such other S1 Shares and securities).

The foregoing proxies are effective only upon acceptance for exchange of S1 Shares tendered pursuant to the Exchange Offer. The Exchange Offer does not constitute a solicitation of proxies (absent an exchange of S1 Shares) for any meeting of S1 stockholders, which will be made only pursuant to separate proxy materials complying with the requirements of the rules and regulations of the SEC.

Backup Withholding. Under the "backup withholding" provisions of federal income tax law, the exchange agent may be required to withhold (currently at a rate of 28%) on any cash payments pursuant to the Exchange Offer or the Second-Step Merger. In order to prevent backup withholding with respect to payments to certain S1 stockholders for S1 Shares sold pursuant to the Exchange Offer or exchanged pursuant to the Second-Step Merger, each such S1 stockholder must timely provide the exchange agent with such S1 stockholder's correct taxpayer identification number (the "TIN") and certify that such stockholder is not subject to backup withholding by completing the substitute Form W-9 in the letter of election and transmittal, or otherwise establish an exemption. Certain S1 stockholders (including, among others, all corporations and certain non-U.S. individuals and entities) are not subject to backup withholding. If an S1 stockholder does not provide timely its correct TIN or fails to provide the certifications described above, the Internal Revenue Service may impose a penalty on the stockholder and payment of cash to the S1 stockholder pursuant to the Exchange Offer or the Second-Step Merger that are U.S. persons for federal income tax purposes should complete and sign the substitute Form W-9 included in the letter of election and transmittal to provide the information necessary to avoid backup withholding. Non-U.S. S1 stockholders should complete and sign an applicable Form W-8 (a copy of which may be obtained from the exchange agent) in order to avoid backup withholding.

Withdrawal Rights

Tenders of S1 Shares made pursuant to the Exchange Offer are irrevocable except that such S1 Shares may be withdrawn at any time prior to the Expiration Time. If Offeror elects to extend the Exchange Offer, is delayed in its acceptance for exchange of S1 Shares or is unable to accept S1 Shares for exchange pursuant to the Exchange Offer for any reason, then, without prejudice to ACI's or Offeror's rights under the Exchange Offer, the exchange agent may, on behalf of Offeror, retain tendered S1 Shares, and such S1 Shares may not be withdrawn except to the extent that tendering S1 stockholders are entitled to withdrawal rights as described in this section. Any such delay will be by an extension of the Exchange Offer to the extent required by law. If Offeror decides to include a subsequent offering period, S1 Shares tendered during the subsequent offering period may not be withdrawn. Please see the section of this prospectus/offer to exchange titled "The Exchange Offer — Extension The Exchange Offer Decides of this prospectus/offer to exchange titled "The Exchange Offer — Extension The Exchange Offer Decides of the Exchange Offer D

For a withdrawal to be effective, a written or facsimile transmission notice of withdrawal must be received by the exchange agent at one of its addresses set forth on the back cover page of this prospectus/offer to exchange. Any such notice of withdrawal must specify the name of the person who tendered the S1 Shares to be withdrawn, the number of S1 Shares to be withdrawn and the name of the registered holder of such S1 Shares, if different from that of the person who tendered such S1 Shares. If certificates evidencing S1 Shares to be withdrawn have been delivered or otherwise identified to the exchange agent, then, prior to the physical release of such certificates, the serial numbers shown on such certificates must be submitted to the exchange agent and, unless such S1 Shares have been tendered by or for the account of an Eligible Institution, the signature(s) on the notice of withdrawal must be guaranteed by an Eligible Institution. If

S1 Shares have been tendered pursuant to the procedure for book-entry transfer as set forth in the section of this prospectus/offer to exchange titled "The Exchange Offer — Procedure for Tendering," any notice of withdrawal must specify the name and number of the account at the Book-Entry Transfer Facility to be credited with the withdrawn S1 Shares.

Withdrawals of S1 Shares may not be rescinded. Any S1 Shares properly withdrawn will thereafter be deemed not to have been validly tendered for purposes of the Exchange Offer. However, withdrawn S1 Shares may be re-tendered at any time prior to the Expiration Time (or during the subsequent offering period, if one is provided) by following one of the procedures described in the section of this prospectus/offer to exchange titled "The Exchange Offer — Procedure for Tendering" (except S1 Shares may not be re-tendered using the procedures for guaranteed delivery during any subsequent offering period).

All questions as to the form and validity (including time of receipt) of any notice of withdrawal will be determined by Offeror, in its discretion, whose determination will be final and binding to the fullest extent permitted by law. None of ACI, Offeror or any of their respective affiliates or assigns, the exchange agent, the information agent or any other person will be under any duty to give any notification of any defect or irregularity in any notice of withdrawal or incur any liability for failure to give any such notification.

Announcement of Results of the Exchange Offer

ACI will announce the final results of the Exchange Offer, including whether all of the conditions to the Exchange Offer have been fulfilled or waived and whether Offeror will accept the tendered S1 Shares for exchange after the Expiration Time. The announcement will be made by a press release.

Ownership of ACI After the Exchange Offer

Based on ACI's and S1's respective capitalizations as of October 12, 2011 and assuming ACI issues 5.9 million ACI Shares pursuant to the Exchange Offer and the Second-Step Merger, former S1 stockholders would own, in the aggregate, approximately 14.4% of the aggregate ACI Shares on a fully diluted basis.

Material Federal Income Tax Consequences

The following is a general summary of the material United States Federal income tax consequences to S1 stockholders that exchange S1 Shares for ACI Shares and/or cash pursuant to the Exchange Offer and the Second-Step Merger. This discussion is based on provisions of the Internal Revenue Code, Treasury regulations promulgated thereunder, and administrative and judicial interpretations thereof, all as in effect as of the date hereof and all of which are subject to change, possibly with retroactive effect. This discussion does not address all aspects of United States Federal income taxation that may be applicable to S1 stockholders in light of their particular circumstances or to S1 stockholders subject to special treatment under United States Federal income tax law including, without limitation:

- partnerships;
- foreign persons;
- · certain financial institutions;
- · insurance companies;
- · tax-exempt entities;
- · dealers in securities;
- · traders in securities that elect to apply a mark-to-market method of accounting;
- · certain U.S. expatriates;
- persons that hold S1 Shares as part of a straddle, hedge, conversion transaction or other integrated investment;

- · S1 stockholders whose functional currency is not the United States dollar; and
- · S1 stockholders who acquired S1 Shares through the exercise of employee stock options or otherwise as compensation.

This discussion is limited to S1 stockholders that hold their S1 Shares as capital assets and does not consider the tax treatment of S1 stockholders that hold S1 Shares through a partnership or other pass-through entity. Furthermore, this summary does not discuss any aspect of state, local or foreign taxation.

Treatment of the Exchange Offer and Second-Step Merger as part of an integrated transaction that does not qualify as a reorganization.

If, as currently expected, the Exchange Offer and the Second-Step Merger are treated as a single integrated transaction that does not qualify as a reorganization, the following are the material federal income tax consequences of the exchange of S1 Shares for the Cash Consideration and/or Stock Consideration pursuant to the Exchange Offer and/or the Second-Step Merger:

- An S1 stockholder that receives Stock Consideration and/or the Cash Consideration in exchange for its S1 Shares pursuant to the Exchange Offer or the Second-Step Merger will recognize gain
 or loss equal to the difference between the sum of the fair market value of the ACI Shares and the amount of cash received and such S1 stockholder's adjusted tax basis in the S1 Shares
 exchanged therefor.
- Such recognized gain or loss will constitute capital gain or loss, and will constitute long-term capital gain or loss if the S1 stockholder's holding period for the S1 Shares exchanged is greater than one year as of the date of the exchange.
- The basis of any ACI Shares received will be equal to their fair market value on the date of the exchange, and their holding period will begin on the day following the date of the exchange.

Treatment as a Reorganization

Although it is not expected, it is possible that the Exchange Offer and the Second-Step Merger may be treated as component parts of an integrated transaction that qualifies as a reorganization within the meaning of Section 368(a) of the Internal Revenue Code. In order to be so treated, certain facts relating to the Exchange Offer and the Second-Step Merger must exist, including, among others, that:

- (1) the direction of the Second-Step Merger is reversed, such that S1 merges into Offeror;
- (2) the value of the ACI Shares issued to S1 stockholders pursuant to the Exchange Offer and the Second-Step Merger as a percentage of the total consideration furnished to S1 stockholders in connection with the Exchange Offer and the Second-Step Merger (including cash paid to dissenters, if any) satisfies the continuity of stockholder interest requirement for corporate reorganizations, which will generally be satisfied if the percentage is 40 or more, taking into account any acciunt any acquisitions by ACI, Offeror or any party related to ACI or Offeror, in connection with the Exchange Offer and the Second-Step Merger, of ACI Shares issued to S1 stockholders. Depending upon the facts, the applicable percentage may be determined using the value of ACI Shares on the date of announcement of the Exchange Offer or at certain other times, but not later than as of the closing date of the transaction. If market prices for ACI Shares upon consummation of the Exchange Offer are less than \$41.48, the Stock Consideration would represent less than 40% of the total value of the Exchange Offer consideration. You are urged to obtain current trading price information prior to making any decision with respect to the Exchange Offer;
 - (3) ACI will continue S1's historic business or will use a significant portion of S1's historic business assets in a business;
 - (4) the Exchange Offer and the Second-Step Merger will be consummated in accordance with the terms of this prospectus/offer to exchange.

We will not seek a ruling from the IRS with regard to the transactions. Accordingly, there can be no certainty that the IRS will not challenge the conclusions described below or that a court would not sustain such a challenge.

If the Exchange Offer and the Second-Step Merger are properly treated as part of an integrated transaction that qualifies as a reorganization within the meaning of Section 368(a) of the Internal Revenue Code, the following are the material federal income tax consequences of the exchange of S1 Shares for cash and/or ACI Shares pursuant to the Exchange Offer and/or the Second-Step Merger:

- An S1 stockholder that receives solely cash in exchange for its S1 Shares will generally recognize capital gain or loss equal to the difference, if any, between the amount of cash received and the
 adjusted tax basis of the S1 Shares. Such gain or loss will be long-term capital gain or loss if the S1 stockholder's holding period for the S1 Shares exchanged is greater than one year on the date
 of the exchange.
- An S1 stockholder that receives solely ACI Shares (or stock and cash in lieu of fractional ACI Shares) in exchange for its S1 Shares will not recognize gain or loss on the exchange except with respect to the cash received in lieu of fractional ACI Shares, which will be treated as described below.
- An S1 stockholder that receives ACI Shares and cash in exchange for its S1 Shares will recognize gain equal to the lesser of: (i) the excess, if any, of the sum of the fair market value of the ACI Shares and the amount of cash received over the adjusted tax basis of the S1 Shares, or (ii) the amount of cash received (excluding cash received in lieu of fractional ACI Shares, which will be treated as described below).
- Such recognized gain will constitute capital gain, unless the receipt of the cash has the effect of a distribution of a dividend as discussed below; in which case such recognized gain will be treated as ordinary dividend income to the extent of the S1 stockholder's ratable share of ACI's accumulated earnings and profits.
- . Any capital gain recognized will constitute long-term capital gain if the S1 stockholder's holding period for the S1 Shares exchanged is greater than one year as of the date of the exchange.
- An S1 stockholder that receives ACI Shares and cash will recognize no loss on the exchange (except, possibly, in connection with cash received in lieu of fractional ACI Shares, as discussed below).
- The aggregate tax basis of the ACI Shares received by an S1 stockholder, including for this purpose any fractional ACI Share for which cash is received, in exchange for S1 Shares will be the same as the aggregate tax basis of the S1 Shares surrendered in exchange therefor, decreased by the amount of any cash received (excluding any cash received in lieu of fractional ACI Shares) and increased by the amount of any gain recognized.
- · The holding period of ACI Shares received in exchange for S1 Shares will include the holding period of the S1 Shares surrendered in exchange therefor.

Possible treatment of cash as a dividend. In general, the determination of whether the gain recognized by an S1 stockholder will be treated as capital gain or ordinary dividend income distribution will depend upon whether and to what extent the exchange reduces the S1 stockholder sole demend percentage stock ownership interest in ACI. For purposes of this determination, an S1 stockholder will be treated as if such S1 stockholder first exchanged all of such S1 stockholder schange for the cash that the S1 stockholder actually received. The gain recognized in the exchange followed by a deemed redemption will be treated as capital gain if, with respect to the S1 stockholder, the deemed redemption is (1) "substantially disproportionate" or (2) "not essentially equivalent to a dividend." In general, the deemed redemption will be "substantially disproportionate" with respect to an S1 stockholder if the percentage described in (a) below is less than 80% of the percentage described in (a) below is essentially equivalent to a dividend," with respect to an S1 stockholder will depend on the S1 stockholder's particular circumstances. In order for the deemed redemption to be "not essentially equivalent to a dividend," the deemed redemption must

result in a "meaningful reduction" in such S1 stockholder's deemed percentage stock ownership of ACI Shares. In general, that determination requires a comparison of (a) the percentage of the outstanding voting stock of ACI that such S1 stockholder is deemed actually and constructively to have owned immediately before the deemed redemption by Offeror and (b) the percentage of the outstanding voting stock of ACI actually and constructively owned by such stockholder immediately after the deemed redemption by Offeror. In applying the foregoing tests, a stockholder may be deemed to own stock that is owned by other persons in addition to stock actually owned. Because the constructive ownership rules are complex, each stockholder should consult its own tax advisor as to the applicability of these rules. The Internal Revenue Service has ruled that a minority stockholder in a publicly traded corporation whose relative stock interest is minimal and that exercises no control with respect to corporate affairs is considered to have a "meaningful reduction" if such stockholder has any reduction in such stockholder's percentage stock ownership.

Cash received in lieu of fractional shares. Cash received in lieu of a fractional ACI Share will be treated as received in redemption of such fractional share interest, and an S1 stockholder likely will recognize capital gain or loss on the deemed redemption measured by the difference between the amount of cash received and the portion of the basis of the ACI Shares allocable to such fractional interest, although it is possible that the deemed redemption payment could be treated as a dividend, as described above. Such capital gain or loss will be long-term capital gain or loss if the S1 stockholder's holding period in the S1 Shares exchanged was greater than one year as of the date of the exchange.

Failure of the Exchange Offer to be treated as part of an integrated transaction.

Treatment of S1 stockholders who tender their shares pursuant to the Exchange Offer. In the unlikely event that the Exchange Offer and the Second-Step Merger are not treated as a single integrated transaction or if the Exchange Offer is completed but the Second-Step Merger does not occur, the Exchange Offer would fail to qualify as a reorganization within the meaning of Section 368(a) of the Internal Revenue Code. In that case, S1 stockholders who tender their shares pursuant to the Exchange Offer would recognize gain or loss, take a fair market value basis in the ACI Shares received, and their holding period for the ACI Shares would begin on the day following the exchange, as described above in "Treatment of the Exchange Offer and Second-Step Merger as part of an integrated transaction that does not qualify as a reorganization."

Treatment of stockholders who exchange their shares pursuant to the Second-Step Merger. If the Exchange Offer and the Second-Step Merger are both consummated but are not treated as part of an integrated transaction, the treatment described above in "Treatment as a Reorganization" would likely apply to S1 stockholders who exchange their shares pursuant to the Second-Step Merger.

THE FOREGOING DISCUSSION IS INTENDED ONLY AS A SUMMARY AND DOES NOT PURPORT TO BE A COMPLETE ANALYSIS OR LISTING OF ALL POTENTIAL FEDERAL INCOME TAX CONSEQUENCES OF THE EXCHANGE OFFER AND THE SECOND-STEP MERGER. SI STOCKHOLDERS ARE URGED TO CONSULT THEIR TAX ADVISORS CONCERNING THE UNITED STATES FEDERAL, STATE, LOCAL AND FOREIGN TAX CONSEQUENCES OF THE EXCHANGE OFFER AND THE SECOND-STEP MERGER TO THEM.

Purpose and Structure of the Exchange Offer

The Exchange Offer is intended to allow ACI, through Offeror, to acquire all of the issued and outstanding S1 Shares. We intend to, promptly after completion of the Exchange Offer, consummate the Second-Step Merger of S1 with a wholly owned subsidiary of ACI pursuant to the DGCL. The purpose of the Second-Step Merger is for ACI, through Offeror, to acquire all outstanding S1 Shares that are not acquired in the Exchange Offer. In this Second-Step Merger, each remaining S1 Share (other than shares held in treasury by S1 and other than shares held by S1 stockholders who properly exercise applicable dissenters' rights under Delaware law) will be cancelled and exchanged for the Proration Amount of Cash and Stock Consideration. After this Second-Step Merger, ACI will own all of the issued and outstanding S1 Shares. Please see the sections of this prospectus/offer to exchange offer — Purpose and Structure of the Exchange Offer"; "The Exchange Offer — Second-Step Merger"; and "The Exchange Offer — Plans for S1."

Second-Step Merger

Under the DGCL, if ACI, through Offeror, acquires, pursuant to the Exchange Offer or otherwise, at least 90% of the S1 Shares, Offeror will be able to effect the Second-Step Merger as a "short form" merger without approval of the S1 Board or a vote of the remaining S1 stockholders. In such event, ACI is required under the Transaction Agreement to take all necessary and appropriate action to cause the Second-Step Merger to become effective as promptly as reasonably practicable after such acquisition, without a meeting of S1 stockholders.

The exact timing of the Second-Step Merger will necessarily depend upon a variety of factors, including the number of S1 Shares Offeror acquires pursuant to the Exchange Offer and ACI's exercise of the Top-Up Option.

Appraisal/Dissenters' Rights

S1 stockholders do not have appraisal rights in connection with the Exchange Offer. However, upon consummation of the Second-Step Merger, S1 stockholders who have not tendered their S1 Shares in the Exchange Offer and who, if a stockholder vote is required, did not vote for, or consent to, the approval of the Second-Step Merger will have rights under Delaware law to dissent from the Second-Step Merger and demand appraisal of their S1 Shares. S1 stockholders at the time of a "short form" merger under Delaware law would also be entitled to exercise dissenters' rights pursuant to such a "short form" merger. Stockholders who perfect dissenters' rights by complying with the procedures set forth in Section 262 of the DGCL will be entitled to receive a cash payment equal to the "fair value" of their S1 Shares, as determined by a Delaware court. Because appraisal rights are not available in connection with the Exchange Offer, no demand for appraisal under Section 262 of the DGCL may be made at this time. Any such judicial determination of the fair value of the S1 Shares could be based upon considerations other than or in addition to the consideration paid in the Exchange Offer, the Second-Step Merger and the market value of the S1 Shares. S1 stockholders should recognize that the value so determined could be higher or lower than, or the same as, the consideration per share paid pursuant to the Exchange Offer or the Second-Step Merger. Moreover, we may argue in an appraisal proceeding that, for purposes of such a proceeding, the fair value of the S1 Shares is less than the consideration paid in the Exchange Offer.

FAILURE TO FOLLOW THE STEPS REQUIRED BY SECTION 262 OF THE DGCL FOR PERFECTING APPRAISAL RIGHTS MAY RESULT IN THE LOSS OF SUCH RIGHTS. BECAUSE OF THE COMPLEXITY OF DELAWARE LAW RELATING TO APPRAISAL RIGHTS, WE ENCOURAGE YOU TO SEEK THE ADVICE OF YOUR OWN LEGAL COUNSEL. THE FOREGOING DISCUSSION IS NOT A COMPLETE STATEMENT OF THE DGCL AND IS QUALIFIED IN 1TS ENTIRETY BY REFERENCE TO THE DGCL. IN PARTICULAR, THE DESCRIPTION OF SECTION 262 ABOVE IS QUALIFIED IN 1TS ENTIRETY BY REFERENCE TO SUCH SECTION.

Plans for S1

The purpose of the Exchange Offer is for ACI to acquire control of, and ultimately the entire interest in, S1. ACI intends, promptly following Offeror's acceptance for exchange, and the exchange of S1 Shares in the Exchange Offer, to consummate the Second-Step Merger of Offeror with and into S1. In the Second-Step Merger, each remaining S1 Share (other than S1 Shares held in treasury by S1 or owned by ACI or its wholly owned subsidiaries, certain restricted S1 Shares converted into restricted ACI Shares pursuant to the Transaction Agreement and S1 Shares held by S1 stockholders who properly exercise applicable dissenters' rights under Delaware law) will be converted into the right to receive the Proration Amount of Cash and Stock Consideration. If the Exchange Offer is successful, ACI intends to consummate the Second-Step Merger as promptly as practicable.

Effective as of the consummation of the Exchange Offer, Offeror will elect directors to the S1 Board proportionate to its ownership interest as provided in the Transaction Agreement. Offeror's director designees will be one or more persons identified on Appendix A to this document.

If, and to the extent that ACI, Offeror and/or any of ACI's subsidiaries acquires control of S1, ACI intends to conduct a detailed review of S1's business, operations, capitalization and management and consider and determine what, if any, changes would be desirable in light of the circumstances which then exist. ACI intends to eliminate S1's public company infrastructure and restructure the combined company's legal entity organization, including restructuring S1's non-U.S. subsidiaries. In addition, it is expected that, initially following the Second-Step Merger, the business and operations of S1 will, except as set forth in this prospectus/offer to exchange, be continued substantially as they are currently being conducted, but ACI expressly reserves the right to make any changes that it deems necessary, appropriate or convenient to optimize potential in conjunction with ACI's businesses and ACI's review or in light of future developments. Such changes could include, among other things, changes in S1's business, corporate and legal structure, assets, properties, marketing strategies, capitalization, management, personnel or dividend policy and changes to S1's restated certificate of incorporation and its amended and restated by-laws.

Except as indicated in this prospectus/offer to exchange, neither ACI nor any of ACI's subsidiaries has any current plans or proposals that relate to or would result in (1) any extraordinary transaction, such as a merger, reorganization or liquidation of S1 or any of its subsidiaries, (2) any purchase, sale or transfer of a material amount of assets of S1 or any of its subsidiaries, (3) any material change in the present dividend rate or policy, or indebtedness or capitalization of S1 or any of its subsidiaries, (4) any change in the current board of directors or management of S1 or any change to any material term of the employment contract of any executive officer of S1, (5) any other material change in S1's corporate structure or business, (6) any class of equity security of S1 being delisted from a national stock exchange or ceasing to be authorized to be quoted in an automated quotation system operated by a national securities association, or (7) any class of equity securities of S1 becoming eligible for termination of registration under Section 12(g)(4) of the Exchange Act.

Effect of the Exchange Offer on the Market for S1 Shares; NASDAQ Listing; Registration Under the Securities Exchange Act of 1934; Margin Regulations

Effect of the Exchange Offer on the Market for the S1 Shares

The exchange of S1 Shares by Offeror pursuant to the Exchange Offer will reduce the number of S1 Shares that might otherwise trade publicly and will reduce the number of S1 stockholders, which could adversely affect the liquidity and market value of the remaining S1 Shares held by the public. The extent of the public market for S1 Shares and the availability of quotations reported in the over-the-counter market depends upon the number of S1 stockholders, the aggregate market value of the S1 Shares remaining at such time, and the interest of maintaining a market in the S1 Shares on the part of any securities firms and other factors.

NASDAQ Listing

S1 Shares are listed on the NASDAQ. Depending upon the number of S1 Shares exchanged pursuant to the Exchange Offer and the aggregate market value of any S1 Shares not purchased pursuant to the Exchange Offer, S1 Shares may no longer meet the standards for continued listing on the NASDAQ and may be delisted from the NASDAQ. The published guidelines of the NASDAQ indicate that it would consider delisting the S1 Shares if, among other things, (1) the number of round lot S1 stockholders falls below 400, (2) the number of publicly held S1 Shares falls below 750,000 or (3) the market value of publicly held S1 Shares falls below 55,000,000.

If, as a result of the exchange of S1 Shares pursuant to the Exchange Offer or otherwise, S1 Shares no longer meet the requirements of the NASDAQ for continued listing and the listing of S1 Shares is discontinued, the market for S1 Shares could be adversely affected. If the NASDAQ were to delist S1 Shares, it is possible that S1 Shares would continue to trade on another securities exchange or in the over-the-counter market and that price or other quotations would be reported by such exchange or other sources. The extent of the public market therefor and the availability of such quotations would depend, however, upon such factors as the number of S1 stockholders and/or the aggregate market value of such securities remaining at such time, the interest in maintaining a market in S1 Shares on the part of securities firms, the possible termination of registration under the Exchange Act as described below, and other factors. ACI cannot predict whether the

reduction in the number of S1 Shares that might otherwise trade publicly would have an adverse or beneficial effect on the market price for or marketability of S1 Shares or whether it would cause future market prices to be greater or less than the consideration being offered in the Exchange Offer. If S1 Shares are not delisted prior to the Second-Step Merger, then S1 Shares will cease to be listed on the NASDAQ upon consummation of the Second-Step Merger.

Registration Under the Securities Exchange Act of 1934

S1 Shares are currently registered under the Exchange Act. This registration may be terminated upon application by S1 to the SEC if S1 Shares are not listed on a "national securities exchange" and there are fewer than 300 record holders. Termination of registration would substantially reduce the information required to be furnished by S1 to holders of S1 Shares and to the SEC and would make certain provisions of the Exchange Act, such as the short-swing profit recovery provisions of Section 16(b), the requirement of furnishing a proxy statement in connection with stockholders' meetings and the requirements of Exchange Act Rule 13e-3 with respect to "going private" transactions, no longer applicable to S1. In addition, "affiliates" of S1 and persons holding "restricted securities" of S1 may be deprived of the ability to dispose of these securities pursuant to Rule 144 under the Securities Act. If registration of S1 Shares is not terminated prior to the Second-Step Merger, then the registration of S1 Shares under the Exchange Act will be terminated upon consummation of the Second-Step Merger.

Marain Regulations

S1 Shares are currently "margin securities," as such term is defined under the rules of the Board of Governors of the Federal Reserve System (the "Federal Reserve Board"), which has the effect, among other things, of allowing brokers to extend credit on the collateral of such securities. Depending upon factors similar to those described above regarding listing and market quotations, following the Exchange Offer it is possible that S1 Shares would no longer constitute "margin securities" for purposes of the margin regulations of the Federal Reserve Board, in which event S1 Shares would no longer be used as collateral for loans made by brokers. In addition, if registration of S1 Shares under the Exchange Act were terminated, S1 Shares would no longer constitute "margin securities."

Conditions of the Exchange Offer

Subject to the terms of the Transaction Agreement, Offeror will not be required to accept for exchange any S1 Shares tendered pursuant to the Exchange Offer and will not be required to make any exchange for S1 Shares accepted for exchange if, immediately prior to the Expiration Time (or substantially concurrently therewith), in the judgment of ACI, any one or more of the following conditions shall not have been satisfied or waived by Offeror:

Minimum Tender Condition

S1 stockholders shall have validly tendered and not properly withdrawn prior to the Expiration Time at least that number of S1 Shares (together with the S1 Shares then owned by ACI, Offeror or any of ACI's other subsidiaries), shall constitute a majority of S1 Shares issued and outstanding on a fully diluted basis.

Registration Statement Condition

The registration statement of which this prospectus/offer to exchange and the accompanying letter of election and transmittal is a part shall have been declared effective under the Securities Act, and no stop order suspending the effectiveness of the registration statement shall have been issued and no proceedings for that purpose shall have been initiated or threatened by the SEC and ACI shall have received all necessary state securities law or "blue sky" authorizations.

HSR Condition

Any applicable waiting period under the HSR Act, and, if applicable, any agreement with the FTC or the Antitrust Division not to accept S1 Shares for exchange in the Exchange Offer, shall have expired or shall have been terminated prior to the Expiration Time.

Other Regulatory Approvals Condition

Any clearance, approval, permit, authorization, waiver, determination, favorable review or consent of any Governmental Authority, other than the HSR Condition, shall have been obtained and such approvals shall be in full force and effect, or any applicable waiting periods for such clearances or approvals shall have expired, except for any failures that would not reasonably be expected to have a material adverse effect on ACI or S1.

Other Conditions

Additionally, Offeror will not be required to accept for exchange any S1 Shares tendered pursuant to the Exchange Offer if any of the following fail to be true:

- (1) the Fundamental S1 Corporate Representations were true and correct as of October 3, 2011 and will be true and correct on and as of the Expiration Time with the same force and effect as if made at the Expiration Time (in either case other than those representations and warranties which address matters only as of a particular date, which representations and warranties shall have been true and correct as of such particular date), except in either case contemplated by this clause (1) for de minimis inaccuracies and (2) the other representations and warranties of S1 set forth in the Transaction Agreement were true and correct as of October 3, 2011 and will be true and correct on and as of the Expiration Time with the same force and effect as if made on the Expiration Time (in either case other than those representations and warranties which address matters only as of a particular date, which representations shall have been true and correct as of such particular date), except in either case contemplated by this clause (2) where the failure of such representations and warranties to be true and correct (disregarding all qualifications or limitations as to materially, material adverse effect or words of similar import set forth therein) has not had and would not reasonably be expected to have a material adverse effect on S1;
- S1 has performed or complied in all material respects with all agreements and covenants required by the Transaction Agreement to be performed or complied with by it on or prior to the
- since October 3, 2011, there shall not have occurred any material adverse change in the business, financial condition or continuing results of S1 and its subsidiaries, taken as a whole (excluding certain events specified in the Transaction Agreement).

The foregoing conditions are for the sole benefit of Offeror and ACI and may be asserted by Offeror or ACI regardless of the circumstances giving rise to any such condition, in whole or in part at any applicable time or from time to time in their sole discretion prior to the expiration of the Offer, except that the conditions relating to receipt of any approvals from any Governmental Authority may be asserted at any time prior to the acceptance for payment of S1 Shares, and all conditions (except for the Minimum Tender Condition) may be waived by ACI or Offeror in their sole discretion in whole or in part at any applicable time or from time to time, in each case subject to the terms and conditions of the Transaction Agreement and the applicable rules and regulations of the SEC. The failure of ACI or Offeror at any time to exercise any of the foregoing rights will not be deemed a waiver of any such right and each such right will be deemed an ongoing right that may be asserted at any time and from time to time.

Source and Amount of Funds

ACI estimates that the aggregate consideration to be paid to S1 stockholders in connection with the Exchange Offer and Second-Step Merger will consist of \$415 million in cash (less applicable withholding taxes and without

interest) and that number of ACI Shares determined in accordance with the exchange ratio. In addition, S1 stockholders will receive cash in lieu of any fractional ACI Shares to which they may be entitled.

No other plans or arrangements have been made to finance or repay such financing after the consummation of the Exchange Offer and the Second-Step Merger. No alternative financing arrangements or alternative financing plans have been made in the event such financings fail to materialize at this time; however, in the event we pursue alternative financing, we will amend this prospectus/offer to exchange to describe such alternative financing.

Amount of Cash Required

ACI estimates that the total amount of cash required to complete the transactions contemplated by the Exchange Offer and the Second-Step Merger will be approximately \$415 million, which estimated total amount includes:

- payment of the cash portion of the Exchange Offer consideration required to acquire all of the S1 Shares pursuant to the Exchange Offer and the Second-Step Merger (including the cash payments due in lieu of the issuance of fractional ACI Shares);
- · any cash that may be required to be paid in respect of dissenters' or appraisal rights; and
- · payment of any fees, expenses and other related amounts incurred in connection with the Exchange Offer and Second-Step Merger.

We expect to have sufficient funds to complete the transactions contemplated by the Exchange Offer and the Second-Step Merger and to pay fees, expenses and other related amounts through a combination of (1) ACI's and S1's cash on hand and (2) borrowings under the proposed commitments described below.

Commitments

We have obtained commitments from Wells Fargo to arrange, and Wells Fargo Bank to provide, subject to certain conditions, senior bank financing consisting of up to \$450 million under a proposed new secured credit facility, comprised of a \$200 million senior secured term loan (the "Term Facility") and a \$250 million senior secured revolving credit facility (the "Revolving Facility" and, together with the Term Facility, the "Facility") for financing a portion of the cash component of the consideration to be paid to \$1 stockholders in connection with the Exchange Offer. ACI plans to fund the remaining cash portion of the cash component of the consideration to be paid to \$1 stockholders in connection with the Exchange Offer through the cash on ACI's balance sheet (the "Cash Contribution"), provided that the Cash Contribution shall be deemed to be reduced by the amount of cash on the balance sheet of ACI used by ACI prior to the Expiration Time solely for purposes of acquiring outstanding capital stock of \$1. Additionally, ACI will have the right, but not the obligation, to increase the amount of the Facility by incurring an incremental term loan facility or increasing the Revolving Facility in an aggregate principal amount not to exceed \$75 million, subject to certain conditions and under terms to be determined.

Interest; Letter of Credit Fees; Unused Commitment Fees

Each loan made under the Facility will bear interest at an Adjusted LIBOR Rate or Alternate Base Rate (as contemplated by the commitment letter relating to the Facility) plus the margin described in the chart below. Interest periods on Adjusted LIBOR Rate-based loans may be one, two, three or six months, at ACI's option. In the case of Adjusted LIBOR Rate-based loans, interest will accrue on the basis of a 360-day year, and will be payable on the last day of each relevant interest period and, for any interest period longer than three months, on each successive date three months after the first day of such interest period. Interest will accrue on Alternate Base Rate-based loans on the basis of a 365/366-day year (or 360-day year if based on the Adjusted LIBOR Rate) and shall be payable quarterly in arrears.

Unused loan commitments will be subject to an unused commitment fee, as described in the chart below.

Category	Leverage Ratio	Commitment Fee Rate	Eurodollar Spread	ABR Spread
Category 1	³ 3.25:1.00	0.50%	2.50%	1.50%
Category 2	32.75:1.00 and <3.25:1.00	0.40%	2.25%	1.25%
Category 3	32.00:1.00 and <2.75:1.00	0.35%	2.00%	1.00%
Category 4	31.00:1.00 and <2.75:1.00	0.30%	1.75%	0.75%
Category 5	<1.00:1.00	0.25%	1.50%	0.50%

Letter of Credit fees will be payable quarterly in arrears and will equal an amount equal to (x) the applicable margin in effect for Adjusted LIBOR Rate-based loans times (y) the average daily maximum aggregate amount available to be drawn under all Letters of Credit. In addition, fronting fees will be payable quarterly in arrears to the issuers of any Letters of Credit.

Conditions to Borrowing

Borrowing under the Facility will be subject to certain conditions. Set forth below is a description of certain conditions precedent to borrowing under the Facility:

- the satisfactory negotiation, execution and delivery of definitive loan documents relating to the Facility (to be based upon and substantially consistent with the terms set forth in the commitment letter and the fee letter) in the discretion of each of the arranger and ACI;
- the terms of the applicable acquisition documents (including the exhibits, schedules and all related documents) will be reasonably satisfactory to the arranger;
- since December 31, 2010, there shall not have been, as determined by Wells Fargo in its reasonable discretion (1) any event, change, effect, development, condition or occurrence (a "Combined Material Adverse Event"), that is materially adverse on or with respect to the business, financial condition or continuing results of operations of ACI and its subsidiaries, taken as a whole, on a pro forma basis after giving effect to the transactions contemplated to occur on the closing date of the Facility, other than any event, change, effect, development, condition or occurrence: (a) in or generally affecting the economy or the financial, commodities or securities markets in the United States or elsewhere in the world or the industry or industries in which ACI or such subsidiaries operate generally or (b) resulting from or arising out of (i) any natural disasters or weather-related or other force majeure event or (ii) any changes in national or international political conditions, including any engagement in hostilities, whether or not pursuant to the declaration of a national emergency or war, the outbreak or escalation of hostilities or acts of war, sabotage or terrorism, in each case, to the extent that such event, change, effect, development, condition or occurrence does not affect ACI and such subsidiaries, taken as a whole, in a materially disproportionate manner relative to other participants in the business, industries and geographic region or territory in which ACI and such subsidiaries operate, or as determined by ACI in its reasonable discretion; or (2) any event, change, effect, development, condition or occurrence that is materially adverse on or with respect to the business, financial condition or continuing results of operations of S1 and its subsidiaries, taken as a whole (an "Acquired Business Material Adverse Effect" will immediately upon, or promptly following, execution of the acquisition documents, be replaced by the corresponding definitions in the acquisition documents with such modifications to
- there will not exist (pro forma for the acquisition and the financing thereof) any default or event of default under any of the definitive loan documents relating to the Facility, or under any other material indebtedness of ACI or its subsidiaries;
- the Exchange Offer shall have been completed concurrently with the funding of the Term Facility (other than in the event of a funding demand by Wells Fargo prior to the completion of the Exchange

Offer), in each case, in accordance with the applicable acquisition documents without amendment or waiver (except to the extent such waiver (including any consent or discretionary determination as to the satisfaction of any condition) is not materially adverse to Wells Fargo or the lenders) or other modification of any of the terms or conditions thereof (including any change in (x) the dollar amount of the acquisition consideration constituting the acquisition cash consideration, (y) the aggregate number of shares of common stock of ACI constituting the Stock Consideration and (z) the percentage of the shares of S1 that can be exchanged for common stock of ACI or the percentage of the shares of S1 that can be exchanged for the Cash Consideration);

- Wells Fargo Bank shall have received (1) at least five days prior to the closing date of the Facility, audited financial statements of ACI and S1 for each of the three fiscal years ended at least
 45 days prior to the closing date of the Facility; (2) as soon as internal financial statements are available to S1, and in any event at least five days prior to the closing date of the Facility, unaudited
 financial statements for any interim period or periods of ACI and S1 ended after the date of the most recent audited financial statements and more than 45 days prior to the closing date of the
 Facility; (3) customary additional audited and unaudited financial statements for all recent, probable or pending acquisitions; and (4) customary pro forma financial statements, in each case
 meeting the requirements of Regulation 5-X for Form S-1 registration statements or otherwise reasonably satisfactory to the arranger;
- all costs, fees, expenses and other compensation then due with respect to the Facility shall have been paid and ACI shall have complied in all material respects with all of its other obligations under the commitment letter and the fee letter relating to the Facility;
- Wells Fargo shall have received (1) legal opinions, evidence of authority, corporate records and documents from public officials, lien searches and solvency and officer's certificates reasonably satisfactory to the arranger; (2) confirmation satisfactory to the arranger of (a) repayment using cash and cash equivalents and/or a draw on the Revolving Facility and termination of the \$150,000,000 revolving credit facility under that certain Credit Agreement (the "Existing Credit Agreement") dated as of September 29, 2006 (as it may be refinanced or replaced prior to the closing date of the Facility with a revolving credit facility arranged by Wells Fargo), and (b) termination or release of all liens or security interests relating thereto, in each case on terms satisfactory to the arranger; (3) evidence of requisite approval of the board of directors of \$1\$ and material third party and governmental consents necessary in connection with the acquisition, the related transactions or the financing thereof; (4) possessory collateral and financing statements sufficient when properly filed to perfect liens, pledges, and mortgages on the collateral securing the Facility; (5) evidence of satisfactory commitments for title insurance and evidence of insurance; and (6) at least 10 days prior to the closing date of the Facility, all documentation and other information required by bank regulatory authorities under applicable "know-your-customer" and anti-money laundering rules and regulations, including the PATRIOT Act documentation and information:
- the Stock Consideration, together with the proceeds of the Cash Contribution (which shall have been used in full to pay the Cash Consideration or transaction costs prior to or substantially simultaneously with the initial funding of the Facility, other than in the event of a funding demand by Wells Fargo prior to the completion of the Exchange Offer) and the proceeds from the borrowings made on the closing date of the Facility, will be the sole and sufficient sources of funds to consummate the transactions contemplated to occur on such date, refinance certain existing indebtedness of ACI and its subsidiaries (including the Existing Credit Agreement) and S1 and to pay the transaction costs (and after the application of proceeds from the borrowings on the closing date of the Facility, none of ACI, its subsidiaries or S1 will have any material indebtedness for borrowed money other than the Facility);
- accuracy of representations and warranties (1) under the Facility (subject to materiality thresholds and, in the case of S1, only with respect to the Specified Representations referred to below) and (2) made by or with respect to S1 in the acquisition documents as are material to the interest of the lenders (but only to the extent that ACI or one of its affiliates has the right to terminate its obligations under the acquisition agreement as a result of a breach of such representations in the acquisition agreement);

- ACI will, and after completion of the Exchange Offer will use commercially reasonable efforts to cause S1 to, cooperate with Wells Fargo (1) in the preparation of one or more information packages regarding the business, operations, financial projections and prospects of ACI and S1 and all information relating to the transactions contemplated under the Facility commitment letter deemed reasonably necessary by Wells Fargo to complete the syndication of the Facility (the "Confidential Information Memorandum") and (2) the presentation of one or more information packages acceptable in format and content to Wells Fargo (the "Lender Presentation") in connection with the syndication of the Facility by a date sufficient to afford Wells Fargo a period of at least 15 consecutive days (excluding traditional blackout and holiday periods in the bank market) following the general launch of the general syndication of the Facility at the primary bank meeting for prospective lenders (the "Lender Meeting") (which shall occur on or prior to September 9, 2011) to syndicate the Facility prior to the closing date of the Facility; provided that the closing date of the Facility shall not occur prior to September 28, 2011;
- the delivery by ACI to Wells Fargo of a Confidential Information Memorandum and a Lender Presentation on or before September 5, 2011; and
- · the Lender Meeting having occurred on or prior to September 9, 2011.

Notwithstanding any of the conditions outlined above, ACI and Wells Fargo agree that the completion of the syndication of the Facility will not constitute a condition precedent to the closing of the Facility and it is acknowledged and agreed since ACI delivered the Confidential Information Memorandum and Lender Presentation on or prior to September 5, 2011 and the Lender Meeting occurred on or prior to September 9, 2011, then, provided that the other conditions set forth in the commitment letter are satisfied, nothing in the commitment letter will impair the availability of the Facility on or after September 28, 2011.

Maturity

ACI expects that the contemplated Facility will mature on the five-year anniversary of the closing date of the Facility.

Prepayments and Repayments

The loans made under the Facility may be voluntarily repaid without premium or penalty, subject to ACI's payment of breakage costs in connection with any Adjusted LIBOR Rate-based loans.

Subject to certain exceptions and reductions, loans made under the Term Facility (and after payment in full of the Term Facility, loans under the Revolving Facility (without a permanent reduction of commitments)) will be mandatorily prepaid with (1) 100% of the net cash proceeds of any sale or other disposition of any property or assets of ACI or any of its subsidiaries, (2) 100% of the net cash proceeds of insurance paid on account of any loss of any property or assets of ACI or any of its subsidiaries, (3) 50% of the net cash proceeds of any issuance of equity by ACI, (4) 100% of the net cash proceeds of any incurrence of indebtedness for borrowed money by ACI or any of its subsidiaries, (5) 50% of excess cash flow (to be defined in the loan documents) if the Leverage Ratio (as defined in the commitment letter relating to the Facility) is greater than 2.50:1.00, and (6) an amount equal to the balance of the proceeds held in the Escrow Account (defined below) no later than the first business day following the earlier to occur of (a) the abandonment or termination of the Exchange Offer and, to the extent entered into, either of the acquisition documents and (b) the date that is six months after the date of the commitment letter.

Guarante

All obligations of ACI under the Facility will be unconditionally guaranteed by each of ACI's material existing and subsequently acquired or organized domestic direct and indirect subsidiaries, including S1 (but, excluding, to the extent necessary to comply with margin regulations, Offeror and S1 prior to the closing date of the Second-Step Merger).

Security

All obligations of ACI and any guarantor under the Facility and any interest rate and/or currency hedging obligations of ACI or any guarantor owed to the arranger, any agent or lender, or any affiliate of the arranger, any agent or lender will be secured by first priority security interests in all assets of ACI (including 100% of the capital stock of each material domestic subsidiary and 65% of the capital stock of each material first-tier foreign subsidiary of ACI and all intercompany debt, but prior to the Second-Step Merger, excluding any shares of S1 held be ACI to the extent constituting margin stock) and any guarantor (except as otherwise agreed to by Wells Fargo).

To the extent that the proceeds of the Term Facility (when taken together with the Cash Contribution) funded on the closing date of the Exchange Offer exceed 62% of the total consideration payable in accordance with the Exchange Offer documents in respect of the shares accepted in the Exchange Offer plus the associated transaction costs then due and payable, the excess proceeds of the Term Facility shall be funded directly into a blocked account of ACI held at Wells Fargo which account shall be subject to a perfected first priority security interest to secure the obligations of ACI in respect of the Facility pursuant to arrangements and documentation (including, without limitation, a control agreement) in form and substance satisfactory to Wells Fargo (the "Escrow Account").

Representations and Warranties

The credit agreement for the Facility will contain representations and warranties by ACI (with respect to itself and its subsidiaries and, only on and after the completion of the Exchange Offer, S1) relating to: due organization; requisite power and authority; qualification; equity interests and ownership; due authorization, execution, delivery and enforceability of the loan documents; creation, perfection and priority of security interests; no conflicts; governmental consents; historical and projected financial condition; no material adverse change; no restricted junior payments; absence of material litigation; payment of taxes; title to properties; environmental matters; no defaults under material agreements; Investment Company Act and margin stock matters; ERISA and other employee matters; absence of brokers or finders fees; solvency; compliance with laws; status as senior debt; full disclosure; and PATRIOT Act and other related matters.

On the closing date of the Facility, the only representations and warranties relating to S1, its subsidiaries and business that will be a condition precedent to the initial funding of the Facility will be (1) if acquisition documents have been executed on or prior to the closing date, the representations and warranties made by or with respect to S1 in the acquisition documents as are material to the interest of the lenders (but only to the extent that ACI or one of its affiliates has the right to terminate its obligations under the acquisition agreement as a result of a breach of such representations in the acquisition agreement) and (2) representations and warranties relating to: due organization or formation, requisite power and authority; due authorization, execution, delivery and enforceability of the applicable loan documents; no conflicts with constituent documents, laws and material debt documents; solvency; the absence of material litigation affecting the financing of the acquisition; Investment Company Act and margin stock matters; PATRIOT Act and related matters; and creation, perfection and priority of the security interests granted in the proposed collateral (the representations and warranties specified in this clause (2), the "Specified Representations").

Covenants

The loan documents will contain certain financial, affirmative and negative covenants by ACI with respect to ACI, Offeror and ACI's other subsidiaries. Set forth below is a description of the covenants under the Facility:

- a Minimum Fixed Charge Coverage Ratio (defined as (x) EBITDA minus Capital Expenditures divided by (y) Interest plus Scheduled Principal Payments plus Taxes) to be agreed;
- a Maximum Leverage Ratio of (x) 3:50:1.00, prior to the closing date of the Second-Step Merger) and (y) 3.25:1.00, on or after the closing date of the Second-Step Merger, with step down to 3.00:1.00 on the first anniversary of the closing date of the Facility;

- affirmative covenants in respect of the delivery of financial statements and other reports; maintenance of existence; payment of taxes and claims; maintenance of properties; maintenance of insurance; cooperation with syndication efforts; books and records; inspections; lender meetings; compliance with laws; environmental matters; additional collateral and guarantors (including guarantees and pledges of all assets by S1 on and after the Second-Step Merger); in the event ACI obtains corporate level and/or facility level ratings, maintenance of such rating(s); cash management and further assurances, compliance with material obligations under the acquisition documents; to the extent the Facility is funded prior to the completion of the Exchange Offer, completion of the Exchange Offer concurrently with the release of proceeds of the Facility from the Escrow Account in accordance with applicable law and the acquisition documents, without amendment or waiver or other modification of any of the terms or conditions thereof; using all commercially reasonable efforts to take or cause to be taken all corporate, stockholder and other action necessary to cause the Second-Step Merger to close as soon as practicable thereafter; including, in each case, exceptions and baskets to be mutually agreed upon by ACI and the lenders at all times on and following the completion of the Exchange Offer;
- negative covenants in respect of limitations with respect to other indebtedness (with \$250 million permitted for senior unsecured debt on terms and conditions to be determined); liens; negative pledges (provided that, for so long as the securities of \$1 constitute "margin stock" within the meaning of Regulation U, the negative pledges and restrictions on liens set forth in the loan documents shall not apply to such shares to the extent the value of such shares, together with the value of all other margin stock held by ACI and its subsidiaries, exceeds 25% of the total value of all assets subject to such covenants and agreements); restricted junior payments (with \$50 million permitted per year for dividends or stock repurchases plus, solely in the case of stock repurchases, an additional aggregate amount permitted from the closing date of the Second-Step Merger equal to the amount of qualified equity issued by ACI to the seller(s) of \$1 in connection with the acquisition in excess of \$225 million, in each case, provided (1) no event of default before or after giving effect to such restricted payment, (2) the pro forma Leverage Ratio is <2.00:1.00 and the Revolving Facility has pro forma unused commitments equal to or exceeding \$50 million; provided further that, subject to no event of default, if pro forma Leverage Ratio is <2.00:1.00 and the Revolving Facility has pro forma unused commitments equal to or exceeding \$50 million there will be no restrictions on restricted junior payments); restrictions on subsidiary distributions; investments, mergers and acquisitions (with permitted unlimited domestic acquisitions provided (a) no event of default before or after giving effect to such acquisition, (b) pro forma Leverage Ratio <2.50:1.00 and (c) pro forma leverage Ratio is <2.75:1.00 at the time of such acquisition and (iii) pro forma liquidity of \$50 million); and
- sales of assets (including subsidiary interests); sales and lease-backs; capital expenditures; transactions with affiliates (with a basket for intercompany loans existing as of the closing date of the Facility plus \$50 million incurred after the closing date of the Facility); conduct of business; amendments and waivers of organizational documents, junior indebtedness and other material agreements; and changes to fiscal year, including, in each case, exceptions and baskets to be mutually agreed upon by ACI and the lenders.

Notwithstanding anything to the contrary herein, prior to the closing date of the Second-Step Merger, the covenants set forth above shall be more restrictive in many respects, including, without limitation: (1) with respect to ACI and Offeror, no restricted junior payments; (2) with respect to Offeror, no investments or incurrence of any indebtedness and, except as expressly contemplated by the Commitment Letter, no activity other than as expressly required pursuant to the Exchange Offer documents; provided that there shall be no restrictions on the ability of Offeror to sell any shares so long as (a) such shares are sold for fair value and (b) the proceeds of such sale shall be held by Offeror as cash or approved cash equivalents; and (3) no amendment, waiver or other modification of any of the terms or conditions of the Second-Step Merger documents or any Exchange Offer documents (including, without limitation, changes to the percentage of the acquisition consideration constituting the Cash Consideration).

Events of Default

The loan documents for the Facility will include the following events of default (and, as appropriate, grace periods): failure to make payments when due; defaults under other agreements or instruments of indebtedness (with carve outs for cross default and cross acceleration provisions to other indebtedness that would otherwise subject the loans under the Facility to the requirements of Regulation U); certain events under hedging agreements; noncompliance with covenants; breaches of representations and warranties; bankruptcy; judgments in excess of specified amounts; ERISA; impairment of security interests in collateral; invalidity of guarantees; and "change of control" (to be defined in a mutually agreed upon manner by ACI and the lenders).

Certain Legal Matters; Regulatory Approvals

U.S. Antitrust Clearance

Under the HSR Act and the rules that have been promulgated thereunder, certain acquisition transactions may not be consummated unless certain information has been furnished to the Antitrust Division and the FTC and certain waiting period requirements have been satisfied. The exchange of S1 Shares pursuant to the Exchange Offer is subject to such requirements.

Pursuant to the requirements of the HSR Act, ACI filed a Notification and Report Form and requested early termination of the HSR Act waiting period with respect to the Exchange Offer and the Second-Step Merger with the Antitrust Division and the FTC on July 27, 2011. ACI withdrew its initial filing on August 26, 2011, and refiled it on August 29, 2011 in order to permit the Antitrust Division to have additional time to review the filing. On September 27, 2011, ACI withdrew its initial HSR filing and refiled it on September 28, 2011 in order to permit the Antitrust Division to have additional time to review the filing. The 30-calendar day waiting period recommenced in connection with such refiling so that it now expires, unless terminated earlier or extended, at 11:59 p.m., Eastern time, on October 28, 2011.

You should be aware that all required regulatory approvals may not be obtained in a timely manner, and this could result in a delay in the completion of the Exchange Offer. Although ACI has twice withdrawn and refiled its HSR Act filing prior to the date of this prospectus/offer to exchange in an effort to convince the DOJ staff of ACI's view as to the competitive nature of payment systems marketplace, there can be no assurance that the DOJ will concur with its belief that the transaction should be permitted to close. If ACI again withdraws and refiles its HSR Act filing, the DOJ issues a request for additional information or documentary material or the DOJ institutes an action challenging the transaction, the Expiration Time would be extended and the completion of the Exchange Offer could be prevented.

The Antitrust Division may extend the initial waiting period by issuing a Request for Additional Information and Documentary Material. In such an event, the statutory waiting period would extend until 30 days after ACI has substantially complied with the Request for Additional Information and Documentary Material, unless it is earlier terminated by the applicable antitrust agency. Thereafter, the waiting period can be extended only by court order or as agreed to by ACI. S1 Shares will not be accepted for exchange, or exchanged, pursuant to the Exchange Offer until the expiration or earlier termination of the applicable waiting period under the HSR Act.

At any time before or after the consummation of the Exchange Offer, one of the antitrust agencies could take such action under the antitrust laws as it deems necessary or desirable in the public interest, including seeking to enjoin the purchase of S1 Shares pursuant to the Exchange Offer or seeking divestiture of the shares so acquired or divestiture of ACI's or S1's assets. States and private parties may also bring legal actions under the antitrust laws. There can be no assurance that a challenge to the Exchange Offer on antitrust grounds will not be made, or if such a challenge is made, what the result will be

Other Regulatory Approvals

The Exchange Offer and the Second-Step Merger will also be subject to review by antitrust, insurance and other authorities in jurisdictions outside the U.S. ACI has filed and is in the process of filing as soon as practicable all applications and notifications determined by ACI to be necessary or advisable under the laws of the respective jurisdictions for the consummation of the Exchange Offer and the Second-Step Merger.

No assurance can be given that the required consents and approvals of the applicable governmental authorities to complete the Exchange Offer and Second-Step Merger will be obtained, and, if all required consents and approvals are obtained, no assurance can be given as to the terms, conditions and timing of the consents and approvals. If ACI agrees to any material requirements, limitations, costs, divestitures or restrictions in order to obtain any consents or approvals required to consummate the Exchange Offer, these requirements, limitations, additional costs or restrictions could adversely affect ACI's ability to integrate the operations of ACI and S1 or reduce the anticipated benefits of the combination contemplated by the Exchange Offer and Second-Step Merger.

Please see the sections of this prospectus/offer to exchange titled "Risk Factors" and "The Exchange Offer — Conditions of the Exchange Offer."

Other State Takeover Statutes

A number of other states have adopted laws and regulations applicable to attempts to acquire securities of corporations which are incorporated, or have substantial assets, stockholders, principal executive offices or principal places of business, or whose business operations otherwise have substantial economic effects, in such states. To the extent that these state takeover statutes (other than Section 203 of the DGCL) purport to apply to the Exchange Offer or the Second-Step Merger, ACI believes that there are reasonable bases for contesting such laws. In Edgar v. MITE Corp., the Supreme Court of the United States invalidated on constitutional grounds the Illinois Business Takeover Statute, which, as a matter of state securities law, made takeovers of corporations meeting certain requirements more difficult. However, in 1987 in CTS Corp. v. Dynamics Corp. of America, the Supreme Court held that the State of Indiana may, as a matter of corporate law and, in particular, with respect to those aspects of corporate law concerning corporate governance, constitutionally disqualify a potential acquiror from voting on the affairs of a target corporation without the prior approval of the remaining stockholders. The state law before the Supreme Court was by its terms applicable only to corporations that had a substantial number of stockholders in the state and were incorporated there. Subsequently, in TLX Acquisition Corp. v. Telex Corp., a Federal district court in Oklahoma ruled that the Oklahoma statutes were unconstitutional insofar as they apply to corporations incorporated outside Oklahoma because they would subject those corporations to inconsistent regulations. Similarly, in Tyson Foods, Inc. v. McReynolds, a federal district court in Tennessee ruled that four Tennessee takeover statutes were unconstitutional as applied to corporations incorporated outside Tennessee. This decision was affirmed by the United States Court of Appeals for the Sixth Circuit. In December 1988, a federal district court in Corporations i

S1, directly or through its subsidiaries, conducts business in a number of states throughout the United States, some of which have enacted takeover laws. ACI does not know whether any of these laws will, by their terms, apply to the Exchange Offer or the Second-Step Merger and has not complied with any such laws. Should any person seek to apply any state takeover law, ACI will take such action as then appears desirable, which may include challenging the validity or applicability of any such statute in appropriate court proceedings. In the event it is asserted that one or more state takeover laws is applicable to the Exchange Offer or the Second-Step Merger, and an appropriate court does not determine that it is inapplicable or invalid as applied to the Exchange Offer, ACI might be required to file certain information with, or receive approvals from, the relevant state authorities. In addition, if enjoined, ACI might be unable to accept for exchange any S1 Shares tendered pursuant to the Exchange Offer, or be delayed in continuing or consummating the

Exchange Offer and the Second-Step Merger. In such case, ACI may not be obligated to accept for exchange any S1 Shares tendered. Please see the section of this prospectus/offer to exchange titled "The Exchange Offer — Conditions of the Exchange Offer."

Going Private Transaction

The SEC has adopted Rule 13e-3 under the Exchange Act which is applicable to certain "going private" transactions and which may under certain circumstances be applicable to the Second-Step Merger or another business combination following the exchange of S1 Shares pursuant to the Exchange Offer in which ACI seeks to acquire the remaining S1 Shares not held by it. ACI believes that Rule 13e-3 should not be applicable to the Second-Step Merger; however, the SEC may take a different view under the circumstances. Rule 13e-3 requires, among other things, that certain financial information concerning S1 and certain information relating to the fairness of the proposed transaction and the consideration offered to minority stockholders in such transaction be filed with the SEC and disclosed to stockholders prior to consummation of the transaction.

Other

S1 and its subsidiaries conduct business in a number of jurisdictions outside of the United States. In connection with the acquisition of S1 Shares pursuant to the Exchange Offer, the laws of certain of these jurisdictions outside of the United States may require the filing of information with, or the obtaining of the approval of, governmental authorities therein. We intend to take such action as they may require, but no assurance can be given that such approvals will be obtained. If any action is taken before completion of the Exchange Offer by any such governmental authority, we may not be obligated to accept for payment or pay for any tendered S1 Shares. Please see the section of this prospectus/offer to exchange titled "The Exchange Offer — Conditions of the Exchange Offer."

Certain Relationships With S1 and Interests of ACI in the Exchange Offer

As of the date of the Exchange Offer, ACI beneficially owns 1,107,000 S1 Shares, representing approximately 2.0% of the outstanding S1 Shares. Purchase of these S1 Shares is described on Appendix B to this prospectus/offer to exchange. With the exception of the foregoing, ACI has not effected any transaction in securities of S1 in the past 60 days.

The name, citizenship, business address, business telephone number, principal occupation or employment, and five-year employment history for each of the directors and executive officers of ACI and Offeror and certain other information is set forth in Appendix A and Appendix B to this prospectus/offer to exchange. Except as described in this prospectus/offer to exchange and in Appendix A and Appendix B hereto, none of ACI, Offeror, or, after due inquiry and to the best of our knowledge and belief, any of the persons listed on Appendix A or Appendix B to this prospectus/offer to exchange, has during the last five years (i) been convicted in a criminal proceeding (except for matters that were dismissed without sanction or settlement) that resulted in a judgment, decree or final order enjoining the person from future violations of, or prohibiting activities subject to, federal or state securities laws or finding any violation of such laws. Except as set forth in this prospectus/offer to exchange and set forth in Appendix C to this prospectus/offer to exchange, after due inquiry and to the best of our knowledge and belief, none of the persons listed on Appendix A or Appendix B hereto, nor any of their respective associates or majority owned subsidiaries, beneficially owns or has the right to acquire any securities of S1 or has effected any transaction in securities of S1 during the past 60 days.

ACI does not believe that the Exchange Offer and the Second-Step Merger will result in a change in control under any of ACI's stock option plans or any employment agreement between ACI and any of its employees. As a result, no options or other equity grants held by such persons will vest as a result of the Exchange Offer and the Second-Step Merger.

Interest of Executive Officers and Directors of S1 in the Exchange Offer

In considering the recommendation of the S1 Board regarding the Exchange Offer and the Second-Step Merger, S1 stockholders should be aware that certain directors and officers of S1 may be deemed to have interests in the Exchange Offer and the Second-Step Merger that are different from or in addition to the interests of other S1 stockholders. The S1 Board was aware of these interests and considered them, among other matters, in approving the Transaction Agreement, the Exchange Offer and the Second-Step Merger and recommending that S1 stockholders accept the Exchange Offer by tendering their S1 Shares into the Exchange Offer and, if required by applicable law, approve the Second-Step Merger.

As a result of these interests, S1 directors and officers may have reasons for tendering their S1 Shares and, if necessary, voting to approve the Second-Step Merger that are not the same as your interests. S1 stockholders should consider whether these interests may have influenced these directors and officers to support or recommend the Exchange Offer and the Second-Step Merger.

Information on the interests of executive officers and directors of S1 in the Exchange Offer and the Second-Step Merger is more fully described in Amendment No. 2 to S1's Solicitation/Recommendation Statement on Form 14D-9.

Fees and Expenses

ACI has engaged Wells Fargo as a financial advisor with respect to the transaction. In connection with Wells Fargo's services as a financial advisor to ACI in connection with the transaction, ACI has agreed to pay Wells Fargo an aggregate fee of up to \$4 million. In addition, ACI will reimburse Wells Fargo for its reasonable out-of-pocket expenses, including the reasonable fees and expenses of its legal counsel. ACI has also agreed to indemnify Wells Fargo and its affiliates in connection with Wells Fargo's service as a financial advisor against certain liabilities in connection with their engagement.

ACI has also engaged Wells Fargo and Wells Fargo Bank to provide financing for the Exchange Offer and ACI has agreed to pay Wells Fargo and Wells Fargo Bank customary fees in respect thereof. As part of this engagement, ACI has agreed that Wells Fargo Bank will have the right to act as, among other roles, lead managers and lead left bookrunners in connection with any public or Rule 144A offering.

ACI has retained Innisfree M&A Incorporated ("Innisfree") as information agent in connection with the Exchange Offer. The information agent may contact holders of S1 Shares by mail, telephone, facsimile, telegraph, the internet, e-mail, newspapers and other publications of general distribution and in person and may request brokers, dealers, commercial banks, trust companies and other nominees to forward materials relating to the Exchange Offer to beneficial owners of S1 Shares. ACI will pay the information agent up to \$250,000 for these services and the solicitation and advisory services described below, in addition to reimbursing the information agent for its reasonable out-of-pocket expenses. ACI agreed to indemnify the information agent against certain liabilities and expenses in connection with the Exchange Offer.

ACI has also retained Innisfree for solicitation and advisory services in connection with certain solicitations described in this prospectus/offer to exchange, for which Innisfree will receive a customary fee. ACI has also agreed to reimburse Innisfree for out-of-pocket expenses and to indemnify Innisfree against certain liabilities and expenses, including reasonable legal fees and related charges.

In addition, ACI has retained Wells Fargo Bank as the exchange agent in connection with the Exchange Offer. ACI will pay the exchange agent reasonable and customary compensation for its services in connection with the Exchange Offer, will reimburse the exchange agent for its reasonable out-of-pocket expenses and will indemnify the exchange agent against certain liabilities and expenses.

Except as set forth above, ACI will not pay any fees or commissions to any broker, dealer or other person for soliciting tenders of shares pursuant to the Exchange Offer. ACI will reimburse brokers, dealers, commercial banks and trust companies and other nominees, upon request, for customary clerical and mailing expenses incurred by them in forwarding offering materials to their customers.

Accounting Treatment

The acquisition of S1 Shares by ACI will be accounted for under the acquisition method of accounting in accordance with Financial Accounting Standards Board's Accounting Standards Codification (ASC) 805, Business Combinations, and use the fair value concepts defined in ASC 820, Fair Value Measurements and Disclosures. ACI will be considered the acquirer of S1 for accounting purposes. In determining the acquirer for accounting purposes, ACI considered the factors required under U.S. GAAP.

ASC 805 requires, among other things, that assets acquired and liabilities assumed be recognized at their fair values as of the consummation of the offer. In addition, ASC 805 establishes that the consideration transferred be measured at the consummation of the offer at the then-current market price; this particular requirement will likely result in a per share equity component that is different from the amount assumed in the pro forma financial statements.

ASC 820 defines the term "fair value" and sets forth the valuation requirements for any asset or liability measured at fair value, expands related disclosure requirements and specifies a hierarchy of valuation techniques based on the nature of the inputs used to develop the fair value measures. Fair value is defined in ASC 820 as "the price that would be received to sell an asset or paid to transfer a liability in an orderly transaction between market participants at the measurement date." This is an exit price concept for the valuation of the asset or liability. In addition, market participants are assumed to be buyers and sellers in the principal (or the most advantageous) market for the asset or liability. Fair value measurements for an asset assume the highest and best use by these market participants. As a result of these standards, ACI may be required to record assets which are not intended to be used or sold and/or to value assets at fair value measures that do not reflect ACI's intended use of those assets. Many of these fair value measurements can be highly subjective and it is also possible that other professionals, applying reasonable judgment to the same facts and circumstances, could develop and support a range of alternative estimated amounts.

Under ASC 805 acquisition-related transaction costs (e.g., advisory, legal, valuation, other professional fees) and certain acquisition-related restructuring charges impacting the target company are not included as a component of consideration transferred but are accounted for as expenses in the periods in which the costs are incurred.

DESCRIPTION OF ACI CAPITAL STOCK

ACI's authorized capital stock consists of 70,000,000 shares of common stock, par value \$0.005 per share, and 5,000,000 authorized shares of preferred stock, par value \$0.01 per share. As of September 29, 2011, there were 33,503,529 shares of common stock outstanding (including 7,317,987 shares held in treasury) and no shares of preferred stock were outstanding. As of February 16, 2011, there were 215 holders of record of ACI's common stock.

The following description of the terms of the common stock and preferred stock of ACI is not complete and is qualified in its entirety by reference to ACI's amended and restated certificate of incorporation and its amended and restated by-laws. To find out where copies of these documents can be obtained, see "Where to Obtain More Information."

Common Stock

ACI's outstanding capital stock consists of a single class of common stock. Each share of common stock is entitled to one vote upon each matter subject to a stockholders vote and to dividends if and when declared by the ACI board of directors.

ACI Shares are listed on the NASDAQ Global Select Market under the symbol "ACIW."

Preferred Stock

The ACI Board is authorized to issue up to 5,000,000 shares of preferred stock in such series and to fix from time to time before issuance the number of shares to be included in any such series and the designation, relative powers, preferences, rights and qualifications, limitations or restrictions of such series. The ACI board has the power to fix the following terms of any series of the preferred stock:

- · the number of shares of any series and the designation to distinguish the shares of such series from the shares of all other series;
- · the voting powers, if any, and whether such voting powers are full or limited in such series;
- the redemption provisions, if any, applicable to such series, including the redemption price or prices to be paid;
- · whether dividends, if any, will be cumulative or noncumulative, the dividend rate of such series, and the dates and preferences of dividends on such series;
- · the rights of such series upon the voluntary or involuntary dissolution of, or upon any distribution of the assets of, ACI;
- the provisions, if any, pursuant to which the shares of such series are convertible into, or exchangeable for, shares of any other class or classes or of any other series of the same or any other class or classes of stock, or any other security, of ACI or any other corporation or other entity and the rates or other determinants of conversion or exchange applicable thereto;
- the right, if any, to subscribe for or to purchase any securities of ACI or any other corporation or other entity;
- $\bullet \hspace{0.4cm}$ the provisions, if any, of a sinking fund applicable to such series; and
- · any other relative, participating, optional or other special powers, preferences or rights and qualifications, limitations or restrictions thereof.

Organizational Documents

Various provisions contained in ACI's amended and restated certificate of incorporation and amended and restated by-laws could delay or discourage some transactions involving an actual or potential change in control

of ACI or its management and may limit the ability of ACI stockholders to remove current management or approve transactions that ACI stockholders may deem to be in their best interests. These provisions:

- authorize ACI's board of directors to establish one or more series of undesignated preferred stock, the terms of which can be determined by the board of directors at the time of issuance;
- provide an advanced written notice procedure with respect to stockholder proposals and the nomination of candidates for election as directors, other than nominations made by or at the direction of ACI's board of directors:
- state that special meetings of ACI's stockholders may be called only by the chairman of its board of directors, the president or the secretary; and
- allow ACI's directors, and not its stockholders, to fill vacancies on its board of directors, including vacancies resulting from removal or enlargement of the board.

COMPARISON OF STOCKHOLDERS' RIGHTS

As a result of the offer and the Second-Step Merger, holders of S1 Shares will become holders of ACI Shares. Both companies are Delaware corporations and are governed by the DGCL, so many of the differences between the rights of the stockholders of ACI and the current rights of the stockholders of S1 arise primarily from differences in their respective constituent documents.

The following is a summary of the material differences between the current rights of S1 stockholders and the current rights of ACI stockholders under Delaware law and their respective constituent documents. It is not a complete statement of the provisions affecting, and the differences between, the rights of S1 stockholders and ACI stockholders. This summary is qualified in its entirety by reference to Delaware law, as well as to ACI's amended and restated certificate of incorporation, its amended and restated by-laws, S1's amended and restated certificate of incorporation (as amended) and its amended and restated by-laws. Copies of these documents have been filed with the SEC and to find out where copies may be obtained, see the section entitled "Where You Can Find More Information."

	ACI	S1	
Authorized Capital	The authorized capital stock of ACI is (1) 70,000,000 shares of common stock, \$0.005 par value per share, and (2) 5,000,000 shares of preferred stock, \$0.01 par value per share.	The authorized capital stock of S1 is (1) $350,000,000$ shares of common stock, $\$0.01$ par value per share, and (2) $25,000,000$ shares of preferred stock, $\$0.01$ par value per share.	
Number of Directors	ACI's by-laws provide that, subject to the rights of any series of preferred stock to elect additional directors, the number of directors constituting the whole board shall be not less than three and not more than nine. ACI currently has eight directors.	S1's by-laws provide that the number of directors constituting the whole board will be not less than four and not more than fifteen as may be fixed from time to time by its board of directors. S1 currently has seven directors and one vacancy.	
Structure of Board of Directors; Term of Directors	ACI has one class of directors, and ACI's charter does not provide for a classified board. ACI's directors are elected for a one year term.	S1's charter provides for a classified board divided into three classes. $S1$'s directors are elected for a term of three years.	
Removal of Directors	Any director of ACI may be removed, with or without cause, by the holders of a majority of the shares then entitled to vote at an election of directors.	S1's charter provides that no director may be removed except for cause and then only by an affirmative vote of at least two-thirds of the voting stock of S1 at a duly constituted meeting of stockholders called for such purpose. At least 30 days prior to such meeting of stockholders, written notice will be sent to the director or directors whose removal will be considered at such meeting.	
Vacancies on the Board of Directors	ACI's charter provides that vacancies and newly created directorships shall be filled solely by a majority vote of the remaining directors then in office, although fewer than a quorum, or by a sole remaining director.	S1's by-laws provide that vacancies and newly created directorships may be filled by the stockholders or by a majority of the directors then in office, although fewer than a quorum, or by a sole remaining director.	

Special Meetings of Stockholders

Action by Written Consent

ACI ACI's by-laws provide that special meetings of the stockholders may be called only by (1) the chairman of the board, (2) the president, or (3) the secretary within 10 calendar days after receipt of the written request of a majority of the total number of directors that ACI would have if there were no vacancies. Any such request must be sent to the chairman of the board and the secretary and must state the purpose or purpose of the propose meeting. Special meetings of holders of the outstanding preferred stock of ACI, if any, may be called in the manner and for the purposes provided in the applicable preferred stock designation (as defined in ACI's chatter).

Pursuant to ACI's by-laws, ACI's stockholders are permitted to take action by written consent, in lieu of a stockholders' meeting, if such written consent is signed by persons who hold shares having voting power to cast not less than the minimum number of votes necessary to authorize such action at a stockholder meeting at which all stockholders entitled to vote were present and voted.

S1 S1's by-laws provide that special meetings may be called by the chairman of the board, the president or a majority of the board of directors, and will be called by the chairman of the board, the president, or, the secretary upon the written request of the holders of not less than one tenth of all of the outstanding parlial stock of 51 entitled to vote at the meeting. Such written request will state the purpose of the meeting and will be delivered to the principal office of \$1 addressed to the chairman of the board, the president or the secretary. Special meetings relating to change in control of \$1 or amendments to its charrer will be called only by the board of directors. Written notice to each stockholder is required not less than ten nor more than 60 days before the meeting.

Pursuant to S1's by-laws, S1 stockholders are permitted to take action by written consent, in lieu of a stockholders' meeting, if such written consent is signed by persons who hold shares having voting power to cast not less than the minimum number of votes necessary to authorize such action at a stockholder meeting at which all stockholders entitled to vote were present and voted.

Stockholder Proposals

ACI

ACI and ACI's by-laws provide that business to be brought before an annual meeting must be (1) specified in the notice of the annual meeting (or any supplement thereto) given by or at the direction of the board of directors, (2) otherwise properly brought before the annual meeting by the presiding officer or by or at the direction of a majority of the board of directors, or (3) otherwise properly requested to be brought before the annual meeting by a stockholder. To properly bring business (a) the stockholder usb to a stockholder of ACI of record at the time of the giving of the notice for such annual meeting, (b) the stockholder must be entitled to vote at such meeting, (c) the stockholder must be use given timely notice thereof in writing to the secretary, and (d) if the stockholder, or the beneficial owner on whose behalf any business is brought before the meeting, has provided ACI with a proposal solicitation notice, such stockholder or beneficial owner must have delivered a proxy statement and form of proxy to the holders of at the least the percentage of shares of ACI entitled to vote that are required to approve such business that the stockholder proposes to bring before the annual meeting and included in such materials.

included in such materials.

To be timely, a stockholder's notice must be delivered to or mailed and received at the principal executive offices of ACI not less than 90 calendar days nor greater than 120 calendar days prior to the first anniversary of the date of the immediately preceding year's annual meeting of stockholders, provided, however, that if the date of the annual meeting is advanced more than 30 calendar days prior to or delayed by more than 30 calendar days after the anniversary of the preceding year's annual meeting, notice by the stockholder to be timely must be so delivered not later than the close of business on the later of (1) the 90th calendar day prior to such annual meeting and (2) the 10th calendar day following the day on which public disclosure of the date of such meeting is first made. In no event shall the public disclosure of an adjournment of an annual meeting commence a new time period for the giving of a stockholder's notice as described above.

S1's by-laws provide that business to be brought before an annual meeting must be (1) specified in the notice of the meeting, (2) brought before the meeting by the board of directors, or (3) otherwise properly requested by a stockhold or the stock of t

otherwise properly requested by a stockholder.

To properly bring business, the stockholder generally must deliver a notice to the secretary at the principal executive offices not less than 90 nor more than 120 calendar days prior to the first anniversary of the previous year's annual meeting; provided, however, that if the annual meeting is called for a date (1) not within 60 calendar days before or after to the anniversary date and (2) less than 60 days notice or public disclosure of the date of the meeting is given to of the annual meeting was mailed or publicly disclosure of the notice must identify (a) a brief description of the business desired to be brought before the annual meeting and the reasons for conducting such business at the annual meeting, (b) the name and address, as they appear on S1's books, of the stockholder proposing such business, (c) the class and number of shares of S1 which are beneficially owned by the stockholder, and (d) any material interest of the stockholder in such business. The chairman of an annual meeting will, if the facts warrant, determine and declare to the annual meeting that a matter of business was not properly brought before the meeting, and if he should so determine, he will so declare to the meeting and any such business not properly brought before the meeting, and different and was not properly brought before the meeting and any such business not properly brought before the meeting and any such business are not properly brought before the meeting and any such business was not properly brought before the meeting and any such business.

ACI

Astockholder's notice to the secretary must set forth as to each matter the stockholder proposes to bring before the annual meeting (1) a description in reasonable detail of the business desired to be brought before the annual meeting and the reasons for conducting such business at the annual meeting. (2) the name and address, as they appear on ACI's books, of the stockholder proposing such business and the beneficial owner, if any, on whose behalf the proposal is made, (3) the class and series and number of shares of capital stock of ACI that are owned beneficially and of record by the stockholder proposing such business and by the beneficial owner, if any, on whose behalf the proposal is made, (4) a description of all arrangements or understandings among such stockholder and any other person or persons (including their names) in connection with the proposal of such business by such stockholder and any material interest of such stockholder a proxy statement and form of proxy to holders of at least the percentage of shares of ACI entitled to vote that are required to approve the proposal, and (6) a representation that such stockholder intends to appear in person or by proxy at the annual meeting to bring such business before the annual meeting. These requirements are in addition to any SEC requirements.

requirements.

For purposes of this provision, "public disclosure" means disclosure in a press release reported by the Dow. Jones News Service, Associated Press or comparable national news service or in a document filed by ACL with the SEC pursuant to the Exchange Act or furnished by ACL to stockholders. Nothing in this provision of ACI's by-laws will be deemed to affect any rights of stockholders to request inclusion of proposals in ACI's proxy statement pursuant to Rule 14a-8 under the Exchange Act.

Stockholder Nominations

ACI's by-laws provide that any stockholder entitled to vote in the election of directors generally may nominate one or more persons for election as directors at a meeting only if written notice of such stockholder's intent to make such nomination or nominations has been received by the secretary of ACI not less than 90 calendar days nor greater than 120 calendar days prior to the first anniversary of the date of the immediately preceding year's annual meeting of stockholders; provided, however, that if the date of the annual meeting is advanced more than 30 calendar days prior to or delayed by more than 30 calendar days after the anniversary of the preceding year's annual meeting, notice by the stockholder to be timely must be so delivered not later than the close of business on the later of (1) the 90th calendar day prior to such annual meeting and (2) the 10th calendar day following the day on which public disclosure of the date of such meeting is first made. In no event shall the public disclosure of an adjournment of an annual meeting commence a new time period for the giving of a stockholder's notice as described above.

described above.

Each such notice shall set forth: (1) the name and address of the stockholder who intends to make the nomination and of the beneficial owner, if any, on whose behalf the nomination is made; (2) a representation that the stockholder is a holder of record of ACI's common stock entitled to vote for the election of directors on the date of such notice and intends to appear in person or by proxy at the meeting to nominate the person or persons specified in the notice; (3) the class and number of shares owned beneficially and of record by the stockholder giving notice and by the beneficial owner, if any, on whose behalf the nomination is made; (4) a description of all arrangements or understandings between or among the stockholder, the beneficial owner on whose behalf the notice is given and each nominee and any other person to persons (naming such person or persons) pursuant to which the nomination or nominations are to be made by the stockholder; (5) such other information regarding each nominee proposed by such stockholder as would be required to be included in a proxy statement filed pursuant to be proxy rules of the SEC, had the nominee been nominated, or intended to be nominated, by ACI's board of directors; (6) the consent of each nominee to serve as a director of ACI if so elected; and (7) whether the stockholder, or the beneficial owner on whose behalf the nomination is made, intends to deliver a proxy statement and form of proxy to holders of at least the percentage of shares of our common stock entitled to vote that are required to elect the nominee(s).

S1
S1's by-laws provide that any stockholder who is a stockholder of record at the time of giving the requisite notice and on the record date for the determination of stockholders entitled to notice of and to vote at such meeting and who gives proper notice may nominate candidates to stand for election as directors. To be proper, a stockholder's notice with respect to an annual meeting generally must be delivered to \$1's principal executive offices not less than 90 nor more than 120 calendar days prior to the first anniversary of the previous year's annual meeting; provided, however, that if (1) the annual meeting is called for a date not within 60 calendar days before or after the anniversary date and (2) less than 60 days notice or prior public disclosure of the date of the meeting is given to stockholders, the notice must be received within 10 days of the date on which notice of the date of the annual meeting was mailed or publicly disclosed.

The notice must contain certain information, including information regarding the stockholder and the nominee. These requirements are in addition to any SEC requirements. The chairman of the meeting will, if the facts warrant, determine and declare to the meeting that a nomination was not made in accordance with procedures prescribed by the by-laws, and if he should so determine, he will so declare to the meeting and the defective nomination will be disregarded.

ACI

ACI
In addition to the name and address of the stockholder making the nomination, as they appear on ACI's books, the notice must also include the name and principal business address of all (1) persons controlling, directly or indirectly, or acting in concert with, such stockholder, (2) beneficial owners of shares of stock of ACI owned of record or beneficially by such stockholder (with the term 'beneficial ownership'' as used herein to have the meaning given to that term in Rule 13d-3 under the Exchange Acr) and (3) persons controlling, controlled by, or under common control with, any person specified in the foregoing clause (1) or (2) (with the term 'control' as used herein to have the meaning given to that term in Rule 405 under the Securities Act of 1933, as amended) (any such person or beneficial owners set forth in the foregoing clauses (1), (2) and (3) shall be a "Stockholder Associated Person").

The stockholder notice must also disclose (1) any derivative positions related to any class or series of securities of ACI held or beneficially held by the stockholder and each Stockholder Associated Person; and (2) whether and the extent to which any hedging, swap or other transactions or series of transactions have been entered into by or on behalf of, or any other agreement, arrangement or understanding (including any short position or any borrowing or lending of shares of stock) has been made, the effect or intent of which is to mitigate loss to, or manage risk of stock price changes for, or to increase the voting power of, the stockholder or any Stockholder Associated Person with respect to any shares of stock to ACI.

To be eligible to be a nominee for election or re-election as a director of ACI, the board of directors may require a person to deliver to the secretary at the principal executive offices of ACI, a written questionnaire with respect to the identity, background and qualification of such person and the background of any other person or entity on whose behalf the nomination is being made and a written representation and agreement regarding certain matters.

oeing inate and a written representation and agreement regarding certain inaters.

ACI's charter provides that certain provisions of ACI's charter relating to directors may only be amended by the majority vote of all classes of voting stock. Under Delaware law, ACI's board of directors must adopt a resolution recommending an amendment and call a special meeting of the stockholders (or propose consideration of the resolution at the next annual meeting) to approve the amendment.

	ACI	S1
Amendment of By-Laws	ACI's by-laws may be amended by its stockholders or its board of directors, provided that no amendment adopted by the board of directors may vary or conflict with any amendment adopted by the stockholders. Amendment of certain by-laws requires a majority vote of all classes of voting stock issued and outstanding.	S1's by-laws may be amended by its board of directors or the stockholders as provided under the DGCL.
Limitations on Director Liability	To the fullest extent permitted by the DGCL or any other applicable law, no director of ACI will be personally liable to ACI or its stockholders for or with respect to any acts or omissions in the performance of his or her duties as a director of ACI.	No director of S1 will be liable to S1 or its stockholders for monetary damages for breach of fiduciary duty as a director, provided that this provision will not eliminate or limit the liability of a direct or [1] for any breach of the director's duty of loyalty to S1 or its stockholders, (2) for acts or omissions not in good faith or which involve intentional misconduct or a knowing violation of law, (3) for the types of liability set forth in Section 174 of the DGCL, or (4) for any transaction from which the director received any improper personal benefit.
Dividends	ACI does not declare regular cash dividends.	S1 does not declare regular cash dividends.
Stockholder Rights Plan	ACI does not have a stockholder rights plan.	S1 does not have a stockholder rights plan.
Restrictions on Transactions With "Interested Stockholders"	$\ensuremath{\mathrm{ACI}}$ has not opted out from Section 203, and therefore Section 203 of the DGCL is applicable to ACI.	${\rm S1}$ has not opted out from Section 203, and therefore Section 203 of the DGCL is applicable to ${\rm S1}.$

UNAUDITED CONDENSED COMBINED PRO FORMA FINANCIAL INFORMATION

The following unaudited pro forma condensed combined statements of operations for the year ended December 31, 2010 and for the six months ended June 30, 2011 are presented on a pro forma basis to give effect to the Exchange Offer and related transactions as if they had been completed on January 1, 2010. The following unaudited pro forma condensed combined balance sheet as of June 30, 2011 is presented on a pro forma basis to give effect to the Exchange Offer and related transactions as if they had been completed on June 30, 2011.

The following unaudited pro forma condensed combined financial statements, or the "pro forma financial statements," were derived from and should be read in conjunction with:

- the consolidated financial statements of ACI as of and for the year ended December 31, 2010 and the related notes included in the ACI 10-K, which is incorporated by reference into this prospectus/offer to exchange;
- the consolidated financial statements of S1 as of and for the year ended December 31, 2010 and the related notes included in the S1 10-K, which is incorporated by reference into this prospectus/offer to exchange;
- the consolidated financial statements of ACI as of and for the six months ended June 30, 2011 and the related notes included in the ACI 10-Q, which is incorporated by reference into this prospectus/offer to exchange; and
- the consolidated financial statements of S1 as of and for the six months ended June 30, 2011 and the related notes included in the S1 10-Q, which is incorporated by reference into this prospectus/offer to exchange.

The consolidated financial statements of ACI and S1 as of and for the six months ended June 30, 2011 and for the year ended December 31, 2010 have been adjusted in the pro forma financial statements to give effect to items as disclosed in Note 4. The pro forma financial statements should be read in conjunction with the accompanying notes to the pro forma financial statements.

The unaudited pro forma adjustments were based on publicly available information, including the ACI 10-K, the ACI 10-Q, the S1 10-K and the S1 10-Q. The unaudited pro forma adjustments were also based on certain assumptions and estimates that ACI believes are reasonable based on such publicly available information. Additional information may exist that could materially affect the assumptions and estimates and related pro forma adjustments. Pro forma adjustments have been included only to the extent appropriate information is known, factually supportable and reasonably available to ACI.

The pro forma financial statements assume, among other things, that upon consummation of the offer all outstanding S1 Shares are acquired by ACI for \$9.68 with S1 stockholders making a cash and stock election, subject to proration of 66.2% Cash Consideration and 33.8% Stock Consideration.

The pro forma financial statements have been presented for informational purposes only. The pro forma financial statements are not necessarily indicative of what the combined company's financial position or results of operations actually would have been had the Exchange Offer been completed as of the dates indicated. In addition, the pro forma financial statements do not purport to project the future financial position or operating results of the combined company. There were no material transactions between ACI and S1 during the periods presented in the pro forma financial statements that would need to be eliminated.

The pro forma financial statements have been prepared using the acquisition method of accounting under U.S. GAAP. ACI has been treated as the acquirer in the Exchange Offer for accounting purposes. Assumptions and estimates underlying the pro forma adjustments are described in the accompanying notes, which should be read in conjunction with the pro forma financial statements.

Acquisition accounting is dependent upon certain valuations and other studies that have not yet begun or are not yet completed, and will not be completed until after the closing of the offer. Accordingly, the pro forma adjustments are preliminary and have been made solely for the purpose of preparing the pro forma

financial statements and are based upon preliminary information available at the time of the preparation of this prospectus/offer to exchange. Differences between these preliminary estimates and the final acquisition accounting will occur and these differences could have a material impact on the pro forma financial statements and the combined company's future results of operations and financial position.

The pro forma financial statements do not reflect any cost savings or other synergies that the combined company may achieve as a result of the offer or the costs to integrate the operations of ACI and S1 or the costs necessary to achieve these cost savings and other synergies. The effects of the foregoing items could, individually or in the aggregate, materially impact the pro forma financial statements.

The following table presents unaudited condensed combined pro forma balance sheet data at June 30, 2011 (expressed in thousands of U.S. dollars, except share and per share data) giving effect to the proposed acquisition of S1 Shares as if such acquisition had occurred at June 30, 2011:

Unaudited Pro Forma Condensed Combined Balance Sheet As of June 30, 2011

	Wor	ACI Worldwide, Inc.		Reclassified S1 Corporation (Note 2)		Pro Forma Adjustments (Note 4)		Pro Forma Combined
ASSETS		_			<u> </u>			
Current assets								
Cash and cash equivalents	\$	170,807	\$	71,720	\$	(100,000)(a)	\$	142,527
Billed receivables, net of allowances for doubtful accounts		71,256		45,092		_		116,348
Accrued receivables		9,824		9,257		_		19,081
Income taxes receivable		_		1,953				1,953
Deferred income taxes, net		11,292		2,639		_		13,931
Prepaid expenses		14,531		4,612		_		19,143
Other current assets		10,470		4,167		_		14,637
Total current assets		288,180		139,440		(100,000)		327,620
Property and equipment, net		22,292		21,196		_		43,488
Software, net		25,357		3,098		_		28,455
Goodwill		219,315		148,236		290,714(b)		658,265
Other intangible assets, net		21,762		7,313				29,075
Deferred income taxes, net		28,776		_		_		28,776
Other noncurrent assets		7,965		7,830		11,688(c)		27,483
TOTAL ASSETS	\$	613,647	\$	327,113	\$	202,402	\$	1,143,162
LIABILITIES AND STOCKHOLDERS' EQUITY					_		_	
Current liabilities								
Accounts payable	\$	12,703	\$	11,975	\$	_	\$	24,678
Accrued employee compensation		23,127		14,249		2,094(d)		39,470
Deferred revenue		131,735		50,018				181,753
Income taxes payable		1,784		375		_		2,159
Alliance agreement liability		1,600		_		_		1,600
Note payable under credit facility		75,000		36		(75,036)(e)		_
Current portion of note payable		_		_		8,750(f)		8,750
Accrued and other current liabilities		19,722		3,693		_		23,415
Total current liabilities		265,671		80,346		(64,192)		281,825
Deferred revenue		30,035						30,035
Long term note payable		_		_		356,733(f)		356,733
Alliance agreement noncurrent liability		20,667		_				20,667
Other noncurrent liabilities		17,734		3,084		_		20,818
Total liabilities		334,107		83,430		292,541		710,078
Stockholders' equity	·					<u></u>		
Preferred stock		_		_		_		_
Common stock		204		539		(480)(g)		263
Common stock warrants		24,003		_		_		24,003
Treasury stock		(167,286)		_		_		(167,286)
Additional paid-in capital		316,695		1,805,627		(1,637,374)(h)		484,948
Retained earnings		116,711		(1,561,628)		1,546,860(i)		101,943
Accumulated other comprehensive loss		(10,787)		(855)		855(j)		(10,787)
Total stockholders' equity		279,540		243,683		(90,139)		433,084
TOTAL LIABILITIES AND STOCKHOLDERS' EQUITY	\$	613,647	\$	327,113	\$	202,402	\$	1,143,162

See accompanying Notes to Unaudited Pro Forma Condensed Combined Financial Statements, which are an integral part of these statements.

The following table sets forth unaudited condensed consolidated pro forma results of operations for the year ended December 31, 2010 (expressed in thousands of U.S. dollars, except share and per share data) giving effect to the proposed acquisition of S1 Shares as if such acquisition had occurred at January 1, 2010:

Unaudited Pro Forma Condensed Combined Statement of Operations For the Year Ended December 31, 2010

	Wor	ACI rldwide, Inc.	S1 Corporation		Pro Forma Adjustments n (Note 4)		ro Forma combined
Revenues:							
Software license fees	\$	164,559	\$	26,237	\$	_	\$ 190,796
Maintenance fees		135,523		63,034		_	198,557
Services		73,989		65,180		_	139,169
Software hosting fees		44,353		54,635			 98,988
Total revenues		418,424		209,086		_	627,510
Expenses:							,
Cost of software license fees(1)		12,591		2,242		_	14,833
Cost of maintenance, services, and hosting fees(1)		117,132		82,778		27,595(k)	227,505
Cost of hosting		_		27,595		(27,595)(k)	_
Research and development		74,076		35,508		_	109,584
Selling and marketing		70,553		28,172		_	98,725
General and administrative		70,096		27,134		_	97,230
Depreciation and amortization		20,328		10,161			 30,489
Total expenses		364,776		213,590		_	578,366
Operating income (loss)		53,648		(4,504)			49,144
Other income (expense):							
Interest income		665		214		_	879
Interest expense		(1,996)		(455)		(9,333)(1)	(11,784)
Other, net		(3,615)		(1,367)		_	(4,982)
Total other income (expense)		(4,946)		(1,608)		(9,333)	(15,887)
Income (loss) before income taxes		48,702		(6,112)		(9,333)	33,257
Income tax expense (benefit)		21,507		171		(3,267)(m)	18,411
Net income (loss)	\$	27,195	\$	(6,283)	\$	(6,066)	\$ 14,846
Income (loss) per share information							
Weighted average shares outstanding							
Basic		33,560		52,495		5,850(n)	39,410
Diluted		33,870		52,495		5,850(n)	39,720
Income (loss) per share							
Basic	\$	0.81	\$	(0.12)			\$ 0.38
Diluted	\$	0.80	\$	(0.12)			\$ 0.37

⁽¹⁾ The cost of software license fees excludes charges for depreciation but includes amortization of purchased and developed software for resale. The cost of maintenance, services, and hosting fees excludes charges for depreciation.

See accompanying Notes to Unaudited Pro Forma Condensed Combined Financial Statements, which are an integral part of these statements.

The following table sets forth unaudited condensed consolidated pro forma results of operations for the six months ended June 30, 2011 (expressed in thousands of U.S. dollars, except share and per share data) giving effect to the proposed acquisition of S1 Shares as if such acquisition had occurred at January 1, 2010:

Unaudited Pro Forma Condensed Combined Statement of Operations For the Six Months Ended June 30, 2011

	Wor	ACI Worldwide, Inc.		S1 Pro Forma Adjustments Corporation (Note 4)		justments	ro Forma combined
Revenues:							
Software license fees	\$	89,809	\$	17,959	\$	_	\$ 107,768
Maintenance fees		72,265		33,108		_	105,373
Services		34,044		41,826		_	75,870
Software hosting fees		21,791		28,272			 50,063
Total revenues		217,909		121,165			 339,074
Expenses:							
Cost of software license fees(1)		7,578		1,124		_	8,702
Cost of maintenance, services, and hosting fees(1)		61,425		48,056		14,376(k)	123,857
Cost of hosting		_		14,376		(14,376)(k)	_
Research and development		46,914		17,320		_	64,234
Selling and marketing		41,085		14,489		_	55,574
General and administrative		32,166		16,312		_	48,478
Depreciation and amortization		10,821		5,108			 15,929
Total expenses		199,989		116,785			 316,774
Operating income		17,920		4,380		_	 22,300
Other income (expense):							
Interest income		434		113		_	547
Interest expense		(1,017)		(206)		(4,511)(l)	(5,734)
Other, net		(42)		(928)			 (970)
Total other income (expense)		(625)		(1,021)		(4,511)	(6,157)
Income (loss) before income taxes		17,295		3,359		(4,511)	16,143
Income tax expense (benefit)		5,873		1,170		(1,579)(m)	5,464
Net income (loss)	\$	11,422	\$	2,189	\$	(2,932)	\$ 10,679
Income (loss) per share information							,
Weighted average shares outstanding							
Basic		33,383		53,475		5,850(n)	39,233
Diluted		34,120		54,277		5,850(n)	39,970
Income (loss) per share							
Basic	\$	0.34	\$	0.04			\$ 0.27
Diluted	\$	0.33	\$	0.04			\$ 0.27

⁽¹⁾ The cost of software license fees excludes charges for depreciation but includes amortization of purchased and developed software for resale. The cost of maintenance, services, and hosting fees excludes charges for depreciation.

See accompanying Notes to Unaudited Pro Forma Condensed Combined Financial Statements, which are an integral part of these statements.

Notes to the Unaudited Pro Forma Condensed Combined Financial Statements

1. Description of Transaction

On October 3, 2011, ACI and S1 entered into the Transaction Agreement, in which ACI agreed to acquire S1 pursuant to the Exchange Offer for a price per S1 Share equal to, at the election of the holder thereof, 0.3148 ACI Shares or \$10.00 in cash for each S1 Share (less applicable withholding taxes and without interest), subject to proration.

At October 12, 2011, the last trading day prior to the date of this prospectus/offer to exchange, the closing trading price for ACI Shares was \$28.77 per share. Based on the closing trading ACI Share price as of October 12, 2011, the Cash-Stock Consideration had a blended value of \$9.68 as of such date, assuming full proration.

The pro forma financial statements do not give effect to ACI's beneficial ownership of S1 Shares as of the date of this prospectus/offer to exchange.

2. Basis of Presentation

These pro forma financial statements were prepared using the acquisition method of accounting in accordance with Financial Accounting Standards Board's Accounting Standards Codification (ASC) 805, Business Combinations, and use the fair value concepts defined in ASC 820, Fair Value Measurements and Disclosures. Certain reclassifications have been made to the historical financial statements of \$1\$ to conform

with ACI's presentation, primarily related to showing balances on the balance sheet that S1 only shows in their footnotes as detailed in the following table:

		As of June 30, 2011				
		Historical S1 Corporation		nssification Is and unaudited)		eclassified Corporation
ASSETS						
Current assets						
Cash and cash equivalents	\$	71,720	\$	_	\$	71,720
Billed receivables, net of allowances for doubtful accounts		_		45,092		45,092
Accrued receivables		_		9,257		9,257
Accounts receivable, net		54,349		(54,349)		_
Income taxes receivable		_		1,953		1,953
Deferred income taxes, net		_		2,639		2,639
Prepaid expenses		4,612		_		4,612
Other current assets		8,759		(4,592)		4,167
Total current assets		139,440				139,440
Property and equipment, net		21,196		_		21,196
Software, net		_		3,098		3,098
Goodwill		148,236		_		148,236
Other intangible assets, net		10,411		(3,098)		7,313
Other noncurrent assets		7,830		_		7,830
TOTAL ASSETS	\$	327,113	\$	_	\$	327,113
LIABILITIES AND STOCKHOLDERS' EQUITY Current liabilities						
Accounts payable	\$	11,975	\$	_	\$	11,975
Accrued employee compensation		14,249		_		14,249
Deferred revenue		50,018		_		50,018
Income taxes payable		375		_		375
Accrued restructuring		412		(412)		_
Note payable under credit facility		_		36		36
Current portion of debt obligation		36		(36)		_
Current liabilities		3,281		412		3,693
Total current liabilities		80,346				80,346
Other noncurrent liabilities		3,084		_		3,084
Total liabilities		83,430		_		83,430
Stockholders' equity						
Common stock		539		_		539
Additional paid-in capital		1,805,627		_		1,805,627
Retained earnings		(1,561,628)		_		(1,561,628)
Accumulated other comprehensive loss		(855)		_		(855)
Total stockholders' equity		243,683		_		243,683
TOTAL LIABILITIES AND STOCKHOLDERS' EQUITY	\$	327,113	\$		\$	327,113
	<u>* </u>	02.,0				02.,.10

ASC 805 requires, among other things, that assets acquired and liabilities assumed be recognized at their fair values as of the consummation of the combination. In addition, ASC 805 establishes that the consideration transferred be measured at the consummation of the combination at the then-current market price; this particular requirement will likely result in a per share equity component that is different from the amount assumed in the pro forma financial statements.

ASC 820 defines the term "fair value" and sets forth the valuation requirements for any asset or liability measured at fair value, expands related disclosure requirements and specifies a hierarchy of valuation techniques based on the nature of the inputs used to develop the fair value measures. Fair value is defined in ASC 820 as "the price that would be received to sell an asset or paid to transfer a liability in an orderly transaction between market participants at the measurement date." This is an exit price concept for the valuation of the asset or liability. In addition, market participants are assumed to be buyers and sellers in the principal (or the most advantageous) market for the asset or liability. Fair value measurements for an asset assume the highest and best use by these market participants. As a result of these standards, ACI may be required to record assets which are not intended to be used or sold and/or to value assets at fair value measures that do not reflect ACI's intended use of those assets. Many of these fair value measurements can be highly subjective and it is also possible that other persons, applying reasonable judgment to the same facts and circumstances, could develop and support a range of alternative settingted amounts.

Under ASC 805 acquisition-related transaction costs, such as advisory, legal, valuation, other professional fees, and certain acquisition-related restructuring charges impacting the target company are not included as a component of consideration transferred but are accounted for as expenses in the periods in which the costs are incurred. Total advisory, legal, regulatory, valuation costs and change in control payments expected are estimated to be approximately \$26.5 million. These anticipated costs for ACI are reflected in the unaudited pro forma condensed combined balance sheet as a reduction to retained earnings and an increase in debt.

3. Estimate of Consideration Expected to be Transferred

The following is a preliminary estimate of consideration expected to be transferred to consummate the offer:

	ousands, except share d per share amounts
S1 Basic Shares of Common Stock Outstanding	54,983,593
Exchange Ratio	 0.1064
Total Shares of ACI to be Issued	5,850,254
ACI Closing Share Price on October 12, 2011	\$ 28.77
Total Value of ACI Common Shares to be Issued	168,312
Total Cash Consideration	 363,991
Total Purchase Price	\$ 532,303

"S1 Basic Shares of Common Stock Outstanding" in the above table is based upon the 54,983,593 shares outstanding as of September 29, 2011 per the Transaction Agreement.

The preliminary purchase price will fluctuate with changes in the trading price of ACI Shares. A 10% increase or decrease in the \$28.77 price of ACI Shares as of October 12, 2011 used in the preliminary purchase price calculation above would increase or decrease the purchase price by approximately \$16.8 million.

The table below represents a preliminary allocation of the purchase price as of June 30, 2011:

	(III tilousalius)
Book value of net assets acquired as of June 30, 2011	\$ 243,683
Adjustments to:	
Eliminate S1's historical goodwill	(148,236)
Increase liability for S1's stock appreciation rights	(2,094)
Allocate excess purchase price to goodwill	438,950
	\$ 532,303

The table above does not reflect adjustments for the fair value of intangible assets acquired. ACI expects to allocate a portion of the purchase price to developed technology, trade names and customer relationships. In addition, the pro forma statement of operations does not reflect amortization of the fair value adjustments to

other intangible assets, including developed technology, trade names and customer relationships. For each \$1.0 million of purchase price allocated to intangible assets assuming a 7 year estimated life, amortization expense will increase by \$0.1 million and income before taxes will decrease by \$0.1 million.

The table above also does not reflect adjustments to deferred taxes related to book/tax basis differences that may result from the purchase price allocation.

4. Pro Forma Adjustments

Adjustments included in the column under the heading "Pro Forma Adjustments" represent the following:

- (a) To adjust cash for the \$100 million in ACI cash on hand expected to be paid as a part of the acquisition.
- (b) To adjust goodwill as follows:

	(III tilousullus)
Eliminate S1 existing goodwill	\$ (148,236)
Goodwill for acquisition of S1	438,950
Total	\$ 290,714

- (c) To adjust other noncurrent assets for \$11.7 million in debt issuance costs on the revolving credit facility and senior note secured for financing of the transaction.
- (d) To adjust accrued employee compensation for the additional liability to cash settle S1's outstanding stock appreciation rights based upon S1's June 30, 2011 closing price of \$7.48 and the blended value of the Cash-Stock Consideration of \$9.68 as of October 12, 2011.
- (e) To adjust for the payoff of S1 and ACI's existing outstanding debt.
- (f) To adjust for ACI's new revolving credit facility and senior note secured to finance the transaction, including the current portion under the senior note.

ACI has obtained commitments from Wells Fargo to arrange, and Wells Fargo Bank to provide, subject to certain conditions, senior bank financing consisting of up to \$450 million under a proposed new secured credit facility, comprising of the Term Facility and the Revolving Facility (together, the Facility) for financing the cash component of the consideration to be paid to \$1's stockholders in connection with the Exchange Offer. Additionally, ACI will have the right, not the obligation, to increase the amount of the Facility by incurring an incremental term loan facility or increasing the Revolving Facility in an aggregate principal amount not to exceed \$75 million, subject to certain conditions and under terms to be determined.

Each loan made under the Facility will bear interest at an Adjusted LIBOR Rate or Alternate Base Rate (as contemplated by the commitment letter relating to the Facility) plus the margin described in the chart below. Interest periods on Adjusted LIBOR Rate-based loans may be one, two, three or six months, at ACI's option. In the case of Adjusted LIBOR Rate-based loans, interest will accrue on the basis of a 360-day year, and will be payable on the last day of each relevant interest period and, for any interest period longer than three months, on each successive date three months after the first day of such interest period. Interest will accrue on Alternate Base Ratebased loans on the basis of a 365/366-day year (or 360-day year if based on the Adjusted LIBOR Rate) and shall be payable quarterly in arrears.

Unused loan commitments will be subject to an unused commitment fee, as described in the chart below.

Category	Leverage Ratio	Commitment Fee Rate	Eurodollar Spread	ABR Spread
Category 1	33.25:1.00	0.50%	2.50%	1.50%
Category 2	32.75:1.00 and £3.25:1.00	0.40%	2.25%	1.25%
Category 3	32.00:1.00 and £2.75:1.00	0.35%	2.00%	1.00%
Category 4	31.00:1.00 and <2.00:1.00	0.30%	1.75%	0.75%
Category 5	<1.00:1.00	0.25%	1.50%	0.50%

(g) To record the stock portion of the offer consideration, at par, and to eliminate S1's Shares, at par, as follows:

			(In thousands)
Elimina	tion of	S1's common stock outstanding	\$ (539)
Issuance	e of A	CI's common stock(1)	59
Total			\$ (480)
	(1)	Represents the issuance of approximately 5.9 million shares associated with the acquisition of S1	
	(h)	To record the stock portion of the Exchange Offer consideration, at fair value less par, and to eliminate S1's additional paid-in capital, as follows:	
			(In thousands)
Elimina	tion of	S1's additional paid-in capital	\$ (1,805,627)
Issuance	e of A	CI common stock	 168,253
Total			\$ (1,637,374)
	(i)	To eliminate S1's accumulated deficit, and to record estimated non-recurring costs of ACI for advisory, legal, regulatory and valuation costs, as follows:	
			(In thousands)
Elimina	tion of	S1's accumulated deficit	\$ 1,561,628
Estimat	ed rem	aining offer related transaction costs	(14,768)
Total			\$ 1,546,860

- (j) To eliminate S1's accumulated other comprehensive loss.
- (k) To reclassify S1's cost of hosting line to ACI's cost of maintenance, services and hosting fees line on the pro forma condensed combined statement of operations.
- (l) To adjust interest expense as follows:

	Six Months Ended June 30, 2011			Year Ended December 31, 2010	
	(In thous				
Elimination of S1's interest on existing debt	\$	(206)	\$	(455)	
Elimination of ACI's interest on existing debt		(377)		(778)	
Estimated interest on debt secured for acquisition		4,093		8,564	
Elimination of amortization on ACI's existing debt issuance costs		(168)		(336)	
Estimated amortization of debt issuance costs for new debt		1,169		2,338	
Total	\$	4,511	\$	9,333	

For purposes of calculating the pro forma interest expense, ACI used a rate of 2.24% and 2.34% for the six months ended June 30, 2011 and the year ended December 31, 2010, respectively. A change in the interest rate of 0.25% would change interest expense by approximately \$0.5 million and \$0.9 million for the six months ended June 30, 2011 and the year ended December 31, 2010, respectively.

- (m) Reflects the income tax benefit of the adjustments described in (l) above at ACI's domestic effective tax rate of 35%.
- (n) Reflects the conversion and exchange of each of the 55.0 million S1 Shares for 0.1064 ACI Shares.

FORWARD-LOOKING STATEMENTS

This prospectus/offer to exchange and the accompanying letter of transmittal contains forward-looking statements based on current expectations that involve a number of risks and uncertainties. All opinions, forecasts, projections, future plans or other statements other than statements of historical fact, are forward-looking statements and include words or phrases such as "believes," "will," "expects," "would" and words and phrases of similar impact. The safe harbor provisions of the Private provision Act of 1995 do not apply to any forward-looking statements made in connection with an exchange offer, including forward-looking statements from ACI's or S1's Form 10-K which are incorporated by reference into the prospectus/offer to exchange or in any Form 425 filed in the future.

We can give no assurance that such expectations will prove to have been correct. Actual results could differ materially as a result of a variety of risks and uncertainties, many of which are outside of the control of management. These risks and uncertainties include, but are not limited to the following: (1) that a transaction with S1 may not be completed on a timely basis; (2) negative effects on our business or S1's business resulting from the pendency of the Exchange Offer; (3) that we may not achieve the synergies and other expected benefits within the expected time or in the amounts we anticipate; (4) that we may not be able to promptly and effectively integrate the merged businesses after closing; and (5) that the committed financing may not be available. Other factors that could materially affect our business and actual results of operations are discussed in our most recent 10-Ks as well as other filings with the SEC available at www.sec.gov.

LEGAL MATTERS

Jones Day has provided an opinion regarding the validity of the ACI Shares to be issued pursuant to the Exchange Offer.

EXPERTS

The consolidated financial statements of ACI and subsidiaries as of and for the years ended December 31, 2010 and 2009, incorporated in this prospectus by reference from ACI's Annual Report on Form 10-K for the year ended December 31, 2010 and the effectiveness of ACI's internal control over financial reporting as of December 31, 2010, have been audited by Deloitte & Touche LLP, an independent registered public accounting firm, as stated in their reports, which are incorporated herein by reference. Such financial statements have been so incorporated in reliance upon the reports of such firm given upon their authority as experts in accounting and auditing.

The consolidated financial statements of ACI and subsidiaries for the year ended December 31, 2008, have been incorporated by reference herein in reliance upon the report of KPMG LLP, independent registered public accounting firm, incorporated by reference herein, and upon the authority of said firm as experts in accounting and auditing.

The audit report on the December 31, 2008 consolidated financial statements contains an explanatory paragraph that refers to the adoption of the Financial Accounting Standards Board (FASB) Interpretation No. 48, Accounting for Uncertainty in Income Taxes — an interpretation of FASB Statement No. 109 (now codified as Accounting Standards Codification (ASC) 740, Income Taxes).

The financial statements and management's assessment of the effectiveness of internal control over financial reporting (which is included in Management's Report on Internal Control over Financial Reporting) incorporated in this prospectus/offer to exchange by reference to the S1 10-K have been so incorporated in reliance on the report of PricewaterhouseCoopers LLP, an independent registered public accounting firm, given on the authority of said firm as experts in auditing and accounting.

ADDITIONAL NOTE REGARDING THE EXCHANGE OFFER

The Exchange Offer is being made solely by this prospectus/offer to exchange and the accompanying letter of election and transmittal and is being made to S1 stockholders. ACI and Offeror are not aware of any jurisdiction where the making of the Exchange Offer or the tender of S1 Shares in connection therewith would not be in compliance with the laws of such jurisdiction. In any jurisdiction where the securities, blue sky or other laws require the Exchange Offer to be made by a licensed broker or dealer, the Exchange Offer shall be deemed to be made on behalf of ACI, through Offeror, or by one or more registered brokers or dealers licensed under the laws of such jurisdiction.

Unless otherwise specifically noted herein, all references to "dollars" and "\$" shall refer to U.S. dollars.

WHERE YOU CAN FIND MORE INFORMATION

ACI and S1 file annual, quarterly and current reports, proxy statements and other information with the SEC. You may read and copy any of this information filed with the SEC at the SEC's public reference room:

Public Reference Room 100 F Street NE Room 1580 Washington, D.C. 20549

For information regarding the operation of the Public Reference Room, you may call the SEC at 1-800-SEC-0330. These filings made with the SEC are also available to the public through the website maintained by the SEC at http://www.sec.gov or from commercial document retrieval services.

ACI has filed a registration statement on Form S-4 to register with the SEC the offering and sale of ACI Shares to be issued in the Exchange Offer and the Second-Step Merger. This prospectus/offer to exchange is a part of that registration statement. We may also file additional amendments to the registration statement. In addition, ACI filed with the SEC a Tender Offer Statement on Schedule TO under the Exchange Act, together with exhibits, to furnish certain information about the Exchange Offer, and we may also file amendments to the Schedule TO. You may obtain copies of the Form S-4 and Schedule TO (and any amendments to those documents) by contacting the information agent as directed on the back cover of this prospectus/offer to exchange.

Some of the documents previously filed with the SEC may have been sent to you, but you can also obtain any of them through ACI, the SEC or the SEC's website as described above. Documents filed with the SEC are available from ACI without charge, excluding all exhibits, except that, if ACI has specifically incorporated by reference an exhibit in this prospectus/offer to exchange, the exhibit will also be provided without charge.

You may obtain documents filed with the SEC by requesting them in writing or by telephone from ACI's Information Agent for the Exchange Offer, Innisfree M&A Incorporated at the following addresses:

501 Madison Avenue, 20th Floor New York, New York 10022 Stockholders May Call Toll Free: (888) 750-5834 Banks and Brokers May Call Collect: (212) 750-5833

If you would like to request documents, in order to ensure timely delivery, you must do so at least five business days before the Expiration Time. This means you must request this information no later than the later of October 24, 2011 and five business days prior to any extended Expiration Time. ACI will mail properly requested documents to requesting stockholders by first class mail, or another equally prompt means, within one business day after receipt of such request.

You can also get more information by visiting ACI's website at http://www.aciworldwide.com and S1's website at http://www.s1.com.

Materials from these websites and other websites mentioned in this prospectus/offer to exchange and the accompanying letter of election and transmittal are not incorporated by reference in this prospectus/offer to exchange. If you are viewing this prospectus/offer to exchange in electronic format, each of the URLs mentioned in this prospectus/offer to exchange is an active textual reference only.

The SEC allows ACI to incorporate information into this prospectus/offer to exchange "by reference," which means that ACI can disclose important information to you by referring you to another document filed separately with the SEC. The information incorporated by reference is deemed to be part of this prospectus/offer to exchange, except for any information superseded by information contained directly in this prospectus/offer to exchange. This prospectus/offer to exchange incorporates by reference the documents set forth below that ACI and S1 have previously filed with the SEC. These documents contain important information about ACI and S1 and their financial condition, business and results.

ACI Filings (Commission File No. 0-25346):

Annual Report on Form 10-K

Quarterly Reports on Form 10-Q

Current Reports on Form 8-K

Proxy Statement on Schedule 14A

Description of common stock as contained in ACI's registration statement on Form 8-A registering ACI's common stock under Section 12 of the Exchange Act

S1 Filings (Commission File No. 000-24931):

Annual Report on Form 10-K

Quarterly Reports on Form 10-Q

Current Reports on Form 8-K

Proxy Statement on Schedule 14A

Solicitation/Recommendation on Schedule 14D-9

Period

For fiscal year ended December 31, 2010, filed on February 18, 2011

For the quarterly period ended June 30, 2011, filed on August 1, 2011, and for the quarterly period $\frac{1}{2}$ ended March 31, 2011, filed on April 29, 2011

Filed on October 3, 2011, September 28, 2011, September 7, 2011, August 30, 2011, August 25, 2011, August 15, 2001, August 2, 2011, July 26, 2011 and June 17, 2011

Filed on August 25, 2011 and April 27, 2011

Filed on January 9, 1995 and amended by Amendment No. 1 to the Form 8-A, filed on March 10, $\,$

Period

For fiscal year ended December 31, 2010, filed on March 11, 2011

For the quarterly period ended June 30, 2011, filed on August 4, 2011, and for the quarterly period ended March 31, 2011, filed on May 5, 2011

Filed on October 3, 2011, August 26, 2011, August 22, 2011, August 15, 2011, August 11, 2011, August 2, 2011, July 27, 2011, July 14, 2011, June 28, 2011 and May 26, 2011

Filed on August 22, 2011 and April 8, 2011

Filed on September 13, 2011, as it may be amended from time to time

ACI also hereby incorporates by reference any additional documents that it or S1 may file with the SEC pursuant to Section 13(a), 13(c), 14 or 15(d) of the Exchange Act from the date of this prospectus/offer to exchange to the termination of the Exchange Offer. Nothing in this prospectus shall be deemed to incorporate information furnished but not filed with the SEC.

ACI and S1 stockholders may obtain any of these documents without charge upon written or oral request to the information agent at Innisfree M&A Inc., 501 Madison Avenue, 20th Floor, New York, New York 10022, stockholders call toll-free (888) 750-5834 (banks and brokerage firms call collect (212) 750-5833), or from the SEC at the SEC's website at http://www.sec.gov

IF YOU WOULD LIKE TO REQUEST DOCUMENTS FROM ACI, PLEASE CONTACT THE INFORMATION AGENT NO LATER THAN OCTOBER 24, 2011, OR FIVE BUSINESS DAYS BEFORE THE EXPIRATION TIME, WHICHEVER IS LATER, TO RECEIVE THEM BEFORE THE EXPIRATION TIME. If you request any incorporated documents, the Information Agent will mail them to you by first-class mail, or other equally prompt means, within one business day of receipt of your request.

YOU SHOULD RELY ONLY ON THE INFORMATION CONTAINED OR INCORPORATED BY REFERENCE IN THIS PROSPECTUS/OFFER TO EXCHANGE IN MAKING YOUR DECISION WHETHER TO TENDER YOUR S1 SHARES INTO THE EXCHANGE OFFER. ACI HAS NOT AUTHORIZED ANYONE TO PROVIDE YOU WITH INFORMATION THAT DIFFERS FROM THAT CONTAINED IN THIS PROSPECTUS/OFFER TO EXCHANGE. THIS PROSPECTUS/OFFER TO EXCHANGE IS DATED OCTOBER 13, 2011. YOU SHOULD NOT ASSUME THAT THE INFORMATION CONTAINED IN THIS PROSPECTUS/OFFER TO EXCHANGE IS ACCURATE AS OF ANY DATE OTHER THAN THAT DATE, AND NEITHER THE MAILING OF THIS PROSPECTUS/OFFER TO EXCHANGE TO STOCKHOLDERS NOR THE ISSUANCE OF ACI SHARES IN THE EXCHANGE OFFER SHALL CREATE ANY IMPLICATION TO THE CONTRARY.

APPENDIX A

DIRECTORS AND EXECUTIVE OFFICERS OF ACI

The name, age, current principal occupation or employment and material occupations, positions, offices or employment for the past five years, of each director and executive officer of ACI are set forth below. References in this Appendix A to "ACI" mean ACI. Unless otherwise indicated below, the current business address of each director and executive officer is c/o ACI, 120 Broadway, Suite 3350, New York, New York 10271. Unless otherwise indicated below, the current business telephone of each director and executive officer is (646) 348-6700. Where no date is shown, the individual has occupied the position indicated for the past five years. Unless otherwise indicated, each occupation set forth opposite an individual's name refers to employment with ACI. Except as described in this Appendix A, none of the directors and executive officers of ACI listed below has, during the past five years, (1) been convicted in a criminal proceeding (excluding traffic violations or similar misdemeanors) or (2) been a party to any judicial or administrative proceeding that resulted in a judgment, decree or final order enjoining the person from future violations of, or prohibiting activities subject to, federal or state securities laws, or a finding of any violation of federal or state securities laws. Each of the directors and executive officers of ACI is a citizen of the United States of America.

DIRECTORS

Age $\underline{\underline{P}}resent$ Principal Occupation and Five-Year Employment History Alfred R. Berkeley, III Mr. Berkeley has been a director since 2007. He currently serves as chairman of Pipeline Financial Group, Inc., the parent of Pipeline Trading Systems, L.L.C., an equity trading brokerage services firm and also served as CEO until March 2010. He also serves as Vice Chairman of the National Infrastructure Advisory Council for the President and as a board member of XBRL US, the non-profit organization established to set data standards for the modernization of the SEC's EDGAR reporting system, and of XBRL International, the international standards organization which develops and maintains the XBRL specification. Mr. Berkeley is also a director of RealPage, Inc. (NASDAQ: RP), a provider of on demand software solutions for the rental housing industry; Fortegra Financial Corp. (NYSE: FRF), an insurance services company that provides distribution and administration services and insurance-related products to insurance companies, insurance brokers and agents and other financial services companies; and EDGAR Online, Inc. (NASDAQ: EDGR), Global provider of XBRL (eXtensible Business Reporting Language) solutions. Mr. Berkeley was the Vice Chairman of NASDAQ from July 2000 through July 2003 and President of NASDAQ from 1996 until 2000 and served as the Chairman of XBRL US until 2010. He served in a number of capacities at Alex. Brown & Sons Inc. from 1972 to 1996, including serving as Managing Director in the corporate finance department where he financed computer software and electronic commerce companies. Mr. Berkeley served as Vice Chairman of the Nomination Evaluation Committee for the National Medal of Technology and Innovation which makes candidate recommendations to the Secretary of Commerce from 2003 to 2009. He was previously a director of Kintera, Inc. (NASDAQ: KNTA), a provider of software for nonprofit organizations, from September 2003 until it was acquired by Blackbaud, Inc. (NASDAQ: BLKB); Webex Communications Inc. (NASDAQ: WEBX), a provider of meeting and web event software, until it was acquired by Cisco Systems, Inc. (NASDAQ: CSCO) and National Research Exchange Inc., a registered broker dealer, until it ceased

operations.

Name	Age	Present Principal Occupation and Five-Year Employment History
John D. Curtis	70	Mr. Curtis has been a director since 2003. He has been the Senior Vice President, General Counsel and Corporate Secretary of The Warranty Group, Inc., a single-source provider for the underwriting, administration and marketing of service contracts and related benefits, since February 2011. He previously worked as an attorney providing legal and business consulting services from August 2002 to February 2011 and served as General Counsel of Combined Specialty Corporation and a director of Combined Specialty Insurance Company, wholly owned subsidiaries of Aon Corporation (NYSE: AOC) from July 2001 to July 2002. He also served as president of First Extended, Inc., a holding company with two principal operating subsidiaries: First Extended Service Corporation, an administrator of vehicle extended service contracts and FFG Insurance Company, a property and casualty insurance company from November 1995 to July 2002. Mr. Curtis also serves as a director of The Warranty Group, Inc. board of directors.
Philip G. Heasley	62	Mr. Heasley has been a director and our President and Chief Executive Officer since March 2005. Mr. Heasley has a comprehensive background in payment systems and financial services. From October 2003 to March 2005, Mr. Heasley served as Chairman and Chief Executive Officer of PayPower LLC, an acquisition and consulting firm specializing in financial services and payment services. Mr. Heasley served as Chairman and Chief Executive Officer of First USA Bank from October 2000 to November 2003. Prior to joining First USA Bank, from 1987 until 2000, Mr. Heasley served in various capacities for U.S. Bancorp, including Executive Vice President, and President and Chief Operating Officer. Before joining U.S. Bancorp, Mr. Heasley spent 13 years at Citicorp, including three years as President and Chief Operating Officer of Diners Club, Inc. Mr. Heasley is also a director of Tier Technologies, Inc. (NASDAQ: TIER), a provider of electronic payment biller-direct solutions, and Lender Processing Services, Inc. (NYSE: LPS), a provider of mortgage processing services, settlement services, mortgage performance analytics and default solutions. Mr. Heasley also serves on the National Infrastructure Advisory Board.
James C. McGroddy	74	Mr. McGroddy has been a director since 2008. He is a self-employed consultant and currently serves as Chairman of the Board of MIQS, a Colorado-based healthcare information technology company, Chairman of the Board of Advanced Networks and Service, Inc. He is a member of the U.S. National Academy of Engineering, Mr. McGroddy was employed by International Business Machines Corporation from 1965 through 1996 in various capacities, including seven years as Senior Vice President of Research. He previously served as a director of Paxar Corporation (NYSE: PXR), a provider of merchandising systems for the retail and apparel industry.
		A-2

Name	Age	Present Principal Occupation and Five-Year Employment History
Harlan F. Seymour	61	Mr. Seymour has been a director since 2002 and ACI's Chairman of the Board since September 2002. He is the sole owner of HFS, LLC, a privately-held investment and business advisory firm advising public and private companies particularly in the area of strategic planning services, and a director of Pool Corporation (NASDAQ: POOL), a wholesale distributor of swimming pool supplies and related equipment, and serves on its audit, governance and strategic planning committees. He serves as a member of various private, profit and non-profit boards of directors, including Payformance Corp., an electronic health care claims and settlement solution company and the advisory board of Calvert Street Capital Partners, a private equity firm. He previously served as a director and as Executive Vice President of ENVOY Corporation, which provides electronic processing services, primarily to the health care industry.
John M. Shay, Jr.	64	Mr. Shay has been a director since 2006. He is the President and owner of Fairway Consulting LLC, a business consulting firm. He is a Certified Public Accountant and was previously employed by Ernst & Young LLP, a Big Four accounting firm offering audit, business advisory and tax services from 1972 through March 2006 serving as an audit partner from October 1984 to March 2006 and managing partner of the firm's New Orleans office from October 1998 through June 2005. Mr. Shay also served as an adjunct auditing professor in the graduate business program of the A.B. Freeman School of Business at Tulane University for approximately 10 years.
John E. Stokely	58	Mr. Stokely has been a director since 2003. He is the President of JES, Inc., an investment and consulting firm providing strategic and financial advice to companies in various industries from August 1999 through 2007, and a director of (i) Imperial Sugar Company (NASDAQ: IPSU), a manufacturer that refines, packages and distributes sugar and (ii) Pool Corporation (NASDAQ: POOL), a wholesale distributor of swimming pool supplies and related equipment. He also serves as Lead Independent Director of Pool Corporation (NASDAQ: POOL) and as a member of various private, profit and non-profit boards of directors, including AMF Bowling. Mr. Stokely previously served as President, Chief Executive Officer and Chairman of the Board of Richfood Holdings, Inc., a publicly-traded FORTUNE 500 food retailer and wholesale grocery distributor, from 1996 until August 1999 when it merged with Supervalu Inc. (NYSE: SVU). He also previously served as a director of O'Charley's Inc. (NASDAQ: CHUX), a casual dining restaurant company, Performance Food Group (NASDAQ: PFCG), a foodservice distributor, until it was acquired by affiliates of The Blackstone Group (NYSE: BX) and Wellspring Capital Management, and Nash-Finch Company (NASDAQ: NAFC), a leading food distribution company.
		A-3

Name Age Present Principal Occupation and Five-Year Employment History

69

Jan H. Suwinski

Mr. Suwinski has been a director since 2007. He is a professor of Business Operations at the Samuel Curtis Johnson Graduate School of Management at Cornell University in Ithaca, New York and currently serves as a director of Tellabs, Inc. (NASDAQ: TLAB), a provider of telecommunications networking products, and Thor Industries, Inc. (NYSE: THO), a manufacturer of recreational vehicles and buses. He served in various management positions in technology based businesses at Corning Incorporated from 1996 and as Executive Vice President of the Opto Electronics Group and a member of the operating committee at Corning Incorporated from 1990 to 1996. He also served as Chairman of Siecor Corporation, a Corning joint venture with Siemens AG from 1992 to 1996. Mr. Suwinski previously served as a director of Ohio Casualty Corporation (NASDAQ: OCAS), the holding company of The Ohio Casualty Insurance Company, which is one of six property-casualty insurance companies that make up Ohio Casualty Group, collectively referred to as Consolidated Corporation.

A-4

EXECUTIVE OFFICERS

Name		Present Principal Occupation and Five-Year Employment History
Philip G. Heasley	62	Mr. Heasley has been a director and our President and Chief Executive Officer since March 2005. Mr. Heasley has a comprehensive background in payment systems and financial services. From October 2003 to March 2005, Mr. Heasley served as Chairman and Chief Executive Officer of PayPower LLC, an acquisition and consulting firm specializing in financial services and payment services. Mr. Heasley served as Chairman and Chief Executive Officer of First USA Bank from October 2000 to November 2003. Prior to joining First USA Bank, from 1987 until 2000, Mr. Heasley served in various capacities for U.S. Bancorp, including Executive Vice President, and President and Chief Operating Officer. Before joining U.S. Bancorp, Mr. Heasley spent 13 years at Citicorp, including three years as President and Chief Operating Officer of Diners Club, Inc. Mr. Heasley is also a director of Tier Technologies, Inc. (NASDAQ: TIER), a provider of electronic payment biller-direct solutions, and Lender Processing Services, Inc. (NYSE: LPS), a provider of mortgage processing services, settlement services, mortgage performance analytics and default solutions. Mr. Heasley also serves on the National Infrastructure Advisory Board.
Scott W. Behrens	40	Mr. Behrens serves as Executive Vice President, Chief Financial Officer and Chief Accounting Officer. Mr. Behrens joined ACI in June 2007 as our Corporate Controller and Chief Accounting Officer. Mr. Behrens was appointed Chief Financial Officer in December 2008. Prior to joining ACI, Mr. Behrens served as Senior Vice President, Corporate Controller and Chief Accounting Officer at SITEL Corporation from January 2005 to June 2007. He also served as Vice President of Financial Reporting at SITEL Corporation from April 2003 to January 2005. From 1993 to 2003, Mr. Behrens was with Deloitte & Touche, LIP, including two years as a Senior Audit Manager. Mr. Behrens holds a Bachelor of Science (Honors) from the University of Nebraska — Lincoln.

Dennis P. Byrnes

47 Mr. Byrnes serves as Executive Vice President, Chief Administrative Officer, General Counsel and Secretary. Mr. Byrnes joined the Company in June 2003. Prior to that Mr. Byrnes served as an attorney in Bank One Corporation's technology group from 2002 to 2003. From 1996 to 2002 Mr. Byrnes was an executive officer at Sterling Commerce, Inc., an electronic commerce software and services company, serving as that company's general counsel from 2000. From 1991 to 1996 Mr. Byrnes was an attorney with Baker Hostetler, a national law firm with over 600 attorneys. Mr. Byrnes holds a JD (cum laude) from The Ohio State University College of Law, a Master of Business Administration from Xavier University and a Bachelor of Science in engineering (magna cum laude) from Case Western Reserve University.

Present Principal Occupation and Five-Year Employment History

Mr. Linberg serves as Vice President and Chief Technology Officer. In this capacity he is responsible for the architectural Charles H. Linberg 53 direction of ACI products including the formation of platform, middleware and integration strategies. Mr. Linberg joined the Company in 1988 and has served in various technical management roles including Vice President of Payment Systems, Vice President of Architecture and Technology, Vice President of BASE24 Development and Vice President of Network Systems. Prior to joining ACI, Mr. Linberg was Vice President of Research and Development at XRT, Inc., where he led the development of XRT's proprietary fault-tolerant LAN/WAN communications middleware, relational database and 4GL products. Mr. Linberg holds a Bachelor of Science in Business Administration from the University of Delaware. Mr. Maki serves as Senior Vice President, Treasurer and Chief Corporate Development Officer. Mr. Maki joined ACI in June 2006. Mr. Maki was appointed Treasurer in January 2008. Prior to joining ACI, Mr. Maki served as Senior Vice President for Stephens, Inc. from 1999 through 2006. From 1994 to 1999, Mr. Maki was a Director in the Corporate Finance Craig A. Maki 45 group at Arthur Andersen and from 1991 to 1994, he was a Senior Consultant at Andersen Consulting. Mr. Maki graduated from the University of Wyoming and received his Master of Business Administration from the University of Denver. David N. Morem Mr. Morem joined ACI in June 2005 and serves as Senior Vice President, Global Business Operations. Prior to his appointment as Senior Vice President, Global Business Operations in January 2008, Mr. Morem served as Chief Administrative Officer of ACI. Prior to joining ACI, Mr. Morem held executive positions at GE Home Loans, Bank One Card Services and U.S. Bank. Mr. Morem brings more than 25 years of experience in process management, finance, credit operations, credit policy and change management. Mr. Morem holds a B.A. degree from the University of Minnesota and a Master of Business Administration from the University of St. Thomas. Mr. Peterson serves as Vice President, Corporate Tax and Assistant Treasurer. Mr. Peterson joined ACI in April 2007. Prior to joining ACI, Mr. Peterson served as Senior Vice President, Corporate Tax and Insurance for SITEL Corporation from Bryan A. Peterson 2004 through 2007. From 1989 to 2004, Mr. Peterson served in numerous tax related positions with Schlumberger Limited. Mr. Peterson holds a B.A. degree from Texas Tech University. Stuart Rhodes Mr. Rhodes joined ACI in August 2007 working in Corporate Development. Prior to joining ACI, Mr. Rhodes was an Analyst in the Technology and Services Investment Banking Group at Wachovia Securities (now Wells Fargo Securities) for two years. Prior to Wachovia Securities, Mr. Rhodes graduated from Sewanee: University of the South with a Bachelor of Arts in Economics.

APPENDIX B

DIRECTORS AND EXECUTIVE OFFICERS OF OFFEROR

The name, age, current principal occupation or employment and material occupations, positions, offices or employment for the past five years, of each director and executive officer of Offeror are set forth below. References in this Appendix B to "Offeror" mean Antelope Investment Co. LLC, a Delaware limited liability company and wholly owned subsidiary of ACI. Unless otherwise indicated below, the current business address of each director and executive officer is (646) 348-6700. Where no date is shown, the individual has occupied the position indicated for the past five years. Unless otherwise indicated, each occupation set forth opposite an individual's name refers to employment with ACI. Except as described in this Appendix B, none of the directors and executive officers of Offeror listed below has, during the past five years, (1) been convicted in a criminal proceeding (excluding traffic violations or similar misdemeanors) or (2) been a party to any judicial or administrative proceeding that resulted in a judgment, decree or final order enjoining the person from future violations of, or prohibiting activities subject to, federal or state securities laws, or a finding of any violation of federal or state securities laws. Each of the directors and executive officers of Offeror is a citizen of the United States of America.

Name	Age	Present Principal Occupation and Five-Year Employment History
Scott W. Behrens	40	Mr. Behrens serves as Vice President and Assistant Treasurer of Offeror and as Executive Vice President, Chief Financial Officer and Chief Accounting Officer of ACI. Mr. Behrens joined ACI in June 2007 as our Corporate Controller and Chief Accounting Officer. Mr. Behrens was appointed Chief Financial Officer in December 2008. Prior to joining ACI, Mr. Behrens served as Senior Vice President, Corporate Controller and Chief Accounting Officer at SITEL Corporation from January 2005 to June 2007. He also served as Vice President of Financial Reporting at SITEL Corporation from April 2003 to January 2005. From 1993 to 2003, Mr. Behrens was with Deloitte & Touche, LLP, including two years as a Senior Audit Manager. Mr. Behrens holds a Bachelor of Science (Honors) from the University of Nebraska — Lincoln.
Dennis P. Byrnes	47	Mr. Byrnes serves as President and Director of Offeror and as Executive Vice President, Chief Administrative Officer, General Counsel and Secretary of ACI. Mr. Byrnes joined ACI in June 2003. Prior to that Mr. Byrnes served as an attorney in Bank One Corporation's technology group from 2002 to 2003. From 1996 to 2002
		Mr. Byrnes was an executive officer at Sterling Commerce, Inc., an electronic commerce software and services company, serving as that company's general counsel from 2000. From 1991 to 1996 Mr. Byrnes was an attorney with Baker Hostetler, a national law firm with over 600 attorneys. Mr. Byrnes holds a JD (cum laude) from The Ohio State University College of Law, a Master of Business Administration from Xavier University and a Bachelor of Science in engineering (magna cum laude) from Case Western Reserve University.

Present Principal Occupation and Five-Year Employment History

Name

Craig A. Maki

Mr. Maki serves as Vice President and Treasurer and a director of Offeror and as Senior Vice President, Treasurer and Chief Corporate Development Officer of ACI. Mr. Maki joined ACI in June 2006. Mr. Maki was appointed Treasurer in January 2008. Prior to joining ACI, Mr. Maki served as Senior Vice President for Stephens, Inc. from 1999 through 2006. From 1994 to 1999, Mr. Maki was a Director in the Corporate Finance group at Arthur Andersen and from 1991 to 1994, he was a Senior Consultant at Andersen Consulting, Mr. Maki graduated from the University of Wyoming and received his Master of Business Administration from the University of Denver. 45

Age

APPENDIX C

STOCK TRANSACTIONS IN THE PAST 60 DAYS

Other than: (1) the acquisition by ACI of 1,000 S1 Shares on July 26, 2011 at a price of \$9.34 per share, (2) the acquisition by ACI of 150,000 S1 Shares on August 12, 2011 at a price of \$9.00 per share, (3) the acquisition by ACI of 500 S1 Shares on August 16, 2011 at a price of \$9.00 per share, (4) the acquisition by ACI of 236,500 S1 Shares on August 17, 2011 at a price of \$9.00 per share, and (5) the acquisition by ACI of 719,000 S1 Shares on August 18, 2011 at a price of \$8.98 per share, none of ACI or, after due inquiry and to the best of its knowledge and belief, any of the persons identified on Appendix A or Appendix B (or any of their respective associates or majority-owned subsidiaries) has engaged in any transaction involving S1 Shares in the past 60 days.

TRANSACTION AGREEMENT
by and among
ACI WORLDWIDE, INC.,
ANTELOPE INVESTMENT CO. LLC and
S1 CORPORATION
Dated as of October 3, 2011

D-i

TABLE OF CONTENTS

ARTICLE I. Defin	D-2	
1.1	Certain Defined Terms	D-2
ARTICLE II. THE	COFFER	D-10
2.1	The Offer	D-10
2.2	Proration of Offer Consideration	D-12
2.3	Election Procedures; Allocation of Offer Consideration	D-13
2.4	Company Consent; Schedule 14D-9	D-14
2.5	Top-Up Option	D-14
2.6	Stockholder Lists	D-16
2.7	Directors	D-16
2.8	Surrender of Certificates	D-17
ARTICLE III. Plai	n of Merger	D-19
3.1	The Merger	D-19
3.2	Effective Time; Closing Date	D-19
3.3	Effects of the Merger	D-19
3.4	Surrender of Certificates	D-20
3.5	Company's Transfer Books Closed; No Further Ownership Rights in Company Shares	D-22
3.6	Stock Options; SARs; Restricted Stock	D-22
3.7	Certificate of Incorporation, By-Laws and Directors and Officers of the Surviving Corporation	D-23
3.8	Taking of Necessary Action; Further Action	D-23
3.9	Appraisal Rights/Dissenting Shares	D-24
ARTICLE IV. Rep	resentations and Warranties of the Company	D-24
4.1	Organization, Standing and Power	D-24
4.2	Company Subsidiaries	D-24
4.3	Authority; Execution and Delivery; Enforceability	D-25
4.4	Capital Structure	D-25
4.5	No Conflicts; Consents	D-20
4.6	SEC Documents; Undisclosed Liabilities	D-26
4.7	Absence of Certain Changes or Events	D-27
4.8	Material Contracts	D-27
4.9	Intellectual Property	D-28
4.10	Certain Business Practices	D-30
4.11	Takeover Laws	D-30
4.12	Taxes	D-30
4.13	Benefit Plans	D-3:
4.14	ERISA Compliance; Excess Parachute Payments; Other Benefits Matters	D-32
4.15	Labor and Employment Matters	D-33
4.16	Litigation	D-34
4.17	Compliance with Applicable Laws	D-34
4.18	Brokers	D-34
4.19	Opinion of Financial Advisor	D-34
4.20	Vote Required	D-34
4.21	Related Party Transactions	D-34

4.22	Insurance	D-35
4.23	Environmental Matters	D-35
4.24	Proxy Statement/Prospectus: Schedule 14D-9; Registration Statement	D-35
4.25	No Additional Representations and Warranties	D-35
	presentations and Warranties of Parent and Merger Sub	D-36
5.1	Organization, Standing and Power	D-36
5.2	Parent Subsidiaries	D-36
5.3	Authority; Execution and Delivery; Enforceability	D-36
5.4	Capital Structure	D-37
5.5	No Conflicts; Consents	D-37
5.6	SEC Documents; Undisclosed Liabilities	D-38
5.7	Absence of Certain Changes or Events	D-39
5.8	Material Contracts	D-39
5.9	Intellectual Property	D-39
5.10	Certain Business Practices	D-42
5.11	Takeover Laws	D-42
5.12	Taxes	D-42
5.13	Reserved	D-42
5.14	ERISA Compliance; Excess Parachute Payments; Other Benefits Matters	D-42
5.15	Labor and Employment Matters	D-43
5.16	Litigation	D-44
5.17	Compliance with Applicable Laws	D-44
5.18	Brokers	D-44
5.19	Issuance of Parent Common Stock	D-44
5.20	Funds	D-44
5.21	Related Party Transactions	D-45
5.22	Insurance	D-45
5.23	Environmental Matters	D-45
5.24	Offer Documents; Schedule 14D-9; Registration Statement; Proxy Statement/Prospectus	D-45
5.25	No Additional Representations and Warranties	D-46
	ovenants Relating to Conduct of Business	D-46
6.1	Conduct of Business by the Company	D-46
6.2	Conduct of Business by Parent	D-48
	Additional Agreements	D-49
7.1	Preparation and Mailing of Post-Effective Amendment and Proxy Statement	D-49
7.2	Company Stockholders Meeting; Company Recommendation	D-50
7.3	Notification	D-50
7.4	Confidentiality; Access to Information	D-50
7.5	Non-Solicitation by the Company	D-51
7.6	Regulatory Filings	D-53
7.7	Public Disclosure	D-55
7.8	Takeover Laws	D-55
7.9	Indemnification	D-55
7.10	Nasdaq Listing	D-56
7.11	Employee Matters	D-56

7.12	Section 16 Matters	D-57
7.13	Stockholder Litigation	D-57
7.14	Obligations of Merger Sub	D-57
7.15	Financing	D-57
7.16	No Control of the Company's Business	D-58
7.17	Non-Disparagement	D-58
ARTICLE VIII. Co	onditions to the Merger	D-58
8.1	Conditions to Obligations of Each Party to Effect the Merger	D-58
ARTICLE IX. Terr	mination	D-60
9.1	Termination	D-60
9.2	Notice of Termination; Effect of Termination	D-62
9.3	Fees and Expenses	D-62
ARTICLE X. Gene	eral Provisions	D-63
10.1	Non-Survival of Representations and Warranties	D-63
10.2	Notices	D-63
10.3	Construction	D-63
10.4	Headings; Table of Contents	D-64
10.5	Mutual Drafting	D-64
10.6	Disclosure Schedules	D-64
10.7	Counterparts; Facsimile and Electronic Signatures	D-64
10.8	Entire Agreement; Third Party Beneficiaries	D-64
10.9	Amendment	D-65
10.10	Severability	D-65
10.11	Remedies; Specific Performance	D-65
10.12	Assignment	D-65
10.13	Applicable Law	D-65
10.14	CONSENT TO JURISDICTION AND SERVICE OF PROCESS	D-66
10.15	WAIVER OF JURY TRIAL	D-66

TABLE OF CONTENTS

EXHIBITS

Exhibit A — Conditions to the Offer

Exhibit B — Form of Surviving Corporation Certificate of Incorporation

Exhibit C — Form of Surviving Corporation By-Laws

TRANSACTION AGREEMENT

This TRANSACTION AGREEMENT (this "Agreement") is made and entered into as of October 3, 2011 by and among ACI Worldwide, Inc., a Delaware corporation ("Parent"), Antelope Investment Co. LLC, a Delaware limited liability company and a wholly owned indirect Subsidiary of Parent ("Merger Sub"), and S1 Corporation, a Delaware corporation (the "Company"). Capitalized terms in this Agreement shall have the meanings ascribed to them as provided in Section 1.1.

WITNESSETH:

WHEREAS, Parent and Merger Sub have previously commenced a cash and stock exchange offer (the "Pending Offer") to purchase all of the issued and outstanding Company Shares, subject to certain terms and conditions:

WHEREAS, the Boards of Directors of Parent, the Company, and Merger Sub, have each determined that it is in the best interests of their respective stockholders for Parent and the Company to enter into a business combination transaction upon the terms and subject to the conditions of this Agreement whereby the holders of Company Shares will tender their Company Shares pursuant to the Offer in exchange for the Offer Pirce and, following completion of the Offer, Merger Sub will merge (such merger, the "Merger") with and into the Company, and, upon the effectiveness of the Merger and in accordance with the General Corporation Law of the State of Delaware (the "DGCL") and the Delaware Limited Liability Company Act (the "DLLCA"), Merger Sub will cease to exist, the Company as the Surviving Corporation in the Merger will become a wholly owned Subsidiary of Parent, and the Company Shares will be converted into the right to receive the Merger Consideration as set forth in Section 3 3(a):

WHEREAS, the Board of Directors of the Company: (i) determined that the Transactions are fair to, and in the best interests of, the Company and its stockholders, and that, considering the financial position of the merging companies, no reasonable concern exists that the Surviving Corporation will be unable to fulfill the obligations of the Company to its creditors existing as of immediately prior to the Effective Time; (ii) adopted and approved this Agreement and approved the Transactions; and (iii) determined to recommend that the stockholders of the Company accept the Offer, tender their Company Shares to Merger Sub pursuant to the Offer and, if necessary under applicable Law, adopt the Agreement and approve the Transactions;

WHEREAS, the Board of Directors of Parent has: (i) determined that the Transactions are fair to, and in the best interests of, Parent and its stockholders and that, considering the financial position of the merging companies, no reasonable concern exists that the Surviving Corporation will be unable to fulfill the obligations of Merger Sub to its creditors existing as of immediately prior to the Effective Time and (ii) adopted the Agreement on behalf of Parent, as the sole stockholder of Merger Sub, and approved the Transactions;

WHEREAS, the Board of Directors of Merger Sub has by unanimous written consent: (i) determined that the Transactions are fair to, and in the best interests of, Merger Sub and its sole member and that, considering the financial position of the merging companies, no reasonable concern exists that the Surviving Corporation will be unable to fulfill the obligations of Merger Sub to its creditors existing as of immediately prior to the Effective Time and (ii) adopted and approved this Agreement and approved the Transactions; and

WHEREAS, accordingly in furtherance of the contemplated acquisition of the Company by Parent, on the terms and subject to the conditions set forth in this Agreement, (i) Parent has agreed to amend the Pending Offer (such offer, as it may be amended from time to time, being referred to in this Agreement as the "Offer") to acquire all of the issued and outstanding Company Shares, for, at the election of the holders of Company Shares, the per Company Share consideration of either 0.3148 of a share of Parent Common Stock or \$10.00 in cash, without interest (the "Offer Price"), subject to pro ration pursuant to Section 2.2, and (ii) after acquiring the Company Shares pursuant to the Offer in accordance with the terms hereof, the Merger will occur in which each Non-Election Share outstanding as of the Effective Time (other than Company Shares that will be cancelled pursuant to Section 3.3(b), Company Shares that will be converted into shares of Parent Common Stock pursuant to Schedule 6.1 and any Dissenting Shares) will be converted into the Merger Consideration as set forth in Section 3.3(a).

NOW, THEREFORE, in consideration of the foregoing premises, the mutual covenants, promises and representations set forth herein, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto hereby agree as follows:

ARTICLE I.

Definitions

- 1.1 Certain Defined Terms. As used in this Agreement, the following terms have the meanings specified in this Section 1.1:
 - "1997 Plan" shall have the meaning set forth in Section 3.6(a)(i)(A).
 - "1998 Plan" shall have the meaning set forth in Section 3.6(a)(i)(A).
 - "2003 Plan" shall mean the S1 Corporation 2003 Stock Incentive Plan (as amended and restated effective February 26, 2008).
 - "Acceptance Time" shall have the meaning set forth in Section 2.1(a)(ii).
 - "Affiliate" shall have the meaning set forth in Rule 12b-2 of the Exchange Act.
 - "Agency Litigation" shall have the meaning set forth in Section 7.6(b).
 - "Agreement" shall have the meaning set forth in the Preamble.
 - "Alternative Financing" shall have the meaning set forth in Section 7.15(a).
 - "Antitrust Laws" shall mean any antitrust, competition or trade regulatory Law of any Governmental Entity.
 - "Average Per Share Price" shall have the meaning in this Section 1.1.
- "Blended Value" shall mean the sum of (1) \$6.62 plus (2) an amount equal to the product (rounded to the nearest cent) of (x) 0.1064 times (y) the volume weighted average sales price per share of Parent Common Stock for the ten consecutive days that Parent Common Stock has traded ending on and including the second clear trading day immediately prior to the Effective Time as reported on Nasdaq (the "Average Per Share Price").
 - "Blue Sky Laws" shall have the meaning set forth in Section 4.5(b).
 - "Book-Entry Shares" shall have the meaning set forth in Section 2.5(c).
 - "Business Day" shall mean any day other than a Saturday, Sunday or other day on which the banks in New York City are authorized by Law or executive order to be closed.
 - "Cash Election" shall have the meaning set forth in Section 2.1(c).
 - "Cash Election Share" shall have the meaning set forth in Section 2.2(a).
 - "Cash Price Per Share" shall have the meaning set forth in Section 2.1(a)(i).
 - "Cash Proration Factor" shall have the meaning set forth in Section 2.2(c)(ii).
 - "Certificate" shall have the meaning set forth in Section 3.4(c).
 - "Certificate of Merger" shall have the meaning set forth in Section 3.2.
 - "Closing" shall have the meaning set forth in Section 3.2.
 - "Closing Date" shall have the meaning set forth in Section 3.2.
 - "Code" shall mean the Internal Revenue Code of 1986, as amended.
 - "Company" shall have the meaning set forth in the Preamble.

"Company Acquisition Proposal" shall mean any offer, proposal or indication of interest received from a third party (other than a party to this Agreement) providing for any Company Acquisition Transaction, including any renewal or revision to such a previously made offer, proposal or indication of interest.

"Company Acquisition Transaction" shall mean any transaction or series of transactions involving: (a) any merger, consolidation, share exchange, recapitalization, business combination or similar transaction involving the Company or any of its Subsidiaries; (b) any direct or indirect acquisition of securities, tender offer, exchange offer or other similar transaction in which a Person or "group" (as defined in the Exchange Act) of Persons directly or indirectly acquires beneficial or record ownership of securities representing 20% or more of the outstanding Company Shares; (c) any direct or indirect acquisition of any business or businesses or of assets that constitute or account for 20% or more of the consolidated net revenues, net income or assets of the Company and its Subsidiaries, taken as a whole (based on the fair market value thereof); (d) any liquidation or dissolution of the Company or any material Subsidiary of the Company; or (e) any combination of the foregoing (in each case, other than any of the Transactions).

- "Company Change of Recommendation" shall have the meaning set forth in Section 7.5(d).
- "Company Charter Documents" shall have the meaning set forth in Section 4.1.
- "Company D&O Policy" shall have the meaning set forth in Section 7.9(b).
- "Company Director Plan" shall mean the S1 Corporation Directors' Deferred Compensation Plan, effective as of January 1, 2010.
- "Company Disclosure Schedule" shall mean the disclosure schedule delivered by the Company to Parent prior to the execution of this Agreement.
- "Company Employee" shall mean any current, former or retired employee, officer or director of the Company or any of its Subsidiaries.
- "Company Employee Plans" shall have the meaning set forth in Section 4.13.
- "Company Employment Agreements" shall have the meaning set forth in Section 4.13.
- "Company ERISA Affiliate" shall have the meaning set forth in Section 4.14(a).
- "Company Insiders" shall have the meaning set forth in Section 7.12.
- "Company International Employee Plan" shall have the meaning set forth in Section 4.13.
- "Company Intellectual Property" shall mean all Intellectual Property owned by the Company or any of its Subsidiaries.
- "Company Material Contracts" shall have the meaning set forth in Section 4.8(a).
- "Company Pension Plans" shall have the meaning set forth in Section 4.14(a).
- "Company Preferred Stock" shall have the meaning set forth in Section 4.4(a).
- "Company Recommendation" shall have the meaning set forth in Section 4.3(b).
- "Company SEC Documents" shall have the meaning set forth in the preamble to Article IV.
- "Company Shares" shall mean each share of common stock, par value \$0.01 per share, of the Company.
- "Company Software Products" shall mean all Software products developed and owned by the Company or any of its Subsidiaries that are (i) offered for license by the Company or its Subsidiaries or (ii) used in the conduct of their respective businesses.
 - "Company Stock Option" shall have the meaning set forth in Section 3.6(a)(i)(A).

- "Company Stock Plan" means each of the 1997 Plan, the 1998 Plan, the 2003 Plan and the Company Director Plan,
- "Company Stockholder Approval" shall have the meaning set forth in Section 4.3(a).
- "Company Stockholders Meeting" shall have the meaning set forth in Section 7.2(a).

"Company Superior Offer" shall mean a bona fide written Company Acquisition Proposal (for purposes of this definition, replacing all references in such definition to 20% with 50%) that the Board of Directors of the Company or any committee thereof determines, in good faith, after consultation with outside legal counsel and a financial advisor (i) is on terms that are more favorable from a financial point of view to the Company's stockholders than the Transactions (including any proposal by Parent to amend the terms of this Agreement) after taking into account all of the terms and conditions of such proposal and (ii) is likely to be completed (without material modification of its terms), in each of the cases of clause (i) and (ii) taking into account all financial, regulatory, legal and other aspects of such Company Acquisition Proposal (including the timing and likelihood of consummation thereof) and the payment, if any, of the Company Termination Fee.

- "Company Systems" shall have the meaning set forth in Section 4.9(f).
- "Company Termination Fee" shall mean a fee payable by the Company in the amount of \$19.14 million.
- "Consent Decree" shall have the meaning set forth in Section 7.6(e).
- "Continuing Directors" shall have the meaning set forth in Section 2.7(b).
- "Contract" shall mean any contract, lease, license, indenture, note, bond, agreement, permit, concession, franchise or other instrument.
- "DGCL" shall have the meaning set forth in the Preamble.
- "Disclosure Schedules" shall mean the Company Disclosure Schedule and the Parent Disclosure Schedule.
- "Dissenting Shares" shall have the meaning set forth in Section 3.9(a).
- "DLLCA" shall have the meaning set forth in the Preamble.
- "DOJ" shall mean the United States Department of Justice.
- "DOJ Impediment" shall have the meaning set forth in Section 7.6(e).
- "DOL" shall mean the United States Department of Labor.
- "Effect" shall have the meaning set forth in this Section 1.1.
- "Effective Time" shall have the meaning set forth in Section 3.2.
- "Election" shall have the meaning set forth in Section 2.1(c).
- "Election Deadline" shall have the meaning set forth in Section 3.3(f).
- "Environmental Claim" shall mean any administrative, regulatory or judicial actions, suits, orders, demands, directives, claims, liens, investigations, proceedings or written or oral notices of noncompliance, violation, liability or obligation, by or from any Person alleging liability of whatever kind or nature arising out of, based on or resulting from (i) the presence or Release of, or exposure to, any Hazardous Materials at any location; or (ii) any Environmental Law or any permit issued pursuant to any Environmental Law.
- "Environmental Laws" shall mean any and all Laws which (i) regulate or relate to the protection or clean up of the environment; the use, treatment, storage, transportation, handling, disposal or release of Hazardous Materials, the preservation or protection of waterways, groundwater, drinking water, air,

wildlife, plants or other natural resources; or the health and safety of persons or property, including protection of the health and safety of employees; or (ii) impose liability or responsibility with respect to any of the foregoing.

- "ERISA" shall mean the United States Employee Retirement Income Security Act of 1974.
- "Exchange Act" shall mean the United States Securities Exchange Act of 1934.
- "Exchange Agent" shall have the meaning set forth in Section 2.8(a).
- "Exchange Fund" shall have the meaning set forth in Section 2.8(b).
- "Exercise Period" shall have the meaning set forth in Section 3.6(a)(i)(B).
- "Expiration Date" shall have the meaning set forth in Section 2.1(a)(i).

"Event of Default" shall mean the occurrence of any one or more of the following events (regardless of the reason therefor): (i) Parent's or Merger Sub's failure to pay or discharge its obligations in full in accordance with the terms of the Promissory Note; (ii) Parent's or Merger Sub's failure to observe or perform any material covenant, obligation, condition or agreement contained in the Promissory Note which failure remains unremedied for more than 30 days after written notice of such failure has been delivered to Parent or Merger Sub; (iii) the commission of any act of bankruptcy by Parent or Merger Sub, the execution by Parent or Merger Sub of a general assignment for the benefit of creditors, the filing by or against Parent or Merger Sub of a petition in bankruptcy or any petition for relief under the federal bankruptcy act and the continuation of such petition without dismissal for a period of 90 days or more, or the appointment of a receiver or trustee to take possession of the property or assets of Parent or Merger Sub; or (iv) the liquidation, dissolution or winding up of Parent or Merger Sub, whether voluntary or involuntary.

- "Financing" shall have the meaning set forth in Section 5.20.
- "Financing Agreement" shall have the meaning set forth in Section 5.20.
- "Financing Commitment" shall have the meaning set forth in Section 5.20.
- "Form of Election" shall have the meaning set forth in Section 2.1(c).
- "FTC" shall mean the United States Federal Trade Commission
- "Fundtech" shall mean Fundtech Ltd., a company organized under the laws of Israel.
- "Fundtech Merger Agreement" shall mean the Plan of Merger and Reorganization, dated as of June 26, 2011, by and among the Company, Finland Holdings (2011) Ltd., a company organized under the laws of Israel and a wholly owned subsidiary of the Company, and Fundtech.
 - "GAAP" shall mean United States generally accepted accounting principles.
 - "Governmental Entity" shall mean any United States federal, state of the United States or foreign governmental or regulatory agency, commission, court, body, entity or authority.
- "Hazardous Materials" shall mean any pollutant, chemical, substance and any toxic, infectious, carcinogenic, reactive, corrosive, ignitable or flammable chemical, or chemical compound, or hazardous substance, material or waste, whether solid, liquid or gas, that is subject to regulation, control or remediation under any Environmental Laws, including any quantity of asbestos in any form, urea formaldehyde, PCBs, radon gas, crude oil or any fraction thereof, all forms of natural gas, petroleum products or by-products or derivatives.
 - "HSR Act" shall mean the Hart-Scott-Rodino Antitrust Improvements Act of 1976.
 - "HSR Condition" shall have the meaning set forth in Section 8.1(e).
 - "HSR Filing" shall have the meaning set forth in Section 7.6(a).

- "Independent Directors" shall have the meaning set forth in Section 2.7(d).
- "Initial Expiration Date" shall have the meaning set forth in Section 2.1(a)(i).
- "Injunction" shall have the meaning set forth in Section 7.6(c).
- "Insured Parties" shall have the meaning set forth in Section 7.9(b).

"Intellectual Property" shall mean any and all intellectual property or proprietary rights throughout the world, including all (i) trademarks, service marks, trade names, Internet domain names, web sites, trade dress, logos, slogans, company names and other indicia of source (including any goodwill associated with each of the foregoing) and all registrations and applications for registration of the foregoing; (ii) inventions (whether or not patentable or reduced to practice), patents and industrial designs, patent applications, patent disclosures and related know how and all continuations, continuations-in-part, revisions, divisionals, extensions and reexaminations in connection therewith; (iii) works of authorship (whether or not copyrightable), copyrights, copyrightable works, moral rights, and mask works and all registrations and applications for registration of the foregoing; (iv) Software; (v) trade secrets, know-how, technology, processes, designs, algorithms, methods, formulae, and other confidential and proprietary information (including technical data, customer and supplier lists, pricing and cost information, and business and marketing plans); and (vi) all rights to sue or recover and retain damages and costs and attorneys' fees for past, present and future infringement or misappropriation of any of the foregoing.

"IRS" shall mean the United States Internal Revenue Service.

"Knowledge" shall mean with respect to a party hereto, with respect to any matter in question, the actual knowledge of the chief executive officer, the president, the chief financial officer or the general counsel of such party.

"Law" shall mean any statute, law (including common law), ordinance, rule or regulation or any judgment, decree, injunction or other order.

"Liens" shall have the meaning set forth in Section 4.2(a).

"LT" shall have the meaning set forth in Section 2.1(c).

"Material Adverse Effect" shall mean, with respect to any party hereto, any event, change, effect, development, condition or occurrence (each, an "Effect"), individually or in the aggregate with all other Effects, that (I) is materially adverse on or with respect to the business, financial condition or continuing results of operations of such party and its Subsidiaries, taken as a whole, other than any event, change, effect, development, condition or occurrence: (a) in or generally affecting the economy or the financial, commodities or securities markets in the United States or elsewhere in the world or the industry or industries in which such party or its Subsidiaries operate generally or (b) resulting from or arising out of (i) the pendency or announcement of or compliance with the Transactions (including the loss of any customer, vendor, supplier or prospect or a reduction in the amount of business such customer, vendor or supplier does with such party or its Subsidiaries or any of its Subsidiaries, (iii) any changes in GAAP, Law or accounting standards or interpretations thereof, (iv) any natural disasters or weather-related or other force majeure event, (v) any changes in national or international political conditions, including any engagement in hostilities, whether or not pursuant to the declaration of a national emergency or war, the outbreak or escalation of hostilities or acts of war, sabotage or terrorism, (vi) the failure of such party to meet any internal or published projections, forecasts or revenue or earnings predictions (it being understood that the facts or occurrences giving rise or contributing to such failure may be deemed to constitute, or be taken into account in determining whether there has been or would reasonably be expected to be, a Material Adverse Effect), (viii) any change in the market price or trading volume of such party's securities (it being understood that the facts or occurrences giving rise or contributing to such change may be deemed to constitute, or be taken into accou

stockholders of such party arising from or relating to the Transactions, or (ix) any failure to obtain, or any denial or withdrawal of any application for, any license, consent, approval, order or authorization of, or registration, declaration or filing with, any Governmental Entity in connection with the Transactions, but only to the extent, in each of clauses (a), (b)(iii), (b)(iv) and (b)(v), that such event, change, effect, development, condition or occurrence does not affect such party and its Subsidiaries, taken as a whole, in a materially disproportionate manner relative to other participants in the business, industries or geographic region or territory in which such party and its Subsidiaries operate; or (II) would prevent or materially delay such party from consummating the Transactions or otherwise prevent or materially delay such party from performing its obligations under this Agreement, other than any delay due to the failure to obtain the license, consent, approval, order or authorization of, or registration, declaration or filing with, any Governmental Entity in connection with the Transactions.

- "Measurement Date" shall have the meaning set forth in Section 4.4(a).
- "Merger" shall have the meaning set forth in the Preamble.
- "Merger Consideration" shall have the meaning set forth in Section 3.3(a).
- "Merger Sub" shall have the meaning set forth in the Preamble.
- "Minimum Tender Condition" shall have the meaning set forth in Exhibit A.
- "Nasdaq" shall mean the Nasdaq Stock Market.
- "Non-Disclosure Agreement" shall have the meaning set forth in Section 7.4.
- "Non-Election Shares" shall have the meaning set forth in Section 3.3(f).
- "Notice Period" shall have the meaning set forth in Section 7.5(e).
- "Offer" shall have the meaning set forth in the Preamble.
- "Offer Amendment Date" shall have the meaning set forth in Section 2.1(a)(i).
- "Offer Conditions" shall have the meaning set forth in Section 2.1(a)(i).
- "Offer Documents" shall have the meaning set forth in Section 2.1(b).
- "Offer Price" shall have the meaning set forth in the Preamble.
- "Open Source Software" shall mean any software that is generally available to the public in source code form under licenses substantially similar to those approved by the Open Source Initiative and listed at http://www.opensource.org/licenses, which licenses include the GNU General Public License (GPL), the GNU Library or Lesser General Public License (LGPL) or under any other license that purports to require one to (i) require, or condition the use, license or distribution of any software on, the disclosure, licensing or distribution of any source code and/or (ii) permit any licensee of any software to modify any source code relating to such software.
 - "Option Consideration" shall have the meaning set forth in Section 3.6(a)(ii).
 - "Order" shall have the meaning set forth in Section 8.1(d).
 - "Parent" shall have the meaning set forth in the Preamble.
 - "Parent Certificate" shall have the meaning set forth in Section 5.1.
 - "Parent Charter Documents" shall have the meaning set forth in Section 5.1.
 - "Parent Common Stock" shall mean the common stock, par value \$0.005 per share, of Parent.
 - "Parent Disclosure Schedule" shall mean the disclosure schedule delivered by Parent to the Company prior to the execution of this Agreement.

"Parent Employee Plans" shall mean any collective bargaining agreement or any bonus, pension, profit sharing, deferred compensation, incentive compensation, stock ownership, stock purchase, stock option, phantom stock, retirement, vacation, severance, change in control, retention, disability, death benefit, hospitalization, medical or other plan, arrangement or understanding (whether or not legally binding) providing benefits to any current or former employee, officer, director or independent contractor of Parent or any of its Subsidiaries, together with the Parent Pension Plans, any Parent "employee welfare benefit plans" (as defined in Section 3(1) of ERISA), the Parent International Employee Plans and the Parent Employment Agreements.

"Parent Employment Agreements" shall mean any employment, consulting, severance (other than standard severance agreements with employees entered into in the ordinary course of business consistent with past practice) or termination agreements or arrangements (other than standard employment agreements or offer letters entered into in the ordinary course of business consistent with past practice with employees outside the United States in accordance with local Law) between Parent or any of its Subsidiaries, on the one hand, and any current or former employee, executive officer or director of Parent or any of its Subsidiaries, on the other hand.

"Parent ERISA Affiliate" shall mean Parent or any of its Subsidiaries or any entity that would be treated as a "single employer" with Parent or any of its Subsidiaries within the meaning of Section 414(b), (c), (m) or (o) of the Code.

"Parent Intellectual Property" shall mean all Intellectual Property owned by Parent or any of its Subsidiaries.

"Parent International Employee Plan" shall mean each Parent Employee Plan and each government-mandated plan or program that has been adopted or maintained by Parent or any Parent ERISA Affiliate, whether informally or formally, or with respect to which Parent or any Parent ERISA Affiliate will or may have any liability, for the benefit of Parent Employees who perform services outside the United States.

"Parent Material Contracts" shall have the meaning set forth in Section 5.8(a).

"Parent Pension Plans" shall mean all "employee pension benefit plans" as defined in Section 3(2) of ERISA of Parent or any of its Subsidiaries.

"Parent Preferred Stock" shall have the meaning set forth in Section 5.4(a).

"Parent SEC Documents" shall have the meaning set forth in the preamble to Article V.

"Parent Share Election" shall have the meaning set forth in Section 2.1(c).

"Parent Share Election Share" shall have the meaning set forth in Section 2.2(a).

"Parent Software Products" shall mean all Software products developed and owned by Parent or any of its Subsidiaries that are (i) offered for license by Parent or its Subsidiaries, or (ii) used in the conduct of their respective businesses.

"Parent Stock Options" shall have the meaning set forth in Section 5.4(a).

"Parent Systems" shall have the meaning set forth in Section 5.9(f).

"Pending Offer" shall have the meaning set forth in the Preamble.

"Permitted Lien" shall mean any Lien (i) for Taxes or governmental assessments, charges or claims of payment not yet due, being contested in good faith or for which adequate accruals or reserves have been established; (ii) which is a carriers', warehousemen's, mechanics', materialmen's, repairmen's or other similar lien arising in the ordinary course of business consistent with past practice; (iii) arising from licenses of or other grants of rights to use Intellectual Property not incurred in connection with the borrowing of money; or (iv) which does not and would not reasonably be expected to materially impair the continued use of any owned real property or leased real property.

- "Person" shall mean an individual, corporation (including any non-profit corporation), general partnership, limited partnership, limited liability partnership, joint venture, estate, trust, company (including any limited liability company or joint stock company), firm or other enterprise, association, organization, Governmental Entity or other legal entity.
 - "Post-Effective Amendment" shall have the meaning set forth in Section 7.1(a).
 - "Post-Merger Employees" shall have the meaning set forth in Section 7.11(a).
 - "Promissory Note" shall have the meaning set forth in Section 2.5(b).
 - "Proration Amount of Cash and Stock Consideration" shall mean \$6.62 cash, without interest, plus 0.1064 of a share of Parent Common Stock.
 - "Proxy Statement/Prospectus" shall have the meaning set forth in Section 7.1(a).
- "Registration Statement" shall mean the registration statement on Form S-4 (or similar successor form) in connection with the issuance of Parent Common Stock in the Offer and the Merger, and any amendments or supplement thereto.
- "Related Person" shall mean, with respect to any party, an employee, officer, director, holder of more than 5% of the equity securities of such party, partner or member of such party or of any of such party's Subsidiaries and any member of his or her immediate family or any of their respective Affiliates.
- "Release" shall mean any actual or threatened release, spill, emission, leaking, dumping, injection, pouring, deposit, disposal, discharge, dispersal, leaching or migration into or through the environment (including ambient air, surface water, groundwater, land surface or subsurface strata).
 - "Representatives" of any Person shall mean such Person's accountants, consultants, legal counsel, financial advisors and agents and other representatives.
 - "SARs" shall have the meaning set forth in Section 3.6(b).
 - "SARs Consideration" shall have the meaning set forth in Section 3.6(b).
 - "SEC" shall mean the United States Securities and Exchange Commission.
 - "Schedule 14D-9" shall have the meaning set forth in Section 2.4.
 - "Schedule TO" shall have the meaning set forth in Section 2.1(b).
 - "Second Request" shall have the meaning set forth in Section 7.6(b).
 - "Section 16 Information" shall have the meaning set forth in Section 7.12.
 - "Securities Act" shall mean the United States Securities Act of 1933.
 - "Share Election Number" shall have the meaning set forth in Section 2.2(a).
 - "Significant Company Subsidiary" shall mean any Subsidiary of the Company that constitutes a significant Subsidiary within the meaning of Rule 1-02 of Regulation S-X of the SEC.
- "Significant Parent Subsidiary" shall mean any Subsidiary of Parent other than Merger Sub that constitutes a significant Subsidiary within the meaning of Rule 1-02 of Regulation S-X of the SEC.
- "Software" shall mean any and all (i) computer programs, including any and all software implementations of algorithms, models and methodologies, and all software development tools, whether in source code or object code; (ii) databases and compilations, including any and all data and collections of data, whether machine readable or otherwise; (iii) descriptions, schematics, flow-charts and other work product used to design, plan, organize and develop any of the foregoing; and (iv) documentation, manuals, specifications and training materials relating to the foregoing.
 - "Stock Price Per Share" shall have the meaning set forth in Section 2.1(a)(i).

- "Stock Proration Factor" shall have the meaning set forth in Section 2.2(b)(ii).
- "Subsequent Offering Period" shall have the meaning set forth in Section 2.1(a)(iv).
- "Subsidiary" of any Person shall mean any corporation, partnership, limited liability company, joint venture, trust, association, unincorporated organization or other legal entity of any kind of which such Person (either alone or through or together with one or more of its Subsidiaries) (a) owns, directly or indirectly, 50% or more of the capital stock or other equity interests, the holders of which are (i) generally entitled to vote for the election of the board of directors or other governing body of such legal entity or (ii) generally entitled to share in the profits or capital of such legal entity or (b) otherwise owns, directly or indirectly, an amount of voting securities sufficient to elect at least a majority of the board of directors or other governing body of such legal entity.
 - "Surviving Corporation" shall have the meaning set forth in Section 3.1.
- "Tax Return" shall mean all federal, state, local, provincial and non-United States Tax returns, declarations, statements, reports, schedules, forms and information returns and any amended Tax return relating to Taxes.
- "Taxes" shall mean taxes, governmental fees or like assessments or charges, whenever created or imposed, and whether of the United States or elsewhere, and whether imposed by a local, municipal, governmental, state, non-United States, federal or other Governmental Entity, including all interest, penalties and additions imposed with respect to such amounts.
 - "Termination Date" shall have the meaning set forth in Section 9.1(b)(i).
 - "Third Party Intellectual Property" shall have the meaning set forth in Section 4.9(b).
 - "Top-Up Notice" shall have the meaning set forth in Section 2.5(c).
 - "Top-Up Option" shall have the meaning set forth in Section 2.5(a).
 - "Top-Up Shares" shall have the meaning set forth in Section 2.5(a).
 - "Transactions" shall mean this Agreement, the Offer, the Merger, the Top-Up Option and the other transactions contemplated by this Agreement.
 - "Treasury Regulations" shall mean the rules and regulations promulgated under the Code.
 - "Uncertificated Shares" shall have the meaning set forth in Section 3.4(c).
 - "Voting Company Debt" shall have the meaning set forth in Section 4.4(a).
 - "Voting Parent Debt" shall have the meaning set forth in Section 5.4(a).

ARTICLE II.

THE OFFER

2.1 The Offer. (a) (i) Parent shall cause Merger Sub to, and Merger Sub shall, amend the Pending Offer and related Offer Documents to reflect the execution of this Agreement and the terms hereof as promptly as reasonably practicable after the date of this Agreement (but in no event later than the seventh Business Day following the date of this Agreement). The date on which Merger Sub amends the Pending Offer is referred to in this Agreement as the "Offer Amendment Date." The obligations of Merger Sub (and of Parent to cause Merger Sub) to accept for payment and to pay for any Company Shares tendered pursuant to the Offer shall be subject only to the terms and conditions of this Agreement and the satisfaction or waiver of those conditions set forth in Exhibit A (the "Offer Conditions"). Each Company Share accepted by Merger Sub in accordance with the terms and subject to the conditions of the Offer shall be purchased pursuant to the Offer in exchange for the right to receive, at the election of the Offer the office of Subject to proration as set forth in Section 2.2. Unless extended in accordance with

Section 2.1(a)(iii) or 2.1(a)(iv), the Offer shall expire at 5:00 p.m., New York City time, on October 31, 2011 (the "Initial Expiration Date") or, if the Offer has been extended in accordance with Section 2.1(a)(iii) or 2.1(a)(iv), at the time and date to which the Offer has been sexended (the Initial Expiration Date, or such later time and date to which the Offer has been extended in accordance with Section 2.1(a)(iii) or 2.1(a)(iv), the "Expiration Date"). Merger Sub expressly reserves the right (but shall not be obligated) at any time or from time to time in its sole discretion to waive any Offer Condition or modify or amend the terms of the Offer, except that, without the prior written consent of the Company, (A) the Minimum Tender Condition may not be amended or waived and (B) no change may be made to the Offer that (1) decreases the Offer Price or changes the form of consideration, (2) decreases the number of Company Shares to be purchased by Merger Sub in the Offer, (3) modifies the Offer or the Offer Conditions in a manner that adversely affects or reasonably could adversely affect the holders of Company Shares, (4) adds to the conditions set forth on Exhibit A, or (5) extends the Expiration Date of the Offer except as required or permitted by Sections 2.1(a)(iii) and 2.1(a)(iv).

- (ii) Subject to the terms and conditions of this Agreement and to the satisfaction or waiver (to the extent permitted hereunder) by Merger Sub of the Offer Conditions as of any scheduled Expiration Date, Merger Sub shall, and Parent shall cause Merger Sub to, accept for payment and pay for Company Shares validly tendered and not withdrawn pursuant to the Offer promptly after such scheduled Expiration Date (the date and time of acceptance for payment, the "Acceptance Time").
 - (iii) Merger Sub shall extend the Offer from time to time, including as follows
 - (A) Merger Sub shall extend the Offer for one or more periods of not more than 20 Business Days each if at any otherwise scheduled Expiration Date any of the Offer Conditions is not satisfied, and
- (B) Merger Sub shall extend the Offer for any period required by any rule, regulation, interpretation or position of the SEC or the staff thereof or the Nasdaq applicable to the Offer; provided, however, that in no circumstances shall Merger Sub be required to extend the Offer beyond the Termination Date nor may Merger Sub extend the Offer if at the time of such extension all Offer Conditions have been satisfied (other than pursuant to Section 2.1(a)(iv)).
- (iv) Subject to Section 2.1(a)(i), Merger Sub may, in its sole discretion, provide for one "subsequent offering period" (and one or more extensions thereof) in accordance with Rule 14d-11 under the Exchange Act (a "Subsequent Offering Period"), and Merger Sub shall, and Parent shall cause Merger Sub to, immediately accept and promptly pay for all Shares as they are validly tendered during any Subsequent Offering Period for the Offer in accordance with Rule 14d-11 under the Exchange Act at a price per Company Share equal to the Proration Amount of Cash and Stock Consideration. The Offer Documents shall provide for the possibility of a Subsequent Offering Period in a manner consistent with this Section 2.1(a)(iv). Following the Acceptance Time and the expiration of any Subsequent Offering Period, if applicable, Parent shall use its reasonable best efforts to cause the Merger to become effective as promptly as practicable.
- (collectively with all amendments and supplements thereto, the "Schedule TO") and Registration Statement with respect to the Offer that shall contain the offer to purchase and related letter of transmittal and other ancillary Offer documents and supplements thereto, the "Schedule TO") and Registration Statement with respect to the Offer that shall contain the offer to purchase and related letter of transmittal and other ancillary Offer documents and instruments pursuant to which the Offer shall be made (collectively with any supplements or amendments thereto, the "Offer Documents"). The Company shall use its reasonable best efforts to cause its independent public accountants to provide their consent to the use of the Company's audited financial statements in the Offer Documents. The Company and its counsel shall be given a reasonable opportunity to review and comment on the Offer Documents prior to their filing with the SEC and Parent and Merger Sub shall give due consideration to all reasonable additions, deletions or changes suggested thereto by the Company and its counsel. Parent and Merger Sub agree to (i) provide the Company and its counsel with a copy of any written comments (or a description of any oral comments) received by Parent, Merger Sub or their counsel from the SEC or its staff with respect to the Offer Documents promptly after receipt of such comments, (ii) consult with the Company (and give the Company reasonable opportunity to review) regarding

any such comments prior to responding thereto, and Parent and Merger Sub shall give due consideration to all reasonable additions, deletions or changes suggested thereto by the Company and its counsel, and (iii) provide the Company with copies of any written comments or responses thereto. Parent and Merger Sub shall respond promptly to any comments of the SEC or its staff with respect to the Offer Documents. Parent and Merger Sub agree to take all steps necessary to cause the Offer Documents to be disseminated to holders of Company Shares to the extent required by applicable federal securities Laws. Each of Parent, Merger Sub and the Company agrees to promptly correct any information provided by it for use in the Offer Documents if and to the extent that it has become false or misleading in any material respect. Parent and Merger Sub further agree to take all steps necessary to cause the Offer Documents as so corrected to be filed with the SEC and disseminated to holders of Company Shares, in each case as and to the extent required by applicable federal securities Laws.

- (c) Subject to Section 2.2, in the Offer each holder of Company Shares shall be entitled to elect (i) the number of Company Shares which such holder desires to exchange for the right to receive the Cash Price Per Share (a "Cash Election") and (ii) the number of Company Shares which such holder desires to exchange for the right to receive the Stock Price Per Share (a "Parent Share Election"). Any Cash Election or Parent Share Election shall be referred to herein as an "Election," and shall be made on a form consistent with that included in the Pending Offer or as agreed by Merger Sub and the Company for that purpose (a "Form of Election") included as part of the letter of transmittal included in the Offer Documents (the "LT"). Holders of record who hold Company Shares as nominees, trustees or in other representative capacities may submit multiple Forms of Election on behalf of their respective beneficial holders, but only one such Form of Election for each such beneficial holder.
- (d) Parent shall provide or cause to be provided to Merger Sub on a timely basis the funds and shares of Parent Common Stock necessary to purchase any Company Shares that Merger Sub becomes obligated to purchase pursuant to the Offer.
- (e) Merger Sub shall not terminate the Offer without the prior written consent of the Company, except if this Agreement is terminated pursuant to Article IX. If this Agreement is terminated pursuant to Article IX, Merger Sub shall, and Parent shall cause Merger Sub to, promptly (and in any event within 24 hours of such termination) terminate the Offer and not acquire any Company Shares pursuant thereto. If this Agreement is terminated pursuant to Article IX prior to the acquisition of Company Shares in the Offer, Merger Sub shall promptly return, and shall cause any depositary acting on behalf of Merger Sub to return promptly, in accordance with applicable Law, all tendered Company Shares to the registered holders thereof.
- (f) The Offer Price shall be adjusted to reflect appropriately the effect of any forward or reverse stock split, stock dividend (including any dividend or distribution of securities exercisable or exchangeable for or convertible into Parent Common Stock or Company Shares), stock sale, reorganization, recapitalization, reclassification, combination, exchange of shares or other like change with respect to shares of Parent Common Stock or Company Shares occurring on or after the date of this Agreement and prior to the payment by Merger Sub for Company Shares validly tendered and not properly withdrawn in connection with the Offer.
- (g) No fraction of a share of Parent Common Stock shall be issued to represent any fractional share interests in Parent Common Stock to be delivered hereunder in exchange for Company Shares, but in lieu thereof each holder of Company Shares who would otherwise be entitled to a fraction of a share of Parent Common Stock (after aggregating all fractional shares of Parent Common Stock that otherwise would be received by such holder) shall, upon surrender by such holder of each Certificate held thereby, receive from Parent an amount of cash (rounded to the nearest whole cent), without interest, equal to the product of (i) such fraction, multiplied by (ii) the Average Per Share Price (substituting the Acceptance Time for the Effective Time in the definition thereof). The parties hereto acknowledge that payment of the cash consideration in lieu of issuing fractional shares is not separately bargained for consideration, but merely represents a mechanical rounding off for purposes of simplifying the corporate and accounting complexities that would otherwise be caused by the issuance of fractional shares.
- 2.2 Proration of Offer Consideration. (a) Each Company Share with respect to which a Cash Election has been effectively made and not revoked, and that is not deemed converted into the right to receive the

Stock Price Per Share pursuant to Section 2.2(b) (each a "Cash Election Share"), shall have the right to receive the Cash Price Per Share. Each Company Share with respect to which a Parent Share Election has been effectively made and not revoked, and that is not deemed converted into the right to receive the Cash Price Per Share pursuant to Section 2.2(c) (a "Parent Share Election Share"), shall have the right to receive the Stock Price Per Share. The number of Company Shares to be converted into the right to receive the Stock Price Per Share shall equal 33.8% of the Company Shares validly tendered in the Offer and not withdrawn (the "Share Election Number"). If the aggregate number of Parent Share Election Shares is equal to the Share Election Number, then each such Parent Share Election Share shall be converted into the Stock Price Per Share.

- (b) If the aggregate number of Parent Share Election Shares is less than the Share Election Number, then
 - (i) each Parent Share Election Share shall be exchanged for the Stock Price Per Share;
- (ii) a stock proration factor (the "Stock Proration Factor") shall be determined by dividing (A) the excess of the Share Election Number over the total number of Parent Share Election Shares by (B) the total number of Cash Election Shares;
- (iii) a number of Cash Election Shares of each Company stockholder equal to the product of (A) the Stock Proration Factor and (B) the total number of Cash Election Shares of such Company stockholder (rounded up to the nearest whole number), shall be exchanged for the Stock Price Per Share (and a Parent Share Election shall be deemed to have been made with respect to such Shares); and
 - (iv) each Cash Election Share as to which a Parent Share Election is not deemed made pursuant to Section 2.2(b)(iii) shall be exchanged for the Cash Price Per Share.
- (c) If the aggregate number of Parent Share Election Shares is greater than the Share Election Number, then:
 - (i) each Cash Election Share shall be exchanged for the Cash Price Per Share;
- (ii) a cash proration factor (the "Cash Proration Factor") shall be determined by dividing (A) the excess of the Parent Share Election Shares over the Share Election Number by (B) the total number of Parent Share Election Shares;
- (iii) a number of Parent Share Election Shares of each Company stockholder equal to the product of (A) the Cash Proration Factor and (B) the total number of Parent Share Election Shares of such Company stockholder (rounded up to the nearest whole number), shall be exchanged for the Cash Price Per Share (and a Cash Election shall be deemed to have been made with respect to such Shares); and
 - (iv) each Parent Share Election Share as to which a Cash Election is not deemed made pursuant to Section 2.2(c)(iii) shall be exchanged for the Stock Price Per Share.
- 2.3 Election Procedures; Allocation of Offer Consideration. (a) The Form of Election, the LT and other appropriate and customary transmittal materials (which shall specify that delivery shall be effected, and risk of loss and title to the Certificates shall pass, only upon proper delivery of such Certificates to the Exchange Agent or, in the case of Company Shares represented by Book-Entry Shares, upon adherence to the procedures set forth therein) and pursuant to which each holder of Company Shares may make an Election, shall be mailed as part of the LT included in the Offer Documents.
- (b) Parent shall make available one or more Forms of Election as may reasonably be requested from time to time by all Persons who become holders (or beneficial owners) of Company Shares prior to the Election Deadline, and the Company shall provide to the Exchange Agent all information reasonably necessary for it to perform its duties as specified herein.
- (c) Any Election shall be deemed to have been properly made only if the Exchange Agent has actually received a properly completed Form of Election by the Election Deadline. A Form of Election shall be deemed properly completed only if accompanied by (i) one or more Certificates (or customary affidavits and

indemnification regarding the loss or destruction of such Certificates or the guaranteed delivery of such Certificates) representing all certificated Company Shares covered by such Form of Election or (ii) in the case of Book-Entry Shares, any additional documents specified by the procedures set forth in the Form of Election, together with a duly executed LT included in the Form of Election. Any Form of Election may be revoked or changed by the Person submitting such Form of Election prior to the Election Deadline. In the event any Form of Election is revoked prior to the Election Deadline, the Company Shares represented by such Form of Election shall become Non-Election Shares and Parent shall cause the Certificates, if any, representing the Company Shares to be promptly returned without charge to the Person submitting the Form of Election upon written request to that effect from the holder who submitted the Form of Election, except to the extent (if any) a subsequent election is properly made with respect to any or all of the applicable Company Shares. Subject to the terms of this Agreement and of the Form of Election, the Exchange Agent shall have reasonable discretion to determine whether any election, revocation or change has been properly or timely made and to disregard immaterial defects in the Forms of Election, and any good faith decisions of the Exchange Agent regarding such matters shall be binding and conclusive. None of Parent, Merger Sub or the Exchange Agent shall be under any obligation to notify any Person of any defect in a Form of Election.

- 2.4 Company Consent; Schedule 14D-9. Contemporaneously with the filing of the amendment to Schedule TO, the Company shall file with the SEC an amendment to its Solicitation/Recommendation Statement on Schedule 14D-9 (together with all amendments and supplements thereto, including amendments pursuant to this Section 2.4, the "Schedule 14D-9") containing, subject to Section 4.3(b), the Company Recommendation. The Company hereby consents to the inclusion of the Company Recommendation in the Offer Documents and to the inclusion of a copy of the Schedule 14D-9 with the Offer Documents mailed or furnished to the holders of Company Shares. Except with respect to any amendments filed in connection with a Company Acquisition Proposal or a Company Change of Recommendation, Parent and Merger Sub, and their respective counsel, shall be given a reasonable opportunity to review and comment on the Schedule 14D-9 prior to its filing with the SEC and the Company shall give due consideration to all reasonable additions, deletions or changes suggested thereto by Parent and its counsel. The Company agrees to (a) provide Parent, Merger Sub and their received by the Company or their counsel from the SEC or its staff with respect to the Schedule 14D-9 promptly after receipt of such comments, (b) consult with Parent and Merger Sub (and give each a reasonable opportunity to review) regarding any such comments prior to responding thereto and the Company shall give due consideration to all reasonable additions, deletions or changes suggested thereto by Parent and its counsel, and (c) provide Parent and Merger Sub with copies of any written comments or responses thereto. The Company shall respond promptly to any comments of the SEC or its staff with respect to the Schedule 14D-9. The Company, Parent and Merger Sub agree to promptly correct any information provided by each for use in the Schedule 14D-9 if and to the extent that such information has become false or misleading in any material respect. The Company further agrees to take all s
- 2.5 *Top-Up Option.* (a) Subject to Sections 2.5(b) and 2.5(c), the Company hereby grants to Merger Sub an irrevocable option (the "*Top-Up Option*"), for so long as this Agreement has not been terminated pursuant to its terms, to purchase from the Company up to the number (but not less than that number) of authorized and unissued Company Shares equal to the lowest number of Company Shares that, when added to the number of Company Shares owned by Parent, Merger Sub or any Subsidiary of Parent at the time of the exercise of the Top-Up Option, constitutes at least one Share more than 90% of the Company Shares (after giving effect to the issuance of Company Shares to be issued upon exercise of the Top-Up Option, the "*Top-Up Shares*")).
- (b) The Top-Up Option may be exercised by Merger Sub only once, in whole but not in part, at any time during the two-Business Day period following the Acceptance Time, or if any Subsequent Offering Period is provided, during the two-Business Day period following the expiration date of such Subsequent Offering Period, and only if Merger Sub owns as of such time more than 50% but less than 90% of the Company Shares outstanding; provided that, notwithstanding anything in this Agreement to the contrary, the Top-Up

Option shall not be exercisable (i) to the extent the number of Company Shares issuable upon exercise of the Top-Up Option would exceed the number of authorized but unissued and unreserved Company Shares, (ii) if any Order then in effect would prohibit the exercise of the Top-Up Option or the delivery of the Top-Up Shares, (iii) unless Parent or Merger Sub has accepted for payment all Company Shares validly tendered in the Offer and not withdrawn, and (iv) unless immediately after such exercise and the issuance of Company Shares pursuant thereto, Merger Sub would own at least 90% of the outstanding Company Shares (assuming the issuance of the Top-Up Shares). In any event, the Top-Up Option shall terminate upon the earlier to occur of (A) the Effective Time and (B) termination of this Agreement in accordance with Article IX. The aggregate purchase price payable for the Top-Up Shares being purchased by Merger Sub pursuant to the Top-Up Option shall be determined by multiplying the number of such Top-Up Shares by the greater of (i) the Closing price of a Company Share on Nasdaq the last trading day prior to the exercise of the Top-Up Option or (ii) the Cash Price Per Share, without interest. Such purchase price may be paid by Merger Sub, at its election, either (1) in cash or (2) by paying in cash an amount equal to not less than the aggregate par value of such Top-Up Shares and be executing and delivering to the Company a promissory note having a principal amount equal to the balance of such purchase price (the "Promissory Note"). Any such Promissory Note shall be full recourse against Parent and Merger Sub, shall bear interest at the rate of 5% per annum, shall mature on the first anniversary of the date of execution and delivery of such Promissory Note and may be prepaid without premium or penalty; provided, however, that upon any Event of Default, all principal and accrued interest thereunder shall immediately become due and payable. Without the prior written consent of the Company, the right to exercise the T

(c) In the event Merger Sub wishes to exercise the Top-Up Option, Merger Sub shall deliver to the Company written notice (the "Top-Up Notice") setting forth (i) the number of Top-Up Shares that Merger Sub intends to purchase pursuant to the Top-Up Option, (ii) the manner in which Merger Sub intends to pay the applicable purchase price, and (iii) the place, date and time at which the closing of the purchase of such Top-Up Shares by Merger Sub is to take place. The Top-Up Notice shall also include an undertaking signed by Parent and Merger Sub that, promptly following such exercise of the Top-Up Option, Merger Sub intends to consummate the Merger in accordance with Section 253 of the DGCL as contemplated by Section 7.2(c). At the closing of the purchase of the Top-Up Shares, Parent and Merger Sub shall cause to be delivered to the Company the consideration required to be delivered in exchange for the Top-Up Shares, and the Company shall cause to be issued to Merger Sub a certificate representing the Top-Up Shares or, at Parent's or Merger Sub's request or otherwise if the Company does not then have certificated Company Shares, the applicable number of uncertificated Company Shares by book-entry (the "Book-Entry Shares"). The parties shall use their reasonable best efforts to cause the closing of the purchase of the Top-Up Shares to occur on the same day that the Top-Up Notice is deemed received by the Company (provided that such notice is received no later than 10:00 a.m., New York City time on such day), pursuant to Section 10.2, and if not so consummated on such day, as promptly thereafter as possible. The parties further agree to use their reasonable best efforts to cause the Merger to be consummated in accordance with Section 253 of the DGCL as contemplated by Section 7.2(c) as close in time as possible to (including, to the extent possible, on the same day as) the issuance of the Top-Up Shares. Parent, Merger Sub and the Company shall cooperate to ensure that any issuance of the Top-Up Shares is accomplished

(d) Parent and Merger Sub understand that the Top-Up Shares shall not be registered under the Securities Act and shall be issued in reliance upon an exemption thereunder for transactions not involving a public offering. Each of Parent and Merger Sub represents and warrants to the Company that Merger Sub is, and shall be upon any exercise of the Top-Up Option, an "accredited investor" as defined in Rule 501 of Regulation D promulgated under the Securities Act. Each of Parent and Merger Sub represents, warrants and agrees that the Top-Up Option is being, and the Top-Up Shares shall be, acquired by Merger Sub for the purpose of investment and not with a view to or for resale in connection with any distribution thereof within the meaning of the Securities Act. Any certificates evidencing Top-Up Shares shall include any legends required by applicable securities Laws.

- (e) Any dilutive impact on the value of the Company Shares as a result of the issuance of the Top-Up Shares shall be taken into account in any determination of the fair value of any Dissenting Shares pursuant to Section 262 as contemplated by Section 3.9.
- 2.6 Stockholder Lists. In connection with the Offer, the Company shall cause its transfer agent to furnish Parent and Merger Sub with mailing labels, security position listings and readily available computer files containing the names and addresses of the record holders of the Company Shares as of the most recent practicable date and shall furnish or cause to be furnished to Parent and Merger Sub such information and assistance (including periodic updates of such information) as Parent or Merger Sub or their agents may reasonably request in communicating the Offer to the record and beneficial holders of the Company Shares. Subject to the requirements of applicable Law, and except for such actions as are reasonably necessary to disseminate the Offer Documents, each of Parent and Merger Sub shall hold all information and documents provided to it under this Section 2.6 in confidence in accordance with the Non-Disclosure Agreement, and shall use such information and documents only in connection with the Offer, and, if this Agreement has been terminated by Parent or Merger Sub, shall promptly deliver to the Company all such information and documents (and all copies thereof).
- 2.7 Directors. (a) Promptly upon the purchase by Merger Sub pursuant to the Offer of such number of Company Shares as would satisfy at least the Minimum Tender Condition, and from time to time thereafter, Merger Sub shall be entitled to designate such number of directors, rounded up to the next whole number, on the Board of Directors of the Company as shall give Merger Sub representation on the Board of Directors of the Company (after giving effect to any increase in the number of directors pursuant to this Section 2.7) and (iii) the percentage that such number of Company Shares so purchased bears to the total number of then-outstanding Company Shares on a fully diluted basis, and the Company shall, upon request by Merger Sub, promptly increase the size of the Board of Directors of the Company or use commercially reasonable efforts to seek the resignations of such number of directors as is necessary to provide Merger Sub with such level of representation and shall use commercially reasonable efforts to cause Merger Sub's designees to be so elected or appointed. Subject to the applicable requirements of Nasdaq, the Company shall also use commercially reasonable efforts to cause individuals designated by Merger Sub to constitute the same percentage of each committee of the Board of Directors of the Company as the percentage of each committee of the Board of Directors of the Company shall be subject to Section 14(f) of the Exchange Act. At the request of Merger Sub, the Company shall file with the SEC and mail to the stockholders the information required by Section 14(f) of the Exchange Act and Rule 14f-1 promulgated thereunder, and Parent and Merger Sub shall supply to the Company shall be solely responsible for such information. For purposes of this Agreement, such information required by Section 14(f) of the Exchange Act and Rule 14f-1 promulgated thereunder, and Parent and Merger Sub shall be solely responsible for such information. For purposes of this Agreement, such information required by Section 14(f) of
- (b) Notwithstanding the provisions of this Section 2.7, the Company, Parent and Merger Sub shall cause the Board of Directors of the Company to include, at all times prior to the Effective Time, at least three of the members of the Board of Directors of the Company, selected by the members of the Board of Directors of the Company, who were directors of the Company on the date of this Agreement ("Continuing Directors"), each of whom shall be an "independent director" as defined by Rule 4200(a)(15) of the Nasdaq Marketplace Rules and eligible to serve on the Company's audit committee under the Exchange Act and the applicable requirements of Nasdaq; provided, however, that if at any time prior to the Effective Time there shall be fewer than three Continuing Directors serving as directors of the Company for any reason, then the Board of Directors of the Company shall take all necessary action (including creating a committee of the Board of Directors of the Company) to cause an individual selected by the remaining Directors of the Company shall take all necessary action (including creating a committee of the Board of Directors of the Company) to cause an individual selected by the remaining Directors of the Company shall take all necessary action (including creating a committee of the Board of Directors of the Company) to cause an individual selected by the remaining Directors of the Company shall take all necessary action (including creating a committee of the Board of Directors of the Company) to cause an individual selected by the remaining Directors of the Company shall take all necessary action (including creating a committee of the Board of Directors of the Company) to cause an individual selected by the remaining Directors of the Company shall take all necessary action (including creating a committee of the Board of Directors of the Company) to cause an individual selected by the remaining Directors of the Company shall take all necessary action (including creating a committee of the Board of Directors of the

Affiliates to be appointed to serve on the Board of Directors of the Company (and such individual shall be deemed to be a Continuing Director for all purposes under this Agreement).

(c) Following the election or appointment of Merger Sub's designees pursuant to Section 2.7(a) and prior to the Effective Time, any action by the Company with respect to any amendment, supplement, modification or waiver of any term of this Agreement, any termination of this Agreement by the Company, any extension of time for the performance of any of the obligations or other acts of Parent or Merger Sub under this Agreement, any waiver of compliance with any of the agreements or conditions under this Agreement that are for the benefit of the Company, any amendment to the Certificate of Incorporation or Bylaws, any material modification of any Company Stock Plan, any authorization of an agreement between the Company and any of its Affiliates, on the one hand, and Parent, Merger Sub or any of their Affiliates, on the other hand, any exercise of the Company's rights or remedies under this Agreement and any action to seek to enforce any obligation of Parent or Merger Sub under this Agreement (or any other action by the Board of Directors of the Company with respect to the Transactions if such other action adversely affects, or could reasonably be expected to adversely affects, any of the holders of Company Shares other than Parent or Merger Sub) may only be authorized by, and shall require the authorization of, a majority of the Continuing Directors should there be only one). For purposes of considering any matter set forth in this Section 2.7(c), the Continuing Directors shall be permitted to meet without the presence of the other directors. The Continuing Directors shall have the authority to retain such counsel (which may include current counsel to the Company) and other advisors at the expense of the Company as determined by the Continuing Directors and shall have the authority to institute any action on behalf of the Company to enforce performance of this Agreement or any of the Company's rights hereunder. The Company shall indemnify and advance expenses to, and Parent shall cause the Company to indemnify and

(d) In the event that Merger Sub's designees are elected or appointed to the Board of Directors of the Company pursuant to Section 2.7(a), until the Effective Time, (i) the Board of Directors of the Company shall have at least such number of directors as may be required by the Nasdaq rules or the federal securities Laws who are considered independent directors within the meaning of such rules and Laws ("Independent Directors") and (ii) each committee of the Board of Directors of the Company that is required (or a majority of which is required) by the Nasdaq rules or the federal securities Laws to be composed solely of Independent Directors shall be so composed; provided, however, if the number of Independent Directors is reduced below the number of directors as may be required by such rules or Laws for any reason, the remaining Independent Director(s) shall be entitled to designate persons to fill such vacancies who shall be deemed to be Independent Directors for purposes of this Agreement or, if no Independent Director then remains, the other directors shall designate such number of directors as may be required by the Nasdaq rules and the federal securities Laws to fill such vacancies who shall not be shareholders or Affiliates of Parent or Merger Sub, and such Persons shall be deemed to be Independent Directors for purposes of this Agreement.

- (e) The provisions of this Section 2.7 are in addition to, and shall not limit, any right that Merger Sub, Parent or any Affiliate of Merger Sub or Parent may have (with respect to the election of directors or otherwise) under applicable Law as a holder or beneficial owner of Company Shares.
 - 2.8 Surrender of Certificates
 - (a) Exchange Agent. Parent has appointed Wells Fargo Bank, N.A. to act as exchange agent (the "Exchange Agent") for the Offer.
- (b) Parent To Provide Offer Consideration. Promptly following the Acceptance Time, Parent shall deposit with the Exchange Agent, for exchange in accordance with this Article II, (i) the Certificates representing the aggregate number of shares of Parent Common Stock issuable pursuant to Section 2.2 and (ii) cash in an amount sufficient for payment in respect of the aggregate Cash Price Per Share payable pursuant to Section 2.2 and for payment in lieu of fractional shares of Parent Common Stock to which holders of Company Shares may be entitled pursuant to Section 2.1(g) (collectively, the "Exchange Fund"). In the event that the shares and/or cash in the Exchange Fund shall be insufficient to fully satisfy all of the payment

obligations to be made by the Exchange Agent hereunder, Parent shall promptly make available to the Exchange Agent the amounts so required to satisfy such payment obligations in full. The Exchange Agent shall deliver the cash payments and Parent Common Stock contemplated to be paid for Company Shares pursuant to this Agreement out of the Exchange Fund as contemplated by this Section 2.8, the Exchange Fund shall not be used for any other purpose. Amounts of cash in the Exchange Fund shall be invested by the Exchange Agent as directed by Parent; provided, however, that: (i) no such investment or losses thereon shall affect the amounts payable to the holders of Company Shares and (ii) such investments shall be in obligations rated A-1 or P-1 or better by United States or America of any agency or instrumentality thereof and backed by the full faith and credit of the United States of America, in commercial paper obligations rated A-1 or P-1 or better by Moody's Investors Service, Inc. or Standard & Poor's Corporation, respectively, or in certificates of deposit, bank repurchase agreements or banker's acceptances of commercial banks with capital exceeding \$1 billion (based on the most recent financial statements of such bank that are then publicly available). Any net profit resulting from, or interest or income produced by, such investments shall be payable to

- (c) Transfers of Ownership. If shares of Parent Common Stock are to be issued in a name other than that in which the Certificates surrendered in exchange therefor are registered, it shall be a condition of the issuance thereof that the Certificates so surrendered shall be properly endorsed and otherwise in proper form for transfer and that the Persons requesting such exchange shall have paid to Parent or any agent designated by it any transfer or other Taxes required by reason of the issuance of shares of Parent Common Stock in any name other than that of the registered holder of the Certificates surrendered, or established to the satisfaction of Parent or any agent designated by it that such Tax has been paid or is not payable.
- (d) Lost, Stolen or Destroyed Certificates. If any Certificate shall have been lost, stolen or destroyed, upon the making of an affidavit of that fact by the Person claiming such Certificate to be lost, stolen or destroyed and, if required by Parent or the Exchange Agent to the extent in accordance with customary practice, the posting by such Person of a bond in such reasonable amount as Parent or the Exchange Agent shall direct as indemnity against any claim that may be made against them with respect to such Certificate, the Exchange Agent shall issue in exchange for such lost, stolen or destroyed Certificate the Offer Price payable to the holder thereof pursuant to Article II.
- (e) Required Withholding. Each of the Exchange Agent and Parent shall be entitled to deduct and withhold from any consideration, or other amounts, payable or otherwise deliverable pursuant to this Agreement to any holder or former holder of Company Shares or any other Person such amounts as are required to be deducted or withheld therefrom under any applicable Law (including any withholding provision of the Code and the Treasury Regulations promulgated thereunder). To the extent amounts are deducted or withheld pursuant to this Section 2.8, such amounts shall be treated for all purposes under this Agreement as having been paid to the Person to whom such amounts would otherwise have been paid and such amounts shall be remitted to the applicable Governmental Entity in accordance with applicable Law and notice thereof shall be provided to the applicable holder of Company Shares. Any purported withholding of Taxes from payments or other deliveries made in accordance with the provisions of this Agreement, the amount of which was forwarded to the relevant Governmental Entity, shall not be deemed a breach of this Agreement and the amounts so withheld shall be treated for all purposes under this Agreement as having been paid to the Person to whom such amounts would otherwise have been paid, notwithstanding that withholding of Taxes might have not been required.
- (f) No Liability. Notwithstanding anything to the contrary in this Section 2.8, none of the Exchange Agent, Parent or any party hereto shall be liable to a holder of Company Shares for any amount properly paid to a public official pursuant to any applicable abandoned property, escheat or similar Law.

ARTICLE III.

PLAN OF MERGER

- 3.1 The Merger. Upon the terms and subject to the conditions set forth in this Agreement, and in accordance with the DGCL and the DLLCA, at the Effective Time, Merger Sub shall be merged with and into the Company. As a result of the Merger, the separate corporate existence of Merger Sub shall cease and the Company shall continue as the surviving corporation (the "Surviving Corporation") and shall become a wholly owned Subsidiary of Parent. In the event Parent and the Company mutually determine prior to the Effective Time that the Merger would qualify as a tax-free reorganization within the meaning of Section 368(a) of the Code and the Treasury Regulations, this Agreement shall be deemed amended without further action of any party hereto or its stockholders to provide that the Merger shall consist of the Company merging with and into the Merger Sub and Merger Sub continuing as the Surviving Corporation, and Exhibit B, Exhibit C and Section 3.7 shall be modified as appropriate.
- 3.2 Effective Time; Closing Date. The closing of the Merger and the other Transactions (the "Closing") shall take place at the offices of Jones Day, 222 East 41st Street, New York, New York 10017, at 9:00 a.m. (local time) on the date that is the second Business Day after the satisfaction or waiver of the conditions set forth in Article VIII (other than those conditions which by their terms are to be satisfied or waived as of the Closing but subject to the satisfaction or waiver thereof), or at such other time, date or location as the parties hereto shall mutually agree. The date upon which the Closing actually occurs is referred to herein as the "Closing Date." On or before the Closing Date, the parties hereto shall cause the Merger to be consummated by filing a certificate of merger (the "Certificate of Merger") with the Secretary of State of the State of Delaware, in such form as is required by, and executed in accordance with, the relevant provisions of the DGCL and the DLLCA (the date and time of such filing of the Certificate of Merger (or such later time as may be agreed by each of the parties hereto and specified in the Certificate of Merger) being the "Effective Time").
 - 3.3 Effects of the Merger. At the Effective Time, without any action on the part of Parent, Merger Sub, the Company or the holders of any of the following securities, the following shall occur:
 - (a) Conversion of Company Shares. Subject to Section 3.4, each Non-Election Share that is outstanding as of the Effective Time (other than Company Shares that will be canceled pursuant to Section 3.3(b), Company Shares that will be converted into shares of Parent Common Stock pursuant to Schedule 6.1 and any Dissenting Shares) shall automatically be converted into the right to receive \$6.62 in cash, without interest, and 0.1064 of a share of Parent Common Stock (collectively, the "Merger Consideration"), upon surrender, in the manner provided in Section 3.4, of the certificate that formerly evidenced such Company Share.
 - (b) Parent-Owned Stock and Stock Held in Treasury. Each Company Share held in the treasury of the Company or owned by Parent or any direct or indirect wholly owned Subsidiary of the Company or of Parent immediately prior to the Effective Time shall be canceled and retired without any conversion or consideration paid in respect thereof and shall cease to exist.
 - (c) Capital Stock of Merger Sub. Each share or membership interest of Merger Sub issued and outstanding immediately prior to the Effective Time shall be converted into one validly issued, fully paid and nonassessable share of the Surviving Corporation. Each certificate evidencing ownership of such shares or membership interests of Merger Sub immediately prior to the Effective Time shall, as of the Effective Time, evidence ownership of such shares of the Surviving Corporation.
 - (d) Adjustments to Merger Consideration. The Merger Consideration shall be adjusted to reflect appropriately the effect of any forward or reverse stock split, stock dividend (including any dividend or distribution of securities exercisable or exchangeable for or convertible into Parent Common Stock or Company Shares), stock sale, reorganization, recapitalization, reclassification, combination, exchange of shares or other like change with respect to shares of Parent Common Stock or Company Shares occurring on or after the date of this Agreement and prior to the Effective Time.

- (e) Fractional Shares. No fraction of a share of Parent Common Stock shall be issued to represent any fractional share interests in Parent Common Stock to be delivered hereunder in exchange for Company Shares, but in lieu thereof each holder of Company Shares who would otherwise be entitled to a fraction of a share of Parent Common Stock (after aggregating all fractional shares of Parent Common Stock that otherwise would be received by such holder) shall, upon surrender by such holder of each Certificate held thereby, receive from Parent an amount of cash (rounded to the nearest whole cent), without interest, equal to the product of (i) such fraction, multiplied by (ii) the Average Per Share Price. The parties hereto acknowledge that payment of the cash consideration in lieu of issuing fractional shares is not separately bargained for consideration, but merely represents a mechanical rounding off for purposes of simplifying the corporate and accounting complexities that would otherwise be caused by the issuance of fractional shares.
- (f) Non-Election Shares. All Company Shares (other than Company Shares that have been validly tendered in the Offer and not validly withdrawn and with respect to which the Exchange Agent has received an effective, properly and fully completed Form of Election on or before the Expiration Date) (or such other time and date as the Company and Parent may agree) (the "Election Deadline") shall be deemed to be "Non-Election Shares"; provided, however, that if the Acceptance Time has not occurred prior to the Effective Time, all outstanding Company Shares as of the Effective Time shall be deemed to be "Non-Election Shares." Notwithstanding any other provision hereof, but subject to Section 3.4, each Company Share that is a Non-Election Share or a Dissenting Share shall by virtue of this Agreement and the Merger be converted into the right to receive the Merger Consideration.
- 3.4 Surrender of Certificates.
- (a) Exchange Agent. Parent has appointed Wells Fargo Bank, N.A. to act as Exchange Agent in the Merger.
- (b) Parent to Provide Merger Consideration. At or prior to the Effective Time, Parent shall deposit with the Exchange Agent in the Exchange Fund, for exchange in accordance with this Article III, (i) the Certificates representing aggregate shares of Parent Common Stock issuable pursuant to Section 3.3 and (ii) cash in an amount sufficient for payment in Ireu of fractional shares of Parent Common Stock to which holders of Company Shares may be entitled pursuant to Section 3.3(e). In the event that the shares and/or cash in the Exchange Fund shall be insufficient to fully satisfy all of the payment obligations to be made by the Exchange Agent hereunder, Parent shall promptly make available to the Exchange Agent the amounts so required to satisfy such payment obligations in full. The Exchange Agent shall deliver the cash payments and Parent Common Stock contemplated to be paid for Company Shares pursuant to this Agreement out of the Exchange Fund as contemplated hereby. Except as contemplated by this Section 3.4, the Exchange Fund shall not be used for any other purpose. Amounts of cash in the Exchange Fund shall be invested by the Exchange Agent as directed by Parent; provided, however, that: (i) no such investment or losses thereon shall affect the Merger Consideration payable to the holders of Company Shares and (ii) such investments shall be in obligations of or guaranteed by the United States or America of any agency or instrumentality thereof and backed by the full faith and credit of the United States of America, in commercial paper obligations rated A-1 or P-1 or better by Moody's Investors Service, Inc. or Standard & Poor's Corporation, respectively, or in certificates of deposit, bank repurchase agreements or banker's acceptances of commercial banks with capital exceeding \$1 billion (based on the most recent financial statements of such bank that are then publicly available). Any net profit resulting from, or interest or income produced by, such investments shall be payable to the Surviving Corporation o
- (c) Exchange Procedures. As soon as reasonably practicable after the Effective Time (but no later than three Business Days thereafter), Parent shall cause the Exchange Agent to mail to each holder of record of Non-Election Shares as of the Effective Time of one or more certificates (each, a "Certificate") or uncertificated Company Shares ("Uncertificated Shares") that immediately prior to the Effective Time represented issued and outstanding Non-Election Shares that were converted into the right to receive Merger Consideration pursuant to Section 3.3: (i) an LT in customary form (which shall specify that delivery shall be effected, and risk of loss and title to the Certificate or Uncertificated Shares shall pass, only upon delivery of

the Certificate or the Uncertificated Shares to the Exchange Agent), that shall also be in such form and have such other provisions as Parent and the Company may reasonably specify, and (ii) instructions for use in effecting the surrender of the Certificate or the transfer of Uncertificated Shares in exchange for the Merger Consideration. Upon (x) surrender of Certificates for cancellation to the Exchange Agent or (y) receipt of an "agent's message" by the Exchange Agent (or such other evidence, if any, of transfer as the Exchange Agent may reasonably request) in the case of a Book-Entry Shares, together with such LT, duly completed and validly executed in accordance with the instructions thereto, the holders of such Certificates or Uncertificated Shares shall be entitled to receive in exchange therefor the Merger Consideration and cash in lieu of fractional shares of Parent Common Stock to which such holder is entitled pursuant to Section 3.3(e), and the Certificates so surrendered or the Uncertificated Shares so transferred shall forthwith be canceled. Until so surrendered or canceled, outstanding Certificates and Uncertificated Shares shall be deemed from and after the Effective Time to evidence only the right to receive, upon surrender and without interest, the Merger Consideration into which the Company Shares theretofore represented by such Certificates shall have been converted pursuant to Section 3.3(e).

- (d) Transfers of Ownership. If shares of Parent Common Stock are to be issued in a name other than that in which the Certificates surrendered in exchange therefor are registered, it shall be a condition of the issuance thereof that the Certificates so surrendered shall be properly endorsed and otherwise in proper form for transfer and that the Persons requesting such exchange shall have paid to Parent or any agent designated by it any transfer or other Taxes required by reason of the issuance of shares of Parent Common Stock in any name other than that of the registered holder of the Certificates surrendered, or established to the satisfaction of Parent or any agent designated by it that such Tax has been paid or is not payable.
- (e) Lost, Stolen or Destroyed Certificates. If any Certificate shall have been lost, stolen or destroyed, upon the making of an affidavit of that fact by the Person claiming such Certificate to be lost, stolen or destroyed and, if required by Parent or the Exchange Agent to the extent in accordance with customary practice, the posting by such Person of a bond in such reasonable amount as Parent or the Exchange Agent may direct as indemnity against any claim that may be made against them with respect to such Certificate, the Exchange Agent shall issue in exchange for such lost, stolen or destroyed Certificate the Merger Consideration into which the Company Shares represented by such Certificate immediately prior to the Effective Time shall have been converted pursuant to Section 3.3 and any cash in lieu of fractional shares of Parent Common Stock payable to the holder thereof pursuant to Section 3.3(e).
- (f) Required Withholding. Each of the Exchange Agent and Parent shall be entitled to deduct and withhold from any consideration, or other amounts, payable or otherwise deliverable pursuant to this Agreement to any holder or former holder of Company Shares or any other Person such amounts as are required to be deducted or withheld therefrom under any applicable Law (including any withholding provision of the Code and the Treasury Regulations promulgated thereunder). To the extent amounts are deducted or withheld pursuant to this Section 3.4, such amounts shall be treated for all purposes under this Agreement as having been paid to the Person to whom such amounts would otherwise have been paid and such amounts shall be remitted to the applicable Governmental Entity in accordance with the applicable Law and notice thereof shall be provided to the applicable holder of Company Shares. Any purported withholding of Taxes from payments or other deliveries made in accordance with the provisions of this Agreement, the amount of which was forwarded to the relevant Governmental Entity, shall not be deemed a breach of this Agreement and the amount so withheld shall be treated for all purposes under this Agreement as having been paid to the Person to whom such amounts would otherwise have been paid, notwithstanding that withholding of Taxes might have not been required. Any holder of former holder of Company Shares or Company Stock Options that instructed Parent or the Exchange Agent to sell any part of the shares of Parent Common Stock due to such holder shall have no claim against Parent or the Exchange Agent and/or their officers or directors with respect to such sale, including the timing of such sale, the price received for the sold shares or the transaction costs.
- (g) Termination of Exchange Fund. Promptly following the date that is one year after the Closing Date, the Surviving Corporation shall be entitled to require the Exchange Agent to deliver to Parent any portion of the Exchange Fund which has not been disbursed to holders of Company Shares (including all interest and

other income received by the Exchange Agent in respect of the Exchange Fund), and thereafter each holder of a Company Share may surrender or transfer, as applicable, such Company Share to Parent or the Surviving Corporation and (subject to abandoned property, escheat and other similar Laws) receive in consideration therefor the Merger Consideration into which such Company Shares shall have been converted pursuant to Section 3.3 and cash in lieu of fractional shares of Parent Common Stock pursuant to Section 3.3(e), in each case without interest, but such holder shall have no greater rights against Parent or the Surviving Corporation under applicable Law.

- (h) No Liability. Notwithstanding anything to the contrary in this Section 3.4, none of the Exchange Agent, Parent or any party hereto shall be liable to a holder of shares of Parent Common Stock or Company Shares for any amount properly paid to a public official pursuant to any applicable abandoned property, escheat or similar Law.
- 3.5 Company's Transfer Books Closed; No Further Ownership Rights in Company Shares. At the Effective Time: (a) the share transfer books of the Company shall be deemed closed, and no transfer of any Company Shares or certificates with respect thereto shall thereafter be made or consummated and (b) all holders of Company Shares shall cease to have any rights as stockholders of the Company except for any right to receive the Merger Consideration and cash in lieu of fractional shares of Parent Common Stock pursuant to Section 3.3(e). The Merger Consideration issued in accordance with the terms hereof (including any cash paid in lieu of fractional Shares of Parent Common Stock pursuant to Section 3.3(e)) shall be deemed to have been issued in full satisfaction of all rights pertaining to such Company Shares.
 - 3.6 Stock Options; SARs; Restricted Stock.
 - (a) Stock Options.
- (i) As soon as reasonably practicable after the date of this Agreement, the Company will take (or will cause to be taken) all actions necessary (including providing such notices, adopting such amendments to the Company Stock Plans and, taking such other actions as are reasonably requested by Parent) such that:
 - (A) All outstanding options or rights to acquire Company Shares (each, a "Company Stock Option") under the S1 Corporation 1997 Stock Option Plan (the "1997 Plan") and the Security First Technologies Corporation 1998 Directors' Stock Option Plan (the "1998 Plan"), if elected by the holder, will be exercised effective as of immediately prior to the Effective Time, with the effect that the Company Shares issuable upon exercise will be deemed for all purposes to be issued and outstanding immediately prior to the Effective Time and will have the right to receive the Merger Consideration.
 - (B) Without limiting the generality or effect of the preceding paragraph (A), the holders of all Company Stock Options under the 1997 Plan and the 1998 Plan will be notified that such Company Stock Options may be exercised at any time during the period that commences on the date of this Agreement and ends on the day before the Effective Time (the "Exercise Period"), provided that (A) any such exercise, to the extent that it relates to a Company Stock Option that would become exercisable only at the Effective Time, will be contingent until, and will become effective only upon, the occurrence of the Effective Time and (B) no Company Stock Option may be exercised after the Exercise Period.

The Company shall provide Parent and its counsel with a reasonable opportunity to review and comment on all documents evidencing the matters set forth in this Section 3.6 prior to finalizing the same and/or providing copies to Company employees, and the Company and Parent shall act in good faith to mutually agree on all reasonable additions, deletions or changes suggested by Parent and its counsel to the extent practicable taking into account any applicable deadlines.

(ii) Each outstanding Company Stock Option under the 1997 Plan or the 1998 Plan that is not exercised before the end of the Exercise Period, and any other Company Stock Option that is outstanding as of immediately before the Effective Time, shall be terminated and canceled at the Effective Time, and the holder of each Company Stock Option under the 2003 Plan and each Company Stock Option that shall have vested as of or prior to the Effective Time pursuant to the terms of the applicable Company Stock Plan and/or related award agreement will, subject to any required Tax withholding, be entitled to receive an amount in cash guilal

to the product of (1) the excess, if any, of (A) the Blended Value over (B) the exercise price per Company Share subject to such Company Stock Option and (2) the total number of Company Shares subject to such Company Stock Option as in effect immediately prior to the Effective Time (the "Option Consideration"); provided, however, that if the Option Consideration is zero or a negative number as of the Effective Time, such Company Stock Option shall be canceled and no amount shall be paid in respect thereof. Parent shall pay or cause to be paid the Option Consideration to the holders of the Company Stock Options in a lump sum as soon as practicable after the Effective Time but in no event later than five Business Days following the Effective Time.

- (b) SARs. At the Effective Time, each stock appreciation right granted under the applicable Company Stock Plan (the "SARs") shall be canceled at the Effective Time, and the holder of each Unit that shall have vested as of or prior to the Effective Time pursuant to the applicable Company Stock Plan will, subject to any required Tax withholding, be entitled to receive an amount in cash equal to the product of (i) the excess, if any, of (A) the Blended Value over (B) the exercise price per share of the Company Shares, if any, subject to such Units and (ii) the total number of Company Shares, if any, subject to such Units and in the total number of Company Shares, if any, subject to such Units and in the total number of Company Shares, if any, subject to such Units and (ii) the total number of Company Shares, if any, subject to such Units and (iii) the total number of Company Shares, if any, subject to such Units and (ii) the total number of Company Shares, if any, subject to such Units and (iii) the total number of Company Shares, if any, subject to such Units and (iii) the total number of Company Shares, if any, subject to such Units and (iii) the total number of Company Shares, if any, subject to such Units and (iii) the total number of Company Shares, if any, subject to such Units and (iii) the total number of Company Shares, if any, subject to such Units and (iii) the total number of Company Shares, if any, subject to such Units and (iii) the total number of Company Shares, if any, subject to such Units and (iii) the total number of Company Shares, if any, subject to such Units and (iii) the total number of Company Shares, if any, subject to such Units and (iii) the total number of Company Shares, if any, subject to such Units and (iii) the total number of Company Shares, if any, subject to such Units and (iii) the total number of Company Shares, if any, subject to such Units and (iii) the total number of Company Shares, if any, subject to such Units and (iii) the total number of Company Shares, if any, subjec
- (c) Restricted Stock and Restricted Units. Except as set forth in Schedule 6.1 of the Company Disclosure Schedule, at the Effective Time, each outstanding restricted Company Share, restricted stock unit and restricted cash unit that shall have vested as of or prior to the Effective Time pursuant to the applicable Company Stock Plan shall be treated as an outstanding fully vested Company Share and shall have the right to receive the Merger Consideration.
- (d) Board Resolutions. At or prior to the Effective Time, the Company, the Board of Directors of the Company and the compensation committee of the Board of Directors of the Company, as applicable, shall adopt any resolutions and take any actions which are necessary to effectuate the provisions of this Section 3.6 including all necessary action to ensure that from and after the Effective Time neither Parent nor the Surviving Corporation shall be required to deliver Company Shares or other capital stock of the Company to any Person pursuant to or in settlement of Company Stock Options, and all Company Stock Plans or other Company Employee Plans conferring any rights to Company Shares or other capital stock of the Company shall be deemed to be amended to be in conformity with this Section 3.6.
 - 3.7 Certificate of Incorporation, By-Laws and Directors and Officers of the Surviving Corporation.
- (a) *Certificate of Incorporation*. At the Effective Time, the certificate of incorporation of the Company shall be amended by virtue of the Merger in the form set forth on Exhibit B and, as so amended, shall be the certificate of incorporation of the Surviving Corporation until thereafter amended as provided by applicable Law and such certificate of incorporation.
- (b) At the Effective Time, the by-laws of the Surviving Company shall be in the form set forth on Exhibit C and shall be the by-laws of the Surviving Corporation until thereafter amended as provided by Law, the certificate of incorporation of the Surviving Corporation and such by-laws.
- (c) Directors and Officers. The initial directors of the Surviving Corporation shall be the directors of Merger Sub immediately prior to the Effective Time, each to hold office in accordance with the certificate of incorporation and by-laws of the Surviving Corporation until their respective successors are duly elected or appointed and qualified. The initial officers of the Surviving Corporation shall be the officers of the Company immediately prior to the Effective Time, each to hold office in accordance with the certificate of incorporation and by-laws of the Surviving Corporation until their respective successors are duly appointed.
- 3.8 Taking of Necessary Action; Further Action. If, at any time after the Effective Time, any further action is necessary or desirable to carry out the purposes of this Agreement and the Transactions, the officers and directors of either the Surviving Corporation or Parent may take any and all such lawful and necessary action.

- 3.9 Appraisal Rights/Dissenting Shares. (a) Notwithstanding any provision of this Agreement to the contrary and to the extent available under the DGCL, all Non-Election Shares that are outstanding immediately prior to the Effective Time and that are held by stockholders of the Company who have neither voted in favor of the Merger nor consented thereto in writing and who have demanded properly in writing appraisal for such Company Shares in accordance with Section 262 of the DGCL (collectively, the "Dissenting Shares") shall not be converted into, or represent the right to receive, the Merger Consideration. Such stockholders shall be entitled to receive payment of the appraised value of Dissenting Shares held by them in accordance with the provisions of Section 262 of the DGCL, except that all Dissenting Shares held by stockholders of the Company who have failed to perfect or who effectively have withdrawn or lost their rights to appraisal of such Dissenting Shares under Section 262 of the DGCL, shall thereupon be deemed to have been converted into, and to have become exchangeable for, as of the Effective Time, the right to receive the Merger Consideration, without any interest thereon, upon surrender, in the manner provided in Article III, of the Certificate or Certificates that formerly evidenced such Company Shares.
- (b) The Company shall give Parent (i) prompt notice of any demands for appraisal received by the Company, withdrawals of such demands and any other similar instruments served pursuant to the DGCL and received by the Company and (ii) the opportunity to direct all negotiations and proceedings with respect to demands for appraisal under the DGCL. The Company shall not, except with the prior written consent of Parent, make any payment with respect to any demands for appraisal or offer to settle or settle any such demands.
- (c) At the Effective Time, the effect of the Merger shall be as provided in the applicable provisions of the DGCL and the DLLCA. Without limiting the generality of the foregoing, at the Effective Time, all of the property, rights, privileges, powers and franchises of the Company and Merger Sub shall vest in the Surviving Corporation, and all debts, liabilities, obligations, restrictions, disabilities and duties of each of the Company and Merger Sub shall become the debts, liabilities, obligations, restrictions, disabilities and duties of the Surviving Corporation.

ARTICLE IV.

REPRESENTATIONS AND WARRANTIES OF THE COMPANY

Except as disclosed in (a) the reports, schedules, forms, statements and other documents filed or furnished by the Company with the SEC since December 31, 2009 (the "Company SEC Documents") filed on or prior to the date five days prior to the date hereof (excluding any risk factor disclosure and disclosure of risks included in any "forward-looking statements" disclaimer or similar statements included in such Company SEC Documents that are predictive, forward-looking or primarily cautionary in nature); provided that this clause (a) shall not qualify Sections 4.1, 4.3, 4.4(a), 4.7, 4.18 or 4.20, or (b) the Company Disclosure Schedule, the Company represents and warrants to Parent and Merger Sub as follows:

- 4.1 Organization, Standing and Power. Each of the Company and each Significant Company Subsidiary is duly organized, validly existing and, to the extent applicable, in good standing under the Laws of the jurisdiction in which it is organized and has full corporate power and authority to conduct its businesses as presently conducted. The Company and each Significant Company Subsidiary is duly qualified to do business in each jurisdiction where the nature of its business or its ownership or leasing of its properties make such qualification necessary and the failure to so qualify has had or would reasonably be expected to have a Material Adverse Effect on the Company. The Company has made available to Parent true and complete copies of its certificate of incorporation and by-laws as amended through the date of this Agreement (together, the "Company Charter Documents") and the comparable charter and organizational documents of each Significant Company Subsidiary, in each case as amended through the date of this Agreement.
- 4.2 Company Subsidiaries. (a) All the outstanding shares of capital stock of each Subsidiary of the Company have been validly issued and are fully paid and nonassessable and, except as set forth in the Company Disclosure Schedule, are owned by the Company, by another Subsidiary of the Company and another Subsidiary of the Company, free and clear of all pledges, liens, charges, mortgages,

encumbrances and security interests of any kind or nature whatsoever (collectively, "Liens") other than Permitted Liens.

- (b) Except for its interests in the Company's wholly owned Subsidiaries and except for the ownership interests set forth in the Company Disclosure Schedule, the Company does not own, directly or indirectly, any capital stock, membership interest, partnership interest, joint venture interest or other equity interest in any other Person.
- 4.3 Authority; Execution and Delivery; Enforceability. (a) The Company has all requisite corporate power and authority to execute and deliver this Agreement. The execution and delivery by the Company of this Agreement and the consummation by the Company of the Transactions have been duly authorized by all necessary corporate action on the part of the Company, subject, in the case of the Merger, only to adoption of this Agreement by the holders of a majority of the outstanding Company Shares entitled to vote on such matter, to the extent required by applicable Law (the "Company Stockholder Approval"). The Company has duly executed and delivered this Agreement, and this Agreement constitutes its legal, valid and binding obligation, enforceable against it in accordance with its terms.
- (b) The Board of Directors of the Company, at a meeting duly called and held in compliance with the requirements of the DGCL, has (i) determined that the Transactions are fair to, and in the best interests of, the Company and its stockholders, and that, considering the financial position of the merging companies, no reasonable concern exists that the Surviving Corporation shall be unable to fulfill the obligations of the Company to its creditors existing as of immediately prior to the Effective Time; (ii) adopted and approved this Agreement and approved the Transactions; (iii) determined to recommend that the stockholders of the Company accept the Offer and tender their Company Shares to Merger Sub in the Offer and, to the extent required by applicable Law, adopt this Agreement and approve the Transactions (the "Company Recommendation"); and (iv) adopted a resolution having the effect of causing the Company not to be subject to any restriction set forth in any state takeover law or similar Law, including Section 203 of the DGCL, that would otherwise apply to the Transactions.
- 4.4 Capital Structure. (a) The authorized share capital of the Company consists of 350,000,000 Company Shares and 25,000,000 shares of preferred stock, \$0.01 par value ("Company Preferred Stock"). At the close of business on September 29, 2011 (the "Measurement Date"), (i) 54,983,593 Company Shares and no shares of Company Preferred Stock were issued and outstanding, (ii) no Company Shares and Company Preferred Stock were held by the Company in its treasure, and (iii) 6,402,815 Company Shares were reserved for issuance under Company Employee Plans, of which 4,283,545 were subject to outstanding options or the grant of rights to purchase Company Shares, 805,241 were restricted Company Shares and 1,314,029 Company Shares were available for future option or restricted share grants. Except as set forth above, at the close of business on the Measurement Date, no shares of capital stock or other voting securities of the Company Shares sare, and all such shares that may be issued prior to the Effective Time shall be when issued, duly authorized, validly issued, fully paid and nonassessable and not subject to or issued in violation of any purchase option, call option, right of first refusal, preemptive right, subscription right or any similar right under any provision of the DGCL, the Company Charter Documents or any Contract to which the Company is a party or otherwise bound. There are no bonds, debentures, notes or other indebtedness of the Company having the right to vote (or convertible into, or exchangeable for, securities having the right to vote) on any matters on which holders of Company Shares may vote ("Voting Company Debt"). Except as set forth above, as of the date of this Agreement, there are no options, warrants, rights, convertible or exchangeable securities, "phantom" stock rights, stock appreciation rights, stock-based performance units, commitments, Contracts, arrangements or undertakings of any kind to which the Company or any of its Subsidiaries to issue, deliver or sell, or cause to be issued, deli

its Subsidiaries to repurchase, redeem or otherwise acquire any shares of capital stock of the Company or any of its Subsidiaries.

- (b) The Company Disclosure Schedule sets forth the following information with respect to each Company Stock Option or restricted Company Share as of the Measurement Date: (i) the name of the grantee; (ii) the particular plan, if applicable, pursuant to which such award was granted, (iii) the number of Company Shares subject to such award; (iv) the exercise price, if any, of such award; (v) the date on which such award was granted; (vi) the applicable vesting schedule, including the vesting commencement date and any accelerated vesting provisions; and (vii) the date on which such award expires. The Company has made available to Parent accurate and complete copies of all equity plans pursuant to which the Company has granted such awards that are currently outstanding and the form of all equity award agreements evidencing such awards. All Company Shares subject to issuance as aforesaid have been duly authorized and, upon issuance on the terms and conditions specified in the instrument pursuant to which they are issuable, shall be validly issued, fully paid and nonassessable. Except as set forth on Schedule 6.1 of the Company Disclosure Schedule, effective as of the Effective Time, each outstanding award of restricted Company Shares then outstanding shall become fully vested and all restrictions therein shall lapse. All outstanding Company Shares, all outstanding Company Stock Options, and all outstanding shares of capital stock of each Subsidiary of the Company Employee Plans.
- 4.5 No Conflicts; Consents. (a) The execution and delivery by the Company of this Agreement does not, and the consummation of the Transactions shall not, conflict with, or result in any violation of or default (with or without notice or lapse of time, or both) under, or give rise to a right of termination, cancelation or acceleration of any obligation or to loss of a material benefit under, or to increased, additional, accelerated or guaranteed rights or entitlements of any Person under, or result in the creation of any Lien upon any of the properties or assets of the Company or any of its Subsidiaries under, any provision of (i)(A) the Company Charter Documents or (B) the comparable charter or organizational documents of any Subsidiary of the Company, (ii) any Company Material Contract (as defined herein) or (iii) subject to the filings and other matters referred to in Section 4.5(b), any material Law applicable to the Company or any of its Subsidiaries or their respective properties or assets other than, in the case of clauses (i)(B), (ii) or (iii) above, any such items that have not had and would not reasonably be expected to have a Material Adverse Effect on the Company.
- (b) The execution and delivery of this Agreement by the Company does not and the consummation of the Transactions do not, and the performance of this Agreement and the Transactions by the Company shall not, require any consent, approval, authorization or permit of, or filing with or notification to, any Governmental Entity or any third party, except (i) for applicable requirements, if any, of the Securities Act, the Exchange Act, state securities Laws ("Blue Sky Laws"), the HSR Act, the requirements of any Governmental Entity under applicable competition, antitrust or non-United States investment Laws, the required approvals of this Agreement by the Company's stockholders pursuant to the DGCL, the filing of the appropriate merger documents as required by the DGCL or the DLLCA, and such other filings, notices, permits, authorizations, consents or approvals as may be required by reason of the status of Parent, Merger Sub or their Affiliates, and (ii) where the failure to obtain such consents, approvals, authorizations or permits, or to make such filings or notifications, would not reasonably be expected to have a Material Adverse Effect on the Company.
- 4.6 SEC Documents; Undisclosed Liabilities. (a) The Company has filed all reports, schedules, forms, statements and other documents required to be filed by the Company with the SEC since December 31, 2009 pursuant to Sections 13(a) and 15(d) of the Exchange Act.
- (b) As of its respective date, each Company SEC Document filed with the SEC complied in all material respects with the requirements of the Exchange Act or the Securities Act, as the case may be, and the rules and regulations of the SEC promulgated thereunder applicable to such Company SEC Document, and did not contain any untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they were made, not misleading. Except to the extent that information contained in any Company SEC Document filed with the SEC has been revised or superseded by a later filed Company SEC Document, none of the Company SEC

Documents filed with the SEC contained any untrue statement of a material fact or omits to state any material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they were made, not misleading. The consolidated financial statements of the Company included in the Company SEC Documents filed with the SEC comply as to form in all material respects with applicable accounting requirements and the published rules and regulations of the SEC with respect thereto, have been prepared in accordance with GAAP applied on a consistent basis during the periods involved (except as may be indicated in the notes thereto) and fairly present in all material respects the consolidated financial position of the Company and its consolidated Subsidiaries as of the dates thereof and the consolidated results of their operations and cash flows for the periods shown (subject, in the case of unaudited statements, to normal year-end audit adjustments).

- (c) Except (i) as reflected or reserved against in the balance sheet (or the notes thereto) as of December 31, 2010 included in the Company SEC Documents, (ii) as permitted or contemplated by this Agreement, (iii) for liabilities and obligations incurred since December 31, 2010 in the ordinary course of business, and (iv) for liabilities or obligations which have been discharged or paid in full in the ordinary course of business, neither the Company nor any of its Subsidiaries has any liabilities or obligations of any nature, whether or not accrued, contingent or otherwise, that would be required by GAAP to be reflected on a consolidated balance sheet of the Company and its consolidated Subsidiaries (or in the notes thereto), other than those that would not reasonably be expected to have a Material Adverse Effect on the Company.
- (d) The Company maintains a system of "internal control over financial reporting" (as defined in Rules 13a-15(f) and 15d-15(f) of the Exchange Act) sufficient to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with GAAP and includes those policies and procedures that (i) pertain to the maintenance of records that in reasonable detail accurately and fairly reflect the transactions and dispositions of the Company's and its Subsidiaries' assets; (ii) provide reasonable assurance that transactions are recorded as necessary to permit preparation of financial statements in accordance with GAAP, and that the Company's and its Subsidiaries' receipts and expenditures are being made only in accordance with authorizations of the Company's management and directors; and (iii) provide reasonable assurance regarding prevention or timely detection of unauthorized acquisition, use or disposition of the Company's and its Subsidiaries' assets that could have a material effect on the Company's financial statements.
- (e) The "disclosure controls and procedures" (as defined in Rules 13a-15(e) and 15d-15(e) of the Exchange Act) utilized by the Company are reasonably designed to ensure that all information (both financial and non-financial) required to be disclosed by the Company in the reports that it files or submits under the Exchange Act is recorded, processed, summarized and reported within the time periods specified in the rules and forms of the SEC and that all such information required to be disclosed is accumulated and communicated to the management of Company, as appropriate, to allow timely decisions regarding required disclosure and to enable the chief executive officer and chief financial officer of the Company to make the certifications required under the Exchange Act with respect to such reports.
- 4.7 Absence of Certain Changes or Events. Since December 31, 2010 until the date of this Agreement, there has not been any event, change, effect or development that has had or would reasonably be expected to have a Material Adverse Effect on the Company.
- 4.8 Material Contracts. (a) Other than the Contracts, including amendments thereto, required to be filed as an exhibit to any report of the Company filed pursuant to the Exchange Act of the type described in Item 601(b)(10) of Regulation S-K promulgated by the SEC, each of which was filed in an unredacted form, the Company Disclosure Schedule sets forth a true and correct list of:
 - (i) each Contract to which the Company or any of its Subsidiaries is a party to or bound that (A) expressly imposes any material restriction on the right or ability of the Company or any of its Subsidiaries to compete with any other Person, (B) contains any right of first refusal, right of first offer or similar term that materially restricts the right or ability of the Company or any of its Subsidiaries to acquire or dispose of the securities of another Person or (C) expressly imposes any material restriction on the right or ability of the Company or any of its Subsidiaries to engage or compete in any line of business

or in any geographic area or that contains exclusivity or non-solicitation provisions (excluding customary employee non-solicitation provisions with customers and partners); or

(ii) each Contract to which the Company or any of its Subsidiaries is a party to or bound that was entered into not in the ordinary course of business and would purport to bind, or purport to be applicable to the conduct of, Parent or its Subsidiaries (other than the Company or its Subsidiaries) in any materially adverse respect (whether before or after the Effective Time).

Contracts, including amendments thereto, required to be filed as an exhibit to any report of the Company filed pursuant to the Exchange Act of the type described in Item 601(b)(10) of Regulation S-K promulgated by the SEC, together with any Contracts of the type described in clauses (i) and (ii) above, are referred to herein as "Company Material Contracts".

- (b) A true and correct copy of each Company Material Contract has previously been made available to Parent and each such Contract is a valid and binding agreement of the Company or its Subsidiary party thereto and, to the Knowledge of the Company, any counterparty thereto, and is in full force and effect, and none of the Company or its Subsidiaries nor, to the Knowledge of the Company, any other party thereto is in default or breach in any respect under the terms of any such Company Material Contract, except for such default or breach as would not reasonably be expected to have a Material Adverse Effect on the Company.
- 4.9 Intellectual Property. (a) To the Knowledge of the Company, the Company and its Subsidiaries exclusively own or possess all right, title and interest in and to, or have the rights to use pursuant to a valid, binding and enforceable license agreement, all Intellectual Property necessary to conduct the business of the Company and its Subsidiaries. The Company Intellectual Property is, to the Knowledge of the Company, valid, subsisting and enforceable and none of such Company Intellectual Property has been misused, withdrawn, canceled or abandoned except as would not adversely affect the operations of the Company as currently conducted. All application and maintenance fees for such Company Intellectual Property for which the Company has applied for or received registration from any Governmental Entity have been paid in full and are current.
- (b) To the Knowledge of the Company, the operation of the business of the Company and its Subsidiaries as currently conducted and as currently proposed to be conducted does not infringe, misappropriate, dilute or otherwise violate or conflict with the Intellectual Property of any other Person ("Third Party Intellectual Property"). Section 4.9(b) of the Company Disclosure Schedule sets forth a list of all suits, actions, proceedings or litigation alleging any of the foregoing that are pending or that have been threatened in writing within two years prior to the date hereof. Neither the Company nor any of its Subsidiaries has received any written notice of any claims or assertions, contesting the ownership, use, validity or enforceability of any Company Intellectual Property. To the Knowledge of the Company, no Person has been engaged, is engaging or is proposed to engage in any activity or use of any Intellectual Property that infringes, misappropriates, dilutes or otherwise violates or conflicts with the Company Intellectual Property.
- (c) The Company and its Subsidiaries have implemented reasonable measures to maintain and protect the secrecy, confidentiality and value of any trade secrets and other confidential information related to the business of the Company and its Subsidiaries. Each current and former employee and independent contractor of, and consultant to, the Company or any of its Subsidiaries has entered into a valid and enforceable written agreement or the Company and its Subsidiaries otherwise has rights enforceable under applicable Law pursuant to which such employee, independent contractor or consultant agrees or is required to maintain the confidentiality of the confidential information of the Company or its Subsidiary and assigns to the Company or its applicable Subsidiary all rights, title and interest in Intellectual Property authored, developed or otherwise created by such employee, independent contractor or consultant in the course of their employment or other relationship with the Company or the applicable Subsidiary of the Company. To the Knowledge of the Company, no employee and no independent contractor or consultant or other third party to any such agreement is in breach thereof.
- (d) The Company and its Subsidiaries have implemented commercially reasonable measures to protect and limit access to the source code for the Company Software Products. Except as set forth on Section 4.9(d)

of the Company Disclosure Schedule or otherwise in the ordinary course of their respective businesses, neither the Company nor any of its Subsidiaries has disclosed, delivered, licensed or otherwise made available, and does not have a duty or obligation (whether present, contingent or otherwise) to disclose, deliver, license or otherwise make available, any source code for any Company Software Products to any Person who was not, as of the date of disclosure or delivery, an employee or contractor of the Company or one of its Subsidiaries.

- (e) Neither the Company nor any of its Subsidiaries has granted, nor agreed or committed to grant, nor given an option to obtain, ownership of or any exclusive license with respect to any Intellectual Property, including any Company Software Products, to any other Person. Immediately following the Effective Time, the Surviving Corporation shall continue to hold the same ownership rights or valid licenses (as applicable) to all of the Company Intellectual Property, in each case, free from less, encumbrances, security interests, orders and arbitration awards, and on the same terms and conditions as in effect with respect to the Company prior to the Effective Time. Neither this Agreement nor the consummation of the Transactions shall result in: (i) Parent's or the Surviving Corporation's granting to any third party any right to or with respect to any Intellectual Property owned by, or licensed to, either of them; (ii) either Parent's or the Surviving Corporation's being bound by, or subject to, any non-compete or other restriction on the operation or scope of their respective businesses; or (iii) either Parent's or the Surviving Corporation's being contractually obligated to pay any royalties or other amounts to any third party in excess of those payable by Parent or the Surviving Corporation, respectively, prior to the Effective Time.
- (f) The Company and its Subsidiaries have implemented commercially reasonable measures to the extent within their control to protect the internal and external security and integrity of all computer and telecom servers, systems, sites, circuits, networks, interfaces, platforms and other computer and telecom assets and equipment used by the Company or its Subsidiaries (the "Company Systems"), and the data stored or contained therein or transmitted thereby including procedures preventing unauthorized access and the introduction of viruses, worms, Trojan horses, "back doors" and other contaminants, bugs, errors or problems that disrupt their operation or have an adverse impact on the operation of other software programs or operating systems, and the taking and storing on-site and off-site of back-up copies of critical data. There have been (i) to the Company's Knowledge, no material unauthorized intrusions or breaches of the security of the Company Systems and (ii) no material failures or interruptions in the Company Systems for the two years prior to the date hereof. All Company Systems are sufficient for the conduct of the business of the Company and its Subsidiaries as currently conducted.
- (g) Except as would not reasonably be expected to have a Material Adverse Effect on the Company, the Company and its Subsidiaries comply with and have at all times (i) complied with and (ii) conducted their business in accordance with all applicable data protection or privacy Laws governing the collection, use, storage, transfer and dissemination of personal information and any privacy policies, programs or other notices that concern the collection or use of personal information by the Company or its Subsidiaries. There have not been any material complaints or notices to, or audits, proceedings or investigations conducted or claims asserted against, the Company and its Subsidiaries by any Person regarding the collection, use, storage, transfer or dissemination of personal information by any Person in connection with the business of the Company or its Subsidiaries by the Company or any of its Subsidiaries with any applicable privacy Laws or privacy policies, programs or other notices. The execution, delivery and performance of this Agreement and the consummation of the Transactions, and any resulting disclosure to and use by Parent and Merger Sub and their Affiliates of, data, personally identifiable information and other information maintained by the Company shall comply in all material respects with the Company's privacy policies and terms of use, any applicable Contracts to which it is party or by which it is bound, and with all applicable Laws relating to privacy and data security (including any such laws in the jurisdictions where the applicable information is collected).
- (h) Except as set forth on Section 4.9(h) of the Company Disclosure Schedule, no government funding, facilities at a university, college, other educational institution or research center or funding from third parties was used in the development of any Company Intellectual Property. To the Knowledge of the Company, no current or former employee, consultant or independent contractor of the Company or any of its Subsidiaries, who was involved in, or who contributed to, the creation or development of any Company Intellectual

Property, has performed services for the government, any university, college or other educational institution or research center during a period of time during which such employee, consultant or independent contractor was also performing services for the Company or any of its Subsidiaries.

- (i) Except as would not reasonably be expected to have a Material Adverse Effect on the Company, to the Knowledge of the Company, (i) there are no defects in any of the Company Software Products that would prevent the same from performing substantially in accordance with its user specifications and (ii) all Company Software Products are free of all undocumented viruses, worms, Trojan horses, "back doors" and other contaminants and do not contain any bugs, errors or problems that disrupt their operation or have an adverse impact on the operation of other software programs or operating systems.
- (j) To the Knowledge of the Company, in no case does the Company's or any of its Subsidiaries' use, incorporation or distribution of Open Source Software give rise to any obligation to disclose or distribute any source code, to license any Company Software Products or Intellectual Property without charge.
- (k) During the preceding two years, neither the Company nor any of its Subsidiaries has received any warranty or indemnity claims related to the Company Software Products that are (i) claims under any "epidemic failure" or similar clause or (ii) other material claims outside the ordinary course of business. During the preceding two years, neither the Company nor any of its Subsidiaries has resolved any warranty or indemnity claims related to the Company Software Products for amounts in excess of \$500,000 over the Company's accounting reserves under GAAP, with respect to any specific customer.
- (I) Except as would not reasonably be expected to have a Material Adverse Effect on the Company, the Company and its Subsidiaries are not, nor shall they be as a result of the execution and delivery of this Agreement or the performance of the obligations of the Company under this Agreement, in breach of any license, sublicense or other agreement relating to Third Party Intellectual Property, and the execution and delivery of this Agreement or the performance of the obligations under this Agreement by the Company shall not result in the loss or impairment of, or give rise to any right of any third party to terminate, any of the Company's or its Subsidiaries' rights to any Third Party Intellectual Property, nor require the consent of any Governmental Entity or third party in respect of any Third Party Intellectual Property.
- 4.10 Certain Business Practices. Neither the Company nor any of its Subsidiaries (nor any of their respective Representatives acting on its behalf) (a) has made or agreed to make any contribution, payment, gift or entertainment to, or accepted or received any contributions, payments, gifts or entertainment from, any government official, employee, political party or agent or any candidate for any federal, state, local or non-United States public office, where either the contribution, payment or gift or the purpose thereof was illegal under the Laws of any federal, state, local or non-United States jurisdiction or (b) has engaged in or otherwise participated in, assisted or facilitated any transaction that is prohibited by any applicable embargo or related trade restriction imposed by the United States Office of Foreign Assets Control or any other United States federal Governmental Entity.
- 4.11 Takeover Laws. The Company is not a party to any rights plan or "poison pill." Assuming the accuracy of the representations of Parent in Section 5.11, (i) no "fair price," "moratorium," "control share acquisition" or similar anti-takeover statute is applicable to this Agreement or the Transactions and (ii) the approvals of the Board of Directors of the Company referred to in Section 4.3(b) constitute approvals of the Transactions under the provisions of Section 203 of the DGCL such that Section 203 does not apply to this Agreement and the Transactions and such that no stockholder of the Company immediately prior to the Effective Time or the purchase of Company Shares pursuant to the Offer as a result of the Transactions.
- 4.12 Taxes. (a) Each of the Company and each of its Subsidiaries has timely filed, or has caused to be timely filed on its behalf, all Tax Returns required to be filed by it, and all such Tax Returns are true, complete and accurate, except to the extent any failure to file or any inaccuracies in any filed Tax Returns have not had and would not reasonably be expected to have a Material Adverse Effect on the Company. All Taxes shown to be due on such Tax Returns, or otherwise owed, have been timely paid, except to the extent

that any failure to pay has not had and would not reasonably be expected to have a Material Adverse Effect on the Company. Each of the Company and each of its Subsidiaries has complied with all applicable Laws relating to Taxes including Laws relating to (i) the withholding and payment over to the appropriate Governmental Entity or other Tax authority of all Taxes required to be withheld by the Company or any of its Subsidiaries, (ii) information reporting with respect to, any payment made or received by the Company or any of its Subsidiaries, and (iii) the keeping of books and records, except to the extent any failure to so comply has not had and would not reasonably be expected to have a Material Adverse Effect on the Company.

- (excluding any reserve for all Taxes payable by the Company and its Subsidiaries (excluding any reserve for deferred Taxes to reflect timing differences between book and Tax items) for all Taxable periods and portions thereof through the date of such financial statements. No deficiency with respect to any Taxes has been proposed, asserted or assessed against the Company or any of its Subsidiaries, and no requests for waivers of the time to assess any such Taxes are pending, except to the extent any such deficiency or request for waiver has not had and would not reasonably be expected to have a Material Adverse Effect on the Company. There is no audit, proceeding or investigation now pending against or with respect to the Company or any of its Subsidiaries in respect of any Tax or Tax asset and neither the Company nor any of its Subsidiaries has received any written notice of any proposed audit, proceeding or investigation with regard to any such Tax or Tax asset, except to the extent that any such pending or proposed audit, proceeding or investigation has not had and would not reasonably be expected to have a Material Adverse Effect on the Company.
- (c) There are no material Liens for Taxes (other than for current Taxes not yet due and payable) on the assets of the Company or any of its Subsidiaries. Neither the Company nor any of its Subsidiaries is bound by any agreement with respect to Taxes.
- (d) Neither the Company nor any of its Subsidiaries has entered into or has been a "material advisor" with respect to any transactions that are or would be part of any "reportable transaction" or that could give rise to any list maintenance obligation under Sections 6011, 6111, or 6112 of the Code (or any similar provision under any state or local Law) or the regulations thereunder.
- 4.13 Benefit Plans. From the date of the most recent audited financial statements included in the Company SEC Documents filed with the SEC prior to the date of this Agreement and other than as set forth on the Company Disclosure Schedule, there has not been any adoption or amendment in any material respect by the Company or any of its Subsidiaries of any collective bargaining agreement or any bonus, pension, profit sharing, deferred compensation, incentive compensation, stock ownership, stock purchase, stock option, phantom stock, retirement, vacation, severance, change in control, retention, disability, death benefit, hospitalization, medical or other plan, arrangement or understanding (whether or not legally binding) providing benefits to any current or former employee, officer, director or independent contractor of the Company or any of its Subsidiaries (collectively, with the Company Pension Plans, any Company "employee welfare benefit plans" (as defined in Section 3(1) of ERISA), the Company International Employee Plans and the Company Employment Agreements, the "Company Employee Plans"), excluding standard employment agreements or offer letters entered into in the ordinary course of business consistent with past practice with employees outside the United States in accordance with local Law and offer letters, severance or employment agreements that have been entered into in the ordinary course of business or that provide for severance or termination agreements or arrangements (other than standard employment agreement, other than as set forth on the Company Disclosure Schedule, there are no employment, consulting, severance or termination agreements or arrangements (other than standard employment agreements or offer letters entered into in the ordinary course of business or that provide for severance or change in control benefits to employeey executive officer or director of director of the Company or any of its Subsidiaries have any general severance plan or policy. For purposes of this Agreement, "Company

Company Employee Plan and each government-mandated plan or program that has been adopted or maintained by Company or any Company ERISA Affiliate, whether informally or formally, or with respect to which Company or any Company ERISA Affiliate shall or may have any liability, for the benefit of Company Employees who perform services outside the United States.

4.14 ERISA Compliance; Excess Parachute Payments; Other Benefits Matters. (a) The Company Disclosure Schedule contains a true and complete list of all "employee pension benefit plans" (as defined in Section 3(2) of ERISA) ("Company Pension Plans"), "employee welfare benefit plans" (as defined in Section 3(1) of ERISA) and all other material Company Employee Plans (other than standard employment agreements or offer letters entered into in the ordinary course of business consistent with past practice with employees outside the United States in accordance with local Law, and offer letters, severance or employment agreements that have been entered into in the ordinary course of business or that provide for severance or change in control benefits with a value of less than \$100,000) maintained, or contributed to, by the Company or any of its Subsidiaries or any entity that would be treated as a "single employer" with the Company or any of its Subsidiaries within the meaning of Section 414(b), (c), (n) or (o) of the Code ("Company ERISA Affiliate") for the benefit of any current or former employees, consultants, officers or directors of the Company or any of its Subsidiaries or with respect to which the Company or any of its Subsidiaries or with respect to which the Company or any of its Subsidiaries or with an instances of noncompliance that, individually and in the aggregate, have not had and would not reasonably be expected to have a Material Adverse Effect on the Company. No proceeding has been threatened, asserted, instituted or, to the Knowledge of the Company, is anticipated against any of the Company. Employee Plans, any trustee or fiduciaries thereof, or any of the assets of any trust of any of the Company.

(b) All Company Pension Plans (other than a Company International Employee Plan) have been the subject of determination letters from the IRS to the effect that such Company Pension Plans are qualified and exempt from federal income taxes under Sections 401(a) and 501(a), respectively, of the Code, and no such determination letter has been revoked nor, to the Knowledge of the Company, has revocation been threatened, nor has any such Company Pension Plan been amended since the date of its most recent determination letter or application therefor in any respect that would adversely affect its qualification or materially increase its costs.

(c) No Company Pension Plan (other than a Company International Employee Plan), had, as of the respective last annual valuation date for each such Company Pension Plan, an "unfunded benefit liability" (as such term is defined in Section 4001(a)(18) of ERISA), based on actuarial assumptions that have been furnished to the Company. None of the Company Pension Plans (other than a Company International Employee Plan) has an "accumulated funding deficiency" (as such term is defined in Section 302 of ERISA or Section 412 of the Code), whether or not waived. None of the Company, any of its Subsidiaries, any officer of the Company Pension Plans, any trusts created thereunder or any trustee or administrator thereof, has engaged in a "prohibited transaction" (as such term is defined in Section 406 of ERISA or Section 4975 of the Code) or any other breach of fiduciary responsibility that could subject the Company, any of its Subsidiaries or any officer or fiduciary of the Company or any of its Subsidiaries to the tax or penalty on prohibited transactions imposed by such Section 4975 or to any liability under Section 502(i) or 502(1) of ERISA. None of such Company Employee Plans (other than a Company International Employee Plan) and related trusts has been terminated, nor has there been any "reportable event" (as that term is defined in Section 4043 of ERISA) with respect to any Company Employee Plan during the last five years. Neither the Company nor any Company ERISA Affiliate has incurred a "complete withdrawal" or a "partial withdrawal" (as such terms are defined in Section 4001(a)(3) of ERISA) (or ERISA) since the effective date of such Sections 4203 and 4205 with respect to any Company Pension Plan that is a "multiemployer plan" within the meaning of Section 4001(a)(3) of ERISA (other than a Company International Employee Plan). All premiums to the Pension Benefit Guaranty Corporation have been timely paid in full for all Company Pension Plans subject to Title IV of ERISA. The Pension Benefit Guaranty Corporation has

terminate any Company Pension Plan subject to Title IV of ERISA and, to the Knowledge of the Company, no condition exists that presents a risk that such proceedings shall be instituted or which would constitute grounds under Section 4042 of ERISA for the termination of, or the appointment of a trustee to administer, such plan.

- (d) With respect to any Company Employee Plan that is an employee welfare benefit plan, (i) no such Company Employee Plan (other than a Company International Employee Plan) is unfunded or funded through a "welfare benefits fund" (as such term is defined in Section 419(e) of the Code); (ii) each such Company Employee Plan (other than a Company International Employee Plan) that is a "group health plan" (as such term is defined in Section 5000(b)(1) of the Code), complies in all material respects with the applicable requirements of Section 4980B(f) of the Code; (iii) each such Company Employee Plan (including any such plan covering retirees or other former employees) may be amended or terminated without material liability to the Company or any of its Subsidiaries on or at any time after the Effective Time; and (iv) no such Company Employee Plan (other than a Company International Employee Plan) provides post-retirement health and welfare benefits to any current or former employee of the Company or any of its Subsidiaries, except as required under Section 4980B of the Code, Part 6 of Title I of ERISA or any other applicable Law.
- (e) Except as set forth in the Company Disclosure Schedule, (i) the consummation of any of the Transactions alone, or in combination with any other event, shall not give rise to any liability under any Company Employee Plan, including liability for severance pay, unemployment compensation, termination pay or withdrawal liability, or accelerate the time of payment or vesting or increase the amount of compensation or benefits due to any employee, officer, director, independent contractor, stockholder or other service provider of the Company or any of its Subsidiaries (whether current, former or retired) or their beneficiaries and (ii) any amount that could be received (whether in cash or property or the vesting of property) as a result of the consummation of any of the Transactions by any employee, officer, director or independent contractor of the Company or any of its Affiliates who is a "disqualified individual" (as such term is defined in proposed Treasury Regulation Section 1.280G-1) under any employment, severance or termination agreement, other compensation arrangement or Company Employee Plan currently in effect would not be characterized as an "excess parachute payment" (as defined in Section 280G(b)(1) of the Code).
- (f) Each Company International Employee Plan has been established, maintained and administered in material compliance with its terms and conditions and with the requirements prescribed by any and all statutory or regulatory laws that are applicable to such Company International Employee Plan. Furthermore, no Company International Employee Plan has unfunded liabilities, that as of the Effective Time shall not be offset by insurance or fully accrued. Except as required by law, no condition exists that would prevent the Company from terminating or amending any Company International Employee Plan at any time for any reason without liability to the Company or its Company ERISA Affiliates (other than ordinary administration expenses or routine claims for benefits).
- (g) The Company has provided or made available to Parent correct and complete copies of: (i) all documents embodying each Company Employee Plan, other than legally-mandated plans, programs and arrangements and each Company Employment Agreement including all amendments thereto and all related trust documents, administrative service agreements, group annuity contracts, group insurance contracts, and policies pertaining to fiduciary liability insurance covering the fiduciaries for each Plan; (ii) the most recent annual actuarial valuations, if any, prepared for each Company Employee Plan; (iii) the three most recent annual reports (Form Series 5500 and all schedules and financial statements attached thereto), if any, required under ERISA or the Code in connection with each Company Employee Plan; (iv) if Company Employee Plan is funded, the most recent annual and periodic accounting of Company Employee Plan; and (vi) all current IRS determination, opinion, notification and advisory letters, and all applications and correspondence to or from the IRS or the DOL with respect to each Company Employee Plan; and (vi) all current IRS determination, opinion, notification and advisory letters, and all applications and correspondence to or from the IRS or the DOL with respect to each Company Employee Plan; and (vi) all current IRS determination, opinion, notification and advisory letters, and all applications and correspondence to or from the IRS or the DOL with respect to any such outstanding application.
- 4.15 Labor and Employment Matters. Neither the Company nor any of its Subsidiaries is a party to or bound by any collective bargaining agreement and there are no labor unions, works councils or other organizations representing, purporting to represent or attempting to represent any employee of the Company or

any of its Subsidiaries. Except as set forth in the Company Disclosure Schedule, (i) no strike, slowdown, picketing, work stoppage, concerted refusal to work overtime or other similar labor activity has occurred or been threatened within the past two years or, to the Knowledge of the Company, is anticipated with respect to any employee of the Company or any of its Subsidiaries, (ii) there are no material labor disputes currently subject to any grievance procedure, arbitration or litigation and there is no material representation pentition pending, threatened or, to the Knowledge of the Company, anticipated with respect to any employee of the Company or any of its Subsidiaries, (iii) to the Knowledge of the Company, neither the Company or any of its Subsidiaries has engaged in any unfair labor practices within the meaning of the National Labor Relations Act, and (iv) the Company and each of its Subsidiaries are in compliance in all material respects with all applicable Laws relating to employment and employment practices, workers' compensation, terms and conditions of employment, worker classification, worker safety, wages and hours, civil rights, discrimination, immigration, collective bargaining, and the Worker Adjustment and Retraining Notification Act, 29 U.S.C. §2109 et seq. or the regulations promulgated thereunder.

- 4.16 Litigation. There is no suit, action or proceeding pending or, to the Knowledge of the Company, threatened against the Company or any of its Subsidiaries that has had or would reasonably be expected to have a Material Adverse Effect on the Company, nor is there any material judgment, order or decree outstanding against the Company or any of its Subsidiaries that has had or would reasonably be expected to have a Material Adverse Effect on the Company.
- 4.17 Compliance with Applicable Laws. To the Knowledge of the Company, the Company and its Subsidiaries are, and during the two years prior to the date hereof have been, in compliance with all applicable Laws, except for instances of noncompliance that have not had and would not reasonably be expected to have a Material Adverse Effect on the Company.
- 4.18 *Brokers*. No broker, investment banker, financial advisor or other Person, other than Raymond James & Associates, Inc., the fees and expenses of which shall be paid by the Company, is entitled to any broker's, financial advisor's or other similar fee or commission in connection with the Transactions based upon arrangements made by or on behalf of the Company.
- 4.19 Opinion of Financial Advisor. The Board of Directors of the Company has received the opinion of Raymond James & Associates, Inc. substantially to the effect that, as of the date of such opinion and subject to the limitations and assumptions set forth therein, (i) the Offer Price to be paid to the holders of Company Shares (except for Parent and its Affiliates) in the Offer and (ii) the Merger Consideration to be paid to the holders of Company Shares (except for Parent and its Affiliates) in the Merger are fair, from a financial point of view, to such holders. The Company shall, promptly following receipt of said opinion in written form, furnish an accurate and complete copy of said opinion to Parent for informational purposes only.
- 4.20 *Vote Required.* To the extent required by applicable Law, the only vote of the holders of any class or series of capital stock of the Company necessary or required in order to adopt this Agreement or approve the Transactions is the Company Stockholder Approval.
- 4.21 Related Party Transactions. Except as would not reasonably be expected to be, individually or in the aggregate, material to the Company, no Related Person of the Company (i) has any right or other interest in any property used in, or pertaining to, the Company's business; (ii) owns (of record or as a beneficial owner) an equity interest or any other financial or profit interest in a Person that has or has had business dealings or a material financial interest in any transaction with the Company; (iii) is a director, officer, employee or partner of, or consultant to, or lender to or borrower from, any Person which is a competitor, supplier, customer, landlord, tenant, creditor or debtor of the Company or any of its Subsidiaries; (iv) has any interest, directly or indirectly, in any Contract to which the Company is a party or subject or by which it or any of its properties is bound or affected, except for expenses incurred in the ordinary course of business consistent with past practice, and, with regard to employees and officers, other than current compensation and benefits incurred in the ordinary course of business consistent with past practice; (v) owes any amount to the Company or any of its Subsidiaries nor does the Company or any of its Subsidiaries owe any amount to, or has the Company or any of its Subsidiaries or faction against the

Company or any of its Subsidiaries. The Company is not a guarantor or indemnitor of any material indebtedness of any Related Person of the Company.

- 4.22 Insurance. Except as has not had and would not reasonably be expected to have a Material Adverse Effect on the Company, each insurance policy of the Company or any of its Subsidiaries is in full force and effect and was in full force and effect during the periods of time such insurance policy is purported to be in effect, and neither the Company nor any of the Company's Subsidiaries is (with or without notice or lapse of time, or both) in breach or default (including any such breach or default with respect to the payment of premiums or the giving of notice) under any such policy.
 - 4.23 Environmental Matters. Except for matters that have not had and would not reasonably be expected to have a Material Adverse Effect on the Company:
 - (a) the Company and its Subsidiaries are now, and have been during the five (5) years prior to the date hereof, in compliance with all Environmental Laws, which compliance includes obtaining and complying with any permits required by Environmental Law for the operations of the Company and its Subsidiaries, and neither the Company nor any of its Subsidiaries has received any written communication from a Person that alleges that the Company or any of its Subsidiaries is in violation of, or has liability or obligations under, any Environmental Law or any permit issued pursuant to Environmental Law;
 - (b) there are no Environmental Claims pending or, to the Knowledge of the Company, threatened against the Company or any of its Subsidiaries;
 - (c) there have been no Releases of any Hazardous Material that would reasonably be expected to form the basis of any Environmental Claim against the Company or any of its Subsidiaries; and
 - (d) neither the Company nor any of its Subsidiaries has retained or assumed, either contractually or, to the Knowledge of the Company, by operation of Law, any liabilities or obligations of another Person that would reasonably be expected to form the basis of any Environmental Claim against the Company or any of its Subsidiaries.
- 4.24 Proxy Statement/Prospectus; Schedule 14D-9; Registration Statement. (a) None of the Proxy Statement/Prospectus or the Schedule 14D-9 will, at the time such documents are filed with the SEC, at the time they are mailed to the holders of Company Shares, and at the time any amendment or supplement thereto is filed with the SEC, contain any untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary in order to make the statements made therein, in light of the circumstances under which they are made, not misleading. Notwithstanding the foregoing, no representation is made by the Company with respect to information supplied by or on behalf of Parent or Merger Sub or any Affiliate of Parent or Merger Sub. The Proxy Statement/Prospectus and the Schedule 14D-9 will, at the time such documents are filed with the SEC, at the time they are mailed to the holders of Company Shares, and at the time any amendment or supplement thereto is filed with the SEC, comply as to form in all material respects with the applicable provisions of the Exchange Act and the Securities Act and the rules and regulations promulgated thereunder.
- (b) None of the information supplied by or on behalf of the Company for inclusion in the Registration Statement will, at the time such document is filed with the SEC, at the time any amendment or supplement thereto is filed with the SEC and at the time the Registration Statement is mailed to the holders of Company Shares and at the time of any Company Stockholder Meeting, the Acceptance Time or the Effective Time, contains any untrue statement of a material fact or omits to state any material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they were made, not misleading.
- 4.25 No Additional Representations and Warranties. The Company acknowledges that neither Parent nor Merger Sub makes any representation or warranty as to any matter whatsoever except as expressly set forth in this Agreement or in any certificate delivered by Parent or Merger Sub to the Company in accordance with the terms hereof, and specifically (but without limiting the generality of the foregoing) that neither Parent nor Merger Sub makes any representation or warranty with respect to (a) any projections, estimates or budgets

delivered or made available to the Company (or any of their respective Affiliates, officers, directors, employees or Representatives) of future revenues, results of operations (or any component thereof), cash flows or financial condition (or any component thereof) of Parent or its Subsidiaries or (b) the future business and operations of Parent or its Subsidiaries.

ARTICLE V.

REPRESENTATIONS AND WARRANTIES OF PARENT AND MERGER SUB

Except as disclosed in (a) the reports, schedules, forms, statements and other documents filed or furnished by Parent with the SEC since December 31, 2009 (the "Parent SEC Documents") filed on or prior to the date five days prior to the date hereof (excluding any risk factor disclosure and disclosure of risks included in any "forward-looking statements" disclaimer or similar statements included in such Parent SEC Documents that are predictive, forward-looking or primarily cautionary in nature); provided that this clause (a) shall not qualify Sections 5.1, 5.3, 5.4(a), 5.7, 5.18, 5.19 or 5.20, or (b) the Parent Disclosure Schedule, Parent and Merger Sub jointly and severally represent and warrant to the Company as follows:

- 5.1 Organization, Standing and Power. Each of Parent and Merger Sub and each Significant Parent Subsidiary is duly organized, validly existing and, to the extent applicable, in good standing under the Laws of the jurisdiction in which it is organized and has full corporate power and authority to conduct its businesses as presently conducted. Parent and Merger Sub and each Significant Parent Subsidiary is duly qualified to do business in each jurisdiction where the nature of its business or its ownership or leasing of its properties make such qualification necessary and the failure to so qualify has had or would reasonably be expected to have a Material Adverse Effect on Parent. Parent has made available to the Company true and complete copies of the certificate of incorporation of Parent, as amended through the date of this Agreement (together with the Parent Certificate, the "Parent Charter Documents") and the certificate of formation and limited liability company agreement of Merger Sub, as amended through the date of this Agreement. Merger Sub is a wholly owned indirect Subsidiary of Parent.
- 5.2 Parent Subsidiaries. (a) All the outstanding shares of capital stock of Merger Sub and each other Subsidiary of Parent have been validly issued and are fully paid and nonassessable and, except as set forth in the Section 5.2 of the Parent Disclosure Schedule, are owned by Parent, by another Subsidiary of Parent or by Parent and another Subsidiary of Parent, free and clear of all Liens other than Permitted Liens.
- (b) Except for its interests in Parent's wholly owned Subsidiaries and except for the ownership interests set forth in Section 5.2 of the Parent Disclosure Schedule, Parent does not own, directly or indirectly, any capital stock, membership interest, partnership interest, joint venture interest or other equity interest in any other Person.
- (c) Since the date of its incorporation, Merger Sub has not carried on any business or conducted any operations other than the Transactions, the execution of this Agreement, the performance of its obligations hereunder and thereunder and matters ancillary thereto.
- 5.3 Authority; Execution and Delivery; Enforceability. (a) Each of Parent and Merger Sub has all requisite corporate power and authority to execute and deliver this Agreement. The execution and delivery by Parent and Merger Sub of this Agreement and the consummation by Parent and Merger Sub of the Transactions: (i) have been duly authorized by all necessary corporate action on the part of Parent and Merger Sub and (ii) do not require the approval of Parent's stockholders. Each of Parent and Merger Sub has duly executed and delivered this Agreement, and this Agreement constitutes its legal, valid and binding obligation, enforceable against it in accordance with its terms.
- (b) The Board of Directors of Parent, at a meeting duly called and held in compliance with the DGCL, has: (i) determined that this Agreement and the Transactions are fair to, and in the best interests of, Parent and its stockholders and that, considering the financial position of the merging companies, no reasonable concern exists that the Surviving Corporation shall be unable to fulfill the obligations of Merger Sub to its creditors

existing as of immediately prior to the Effective Time; and (ii) adopted and approved this Agreement and approved the Transactions.

- (c) The Board of Directors of Merger Sub has by unanimous written consent: (i) determined that this Agreement and the Transactions are fair to, and in the best interests of, Merger Sub and its stockholder and that, considering the financial position of the merging companies, no reasonable concern exists that the Surviving Corporation will be unable to fulfill the obligations of Merger Sub to its creditors existing as of immediately prior to the Effective Time; and (ii) adopted and approved this Agreement and approved the Transactions. Parent, as sole stockholder of Merger Sub, has approved this Agreement
- 5.4 Capital Structure. (a) The authorized share capital of Parent comsists of 70,000,000 shares of Parent Common Stock and 5,000,000 shares of preferred stock ("Parent Preferred Stock"). At the close of business on the Measurement Date, (i) 33,503,529 shares of Parent Common Stock were issued and outstanding, (ii) no Parent Preferred Stock was issued and outstanding, (iii) 7,317,987 shares of Parent Common Stock were reserved for future issuance pursuant to options to purchase Parent Common Stock ("Parent Stock Options") and (v) 2,854,070 common stock warrants were issued and outstanding, Except as set forth above, at the close of business on the Measurement Date, no shares of capital stock or other voting securities of Parent were issued, reserved for issuance or outstanding, All outstanding shares of Parent Common Stock are, and all such shares that may be issued prior to the Effective Time will be when issued, duly authorized, validly issued, fully paid and nonassessable and not subject to or issued in violation of any purchase option, call option, right of first refusal, preemptive right, subscription right or any similar right under any provision of the DGCL, the Parent Charter Documents or any Contract to which Parent is a party or otherwise bound. There are no bonds, debentures, notes or other indebtedness of Parent having the right to vote (or convertible into, or exchangeable for, securities having the right to vote) on any matters on which holders of shares of Parent Common Stock may vote ("Voting Parent Debt"). Except as set forth above, as of the date of this Agreement, there are no options, warrants, rights, convertible or exchangeable securities, "phantom" stock rights, stock appreciation rights, stock-based performance units, commitments, Contracts, arrangements or undertakings of any kind to which Parent or any of its Subsidiaries to issue, deliver or sell, or cause to be issued, delivered or sold, additional shares of capital stock or other equity interests in, or any security convertible or ex
- (b) Parent has made available to the Company accurate and complete copies of all equity plans pursuant to which Parent has granted any awards that are currently outstanding and the form of all equity award agreements. All shares of Parent Common Stock subject to issuance with respect to any award have been duly authorized and, upon issuance on the terms and conditions specified in the instrument pursuant to which they are issuable, will be validly issued, fully paid and nonassessable. All outstanding shares of Parent Common Stock, all outstanding Parent Stock Options, and all outstanding shares of capital stock of each Subsidiary of Parent have been issued and granted (i) in compliance with all applicable securities laws and other applicable Laws and (ii) in material compliance with all applicable requirements set forth in the Parent Employee Plans.
- (c) The authorized capital stock of Merger Sub consists of one unit of membership interests, defined as common stock, which has been validly issued, is fully paid and nonassessable and is owned by Parent free and clear of any Lien.
- 5.5 No Conflicts; Consents. (a) The execution and delivery by each of Parent and Merger Sub of this Agreement does not, and the consummation of any of the Transactions will not, conflict with, or result in any violation of or default (with or without notice or lapse of time, or both) under, or give rise to a right of termination, cancelation or acceleration of any obligation or to loss of a material benefit under, or to increased, additional, accelerated or guaranteed rights or entitlements of any Person under, or result in the creation of any

Lien upon any of the properties or assets of Parent or any of its Subsidiaries under, any provision of (i)(A) the Parent Charter Documents or (B) the comparable charter or organizational documents of any Subsidiary of Parent, (ii) any Contract to which Parent or any of its Subsidiaries is a party or by which any of their respective properties or assets is bound or (iii) subject to the fillings and other matters referred to in Section 5.5(b), any material Law applicable to Parent or any of its Subsidiaries or their respective properties or assets other than, in the case of clauses (i)(B), (ii) or (iii) above, any such items that have not had and would not reasonably be expected to have a Material Adverse Effect on Parent.

- (b) The execution and delivery of this Agreement by Parent and Merger Sub does not and the consummation of the Transactions do not, and the performance of this Agreement and the Transactions by Parent and Merger Sub shall not, require any consent, approval, authorization or permit of, or filing with or notification to, any Governmental Entity or any third party, except (i) for applicable requirements, if any, of the Securities Act, the Exchange Act, Blue Sky Laws, the HSR Act, the requirements of any Governmental Entity under applicable competition, antitrust or non-United States investment Laws, the rules and regulations of Nasdaq, the filing of the Certificate of Merger in accordance with Section 264 of the DGCL and Section 18-209 of the DLLCA and such other filings, notices, permits, authorizations, consents or approvals as may be required by reason of the status of Parent, Merger Sub or their Affiliates, and (ii) where the failure to obtain such consents, approvals, authorizations or permits, or to make such filings or notifications, would not reasonably be expected to have a Material Adverse Effect on Parent.
- 5.6 SEC Documents; Undisclosed Liabilities. (a) Parent has filed all reports, schedules, forms, statements and other documents required to be filed by Parent with the SEC since December 31, 2009 pursuant to Sections 13(a) and 15(d) of the Exchange Act.
- (b) As of its respective date, each Parent SEC Document filed with the SEC complied in all material respects with the requirements of the Exchange Act or the Securities Act, as the case may be, and the rules and regulations of the SEC promulgated thereunder applicable to such Parent SEC Document, and did not contain any untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they were made, not misleading. Except to the extent that information contained in any Parent SEC Document filed with the SEC has been revised or superseded by a later filed Parent SEC Document, none of the Parent SEC Documents filed with the SEC contained any untrue statement of a material fact or omits to state any material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they were made, not misleading. The consolidated financial statements of Parent included in the Parent SEC Documents filed with the SEC comply as to form in all material respects with applicable accounting requirements and the published rules and regulations of the SEC with respect thereto, have been prepared in accordance with GAAP applied on a consistent basis during the periods involved (except as may be indicated in the notes thereto) and fairly present in all material respects the consolidated financial position of Parent and its consolidated Subsidiaries as of the dates thereof and the consolidated results of their operations and cash flows for the periods shown (subject, in the case of unaudited statements, to normal year-end audit adjustments).
- (c) Except (i) as reflected or reserved against in the balance sheet (or the notes thereto) as of December 31, 2010 included in the Parent SEC Documents, (ii) as permitted or contemplated by this Agreement, (iii) for liabilities and obligations incurred since December 31, 2010 in the ordinary course of business, and (iv) for liabilities or obligations which have been discharged or paid in full in the ordinary course of business, neither Parent nor any of its Subsidiaries has any liabilities or obligations of any nature, whether or not accrued, contingent or otherwise, that would be required by GAAP to be reflected on a consolidated balance sheet of Parent and its consolidated Subsidiaries (or in the notes thereto), other than those that would not reasonably be expected to have a Material Adverse Effect on Parent.
- (d) Parent maintains a system of "internal control over financial reporting" (as defined in Rules 13a-15(f) and 15d-15(f) of the Exchange Act) sufficient to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with GAAP and includes those policies and procedures that (i) pertain to the maintenance of records that in reasonable

detail accurately and fairly reflect the transactions and dispositions of Parent's and its Subsidiaries' assets; (ii) provide reasonable assurance that transactions are recorded as necessary to permit preparation of financial statements in accordance with GAAP, and that Parent's and its Subsidiaries' receipts and expenditures are being made only in accordance with authorizations of Parent's management and directors; and (iii) provide reasonable assurance regarding prevention or timely detection of unauthorized acquisition, use or disposition of Parent's and its Subsidiaries' assets that could have a material effect on Parent's financial statements.

- (e) The "disclosure controls and procedures" (as defined in Rules 13a-15(e) and 15d-15(e) of the Exchange Act) utilized by Parent are reasonably designed to ensure that all information (both financial and non-financial) required to be disclosed by Parent in the reports that it files or submits under the Exchange Act is recorded, processed, summarized and reported within the time periods specified in the rules and forms of the SEC and that all such information required to be disclosed is accumulated and communicated to the management of Parent, as appropriate, to allow timely decisions regarding required disclosure and to enable the chief executive officer and chief financial officer of Parent to make the certifications required under the Exchange Act with respect to such reports.
- 5.7 Absence of Certain Changes or Events. Since December 31, 2010 until the date of this Agreement, there has not been any event, change, effect or development that has had or would reasonably be expected to have a Material Adverse Effect on Parent.
- 5.8 Material Contracts. (a) Other than the Contracts, including amendments thereto, required to be filed as an exhibit to any report of Parent filed pursuant to the Exchange Act of the type described in Item 601(b)(10) of Regulation S-K promulgated by the SEC, each of which was filed in an unredacted form, except as set forth in Section 5.8(a) of the Parent Disclosure Schedule, there are no:
 - (i) Contracts to which Parent or any of its Subsidiaries is a party to or bound that (A) expressly imposes any material restriction on the right or ability of Parent or any of its Subsidiaries to compete with any other Person, (B) contains any right of first refusal, right of first offer or similar term that materially restricts the right or ability of Parent or any of its Subsidiaries to acquire or dispose of the securities of another Person; or (C) expressly imposes any material restriction on the right or ability of Parent or any of its Subsidiaries to engage or compete in any line of business or in any geographic area or that contains exclusivity or non-solicitation provisions (excluding customary employee non-solicitation provisions with customers and partners); or
 - (ii) Contracts to which Parent or any of its Subsidiaries is a party to or bound that were entered into not in the ordinary course of business and would purport to bind, or purport to be applicable to the conduct of, the Company or its Subsidiaries in any materially adverse respect (whether before or after the Effective Time).

Contracts, including amendments thereto, required to be filed as an exhibit to any report of Parent filed pursuant to the Exchange Act of the type described in Item 601(b)(10) of Regulation S-K promulgated by the SEC, together with any Contracts of the type described in clauses (i) and (ii) above, are referred to herein as "Parent Material Contracts".

- (b) Each Parent Material Contract is a valid and binding agreement of Parent or its Subsidiary party thereto and, to the Knowledge of Parent, any counterparty thereto, and is in full force and effect, and none of Parent or its Subsidiaries nor, to the Knowledge of Parent, any other party thereto is in default or breach in any respect under the terms of any such Parent Material Contract, except for such default or breach as would not reasonably be expected to have a Material Adverse Effect on Parent.
- 5.9 Intellectual Property. (a) To the Knowledge of Parent, Parent and its Subsidiaries exclusively own or possess all right, title and interest in and to, or have the rights to use pursuant to a valid, binding and enforceable license agreement, all Intellectual Property necessary to conduct the business of Parent and its Subsidiaries. The Parent Intellectual Property is, to the Knowledge of Parent, valid, subsisting and enforceable and none of such Parent Intellectual Property has been misused, withdrawn, canceled or abandoned except as would not adversely affect the operations of Parent as currently conducted. All application and maintenance

fees for such Parent Intellectual Property for which Parent has applied for or received registration from any Governmental Entity have been paid in full and are current.

- (b) To the Knowledge of Parent, the operation of the business of Parent and its Subsidiaries as currently conducted and as currently proposed to be conducted does not infringe, misappropriate, dilute or otherwise violate or conflict with the Intellectual Property of any other Person, and there are no suits, actions, proceedings or litigation alleging any of the foregoing that are pending or that have been threatened in writing within two years prior to the date hereof. Neither Parent nor any of its Subsidiaries has received any written notice of any claims or assertions, contesting the ownership, use, validity or enforceability of any Parent Intellectual Property. To the Knowledge of Parent, no Person has been engaged, is engaging or is proposed to engage in any activity or use of any Intellectual Property that infringes, misappropriates, dilutes or otherwise violates or conflicts with the Parent Intellectual Property.
- (c) Parent and its Subsidiaries have implemented reasonable measures to maintain and protect the secrecy, confidentiality and value of any trade secrets and other confidential information related to the business of Parent and its Subsidiaries. Each current and former employee and independent contractor of, and consultant to, Parent or any of its Subsidiaries has entered into a valid and enforceable written agreement or Parent and its Subsidiaries otherwise has rights enforceable under applicable Law pursuant to which such employee, independent contractor or consultant agrees or is required to maintain the confidentiality of the confidential information of Parent or its Subsidiary and assigns to Parent or the applicable Subsidiary of Parent all rights, title and interest in Intellectual Property authored, developed or otherwise created by such employee, independent contractor or consultant in the course of their employment or other relationship with Parent or its applicable Subsidiary. To the Knowledge of Parent, no employee and no independent contractor or consultant or other third party to any such agreement is in breach thereof.
- (d) Parent and its Subsidiaries have implemented commercially reasonable measures to protect and limit access to the source code for the Parent Software Products. Except in the ordinary course of their respective businesses, neither Parent nor any of its Subsidiaries has disclosed, delivered, licensed or otherwise made available, and does not have a duty or obligation (whether present, contingent or otherwise) to disclose, deliver, license or otherwise make available, any source code for any Parent Software Products to any Person who was not, as of the date of disclosure or delivery, an employee or contractor of Parent or one of its Subsidiaries.
- (e) Neither Parent nor any of its Subsidiaries has granted, nor agreed or committed to grant, nor given an option to obtain, ownership of or any exclusive license with respect to any Intellectual Property, including any Parent Software Products, to any other Person. Immediately following the Effective Time, Parent and its Subsidiaries shall continue to hold the same ownership rights or valid licenses (as applicable) to all of the Parent Intellectual Property, in each case, free from liens, encumbrances, security interests, orders and arbitration awards, and on the same terms and conditions as in effect with respect to Parent and its Subsidiaries prior to the Effective Time. Neither this Agreement nor the consummation of any of the Transactions will result in: (i) Parent's or the Surviving Corporation's granting to any third party any right to or with respect to any Intellectual Property owned by, or licensed to, either of them; (ii) either Parent's or the Surviving Corporation's being bound by, or subject to, any non-compete or other restriction on the operation or scope of their respective businesses; or (iii) either Parent's or the Surviving Corporation's being contractually obligated to pay any royalties or other amounts to any third party in excess of those payable by Parent or the Surviving Corporation, respectively, prior to the Effective Time.
- (f) Parent and its Subsidiaries have implemented commercially reasonable measures to the extent within their control to protect the internal and external security and integrity of all computer and telecom servers, systems, sites, circuits, networks, interfaces, platforms and other computer and telecom assets and equipment used by Parent or its Subsidiaries (the "Parent Systems"), and the data stored or contained therein or transmitted thereby including procedures preventing unauthorized access and the introduction of viruses, worms, Trojan horses, "back doors" and other contaminants, bugs, errors or problems that disrupt their operation or have an adverse impact on the operation of other software programs or operating systems, and the taking and storing on-site and off-site of back-up copies of critical data. There have been (i) to Parent's

Knowledge, no material unauthorized intrusions or breaches of the security of the Parent Systems, and (ii) no material failures or interruptions in the Parent Systems for the two years prior to the date hereof. All Parent Systems are sufficient for the conduct of the business of Parent and its Subsidiaries as currently conducted.

- (g) Except as would not reasonably be expected to have a Material Adverse Effect on the Parent, Parent and its Subsidiaries comply with and have at all times (i) complied with and (ii) conducted their business in accordance with all applicable data protection or privacy Laws governing the collection, use, storage, transfer and dissemination of personal information and any privacy policies, programs or other notices that concern the collection or use of personal information by Parent or its Subsidiaries. There have not been any material complaints or notices to, or audits, proceedings or investigations conducted or claims asserted against, Parent and its Subsidiaries by any Person regarding the collection, use, storage, transfer or dissemination of personal information by any Person in connection with the business of Parent or its Subsidiaries or compliance by Parent or any of its Subsidiaries with any applicable privacy Laws or privacy policies, programs or other notices. The execution, delivery and performance of this Agreement and the consummation of the Transactions, and any resulting disclosure to and use by Parent and Merger Sub and their Affiliates of, data, personally identifiable information and other information maintained by Parent shall comply in all material respects with the Parent's privacy policies and terms of use, any applicable Contracts to which it is party or by which it is bound, and with all applicable Laws relating to privacy and data security (including any such laws in the jurisdictions where the applicable information is collected).
- (h) No government funding, facilities at a university, college, other educational institution or research center or funding from third parties was used in the development of any Parent Intellectual Property. To the Knowledge of Parent, no current or former employee, consultant or independent contractor of Parent or any of its Subsidiaries, who was involved in, or who contributed to, the creation or development of any Parent Intellectual Property, has performed services for the government, any university, college or other educational institution or research center during a period of time during which such employee, consultant or independent contractor was also performing services for Parent or any of its Subsidiaries.
- (i) Except as would not reasonably be expected to have a Material Adverse Effect on Parent, to the Knowledge of Parent, (i) there are no defects in any of the Parent Software Products that would prevent the same from performing substantially in accordance with its user specifications and (ii) all Parent Software Products are free of all undocumented viruses, worms, Trojan horses, "back doors" and other contaminants and do not contain any bugs, errors or problems that disrupt their operation or have an adverse impact on the operation of other software programs or operating systems.
- (j) To the Knowledge of Parent, in no case does Parent's or any of its Subsidiaries' use, incorporation or distribution of Open Source Software give rise to any obligation to disclose or distribute any source code, to license any Parent Software Products or Intellectual Property without charge.
- (k) During the preceding two years, neither Parent nor any of its Subsidiaries has received any warranty or indemnity claims related to the Parent Software Products that are (i) claims under any "epidemic failure" or similar clause or (ii) other material claims outside the ordinary course of business. During the preceding two years, neither Parent nor any of its Subsidiaries has resolved any warranty or indemnity claims related to the Parent Software Products for amounts in excess of \$500,000 over Parent's accounting reserves under GAAP, with respect to any specific customer.
- (1) Except as would not reasonably be expected to have a Material Adverse Effect on Parent or its Subsidiaries, Parent and its Subsidiaries are not, nor will they be as a result of the execution and delivery of this Agreement or the performance of the obligations of Parent under this Agreement, in breach of any license, sublicense or other agreement relating to Intellectual Property, and the execution and delivery of this Agreement or the performance of the obligations under this Agreement by Parent will not result in the loss or impairment of, or give rise to any right of any third party to terminate, any of Parent's or its Subsidiaries' rights to any Intellectual Property, nor require the consent of any Governmental Entity or third party in respect of any Intellectual Property.

- 5.10 Certain Business Practices. Neither Parent nor any of its Subsidiaries (nor any of their respective Representatives acting on its behalf) (a) has made or agreed to make any contribution, payment, gift or entertainment to, or accepted or received any contributions, payments, gifts or entertainment from, any government official, employee, political party or agent or any candidate for any federal, state, local or non-United States public office, where either the contribution, payment or gift or the purpose thereof was illegal under the Laws of any federal, state, local or non-United States jurisdiction or (b) has engaged in or otherwise participated in, assisted or facilitated any transaction that is prohibited by any applicable embargo or related trade restriction imposed by the United States Office of Foreign Assets Control or any other United States federal Governmental Entity.
- 5.11 Takeover Laws. As of the date of this Agreement, Parent, together with its "Affiliates" and "Associates," is not, nor at any time during the last three years has it been, an "Interested Stockholder" of the Company, as such terms are defined in Section 203 of the DGCL. No "fair price," "moratorium," "control share acquisition" or similar anti-takeover statute is applicable to the
- 5.12 Taxes. (a) Each of Parent and each of its Subsidiaries has timely filed, or has caused to be timely filed on its behalf, all Tax Returns required to be filed by it, and all such Tax Returns are true, complete and accurate, except to the extent any failure to file or any inaccuracies in any filed Tax Returns have not had and would not reasonably be expected to have a Material Adverse Effect on Parent. All Taxes shown to be due on such Tax Returns, or otherwise owed, have been timely paid, except to the extent that any failure to pay has not had and would not reasonably be expected to have a Material Adverse Effect on Parent. Each of Parent and each of its Subsidiaries has complied with all applicable Laws relating to Taxes including Laws relating to (i) the withholding and payment over to the appropriate Governmental Entity or other Tax authority of all Taxes required to be withheld by Parent or any of its Subsidiaries, (ii) information reporting with respect to, any payment made or received by Parent or any of its Subsidiaries, and (iii) the keeping of books and records, except to the extent any failure to so comply has not had and would not reasonably be expected to have a Material Adverse Effect on Parent.
- (b) The most recent financial statements contained in the Parent SEC Documents filed with the SEC reflect an adequate reserve for all Taxes payable by Parent and its Subsidiaries (excluding any reserve for deferred Taxes to reflect timing differences between book and Tax items) for all Taxable periods and portions thereof through the date of such financial statements. No deficiency with respect to any Taxes has been proposed, asserted or assessed against Parent or any of its Subsidiaries, and no requests for waivers of the time to assess any such Taxes are pending, except to the extent any such deficiency or request for waiver has not had and would not reasonably be expected to have a Material Adverse Effect on Parent. There is no audit, proceeding or investigation now pending against or with respect to Parent or any of its Subsidiaries in respect of any Tax or Tax asset and neither Parent nor any of its Subsidiaries has received any written notice of any proposed audit, proceeding or investigation with regard to any such Tax or Tax asset, except to the extent that any such pending or proposed audit, proceeding or investigation has not had and would not reasonably be expected to have a Material Adverse Effect on Parent.
- (c) There are no material Liens for Taxes (other than for current Taxes not yet due and payable) on the assets of Parent or any of its Subsidiaries. Neither Parent nor any of its Subsidiaries is bound by any agreement with respect to Taxes.
 - 5.13 Reserved
- 5.14 ERISA Compliance; Excess Parachute Payments; Other Benefits Matters. (a) Each Parent Employee Plan has been administered in compliance with its terms and in accordance with all applicable Laws, other than instances of noncompliance that, individually and in the aggregate, have not had and would not reasonably be expected to have a Material Adverse Effect on Parent. No proceeding has been threatened, asserted, instituted or, to the Knowledge of Parent, is anticipated against any of the Parent Employee Plans, any trustee or fiduciaries thereof, or any of the assets of any trust of any of the Parent Employee Plans that would reasonably be expected to have a Material Adverse Effect on Parent.

- (b) All Parent Pension Plans (other than a Parent International Employee Plan) have been the subject of determination letters from the IRS to the effect that such Parent Pension Plans are qualified and exempt from federal income taxes under Sections 401(a) and 501(a), respectively, of the Code, and no such determination letter has been revoked nor, to the Knowledge of Parent, has revocation been threatened, nor has any such Parent Pension Plan been amended since the date of its most recent determination letter or application therefor in any respect that would adversely affect its qualification or materially increase its costs.
- (c) No Parent Pension Plan (other than a Parent International Employee Plan), had, as of the respective last annual valuation date for each such Parent Pension Plan, an "unfunded benefit liability" (as such term is defined in Section 4001(a)(18) of ERISA), based on actuarial assumptions that have been furnished to Parent. None of the Parent Pension Plans (other than a Parent International Employee Plan) has an "accumulated funding deficiency" (as such term is defined in Section 302 of ERISA or Section 412 of the Code), whether or not waived. None of Parent, any of its Subsidiaries, any officer of Parent or any of the Parent Employee Plans which are subject to ERISA, including the Parent Pension Plans, any trusts created thereunder or any trustee or administrator thereof, has engaged in a "prohibited transaction" (as such term is defined in Section 406 of ERISA or Section 4975 of the Code) or any other breach of fiduciary responsibility that could subject Parent, any of its Subsidiaries or any of fits engaged in a "prohibited transaction" (as such term is defined in Section 406 of ERISA) may be prohibited transactions imposed by such Section 4975 or to any liability under Section 502(i) or 502(i) of ERISA. None of such Parent Employee Plans (other than a Parent International Employee Plan) and related trusts has been terminated, nor has there been any "reportable event" (as that term is defined in Section 4043 of ERISA) with respect to any Parent Employee Plan during the last five years. Neither Parent nor any Parent ERISA Affiliate has incurred a "complete withdrawal" or a "partial withdrawal" (as such terms are defined in Sections 4203 and 4205, respectively, of ERISA) since the effective date of such Sections 4203 and 4205 with respect to any Parent Pension Plan that is a "multiemployer plan" within the meaning of Section 4001(a)(3) of ERISA (other than a Parent International Employee Plan). All premiums to the Pension Benefit Guaranty Corporation have been timely paid in full for all Parent Pension Plans sub
- (d) With respect to any Parent Employee Plan that is an employee welfare benefit plan, (i) no such Parent Employee Plan (other than a Parent International Employee Plan) is unfunded or funded through a "welfare benefits fund" (as such term is defined in Section 419(e) of the Code); (ii) each such Parent Employee Plan (other than a Parent International Employee Plan) that is a "group health plan" (as such term is defined in Section 5000(b)(1) of the Code), complies in all material respects with the applicable requirements of Section 4980B(f) of the Code; (iii) each such Parent Employee Plan (including any such plan covering retirees or other former employees) may be amended or terminated without material liability to Parent or any of its Subsidiaries on or at any time after the Effective Time; and (iv) no such Parent Employee Plan (other than a Parent International Employee Plan) provides post-retirement health and welfare benefits to any current or former employee of Parent or any of its Subsidiaries, except as required under Section 4980B of the Code, Part 6 of Title I of ERISA or any other applicable Law.
- (e) Each Parent International Employee Plan has been established, maintained and administered in material compliance with its terms and conditions and with the requirements prescribed by any and all statutory or regulatory laws that are applicable to such Parent International Employee Plan. Furthermore, no Parent International Employee Plan has unfunded liabilities, that as of the Effective Time, will not be offset by insurance or fully accrued. Except as required by law, no condition exists that would prevent Parent from terminating or amending any Parent International Employee Plan at any time for any reason without liability to Parent or its Parent ERISA Affiliates (other than ordinary administration expenses or routine claims for benefits).
- 5.15 Labor and Employment Matters. Neither Parent nor any of its Subsidiaries is a party to or bound by any collective bargaining agreement and there are no labor unions, works councils or other organizations representing, purporting to represent or attempting to represent any employee of Parent or any of its

Subsidiaries. There is no strike, slowdown, picketing, work stoppage, concerted refusal to work overtime or other similar labor activity has occurred or been threatened within the past two years or, to the Knowledge of Parent, is anticipated with respect to any employee of Parent or any of its Subsidiaries. There are no material labor disputes currently subject to any grievance procedure, arbitration or litigation and there is no material representation petition pending, threatened or, to the Knowledge of Parent, anticipated with respect to any employee of Parent or any of its Subsidiaries. To the Knowledge of Parent, neither Parent nor any of its Subsidiaries are in compliance in all material respects with all applicable Laws relating to employment and employment practices, workers' compensation, terms and conditions of employment, worker classification, worker safety, wages and hours, civil rights, discrimination, immigration, collective bargaining, and the Worker Adjustment and Retraining Notification Act, 29 U.S.C. §2109 et seq. or the regulations promulgated thereunder.

- 5.16 Litigation. There is no suit, action or proceeding pending or, to the Knowledge of Parent, threatened against Parent or any of its Subsidiaries that has had or would reasonably be expected to have a Material Adverse Effect on Parent, nor is there any material judgment, order or decree outstanding against Parent or any of its Subsidiaries that has had or would reasonably be expected to have a Material Adverse Effect on Parent.
- 5.17 Compliance with Applicable Laws. To the Knowledge of Parent, Parent and its Subsidiaries are, and during the two years prior to the date hereof have been, in compliance with all applicable Laws, except for instances of noncompliance that, individually and in the aggregate, have not had and would not reasonably be expected to have a Material Adverse Effect on Parent.
- 5.18 *Brokers*. No broker, investment banker, financial advisor or other Person, other than Wells Fargo Securities, LLC, the fees and expenses of which shall be paid by Parent, is entitled to any broker's, finder's, financial advisor's or other similar fee or commission in connection with the Transactions based upon arrangements made by or on behalf of Parent or Merger Sub.
- 5.19 Issuance of Parent Common Stock. When issued in accordance with the terms of this Agreement, the shares of Parent Common Stock to be issued pursuant to the Offer and the Merger to the holders of Company Shares hereunder shall be duly authorized, validly issued, fully paid and nonassessable and not subject to statutory preemptive rights.
- 5.20 Funds. Parent has delivered to the Company a true, complete and correct copy of (i) the draft Credit Agreement to be entered into among Parent, the Lenders referred to therein, Wells Fargo Bank, N.A., as Administrative Agent, Swingline Lender and Issuing Lender, and Wells Fargo Securities, LLC, as sole Lead Arranger and Sole Book Manager (the "Financing Agreement") and (ii) the executed Commitment Letter, dated August 29, 2011, among Parent, Wells Fargo Bank, National Association and Wells Fargo Securities, LLC (the "Financing Commitment") pursuant to which the lenders party thereto have committed, subject to the terms and conditions set forth therein, to lend the amounts set forth therein for the purposes of financing the Transactions (the "Financing"). The Financing Commitment has not been amended or modified prior to the date of this Agreement and as of the date of this Agreement the respective commitments contained in the Financing Commitment have not been withdrawn or rescinded. Except for fee letters (complete copies of which have been provided to the Company, with only fee amounts and market flex provisions and other customary threshold amounts redacted) and the Financing Commitment, as of the date hereof there are no side letters or Contracts to which Parent or Merger Sub is a party related to the Financing that have not been provided to the Company, Parent has fully paid any and all commitment fees or other fees in connection with the Financing Commitment that are payable on or prior to the date hereof, and as of the date hereof the Financing Commitment is in full force and effect and is the legal, valid, binding and enforceable obligations of Parent, and, to the Knowledge of Parent, each of the other parties thereto. There are no conditions precedent or other contingencies related to the funding of the full amount of the Financing, other than as expressly set forth therein. As of the date hereof, subject to the accuracy of the representations and warranties of the Company set forth in Article IV

Parent and Merger Sub shall have at and after Acceptance Time and the Closing funds sufficient to (a) pay all amounts required to be paid by Parent or Merger Sub pursuant to the Transactions and (b) pay any and all fees and expenses required to be paid by Parent, Merger Sub and the Surviving Corporation in connection with any financing.

- 5.21 Related Party Transactions. Except as would not reasonably be expected to be, individually or in the aggregate, material to Parent, no Related Person of Parent (i) has any right or other interest in any property used in, or pertaining to, Parent's business; (ii) owns (of record or as a beneficial owner) an equity interest or any other financial or profit interest in a Person that has or has had business dealings or a material financial interest in any transaction with Parent; (iii) is a director, officer, employee or partner of, or consultant to, or lender to or borrower from, any Person which is a competitor, supplier, customer, landlord, tenant, creditor or debtor of Parent or any of its Subsidiaries; (iv) has any interest, directly or indirectly, in any Contract to which Parent is a party or subject or by which it or any of its properties is bound or affected, except for expenses incurred in the ordinary course of business consistent with past practice, and, with regard to employees and officers, other than current compensation and benefits incurred in the ordinary course of business consistent with past practice; (v) owes any amount to Parent or any of its Subsidiaries or does Parent or any of its Subsidiaries owe any amount to, or has Parent or any of its Subsidiaries or any of its Subsidiaries or any or any of its Subsidiaries or any capacity or indemnitor of any material indebtedness of any Related Person of Parent.
- 5.22 Insurance. Except as has not had and would not reasonably be expected to have a Material Adverse Effect on Parent, each insurance policy of Parent or any of its Subsidiaries is in full force and effect and was in full force and effect during the periods of time such insurance policy is purported to be in effect, and neither Parent nor any of Parent's Subsidiaries is (with or without notice or lapse of time, or both) in breach or default (including any such breach or default with respect to the payment of premiums or the giving of notice) under any such policy.
 - 5.23 Environmental Matters. Except for matters that have not had and would not reasonably be expected to have a Material Adverse Effect on Parent:
 - (a) Parent and its Subsidiaries are now, and have been during the five years prior to the date hereof, in compliance with all Environmental Laws, which compliance includes obtaining and complying with any permits required by Environmental Law for the operations of Parent and its Subsidiaries, and neither Parent nor any of its Subsidiaries has received any written communication from a Person that alleges that Parent or any of its Subsidiaries is in violation of, or has liability or obligations under, any Environmental Law or any permit issued pursuant to Environmental Law;
 - (b) there are no Environmental Claims pending or, to the Knowledge of Parent, threatened against Parent or any of its Subsidiaries;
 - (c) there have been no Releases of any Hazardous Material that would reasonably be expected to form the basis of any Environmental Claim against Parent or any of its Subsidiaries; and
 - (d) neither Parent nor any of its Subsidiaries has retained or assumed, either contractually or, to the Knowledge of Parent, by operation of Law, any liabilities or obligations of another Person that would reasonably be expected to form the basis of any Environmental Claim against Parent or any of its Subsidiaries.
- 5.24 Offer Documents; Schedule 14D-9; Registration Statement; Proxy Statement/Prospectus. (a) None of the Offer Documents, the Registration Statement nor the Post-Effective Amendment will, at the time such documents are filed with the SEC, at the time they are mailed to the holders of Company Shares, and at the time any amendment or supplement thereto is filed with the SEC, contain any untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary in order to make the statements made therein, in light of the circumstances under which they are made, not misleading. Notwithstanding the foregoing, no representation is made by Parent or Merger Sub with respect to information supplied by or on behalf of the Company or any Affiliate of the Company. The Offer Documents, the Registration Statement and

the Post-Effective Amendment will, at the time such documents are filed with the SEC, at the time the Offer Documents are mailed to the holders of Company Shares, and at the time any amendment or supplement thereto is filed with the SEC, comply as to form in all material respects with the applicable provisions of the Exchange Act and the Securities Act and the rules and regulations promulgated thereunder.

- (b) None of the information supplied by or on behalf of Parent, Merger Sub or any Affiliate of Parent or Merger Sub for inclusion in the Proxy Statement/Prospectus or the Schedule 14D-9 will, at the times such documents are filed with the SEC, at the time any amendment or supplement thereto is filed with the SEC and, in the case of the Proxy Statement/Prospectus, at the time the Proxy Statement/Prospectus is mailed to shareholders of the Company and at the time of any Company Stockholder Meeting, contains any untrue statement of a material fact or omits to state any material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they were made, not misleading.
- 5.25 No Additional Representations and Warranties. Parent and Merger Sub acknowledge that the Company makes no representation or warranty as to any matter whatsoever except as expressly set forth in this Agreement or in any certificate delivered by the Company to Parent or Merger Sub in accordance with the terms hereof, and specifically (but without limiting the generality of the foregoing) that the Company makes no representation or warranty with respect to (a) any projections, estimates or budgets delivered or made available to Parent or Merger Sub (or any of their respective Affiliates, officers, directors, employees or Representatives) of future revenues, results of operations (or any component thereof), cash flows or financial condition (or any component thereof) of the Company or its Subsidiaries or (b) the future business and operations of the Company or its Subsidiaries.

ARTICLE VI.

COVENANTS RELATING TO CONDUCT OF BUSINESS

- 6.1 Conduct of Business by the Company. Except for matters set forth in Section 6.1 of the Company Disclosure Schedule or otherwise contemplated by this Agreement or as required by applicable Law, from the date of this Agreement to the Effective Time, the Company shall, and shall cause each of its Subsidiaries to, conduct its business in the ordinary course consistent with past practice and, to the extent consistent therewith, use commercially reasonable efforts to preserve intact its current business organization, keep available the services of its current officers and employees and maintain its relationships with customers, suppliers, licensors, licensees, distributors and others having business dealings with them. In addition, and without limiting the generality of the foregoing, except for matters set forth in Section 6.1 of the Company Disclosure Schedule or otherwise contemplated by this Agreement or as required by applicable Law, from the date of this Agreement to the Effective Time, the Company shall not, and shall not permit any of its Subsidiaries to, do any of the following without the prior written consent of Parent, which consent shall not be unreasonably withheld, delayed or conditioned:
 - (a) (i) declare, set aside or pay any dividends on, or make any other distributions in respect of, any of its capital stock, other than dividends and distributions by a direct or indirect wholly owned Subsidiary of the Company to its parent company, (ii) split, combine or reclassify any of its capital stock or issue or authorize the issuance of any other securities in respect of, in lieu of or in substitution for shares of its capital stock, or (iii) purchase, redeem or otherwise acquire any shares of capital stock of the Company or any of its Subsidiaries or any other securities thereof or any rights, warrants or options to acquire any such shares or other securities;
 - (b) issue, deliver, sell or grant (i) any shares of its capital stock or other voting securities, (ii) any securities convertible into or exchangeable for, or any options, warrants or rights to acquire, any such shares, voting securities or convertible or exchangeable securities, or (iii) any "phantom" stock, "phantom" stock rights, stock appreciation rights or stock-based performance units, other than the issuance of Company Shares upon the exercise of Company Stock Options by any employee, officer or director of the Company outstanding on the date of this Agreement and in accordance with their present

terms or pursuant to existing deferral elections under the Company Directors' Deferred Compensation Plan;

- (c) amend the Company Charter Documents or other comparable charter or organizational documents of any Subsidiary of the Company;
- (d) in any single transaction or series of related transactions having a purchase price (including any assumed debt) in excess of \$2 million in the aggregate, acquire or agree to acquire (i) any Person or business, whether by merging or consolidating with, or by purchasing a substantial equity interest in or portion of the assets of such Person or business, or otherwise or (ii) any assets that are material, individually or in the aggregate, to the Company and its Subsidiaries, taken as a whole;
- (e) (i) subject to Section 6.1(b), grant or announce any incentive awards or any increase in compensation, severance or termination pay to any employee, officer, director or other service provider of the Company or its Subsidiaries, other than (A) to employees or other service providers with an annual base salary less than \$225,000 in the ordinary course of business consistent with past practice or (B) to the extent required under existing Company Employee Plans or existing Company Employees or by applicable Law, (ii) hire any new employees or officers, except in the ordinary course of business consistent with past practice with respect to employees or officers with an annual base salary and incentive compensation opportunity not to exceed \$200,000 per employee or officer, (iii) establish, adopt, enter into, amend, modify or terminate in any material respect any collective bargaining agreement or Company Employee Plan, or (iv) take any action to accelerate any rights or benefits, pay or agree to pay any pension, retirement allowance, termination or severance pay, bonus or other employee benefit, or make any material determinations not in the ordinary course of business consistent with prior practice under any collective bargaining agreement or Company Employee Plan;
- (f) make any change in accounting methods, principles or practices materially affecting the reported consolidated assets, liabilities or results of operations of the Company, except insofar as may have been required by a change in GAAP;
- (g) sell, lease (as lessor), license or otherwise dispose of or subject to any Lien (other than any Permitted Lien) any properties or assets that are material, individually or in the aggregate, to the Company and its Subsidiaries, taken as a whole, except licenses of or other grants of rights to use Intellectual Property in the ordinary course of business consistent with past practice and sales of inventory and excess or obsolete assets in the ordinary course of business consistent with past practice;
- (h) incur any indebtedness for borrowed money or guarantee any such indebtedness of another Person, issue or sell any debt securities or warrants or other rights to acquire any debt securities of the Company or any of its Subsidiaries, guarantee any debt securities of another Person, enter into any "keep well" or other agreement to maintain any financial statement condition of another Person or enter into any arrangement having the economic effect of any of the foregoing, except for short-term borrowings incurred in the ordinary course of business consistent with past practice;
- (i) make or agree to make any new capital expenditure or expenditures (other than in the ordinary course of business or capital expenditures that are contemplated by the Company's annual budget for 2011 and capital expenditure plan for 2012 which have been made available to Parent) that are in excess of \$2.5 million in the aggregate;
- (j) with respect to any Company Intellectual Property, except in the ordinary course of business consistent with past practice, and except for agreements between or among the Company and its Subsidiaries, (A) encumber, impair, abandon, fail to maintain, transfer, license to any Person (including through an agreement with a reseller, distributor, franchisee or other similar channel partner), or otherwise dispose of any right, title or interest of the Company or any of its Subsidiaries in any Company Intellectual Property or Company Software Products or (B) divulge, furnish to or make accessible any material confidential or other non-public information in which the Company or any of its Subsidiaries has trade secret or equivalent rights within the Company Intellectual Property to any Person who is not

subject to an enforceable written agreement to maintain the confidentiality of such confidential or other non-public information;

- (k) make or change any material Tax election or settle or compromise any Tax liability or claim in excess of \$1 million in the aggregate;
- (1) waive, release, assign, settle or compromise any claim, action or proceeding, other than waivers, releases, assignments, settlements or compromises that (i) involve the payment of monetary damages equal to or lesser than the amounts specifically reserved with respect thereto on the balance sheet as of December 31, 2010 included in the Company SEC Documents or that do not exceed \$1 million individually or in the aggregate, (ii) if involving any non-monetary outcome, will not have a material effect on the continuing operations of the Company and (iii) are with respect to ordinary course customer disputes;
 - (m) enter into any new line of business outside of the Company's existing business;
- (n) take any action that it knows or reasonably should have known would reasonably be expected to result in any of the conditions to the (i) Offer set forth in Exhibit A or (ii) Merger not being satisfied:
- (o) take any action that is intended to result in any of its representations or warranties set forth in this Agreement being or becoming untrue in any material respect at any time prior to the Effective Time such that the conditions to closing set forth in Article VIII cannot be fulfilled; or
 - (p) authorize any of, or commit or agree to take any of, the foregoing actions

For purposes of this Section 6.1, if any action, transaction or omission is permitted by the terms of a subsection hereof that specifically relates to the subject matter of the subsection, such action, transaction or omission shall be deemed permitted under all other subsections of this Section 6.1 even if such action, transaction or omission has ancillary effects on other subject matters contemplated by other subsections of this Section 6.1. Nothing contained in this Agreement is intended to give Parent, directly or indirectly, the right to control or direct the operations of the Company or any of its Subsidiaries prior to the Effective Time. Prior to the Effective Time, the Company shall exercise, consistent with the terms and conditions of this Agreement, complete control and supervision over its and its Subsidiaries' respective operations.

- 6.2 Conduct of Business by Parent. Except as contemplated by this Agreement or as required by applicable Law, from the date of this Agreement to the Effective Time, Parent shall, and shall cause each of its Subsidiaries to, conduct its business in the ordinary course consistent with past practice and, to the extent consistent therewith, use commercially reasonable efforts to preserve intact its current business organization, keep available the services of its current officers and employees and maintain its relationships with customers, suppliers, licensors, licensees, distributors and others having business dealings with them. In addition, and without limiting the generality of the foregoing, except as contemplated by this Agreement or as required by applicable Law, from the date of this Agreement to the Effective Time, Parent shall not, and shall not permit any of its Subsidiaries to, do any of the following without the prior written consent of Company, which consent shall not be unreasonably withheld, delayed or conditioned:
 - (a) (i) declare, set aside or pay any dividends on, or make any other distributions in respect of, any of its capital stock, other than dividends and distributions by a direct or indirect wholly owned Subsidiary of Parent to its parent company, (ii) split, combine or reclassify any of its capital stock or issue or authorize the issuance of any other securities in respect of, in lieu of or in substitution for shares of its capital stock, or (iii) purchase, redeem or otherwise acquire any shares of capital stock of Parent or any of its Subsidiaries or any other securities thereof or any rights, warrants or options to acquire any such shares or other securities;
 - (b) issue, deliver, sell or grant any shares of its capital stock or other voting securities, other than (i) the issuance of shares of Parent Common Stock in settlement of awards outstanding under, or otherwise in accordance with, the Parent Employee Plans, (ii) the exercise, conversion or exchange of securities disclosed in the Parent SEC Documents filed prior to the date of this Agreement, or (iii) Parent's issuance of debt securities or capital stock (other than shares of Parent Common Stock or securities

convertible, exercisable or exchangeable for shares of Parent Common Stock) for what its Board of Directors determines in good faith to be fair value;

- (c) amend the Parent Charter Documents or other comparable charter or organizational documents of any Subsidiary of Parent;
- (d) enter into any new line of business outside of Parent's existing businesses;
- (e) take any action that it knows or reasonably should have known would reasonably be expected to result in any of the conditions to the (i) Offer set forth in Exhibit A or (ii) Merger not being satisfied;
- (f) take any action that is intended to result in any of its representations or warranties set forth in this Agreement being or becoming untrue in any material respect at any time prior to the Effective Time such that the condition to closing set forth in Article VIII cannot be fulfilled; or
 - (g) authorize any of, or commit or agree to take any of, the foregoing actions

For purposes of this Section 6.2, if any action, transaction or omission is permitted by the terms of a subsection hereof that specifically relates to the subject matter of the subsection, such action, transaction or omission shall be deemed permitted under all other subsections of this Section 6.2 even if such action, transaction or omission has ancillary effects on other subject matters contemplated by other subsections of this Section 6.2. Nothing contained in this Agreement is intended to give the Company, directly or indirectly, the right to control or direct the operations of Parent or any of its Subsidiaries prior to the Effective Time. Prior to the Effective Time, Parent shall exercise, consistent with the terms and conditions of this Agreement, complete control and supervision over its and its Subsidiaries' respective operations.

ARTICLE VII.

ADDITIONAL AGREEMENTS

7.1 Preparation and Mailing of Post-Effective Amendment and Proxy Statement. (a) The Company shall use its reasonable best efforts to prepare and file within ten Business Days following the date of this Agreement with the SEC proxy materials that shall constitute the proxy statement relating to the matters to be submitted to the stockholders of the Company at the Company Stockholders Meeting (such proxy or information statement, and any amendments or supplements thereto, the "Proxy Statement/Prospectus"). Promptly following the date the Proxy Statement/Prospectus has been cleared for mailing by the SEC, Parent shall prepare and file as promptly as practicable with the SEC a post-effective amendment to the Form S-4 (the "Post-Effective Amendment") for the offer and sale of shares of Parent Common Stock pursuant to the Merger and in which the Proxy Statement/Prospectus will be included as a prospectus. The Proxy Statement/Prospectus and the Post-Effective Amendment shall comply as to form in all material respects with applicable U.S. federal securities Laws. The Company shall provide Parent and its counsel with a reasonable opportunity to review and comment on the Proxy Statement/Prospectus prior to the initial filing with the SEC and the Company shall give due consideration to all reasonable additions, deletions or changes suggested by Parent and its counsel. Parent and its counsel. Parent and its counsel with a reasonable additions, deletions or changes suggested by the Company and its counsel. Each of the Company and Parent shall use commercially reasonable efforts to have the Post-Effective Amendment declared effective under the Securities Act as promptly as practicable after the Post-Effective Amendment is declared effective under the Securities Act.

(b) Each of Parent and the Company shall, as promptly as practicable after receipt thereof, provide the other party copies of any written comments and advise the other party of any oral comments, with respect to the Post-Effective Amendment or Proxy Statement/Prospectus, as the case may be, received from the SEC. Each of Parent and the Company shall provide the other party with a reasonable opportunity to review and comment on any amendment or supplement to the Post-Effective Amendment or Proxy Statement/Prospectus,

as the case may be, and such party shall give due consideration to all reasonable additions, deletions or changes suggested by the other party, and any communications prior to filing such with the SEC and shall promptly provide the other party with a copy of all such filings and communications made with the SEC.

- (c) If at any time prior to the Effective Time, (i) any event or change occurs with respect to the parties or any of their respective Affiliates, officers or directors, which should be set forth in an amendment of, or a supplement to, the Post-Effective Amendment or Proxy Statement/Prospectus or (ii) any information relating to the parties, or any of their respective Affiliates, officers or directors, should be discovered by any of the parties which should be set forth in an amendment or a supplement to the Post-Effective Amendment or Proxy Statement/Prospectus so that the Post-Effective Amendment or Proxy Statement/Prospectus would not include any misstatement of a material fact or omit to state any material fact necessary to make the statements therein, in light of the circumstances under which they were made, not misleading, Parent or the Company, as the case may be, shall file as promptly as practicable with the SEC an amendment of, or a supplement to, the Post-Effective Amendment or Proxy Statement/Prospectus and, as required by Law, disseminate the information contained in such amendment or supplement to the stockholders of the Company.
- 7.2 Company Stockholders Meeting; Company Recommendation. (a) If the Company Stockholder Approval is required by applicable Law, as promptly as reasonably practicable after the Post-Effective Amendment is declared effective under the Securities Act, the Company shall duly take all necessary actions to duly call, give notice of, convene and hold a meeting of the stockholders of the Company or, if possible, arrange for action by written consent of the Company's stockholders (such meeting, or any adjournments or postponements thereof, or consent, the "Company Stockholders Meeting") for the purpose of obtaining the Company Stockholder Approval and shall, subject to Section 7.2(b), use reasonable best efforts to solicit its stockholders to obtain the Company Stockholder Approval. At the Company Stockholders Meeting or any postponement or adjournment thereof, Parent shall vote, or cause to be voted, all of the Company Shares then owned by it, Merger Sub or any of their respective Subsidiaries in favor of the adoption of this Agreement and to deliver or provide, in its capacity as a stockholder of the Company, any other approvals that are required pursuant to applicable Law to effect the Transactions.
- (b) The Board of Directors of the Company shall, subject to this Section 7.2(b), make the Company Recommendation to the stockholders of the Company and include the Company Recommendation in the Proxy Statement/Prospectus. Neither the Board of Directors of the Company nor any committee thereof, shall, directly or indirectly, withdraw, modify, amend or qualify the Company Recommendation in a manner adverse to Parent or Merger Sub (or publicly propose to take any of the foregoing actions) or execute or enter into, any letter of intent, memorandum of understanding, merger agreement or other written agreement providing for a Company Acquisition Proposal, except in accordance with Section 7.5.
- (c) Following the receipt of the Company Stockholder Approval, if applicable, Parent shall cause the Merger to become effective as promptly as practicable. Notwithstanding anything to the contrary contained in this Agreement, if Parent, Merger Sub or any other Subsidiary of Parent owns, by virtue of the Offer, exercise of the Top-Up Option or otherwise, in the aggregate at least 90% of the outstanding Company Shares, Parent, Merger Sub and the Company shall take all actions necessary and appropriate to cause the Merger to become effective as promptly as practicable following the time such ownership is first obtained, without a stockholders' meeting in accordance with Section 253 of the DGCL.
- 7.3 Notification. Each of Parent and the Company shall give prompt written notice to the other (and shall subsequently keep the other informed on a current basis of any developments related to such notice) upon its becoming aware of the occurrence or existence of any fact, event or circumstance that is reasonably likely to result in any of the conditions set forth in Article VIII or Exhibit A not being able to be satisfied prior to the Termination Date; provided, however, that the delivery of any notice pursuant to this Section 7.3 shall not, and shall not be deemed to, cure any breach of any representation or warranty requiring disclosure of such matter at or prior to the date of this Agreement or affect any of the conditions set forth in Article VIII or otherwise limit or affect the remedies available.
- 7.4 Confidentiality; Access to Information. The parties hereto acknowledge and agree that the Company and Parent have previously executed a Non-Disclosure Agreement, dated as of August 30, 2011 (the

"Non-Disclosure Agreement"), which Non-Disclosure Agreement will continue to be in full force and effect in accordance with its terms except as otherwise provided herein. Each of Parent and the Company shall afford the other parties hereto and the other parties' accountants, counsel and other Representatives reasonable access during normal business hours, upon reasonable notice, to its properties, books, records and personnel during the period prior to the Effective Time to obtain all information concerning its business as the other may reasonably request. Each of the parties hereto shall hold, and shall cause its accountants, counsel and other Representatives to hold, in confidence all documents and information furnished to it by or on behalf of another party to this Agreement in connection with the Transactions pursuant to the terms of the Non-Disclosure Agreement. Notwithstanding the foregoing, neither the Company nor Parent shall be required to afford such access if it would unreasonably likely to result in a violation of any agreement to which such party or any of its Subsidiaries is a party (provided that the Company or Parent, as the case may be, has used its reasonable best efforts to find an alternative way to provide the access or information contemplated by this Section 7.4), would be reasonably likely to result in a violation of any applicable Law. No information or knowledge obtained by a party hereto in any investigation pursuant to this Section 7.4 will affect or be deemed to modify any representation or warranty contained herein or the conditions to the obligations of the parties to consummate the Transactions

7.5 Non-Solicitation by the Company. (a) The Company agrees that neither it nor any of its Subsidiaries, nor any of their respective officers, directors or employees, shall, and that it shall use its reasonable best efforts to cause its and their respective Representatives to), directly or indirectly: (i) solicit, initiate, seek or knowingly encourage the making, submission or announcement of any Company Acquisition Proposal, (ii) furnish any nonpublic information regarding the Company or any of its Subsidiaries to any Person in connection with or in response to a Company Acquisition Proposal, (iii) continue or otherwise engage or participate in any discussions or negotiations with any Person with respect to any Company Acquisition Proposal, (iv) except in connection with a Company Change of Recommendation pursuant to Section 7.5(e), enter into any letter of intent, arrangement, agreement or understanding relating to any Company Acquisition Transaction; provided, however, that this Section 7.5 shall not prohibit (A) the Board of Directors of the Company or any committee thereof, directly or indirectly through any officer, employee or Representative, prior to the earlier of the Acceptance Time or the receipt of the Company Stockholder Approval, from furnishing nonpublic information regarding the Company or any of its Subsidiaries to, or entering into or participating in discussions or negotiations with, any Person in response to an unsolicited, bona fide Company Acquisition Proposal that the Board of Directors of the Company or any committee thereof concludes in good faith, after consultation with outside legal counsel and a financial advisor, constitutes or would reasonably be expected to result in a Company Superior Offer if (1) the Board of Directors of the Company or any committee thereof concludes in good faith, after consultation with its outside legal counsel, that the failure to take such action with respect to such Company Acquisition Proposal would be reasonably likely to result in a breach of its fidu

- (b) The Company shall promptly, and in no event later than 24 hours after its receipt of any Company Acquisition Proposal, or any request for nonpublic information relating to the Company or any of its Subsidiaries in connection with a Company Acquisition Proposal, advise Parent orally and in writing of such Company Acquisition Proposal or request (including providing the identity of the Person making or submitting such Company Acquisition Proposal or request, and, (i) if it is in writing, a copy of such Company Acquisition Proposal and any related draft agreements and (ii) if oral, a reasonably detailed summary thereof that is made or submitted by any Person during the period between the date hereof and the Closing Date). The Company shall keep Parent informed on a prompt basis with respect to any change to the material terms of any such Company Acquisition Proposal (and in no event later than 24 hours following any such change), including providing Parent with a copy of any draft agreements and modifications thereof.
- (c) Upon the execution of this Agreement, the Company shall, and shall cause its Subsidiaries and its and their respective officers, directors and employees, and shall use its reasonable best efforts to cause its and their respective Representatives to, immediately cease and terminate any existing activities, discussions or negotiations between the Company or any of its Subsidiaries or any of their respective officers, directors, employees or Representatives and any Person that relate to any Company Acquisition Proposal and shall use reasonable best efforts to obtain the prompt return or destruction of any confidential information previously furnished to such Persons with respect thereto within 12 months prior to the date hereof.
- (d) Except as otherwise provided in Section 7.5(e), the Board of Directors of the Company (or any committee thereof) may not (i) withhold, withdraw or modify, or publicly propose to withhold, withdraw or modify, the Company Recommendation in a manner adverse to Parent or make any statement, filing or release, in connection with obtaining the Company Stockholder Approval or otherwise, inconsistent with the Company Recommendation, (ii) approve, endorse or recommend any Company Acquisition Proposal (any of the foregoing set forth in clauses (i) and (ii), a "Company Change of Recommendation"), or (iii) enter into a written definitive agreement providing for a Company Acquisition Transaction.
- (e) The Board of Directors of the Company or any committee thereof may at any time prior to the earlier of the Acceptance Time or receipt of the Company Stockholder Approval (i) effect a Company Change of Recommendation in respect of a Company Acquisition Proposal, and/or (ii) if it elects to do so in connection with or following a Company Change of Recommendation, terminate this Agreement pursuant to Section 9.1(d/ii) in order to enter into a written definitive agreement providing for a Company Acquisition Transaction, if (and only if): (A) a Company Acquisition Proposal is made to the Company by a third party, and such offer is not withdrawn; (B) the Board of Directors of the Company or such committee thereof determines in good faith after consultation with outside legal counsel, the Board of Directors of the Company Superior Offer; (C) following consultation with outside legal counsel, the Board of Directors of the Company or such committee thereof determines that failure to take such action would be reasonably likely to result in a breach of its fiduciary duties under applicable Law; (D) the Company provides Parent five Business Days' prior written notice of its intention to take such action (such five-Business Day period, the "Notice Period"), which notice shall include the information with respect to such Company Superior Offer that is specified in Section 7.5(e) shall be deemed to be two Business Days); and (E) at the end of the Notice Period described in clause (D), the Board of Directors of the Company or such committee thereof again makes the determination in good faith after consultation with outside legal counsel and a financial advisor (after negotiating in good faith with Parent and its Representatives if requested by Parent during the Notice Period regarding any adjustments or modifications to the terms of this Agreement proposed by Parent and taking into account any such adjustments or modifications) that the Company Acquisition Proposal continues to be a Company Superior Offer and, after consul
- (f) During the period from the date of this Agreement through the Effective Time, neither the Company nor any of its Subsidiaries shall terminate, amend, modify or waive any provision of any confidentiality agreement to which it is a party relating to a proposed business combination involving the Company or any standstill agreement to which it is a party unless the Board of Directors of the Company or any committee

thereof determines in good faith, after consultation with outside legal counsel, that failure to take such action would be reasonably likely to result in a breach of its fiduciary duties under applicable Law. During such period, the Company or its Subsidiaries, as the case may be, shall enforce, to the fullest extent permitted under applicable Law, the provisions of any such agreement, including by obtaining injunctions to prevent any breaches of such agreements and to enforce specifically the terms and provisions thereof in each case except to the extent that the Board of Directors of the Company or any committee thereof determines in good faith, after consultation with outside legal counsel, that taking such action would be reasonably likely to result in a breach of its fiduciary duties under applicable Law.

7.6 Regulatory Filings; Efforts to Close.

(a) Filings and Submission of Information. Prior to the execution of this Agreement, each of Parent and the Company filed with the FTC and the DOJ the notification and report form required under the HSR Act for the Transactions (each, an "HSR Filing"), and representatives of the DOJ informed the parties that the DOJ is the Governmental Entity that would act under the HSR Act in respect of the Transactions. Neither Parent nor the Company will voluntarily withdraw its HSR Filing except that, upon notice to the Company, Parent may withdraw its HSR Filing in the event that counsel to Parent recommends that such action be taken to avoid a Second Request or the initiation of Agency Litigation, in which event Parent shall refile its HSR Filing (and Parent shall pay the fee required under the HSR Act in respect thereof) within two Business Days of its withdrawal; provided, however, that Parent may not voluntarily withdraw and refile its HSR Filing more than two times (excluding, for the avoidance of doubt, any refilling of an HSR Filing made by the Parent prior to the date of this Agreement) without the prior consent of the Company. Each of Parent and the Company shall promptly file with any Governmental Entity other than the DOJ any other filings, reports, information and documentation required in order to complete the Transactions under any Antitrust Laws other than the HSR Act. In the event that the DOJ or any other Governmental Entity requests additional information (other than pursuant to a Second Request or Agency Litigation, provision for which is made in Sections 7.6(b) and 7.6(c)), each of the parties shall respond thereto as promptly as practicable.

(b) Cooperation and Relationship of the Parties. Each of the parties hereto shall (i) furnish to each other's counsel such necessary information and reasonable assistance as such other counsel may request in good faith in connection with its preparation of any filing or submission to any Governmental Entity under the HSR Act and any other Antitrust Laws, (ii) give the other parties prompt notice of any so-called "second request" for information by the DOJ pursuant to 16 C.F.R. §§ 801, et seq. (a "Second Request") or any legal or other proceeding by or before any Governmental Entity with respect to the Transactions ("Agency Litigation"), and respond as promptly as practicable thereto with the objective of causing the Effective Time to occur as promptly as practicable, (iii) promptly inform the other parties of any communication with any Governmental Entity regarding any Second Request, Agency Litigation or the Transactions and keep the other parties informed on a current basis as to the status of any such matter, (iv) consult and cooperate with each other in connection with any analysis, appearance, discussion, presentation, memorandum, brief, argument, opinion or proposal made or submitted to any Governmental Entity in connection with any proceeding or communication relating to the Transactions, and (v) to the extent practicable, except as may be prohibited by any Governmental Entity or by any legal requirement, permit authorized representatives of the other party to be present at each meeting or conference or telephone call with any representative of a Governmental Entity relating to any such proceeding and to have access to any document, opinion or proposal made or submitted to any Governmental Entity in connection with any such proceeding.

(c) Defense of Litigation, Etc. In the event that any administrative or judicial action or proceeding (including Agency Litigation) is instituted (or threatened to be instituted) by a Governmental Entity or private party challenging the Transactions, each of the parties shall cooperate in all respects with each other and use its respective commercially reasonable efforts to contest and resist any such action or proceeding and to have vacated, lifted, reversed or overturned any decree, judgment, injunction or other order, whether temporary, preliminary or permanent, that is in effect and that prohibits, prevents or restricts consummation of the Transactions or delays the Effective Time past the Termination Date (collectively, an "Injunction"); provided, however, that notwithstanding any other provision of this Agreement, including this Section 7.6(a), (b), (d) or (e), (i) Parent shall be entitled to direct the defense of any legal, administrative or judicial action or proceeding

in respect of the Transactions, or negotiations with, any Governmental Entity or other Person relating thereto, or regulatory filings under applicable Antitrust Laws, (ii) the Company shall not make any offer, acceptance or counter-offer to or otherwise engage in negotiations or discussions with any Governmental Entity with respect to any proposed settlement, stay, toll, extension of any waiting period, consent decree, commitment or remedy, or, in the event of litigation, discovery, admissibility of evidence, timing or scheduling, except as specifically requested by or agreed with Parent or its counsel, and (iii) the Company shall use its commercially reasonable efforts to provide full and effective support of Parent and its counsel in all such negotiations and discussions with representatives of any Governmental Entity to the extent requested by counsel to Parent. Each of the parties may, as each deems advisable and necessary, reasonably designate any competitively sensitive material provided to the other pursuant to this Section 7.6 as "Outside Counsel Only Material" and may redact from any information provided to the other party and its counsel any references to such party's valuation of the other party.

(d) Reasonable Best Efforts. Each of the parties hereto shall (i) use its reasonable best efforts to obtain promptly (and in any event no later than the Termination Date) any clearance required under the HSR Act and any other Antitrust Laws for the consummation of the Transactions, (ii) use its reasonable best efforts to avoid or eliminate any impediment under any Antitrust Law, or regulation or rule, that may be asserted by any Governmental Entity, or any other Person, with respect to the Transactions so as to enable the Effective Time to occur expeditiously (and in any event no later than the Termination Date), (iii) use its reasonable best efforts to defend through Agency Litigation or, if applicable, other litigation on the merits any claim asserted in any court, administrative tribunal or hearing that the Transactions would violate any Law, or any regulation or rule of any Governmental Entity, in order to avoid entry of, or to have vacated or terminated, any Injunction, (iv) cause its respective inside and outside counsel to cooperate in good faith with counsel and other representatives of each other party hereto and use its reasonable best efforts to facilitate and expedite the identification and resolution of any such issues and, consequently, the expiration of the applicable HSR Act waiting period and the waiting periods under any other Antitrust Laws at the earliest practicable dates (and in any event no later than the Termination Date), such reasonable best efforts and cooperation to include causing their respective inside and outside counsel (A) to keep each other appropriately informed on a current basis of communications from and to personnel of the reviewing antitrust authority and (B) to confer on a current basis with each other regarding appropriate contacts with and response to personnel of such antitrust authority, (v) use its reasonable best efforts to cause the conditions set forth in Exhibit A and Article VIII to be satisfied on a timely basis, subject to the limitations set forth in this S

(e) Parent's Additional Covenants. Notwithstanding any other provision of this Agreement, if necessary to avoid the entry of an Injunction sought or issued by the DOJ under United States Antitrust Laws (a "DOI Impediment"), Parent's covenants under Section 7.6(e) will include Parent being required hereunder to offer to the DOJ that it or its Subsidiaries take, and, if such offer is accepted by the DOJ, use its best efforts to eliminate any DOJ Impediment: as set forth in the immediately preceding sentence shall require that Parent use its best efforts to effect, such of the following as may be necessary to avoid a DOJ Impediment: (A) the sale, holding separate, licensing, modifying or otherwise disposing of all or any portion of the business, assets or properties of the Company or its Subsidiaries, whether located in or outside the United States, (B) conducting or limiting the conduct of the business, assets or properties of company or its Subsidiaries, whether located in or outside the United States, (B) conducting or limiting the conduct of the business, assets or properties of received to in clause (B) (a "Consent Decree"). The Company shall take, and cause its Subsidiaries to take, all actions reasonably necessary and requested by Parent to satisfy any Consent Decree; provided, however, that the effect thereof may at the Company's or Parent's request be subject to the occurrence of the Acceptance Time, the Effective Time or Parent's waiver of all conditions to its obligations to consummate the Offer under Section 2(c) of Exhibit A hereof. It is understood and agreed by the parties that, for purposes of this Agreement, the effect of any action taken pursuant to this Section 7.6(e) will not, directly or indirectly, be deemed to result in (i) a breach of the

representations and warranties of Parent or the Company in this Agreement or (ii) a failure of any condition set forth in Exhibit A.

- 7.7 Public Disclosure. Except with respect to any (a) Company Change of Recommendation undertaken pursuant to, and in accordance with, Section 7.5, so long as this Agreement is in effect, the parties hereto shall use reasonable best efforts to consult with each other before issuing any press release or making any public announcement primarily relating to the Transactions and, except for any press release or public announcement as may be required by applicable Law or any listing agreement with a securities exchange or quotation system (including Nasdaq) (and then only after as much advance notice and consultation as is feasible), shall not issue any such public announcement without the consent of the other parties hereto, which shall not be unreasonably withheld, conditioned or delayed. Parent and the Company agree to issue a mutually acceptable initial joint press release announcing this Agreement.
- 7.8 Takeover Laws. Each of the Company and its Board of Directors and Parent and its Board of Directors shall, if any takeover statute or similar statute or regulation is or becomes applicable to the Transactions (including the acquisition of Company Shares and shares of Parent Common Stock), use reasonable best efforts to ensure that the Merger, the Offer and the Transactions may be consummated as promptly as practicable and such shares may be held on the terms contemplated hereby and otherwise to minimize the effect of such statute or regulation on this Agreement and the Transactions (including the ownership of such shares without limitation, including following the Acceptance Time and the Effective Time).
- 7.9 Indemnification. (a) From and after the Effective Time, Parent shall, and shall cause the Surviving Corporation to, indemnify and hold harmless all current and former directors, officers and employees, as the case may be, of the Company and its Subsidiaries to the fullest extent permitted by Law for acts or omissions occurring prior to the Effective Time (including acts or omissions occurring in connection with the approval of this Agreement and the consummation of the Transactions) in their capacities as such. Parent shall, and shall cause the Surviving Corporation to, fulfill and honor in all respects the obligations pursuant to any indemnification agreements between the Company or its Subsidiaries, on the one hand, and any current or former directors, officers and employees, as the case may be, of the Company and its Subsidiaries, on the other hand, in effect immediately prior to the date of this Agreement or, subject to the prior approval of Parent, after the date of this Agreement, and any indemnification provisions under the Company Charter Documents or the comparable charter or organizational documents of any of its Subsidiaries as in effect on the date hereof, in each case to the maximum extent permitted by Law, and shall not amend, repeal or otherwise modify any such provision in any manner that would adversely affect the rights of such indemnitee thereunder for any acts or omissions occurring prior to the Effective Time.
- (b) Prior to the Effective Time, the Company shall endeavor to enter into a directors' and officers' liability insurance policy covering those Persons who, as of immediately prior to the Effective Time, are covered by the Company's directors' and officers' liability insurance policy (the "Insured Parties") on terms no less favorable to the Insured Parties than those of the Company's present directors' and officers' liability insurance policy (such policy, a "Company D&O Policy"), for a period of seven years after the Effective Time. If the Company is unable to obtain such a Company D&O Policy prior to the Effective Time, Parent shall cause the Surviving Corporation to maintain in effect, for a period of seven years after the Closing Date, a Company D&O Policy with a creditworthy issuer; provided, however, that in no event shall Parent be required to expend annually more than 250% of the annual premium currently paid by the Company for such coverage (the approximate amount of which is set forth on Section 7.9(b) of the Company Disclosure Schedule) and, if the cost for such coverage is in excess of such amount, Parent shall be required only to maintain the maximum amount of coverage as is reasonably available for 250% of such annual premium.
- (c) In the event that Parent or the Surviving Corporation or any of their respective successors or assigns (i) consolidates with or merges into any other Person and is not the continuing or surviving corporation or entity of such consolidation or merger or (ii) transfers or conveys all or substantially all its properties and assets to any Person, Parent shall cause proper provisions to be made so that the successors and assigns of Parent or the Surviving Corporation assume the obligations set forth in this Section 7.9.

- (d) The obligations of Parent and the Surviving Corporation under this Section 7.9 shall not be terminated or modified in such a manner as to adversely affect any indemnitee and/or Insured Party to whom this Section 7.9 applies without the express written consent of such affected indemnitee and Insured Party. It is expressly agreed that the indemnitees and/or Insured Parties to whom this Section 7.9 applies shall be third party beneficiaries of this Section 7.9.
- (e) Parent shall assume, be jointly and severally liable for, and shall cause the Surviving Corporation and its Subsidiaries to honor, in accordance with their respective terms, each of the covenants contained herein without limit as to time. Parent shall pay all reasonable expenses, including reasonable attorneys' fees, that may be incurred by any indemnitee and/or Insured Party in enforcing the indemnity and other obligations provided hereunder or other applicable indemnification obligation referenced to herein. The rights of each indemnitee and/or Insured Party hereunder shall be in addition to, and not in limitation of, any other rights such Person may have under the Company Charter Documents or the comparable charter or organizational documents of any Subsidiary of the Company, or any other indemnification arrangement or otherwise.
- 7.10 Nasdaq Listing. Parent shall use its reasonable best efforts to cause the shares of Parent Common Stock to be issued in the Offer and the Merger to be approved for listing prior to the issuance thereof on the Nasdaq Global Select Market or on such other national securities exchange as the Parent Common Stock is listed.
- 7.11 Employee Matters. (a) It is Parent's and the Company's intention that, for a period of one year following the Effective Time, individuals who are employed by the Company or any of its Subsidiaries immediately prior to the Effective Time and that continue to be employed by Parent or any of its Subsidiaries, including the Surviving Corporation, following the Effective Time (each, a "Post-Merger Employee") shall be provided with salaries and benefits that are in the aggregate approximately equal to the salaries and benefits (other than equity compensation) they received prior to the Effective Time, it being understood that Parent shall review the salaries and benefits of its and its Subsidiaries' employees from time to time in order to determine the most appropriate way to compensate and incentivize Post-Merger Employees, and accordingly may make such changes in the compensation and benefits that Parent determines to be in the best interests of Parent from time to time.
- (b) It is Parent's and the Company's intention that, each Post-Merger Employee shall be given credit for all service with the Company and its Subsidiaries and their respective predecessors under any employee benefit plan of Parent, the Surviving Corporation or any of their Subsidiaries, including any such plans providing vacation, sick pay, severance and retirement benefits maintained by Parent or its Subsidiaries in which such Post-Merger Employees participate for purposes of eligibility, vesting and entitlement to benefits, including for severance benefits and vacation entitlement (but not for accrual of pension benefits), to the extent past service was recognized for such Post-Merger Employees under the comparable Company Employee Plans immediately prior to the Effective Time. Notwithstanding the foregoing, nothing in this Section 7.11 shall be construed to require crediting of service that would result in (i) duplication of benefits or (ii) service credit for benefit accruals under a defined benefit pension plan.
- (c) In the event of any change in the welfare benefits provided to Post-Merger Employees following the Effective Time, it is Parent's and the Company's intention that Parent shall use its reasonable best efforts to cause (i) the waiver of all limitations as to pre-existing conditions, exclusions and waiting periods with respect to participation and coverage requirements applicable to the Post-Merger Employees (and their eligible dependents) under any welfare benefit plans in which Post-Merger Employees participate following the Effective Time, to the extent that such conditions, exclusions or waiting periods would not apply in the absence of such change, and (ii) for the plan year in which the Effective Time occurs, the crediting of each Post-Merger Employee (or his or her eligible dependents) with any co-payments and deductibles paid prior to any such change in satisfying any applicable deductible or out-of-pocket requirements after such change.
- (d) Nothing in this Section 7.11, express or implied, shall confer upon any Company Employee, or any legal representative or beneficiary thereof, any rights or remedies, including any right to employment or continued employment for any specified period, or compensation or benefits of any nature or kind whatsoever under this Agreement. Nothing in this Section 7.11, express or implied, shall be construed to prevent Parent

from terminating or modifying to any extent or in any respect any benefit plan that Parent may establish or maintain. Notwithstanding anything to the contrary contained in this Section 7.11, nothing contained in this Agreement shall be treated as an amendment to any Company Employee Plan or creation of an employee benefit plan.

- 7.12 Section 16 Matters. Parent and the Company agree that, in order to most effectively compensate and retain the Company Insiders in connection with the Offer and the Merger, both prior to and after the Effective Time, it is desirable that the conversion or exchange of Company Shares into shares of Parent Common Stock in the Offer or the Merger by the Company Insiders be exempted from liability under Section 16 (b) of the Exchange Act to the fullest extent permitted by Law, and for that compensatory and retentive purpose agree to the provisions of this Section 7.12. Assuming that the Company delivers to Parent the Section 16 Information in a timely fashion, the Board of Directors of Parent, or a committee of Non-Employee Directors thereof (as such term is defined for purposes of Rule 16b-3(d) under the Exchange Act), shall adopt a resolution providing that the receipt by the Company Insiders of Parent Common Stock in exchange for Company Shares pursuant to the Transactions and to the extent such securities are listed in the Section 16 Information, are intended to be exempt from liability pursuant to Section 16(b) under the Exchange Act. For the purposes of this Agreement, "Section 16 Information" shall mean information that is accurate in all material respects and provides reasonably specific details regarding the identity of the Company Insiders and the number of Company Shares held by each such Company Insider to be exchanged for Parent Common Stock in the Offer or the Merger; and "Company Insiders" shall mean those officers and directors of the Company who may be subject to the reporting requirements of Section 16(a) of the Exchange Act on or following the Acceptance Time or the Effective Time and who are listed in the Section 16 Information.
- 7.13 Stockholder Litigation. Each party hereto shall promptly advise the other orally and in writing of any litigation commenced or threatened in writing to be commenced by any stockholder of such party against such party and/or any of its directors relating to the Transactions, the transactions with Fundtech and/or the Fundtech Merger Agreement, and shall keep the other party fully informed regarding any such litigation. Each party shall give the other the opportunity to participate in, subject to a customary joint defense agreement, the defense or settlement of any such litigation and shall not settle any such litigation without the prior written consent of the other party (not to be unreasonably withheld, conditioned or delayed).
- 7.14 Obligations of Merger Sub. Parent shall take all action necessary to cause Merger Sub to perform its obligations under this Agreement and to consummate the Offer and the Merger on the terms and subject to the conditions set forth in this Agreement. Parent hereby guarantees the due, prompt and faithful performance and discharge by, and compliance with, all of the obligations, covenants, terms, conditions and undertakings of Merger Sub under this Agreement in accordance with the terms hereof, including any such obligations, covenants, terms, conditions and undertaking that are required to be performed, discharged or complied with following the Effective Time.
- 7.15 Financing. (a) Parent shall take, or cause to be taken, all actions and to do, or cause to be done, all things reasonably necessary to obtain the Financing. If any portion of the Financing becomes unavailable on the terms and conditions (including the flex provisions) contemplated in the Financing Agreement, Parent shall use its reasonable best efforts to arrange and obtain alternative financing from alternative sources in an amount sufficient to consummate the Transactions ("Alternative Financing") as promptly as practicable following the occurrence of such event but no later than the Business Day immediately prior to the Closing Date.
- (b) Prior to the Effective Time, upon the request of Parent, the Company will, and shall instruct its Subsidiaries and its and their directors, officers and other Representatives to, cooperate reasonably with Parent in connection with the Financing or Alternative Financing, including by (i) making senior management of the Company available to participate in meetings and road shows, if any, (ii) providing on a timely basis information reasonably requested by Parent relating to the Financing or Alternative Financing, (iii) preparing in a timely manner business projections and financial statements (including proforma financial statements), (iv) assisting in a timely manner in the preparation of offering memoranda, private placement memoranda,

prospectuses and similar documents, (v) providing such assistance as Parent may reasonably require in procuring a corporate credit rating for Parent from Standard & Poor's Rating Services and a corporate family credit rating for Parent from Moody's Investor Services, Inc., (vi) obtaining the consent of, and customary comfort letters from, PricewaterhouseCoopers LLP (including by providing customary management letters and requesting legal letters to obtain such consent) if necessary or desirable for Parent's use of the Company's financial statements in connection with the Financing or the Alternative Financing, (vii) using commercially reasonable efforts to ensure that the syndication effort of the lenders to Parent benefit materially from the existing lending relationships of the Company, and (viii) providing other financial information and documents as may be reasonably requested by Parent for such the Financing or Alternative Financing, including (A) confirmation of public or non-public nature of information provided, (B) closing certificates and documents and information reasonably necessary to facilitate the pledge of collateral (including the release of any encumbrances on the assets of the Company), and (C) providing such documentation and other information to Parent's lenders that is required by regulatory authorities under applicable "know your customer" and anti-money laundering rules and regulations, including the Patriot Act. Without limiting the generality or effect of the foregoing, all financial and other projections concerning the Company that are made available by the Company to Parent after the date of this Agreement shall be prepared in good faith and shall be based upon assumptions that are reasonable at the time made. Notvithstanding the foregoing, (1) such requested cooperation shall not unreasonably interfere with the ongoing operations of the Company and (2) the Company shall not be required to pay any commitment or other similar fee or incur any other liability in connection with the Fin

- (c) Subsequent Financial Statements. The Company will, if it determines that doing so is practicable, prior to making publicly available its financial results for any period after the date of this Agreement and prior to filing any report or document with the SEC after the date of this Agreement, furnish drafts of such documents to Parent, it being understood that Parent will have no liability by reason of such consultation
- 7.16 No Control of the Company's Business. Nothing contained in this Agreement will give Parent, directly or indirectly, the right to control or direct the Company's or its Subsidiaries' operations prior to the Effective Time. Prior to the Effective Time, the Company shall exercise, consistent with the terms and conditions of this Agreement, complete control and supervision over its and its Subsidiaries' respective operations.
- 7.17 Non-Disparagement. Promptly and in any event within two Business Days after the date of this Agreement, each of the Company and Parent shall instruct its sales and marketing personnel to comply with the principles set forth in Schedule 7.17. Thereafter, if any executive officer of either the Company or Parent obtains actual knowledge that such personnel did not comply with such requirements, the Company or Parent, as applicable, shall take steps that it believes in good faith to be appropriate to remedy the situation.
- 7.18 Dealer Manager. Prior to the Acceptance Time, each of Parent and the Company shall use its reasonable best efforts to comply with the obligations set forth in the dealer manager agreement to which Parent is a party.

ARTICLE VIII.

CONDITIONS TO THE MERGER

8.1 Conditions to Obligations of Each Party to Effect the Merger. The respective obligations of each party to this Agreement to effect the Merger shall be subject to the satisfaction on or prior to the Closing Date

of the following conditions, any of which may be waived, in writing, by mutual agreement of Parent and the Company:

- (a) Registration Statement. The Registration Statement and the Post-Effective Amendment shall have been declared effective by the SEC under the Securities Act and no stop order suspending the effectiveness of the Registration Statement or the Post-Effective Amendment shall have been issued by the SEC and no proceeding for that purpose shall have been initiated by the SEC.
 - (b) Company Stockholder Approval. If required by applicable Law, the Company Stockholder Approval shall have been obtained
 - (c) The Offer. If the Acceptance Time has occurred, Merger Sub shall have purchased all Company Shares validly tendered and not withdrawn pursuant to the Offer.
- (d) No Order. No Governmental Entity shall have enacted, issued, promulgated, enforced or entered any Law, injunction or other order (whether temporary, preliminary or permanent), judgment, decree, executive order or award (an "Order") which is then in effect and has the effect of making the Merger illegal or otherwise prohibiting consummation of the Merger.
- (e) HSR. Any applicable waiting period under the HSR Act, and, if applicable, any agreement with the Federal Trade Commission or DOJ not to acquire Company Shares, shall have expired or shall have been terminated prior to the Closing Date (this condition, the "HSR Condition").
- 8.2 Additional Conditions to Obligations of Parent to Effect the Merger If Acceptance Time Has Not Occurred. If the Acceptance Time has not occurred, the obligation of Parent to effect the Merger shall be subject to the satisfaction on or prior to the Closing Date of the following conditions, any of which may be waived, in writing, by Parent:
 - (a) Governmental Approvals. Any clearance, approval, permit, authorization, waiver, determination, favorable review or consent of any Governmental Entity, other than the HSR Condition, shall have been obtained and shall be in full force and effect, and any applicable waiting periods for such clearances or approvals shall have expired, except for any failures that would not reasonably be expected to have a Material Adverse Effect on Parent or the Company.
 - (b) Representations and Warranties. (A) The representations and warranties of the Company set forth in Sections 4.1, 4.3, 4.4(a), 4.7, 4.18 and 4.20 shall have been true and correct as of the date of this Agreement and shall be true and correct on and as of the Closing Date with the same force and effect as if made at the Closing Date (in either case other than those representations and warranties which address matters only as of a particular date, which representations and warranties shall have been true and correct as of such particular date), except in either case contemplated by this clause (A) for de minimis inaccuracies and (B) the other representations and warranties of the Company set forth in this Agreement shall have been true and correct as of the date of this Agreement and shall be true and correct on and as of the Closing Date with the same force and effect as if made on the Closing Date (in either case other than those representations and warranties which address matters only as of a particular date, which representations shall have been true and correct as of such particular date), except in either case contemplated by this clause (B) where the failure of such representations and warranties to be true and correct (disregarding all qualifications or limitations as to materiality, Material Adverse Effect or words of similar import set forth therein) has not had and would not reasonably be expected to have a Material Adverse Effect on the Company.
 - (c) Covenants. The Company shall have performed or complied in all material respects with all agreements and covenants required by this Agreement to be performed or complied with by it on or prior to the Closing Date.
 - (d) No Material Adverse Effect. Since the date of this Agreement, there shall not have occurred any Material Adverse Effect with respect to the Company which is continuing.

- 8.3 Additional Conditions to Obligations of the Company to Effect the Merger If Acceptance Time Has Not Occurred. If the Acceptance Time has not occurred, the obligation of the Company to effect the Merger shall be subject to the satisfaction on or prior to the Closing Date of the following conditions, any of which may be waived, in writing, by the Company:
 - (a) Governmental Approvals. Any clearance, approval, permit, authorization, waiver, determination, favorable review or consent of any Governmental Entity, other than the HSR Condition, shall have been obtained and shall be in full force and effect, and any applicable waiting periods for such clearances or approvals shall have expired, except for any failures that would not reasonably be expected to have a Material Adverse Effect on Parent.
 - (b) Representations and Warranties. (A) The representations and warranties of Parent and Merger Sub set forth in Sections 5.1, 5.3, 5.4(a), 5.7, 5.18 and 5.19 shall have been true and correct as of the date of this Agreement and shall be true and correct on and as of the Closing Date with the same force and effect as if made at the Closing Date (in either case other than those representations and warranties which address matters only as of a particular date, which representations and warranties shall have been true and correct as of such particular date), except in either case contemplated by this clause (A) for de minimis inaccuracies and (B) the other representations and warranties of Parent and Merger Sub set forth in this Agreement shall have been true and correct as of the date of this Agreement and shall be true and correct on and as of the Closing Date with the same force and effect as if made on the Closing Date (in either case other than those representations and warranties which address matters only as of a particular date, which representations shall have been true and correct as of such particular date), except in either case contemplated by this clause (B) where the failure of such representations and warranties to be true and correct (disregarding all qualifications or limitations as to materiality, Material Adverse Effect or words of similar import set forth therein) has not had and would not reasonably be expected to have a Material Adverse Effect on Parent.
 - (c) Covenants. Each of Parent and Merger Sub shall have performed or complied in all material respects with all of its respective agreements and covenants required by this Agreement to be performed or complied with by it on or prior to the Closing Date.
 - (d) No Material Adverse Effect. Since the date of this Agreement, there shall not have occurred any Material Adverse Effect with respect to the Parent which is continuing.

ARTICLE IX.

TERMINATION

- 9.1 Termination. Notwithstanding anything in this Agreement to the contrary, this Agreement may be terminated and the Transactions abandoned at any time prior to the Acceptance Time:
 - (a) by mutual written consent of Parent and the Company;
 - (b) by either the Company or Parent if:
 - (i) the Merger and the Offer shall not have been consummated prior to July 31, 2012 (the "Termination Date"); provided, however, that the right to terminate this Agreement under this Section 9.1(b) shall not be available to any party whose action or failure to act has been the principal cause of or resulted in the failure of the Merger and the Offer to occur on or before such date; and provided, further that (x) any purported termination of this Agreement by Parent pursuant to this Section 9.1(b)(i) shall be deemed a termination of this Agreement by the Company pursuant to Section 9.1(d)(i)(b) if, at the time of any such intended termination by Parent, the Company is entitled to terminate this Agreement by Parent pursuant to Section 9.1(c)(i) or Section 9.1(c)(ii) or Section 9.1(c)(ii) if, at the time of any such intended termination by the Company, Parent is entitled to terminate this Agreement pursuant to Section 9.1(c)(ii); or Section 9.1(c)(iii); or Section 9.1(c)(iii); or Section 9.1(c)(iii); or Section 9.1(c)(iii

- (ii) a Governmental Entity shall have enacted, issued, promulgated, enforced or entered any Law (including an injunction or other order) or taken any other action, in any case having the effect of permanently restraining, enjoining or otherwise prohibiting the Offer or the Merger, which Law (including any such injunction or other order) or other action shall have become final and nonappealable; or
- (iii) the Offer shall have expired or been terminated without any Company Shares being purchased therein as a result of the failure to satisfy the Minimum Tender Condition; provided that the right to terminate this Agreement pursuant to this Section 9.1(b)(iii) shall not be available to any party whose failure to perform any of its obligations under this Agreement is the primary cause of the failure of the Offer to be consummated by such date, as applicable; or

(c) by Parent, if

- (i) (A) a Company Change of Recommendation, whether or not permitted by the terms hereof, shall have occurred; (B) the Company shall have delivered a notice to Parent of its intent to effect a Company Change of Recommendation in accordance with Section 7.5(e); (C) following the request in writing by Parent, the Board of Directors of the Company shall have failed to reaffirm publicly the Company Recommendation within five Business Days after Parent requests in writing that such recommendation be reaffirmed publicly; provided, however, that Parent shall only be entitled to make such a written request for reaffirmation (and the Company shall only be required to reaffirm publicly the Company Recommendation) an aggregate of three times and thereafter an additional one time for each new Company Acquisition Proposal; or (D) the Company shall have breached, in any material respect, the provisions of Section 7.5; or
- (ii) there shall have been a breach by the Company of any of its representations, warranties, covenants or obligations contained in this Agreement, which breach would result in the failure to satisfy by the Termination Date one or more of the conditions set forth in Exhibit A, and in any such case such breach shall be incapable of being cured or, if capable of being cured, shall not have been cured within 30 days after written notice thereof shall have been received by the Company of such breach; provided, however, the right to terminate this Agreement under this Section 9.1(c)(ii) shall not be available to Parent if at such time the Company would be entitled to terminate this Agreement pursuant to Section 9.1(d)(i)(b) or if Parent is otherwise in material breach of its obligations hereunder; or

(d) by the Company, if

- (i) (a) Merger Sub fails to amend the Pending Offer in violation of Section 2.1 hereof or (b) there shall have been a breach by Parent or Merger Sub of any of its representations, warranties, covenants or obligations contained in this Agreement, which breach would result in the failure to satisfy by the Termination Date one or more of the conditions set forth in Exhibit A, and in any such case such breach shall be incapable of being cured or, if capable of being cured, shall not have been cured within 30 days after written notice thereof shall have been received by Parent or Merger Sub of such breach; provided, however, the right to terminate this Agreement under this Section 9.1(d)(i) shall not be available to the Company if at such time Parent would be entitled to terminate this Agreement pursuant to Section 9.1(c)(ii) or if the Company is otherwise in material breach of its obligations hereunder; or
- (ii) the Company effects a Company Change of Recommendation to accept a Company Acquisition Proposal in accordance with Section 7.5(e), provided that the right to terminate this Agreement pursuant to this Section 9.1(d)(ii) shall not be available to the Company unless the Company pays or has paid the Company Termination Fee to Parent in accordance with Section 9.3 (provided that Parent shall have provided wiring instructions for such payment or, if not, then such payment shall be paid promptly following delivery of such instructions); it being understood that the

Company may enter into any agreement providing for a Company Acquisition Transaction simultaneously with the termination of this Agreement pursuant to this Section 9.1(d)(ii).

A party hereto may only terminate this Agreement if such termination has been duly authorized by an action of the Board of Directors of such party.

- 9.2 Notice of Termination; Effect of Termination. A party desiring to terminate this Agreement shall give written notice of such termination to the other party, specifying the provision pursuant to which such termination is effective. In the event of the termination of this Agreement as provided in Section 9.1, this Agreement shall be of no further force and effect, except (i) each of this Section 9.2, Section 9.3 and Article X shall survive the termination of this Agreement and (ii) nothing herein shall relieve any party from liability for any material breach of this Agreement occurring prior to the termination of this Agreement that was committed intentionally by the breaching party or resulted from the breaching party or specified. No termination of this Agreement shall affect the obligations of the parties contained in the Non-Disclosure Agreement, all of which obligations shall survive termination of this Agreement in accordance with its terms.
- 9.3 Fees and Expenses. (a) Other than as specifically provided in this Section 9.3 or otherwise agreed to in writing by the parties hereto, all costs and expenses incurred in connection with this Agreement and the Transactions shall be paid by the party incurring such costs or expenses, whether or not the Merger is consummated.
 - (b) The Company shall pay to Parent the Company Termination Fee if this Agreement is terminated as follows:
 - (i) if this Agreement is terminated by Parent pursuant to Section 9.1(c)(i), then the Company shall pay the entire Company Termination Fee by the second Business Day following such termination:
 - (ii) if this Agreement is terminated by the Company pursuant to Section 9.1(d)(ii), then the Company shall pay to Parent the entire Company Termination Fee upon such termination (provided that Parent shall have provided wiring instructions for such payment or, if not, then such payment shall be paid promptly following delivery of such instructions); or
 - (iii) (A) if this Agreement is terminated (1) by Parent pursuant to Section 9.1(c)(ii), (2) by the Company or Parent pursuant to Section 9.1(b)(ii), and in any such case of (1), (2) or (3) above, a Company Acquisition Proposal (including a previously communicated Company Acquisition Proposal) shall have been publicly announced or otherwise communicated to a member of senior management or the Board of Directors of the Company (or any Person shall have publicly announced or communicated a bona fide intention, whether or not conditional, to make a Company Acquisition Proposal) at any time after the date of this Agreement and prior to the Acceptance Time, in the case of clause (2), or the date of termination, in the case of clauses (1) or (3), and (B) if within 12 months after the date of such termination, the Company enters into a definitive agreement to consummate, or consummates, any Company Acquisition Transaction, then the Company shall pay to Parent the Company Termination Fee by the second Business Day following the date the Company enters into such definitive agreement or consummates such transaction; provided, however, that, solely for purposes of this Section 9.3(b)(iii), references in the definition of "Company Acquisition Transaction" to 20% shall be deemed to mean 50%.
- (c) All amounts paid pursuant to this Section 9.3 shall be by wire transfer of immediately available funds to an account directed by the party hereto entitled to payment as long as such account has been identified by such party. Each party hereto agrees that the agreements contained in this Section 9.3 are an integral part of the Transactions, and that, without these agreements, the other parties would not enter into this Agreement; accordingly, if any party fails promptly to pay amounts due under this Section 9.3 and, in order to obtain such payment, the other party commences a suit that results in a judgment against the party failing to pay such amounts, then the party failing to pay such amounts shall pay interest on such amounts from the date payment of such amounts was due to the date of actual payment at the prime rate of the Bank of New York in effect on the date such payment was due, together with the reasonable, documented out-of-pocket costs and

expenses of the party seeking collection (including reasonable legal fees and expenses) in connection with such suit.

ARTICLE X.

GENERAL PROVISIONS

- 10.1 Non-Survival of Representations and Warranties. The representations and warranties of the Company, Parent and Merger Sub contained in this Agreement shall terminate and be of no further force and effect as of the Closing, and only the covenants that by their terms contemplate performance after the Closing shall survive the Closing.
- 10.2 Notices. Any notice required to be given hereunder shall be sufficient if in writing, and sent by facsimile or email transmission, by reliable overnight delivery service (with proof of service) or hand delivery (provided that any notice received on any non-Business Day or any Business Day after 5:00 p.m. (addressee's local time) shall be deemed to have been received at 9:00 a.m. (addressee's local time) on the next Business Day unless the notice is required by this Agreement to be delivered within a number of hours or calendar days), addressed as follows (or at such other address, email address or facsimile number for a party as shall be specified by like notice):

(a) if to Parent or Merger Sub, to:

ACI Worldwide, Inc. 6060 Coventry Drive Elkhorn, Nebraska 68022 Attention: General Counsel Facsimile: (402) 778-2567

Email: dennis.byrnes@aciworldwide.com

with a copy (which shall not constitute notice) to each of:

Jones Day 222 East 41st Street New York, New York 10017 Attention: Robert A. Profusek Facsimile: (212) 755-7306 Email: raprofusek@jonesday.com

if to the Company, to: \$1 Corporation 705 Westech Drive Norcross, Georgia 30092 Attention: General Counsel Facsimile: (404) 923-6460 Email: greg.orenstein@s1.com

with a copy (which shall not constitute notice) to each of:

Hogan Lovells US LLP Columbia Square 555 Thirteenth Street, NW Washington, DC 20004 Attention: Daniel Keating Facsimile: (202) 637-5910

Email: daniel.keating@hoganlovells.com

10.3 Construction. For the purposes of this Agreement, except as otherwise expressly provided herein or unless the context otherwise requires: (i) words using the singular or plural number also include the plural

or singular number, respectively, and the use of any gender herein shall be deemed to include the other genders; (ii) references herein to "Articles," "Sections," "subsections," and other subdivisions, without reference to a document are to the specified Articles, Sections, subsection and other subdivisions of this Agreement; (iii) a reference to a subsection or other subdivision without further reference to a Section is a reference to such subsection or subdivision as contained in the same Section in which the reference appears; (iv) the words "herein," "hereof," (iv) all accounting terms used and not expressly defined herein have the respective meanings given to them under GAAP, as applicable; and (vii) any reference herein to any Law or legal requirement (including to any statute, ordinance, code, rule, regulation, or any provision thereof) shall be deemed to include reference to such Law and or to such legal requirement, as amended, and any legal requirements promulgated thereunder or successor thereto.

- 10.4 Headings; Table of Contents. Headings of the Articles and Sections of this Agreement are for convenience of the parties only and shall be given no substantive or interpretive effect whatsoever. The table of contents to this Agreement is for reference purposes only and shall not affect in any way the meaning or interpretation of this Agreement.
- 10.5 Mutual Drafting. The parties hereto agree that they have jointly drafted and have been represented by counsel during the negotiation and execution of this Agreement and, therefore, waive the application of any Law, regulation, holding or rule of construction providing that ambiguities in an agreement or other document will be construed against the party drafting such agreement or document.
- 10.6 Disclosure Schedules. All capitalized terms not defined in the Disclosure Schedules shall have the meanings assigned to them in this Agreement. Each representation and warranty of a party in this Agreement is made and given, and the covenants are agreed to, subject to the disclosures and exceptions set forth on such party's Disclosure Schedule. The disclosure of any matter in any section of a party's Disclosure Schedule shall be deemed to be a disclosure by such party for all purposes of this Agreement and all other sections of such party's Disclosure Schedule to which such disclosure reasonably would be inferred. The listing of any matter on a party's Disclosure Schedule shall expressly not be deemed to constitute an admission by such party, or to otherwise imply, that any such matter is material, is required to be disclosed by such party under this Agreement or falls within relevant minimum thresholds or materiality standards set forth in this Agreement. No disclosure in a party's Disclosure Schedule relating to any possible breach or violation by such party of any Contract or Law shall be construed as an admission or indication that any such breach or violation exists or has actually occurred. In no event shall the listing of any matter in a party's Disclosure Schedule be deemed or interpreted to expand the scope of such party's representations, warranties and/or covenants set forth in this Agreement
- 10.7 Counterparts; Facsimile and Electronic Signatures. This Agreement may be executed in one or more counterparts, each of which shall be deemed an original but all of which together will constitute one and the same instrument and shall become effective when one or more counterparts have been signed by each of the parties and delivered to the other party, it being understood that all parties need not sign the same counterpart. This Agreement or any counterpart may be executed and delivered by facsimile copies or delivered by electronic communications by portable document format (.pdf), each of which shall be deemed an original.
- 10.8 Entire Agreement; Third Party Beneficiaries. This Agreement and the documents and instruments and other agreements among the parties hereto as contemplated by or referred to herein, including the Non-Disclosure Agreement, the Company Disclosure Schedule and the Parent Disclosure Schedule: (a) constitute the entire agreement among the parties with respect to the subject matter hereof and supersede all prior agreements and understandings, both written and oral, among the parties with respect to the subject matter hereof; and (b) except (i) as provided in Section 7.9 (which is intended for the benefit of the Company's former and current officers and directors and other indemnitees, all of whom shall be third party beneficiaries of these provisions), (ii) the right of the Company, on behalf of its stockholders, to pursue damages (including damages for the loss of economic benefits) on their behalf in the event of Parent's breach of this Agreement,

- (iii) from and after the Effective Time, the right of the holders of Company Shares (restricted and unrestricted), Company restricted stock units and restricted cash units, SARs and Company Stock Options to receive the applicable Merger Consideration or other consideration to be paid hereunder as set forth in Article III, and (iv) the financing sources and their respective successors, legal representatives and permitted assigns with respect to their respective rights under Sections 10.9, 10.13, 10.14 and 10.15, nothing in this Agreement shall confer upon any other Person any rights or remedies hereunder.
- 10.9 Amendment. At any time prior to the Effective Time, any provision of this Agreement may be amended or waived if, and only if, such amendment or waiver is in writing and signed, in the case of an amendment, by the Company, Parent and Merger Sub or, in the case of a waiver, by the party against whom the waiver is to be effective; provided, however, that after the Company Stockholder Approval, if any such amendment or waiver shall by applicable Law or in accordance with the rules and regulations of Nasdaq require further approval of the stockholders of the Company; provided, further, that in the case of amendments, modifications, supplements or waivers to Sections 10.13, 10.14, 10.15 and this Section 10.9 (and any amendment, modifications or supplements to any related definitions) which affect the rights of any financing source under the Financing, any such amendment, modifications or waiver shall only be effective with respect to such financing source if such financing source shall have consented thereto. Notwithstanding the foregoing, no failure or delay by the Company or Parent in exercising any right hereunder shall operate as a waiver thereof nor shall any single or partial exercise thereof preclude any other or further exercise of any right hereunder.
- 10.10 Severability. If any provision of this Agreement is held to be illegal, invalid or unenforceable under any present or future Law, and if the rights or obligations of any party hereto under this Agreement will not be materially and adversely affected thereby, (a) such provision will be fully severable, (b) this Agreement shall be construed and enforced as if such provision had never comprised a part hereof, (c) the remaining provisions of this Agreement shall remain in full force and effect and shall not be affected by such provision or its severance herefrom and (d) in lieu of such provision, there shall be added automatically as a part of this Agreement a legal, valid and enforceable provision as similar in terms to such provision as may be possible
- 10.11 Remedies; Specific Performance. Except as otherwise provided herein, any and all remedies herein expressly conferred upon a party hereto (on behalf of itself or third party beneficiaries of this Agreement, including the Company's stockholders) will be deemed cumulative with and not exclusive of any other remedy conferred hereby, or by law or equity upon such party, and the exercise by a party of any one remedy will not preclude the exercise of any other remedy. The parties hereto agree that irreparable damage would occur in the event any provision of this Agreement were not performed in accordance with the terms hereof and that the parties shall be entitled to an injunction, specific performance of the terms hereof or other equitable relief, in addition to any other remedy at Law or equity. Each of the parties agrees that it will not oppose the granting of an injunction, specific performance and other equitable relief sought in accordance with this Section 10.11 on the basis that any other party has an adequate remedy at Law or that any award of specific performance is not an appropriate remedy for any reason at Law or in equity. Any party seeking an injunction or injunctions to prevent breaches of this Agreement and to enforce specifically the terms and provisions hereof in any court having jurisdiction related to this Agreement as provided in Section 10.14 without the necessity of demonstrating damages or posting a bond, this being in addition to any other remedy to which they are entitled at law or in equity.
- 10.12 Assignment. No party hereto may assign either this Agreement or any of its rights, interests, or obligations hereunder without the prior written approval of the other parties. Subject to the preceding sentence, this Agreement shall be binding upon and shall inure to the benefit of the parties hereto and their respective successors and permitted assigns.
- 10.13 Applicable Law. This Agreement shall be governed by and construed and enforced in accordance with the Laws of the State of New York, without giving effect to any choice of law or conflict of law provision or rule that would cause the application of the Laws of any other jurisdiction.

10.14 CONSENT TO JURISDICTION AND SERVICE OF PROCESS. EACH PARTY HERETO CONSENTS TO THE EXCLUSIVE JURISDICTION OF ANY STATE OR FEDERAL COURT LOCATED WITHIN THE COUNTY OF NEW YORK IN THE STATE OF NEW YORK IN THE BOROUGH OF MANHATTAN AND IRREVOCABLY AGREES THAT ALL ACTIONS OR PROCEEDINGS RELATING TO THE FINANCING AGREEMENT OR THE TRANSACTIONS OR THEREBY MAY BE LITIGATED ONLY IN SUCH COURTS. EACH PARTY HERETO ACCEPTS FOR ITSELF AND IN CONNECTION WITH ITS RESPECTIVE PROPERTIES, GENERALLY AND UNCONDITIONALLY, THE JURISDICTION OF SUCH COURTS AND WAIVES ANY DEFENSE OF FORUM NON CONVENIENS, AND IRREVOCABLY AGREES TO BE BOUND BY ANY JUDGMENT RENDERED THEREBY IN CONNECTION WITH THE TRANSACTIONS. EACH PARTY HERETO IRREVOCABLY CONSENTS TO THE SERVICE OF PROCESS OUT OF ANY OF SUCH COURTS IN ANY SUCH ACTION OR PROCEEDING BY THE MAILING OF COPIES THEREOF BY REGISTERED OR CERTIFIED MAIL, POSTAGE PREPAID, TO SUCH PARTY AT THE ADDRESS SPECIFIED IN THIS AGREEMENT, SUCH SERVICE TO BECOME EFFECTIVE 15 CALENDAR DAYS AFTER SUCH MAILING. NOTHING HEREIN SHALL IN ANY WAY BE DEEMED TO LIMIT THE ABILITY OF EITHER PARTY HERETO TO SERVE ANY SUCH LEGAL PROCESS, SUMMONS, NOTICES AND DOCUMENTS IN ANY OTHER MANNER PERMITTED BY APPLICABLE LAW.

10.15 WAIVER OF JURY TRIAL. EACH OF PARENT, MERGER SUB AND THE COMPANY HEREBY IRREVOCABLY WAIVES ALL RIGHT TO TRIAL BY JURY IN ANY ACTION, PROCEEDING OR COUNTERCLAIM (WHETHER BASED ON CONTRACT, TORT OR OTHERWISE) ARISING OUT OF OR RELATING TO THIS AGREEMENT OR THE FINANCING AGREEMENT OR THE ACTIONS OF BUYER, MERGER SUB, THE FINANCING SOURCES UNDER THE FINANCING AGREEMENT OR THE COMPANY IN THE NEGOTIATION, ADMINISTRATION, PERFORMANCE AND ENFORCEMENT HEREOF.

Signature page to follow

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be executed by their duly authorized respective officers as of the date first written above.

ACI WORLDWIDE, INC.

By:

/s/ Philip G. Heasley
Name: Philip G. Heasley
Title: Chief Executive Officer

ANTELOPE INVESTMENT CO. LLC

By: /s/ Dennis P. Byrnes

Name: Dennis P. Byrnes Title: President

S1 CORPORATION

/s/ Johann Dreyer Name: Johann Dreyer Title: CEO

 $Signature\ Page\ to\ Transaction\ Agreement$

CONDITIONS TO THE OFFER

- 1. Merger Sub will not be required to accept for payment or, subject to any applicable rules and regulations of the SEC, including Rule 14e-1(c) under the Exchange Act (relating to Merger Sub's obligation to pay for or return tendered Shares promptly after the termination or withdrawal of the Offer), to pay for any Company Shares tendered pursuant to the Offer, unless, immediately prior to the then-applicable Expiration Time:
 - (a) Company stockholders have validly tendered and not properly withdrawn prior to the Expiration Time at least that number of Company Shares (together with the Company Shares then owned by Parent, Merger Sub or any of Parent's other Subsidiaries), as shall constitute a majority of the Company Shares issued and outstanding on a fully diluted basis on the date of purchase (the "Minimum Tender Condition"):
 - (b) the Registration Statement has been declared effective under the Securities Act, and no stop order suspending the effectiveness of the Registration Statement has been issued or proceedings for that purpose have been initiated or threatened by the SEC, and Parent has received all necessary state securities law or "blue sky" authorizations; and
 - (c) the HSR Condition shall have been satisfied.
- 2. Additionally, Merger Sub will not be required to accept for payment or, subject to any applicable rules and regulations of the SEC, including Rule 14e-1(c) under the Exchange Act (relating to Merger Sub's obligation to pay for or return tendered Shares promptly after the termination or withdrawal of the Offer), to pay for any Company Shares tendered pursuant to the Offer if, immediately prior to the then-applicable Expiration Time, any of the following conditions exists:
 - (a) a Governmental Entity shall have enacted, issued, promulgated, enforced or entered any Law (including any injunction or other order, whether temporary, preliminary or permanent) which is in effect and which has the effect of making the Offer or the Merger illegal or otherwise prohibiting consummation of the Offer or the Merger;
 - (i) any clearance, approval, permit, authorization, waiver, determination, favorable review or consent of any Governmental Entity, other than the HSR Condition, has not been obtained or shall not be in full force and effect, or any applicable waiting periods for such clearances or approvals shall not have expired, except for any failures that would not reasonably be expected to have a Material Adverse Effect on Parent or the Company; or
 - (b) any of the following fail to be true:
 - (i) (A) the representations and warranties of the Company set forth in Sections 4.1, 4.3, 4.4(a), 4.7, 4.18 and 4.20 shall have been true and correct as of the date of this Agreement and shall be true and correct on and as of the Expiration Time with the same force and effect as if made at the Expiration Time (in either case other than those representations and warranties which address matters only as of a particular date, which representations and warranties shall have been true and correct as of such particular date), except in either case contemplated by this clause (A) for de minimis inaccuracies and (B) the other representations and warranties of the Company set forth in this Agreement shall have been true and correct as of the date of this Agreement and shall be true and correct on and as of the Expiration Time with the same force and effect as if made on the Expiration Time (in either case other than those representations and warranties which address matters only as of a particular date, which representations shall have been true and correct as of such particular date), except in either case contemplated by this clause (B) where the failure of such representations and warranties to be true and correct (disregarding all qualifications or limitations as to materiality, Material Adverse Effect or words of similar import set forth therein) has not had and would not reasonably be expected to have a Material Adverse Effect on the Company;

- (ii) the Company shall have performed or complied in all material respects with all agreements and covenants required by this Agreement to be performed or complied with by it on or prior to the Expiration Time; and
 - (iii) since the date of this Agreement, there shall not have occurred any Material Adverse Effect with respect to the Company.

The foregoing conditions are for the sole benefit of Merger Sub and Parent and may be asserted by Merger Sub or Parent regardless of the circumstances giving rise to any such condition, in whole or in part at any applicable time or from time to time in their sole discretion prior to the expiration of the Offer, except that the conditions relating to receipt of any approvals from any Governmental Entity may be asserted at any time prior to the acceptance for payment of Company Shares, and all conditions (except for the Minimum Tender Condition) may be waived by Parent or Merger Sub in their sole discretion in whole or in part at any applicable time or from time to time, in each case subject to the terms and conditions of the Agreement and the applicable rules and regulations of the SEC. The failure of Parent or Merger Sub at any time to exercise any of the foregoing rights shall not be deemed a waiver of any such right and each such right shall be deemed an ongoing right that may be asserted at any time and from time to time.

 $Capitalized \ terms \ used \ in \ this \ \underline{Exhibit \ A} \ and \ not \ otherwise \ defined \ shall \ have \ the \ respective \ meanings \ assigned \ thereto \ in \ this \ Agreement.$

Manually signed facsimile copies of the letter of election and transmittal will be accepted. The letter of election and transmittal and certificates for S1 Shares and any other required documents should be sent to the exchange agent at one of the addresses set forth below:

The exchange agent for the Exchange Offer is:



By Mail

Wells Fargo Bank, N.A. Shareowner Services Voluntary Corporate Actions P.O. Box 64854 St. Paul, MN 55164-0854 For Notice of Guaranteed Delivery

(For Eligible Institutions Only) By Facsimile Transmission: (866) 734-9952 (FAX)

To Confirm Receipt of Notice of Guaranteed Delivery Only: (800) 468-9716

By Hand or Overnight Delivery:

(Until 5:00 p.m. Eastern Time at the Expiration Time) Wells Fargo Bank, N.A. Shareowner Services Voluntary Corporate Actions 161 N. Concord Exchange South St. Paul, MN 55075-1139

Any questions or requests for assistance may be directed to the information agent at the address or telephone numbers set forth below. Additional copies of this prospectus/offer to exchange, the letter of election and transmittal and the Notice of Guaranteed Delivery may be obtained from the information agent at its address and telephone numbers set forth below. Holders of S1 Shares may also contact their brokers, dealers, commercial banks or trust companies or other nominees for assistance concerning the Exchange Offer.

The information agent for the Exchange Offer is:



501 Madison Avenue, 20th Floor New York, New York 10022 Stockholders May Call Toll Free: (888) 750-5834 Banks and Brokers May Call Collect: (212) 750-5833

Until the Expiration Time, or any subsequent offering period, all dealers that effect transactions in these securities, whether or not participating in this offering, may be required to deliver a prospectus/offer to exchange. This is in addition to the dealers' obligation to deliver a prospectus when acting as underwriters and with respect to their unsold allotments or subscriptions.

PART II — INFORMATION NOT REQUIRED IN PROSPECTUS/OFFER TO EXCHANGE

Item 20. Indemnification of Directors and Officers.

Section 145 of the General Corporation Law of the State of Delaware (the "DGCL") permits indemnification by a corporation of certain officers, directors, employees and agents. Consistent therewith, Article Tenth of the Amended and Restated Certificate of Incorporation of the registrant, ACI Worldwide, Inc. ("ACI" or the "Registrant"), provides that ACI shall, to the fullest extent permitted or required by the DGCL, as the same exists or may hereafter be amended (but, in the case of any such amendment, only to the extent that such amendment permits ACI to provide broader indemnification rights than such law permitted ACI to provide prior to such amendment), indemnify a director or officer of ACI or a person who is or was serving at the request of ACI as director, officer, employee or agent of another corporation or of a partnership, joint venture, trust or other enterprise, including service with respect to an employee benefit plan, who was or is made (or threatened to be made) a party to or is otherwise involved in a civil, criminal, administrative or investigative action suit or proceeding (an "indemnified person"). Article Tenth also provides that expenses incurred by an indemnified person will be paid in advance by ACI; provided, however, that, if the DGCL requires, an advancement of expenses incurred by an indemnified person in his or her capacity as a director or officer will be made only if ACI receives an undertaking by or on behalf of the indemnified person to repay all amounts so advanced if it shall ultimately be determined by final judicial decision from which there is no further right to appeal that such indemnified person is not entitled to be indemnified person. Part and the property is a policy is currently in effect.

ACI has entered into Indemnification Agreements with each of its executive officers and certain other employees. Under the Indemnification Agreements, ACI agrees to indemnify the employee to the fullest extent permitted by law if the employee was, is or becomes a party to or witness or other participant in any threatened, pending or completed action, suit, proceeding or alternative dispute resolution mechanism, or any hearing, inquiry or investigation by reason of (or arising in part out of) any event or occurrence related to the fact that the employee is or was a director, officer, employee, agent or fiduciary of ACI, or any subsidiary of ACI, or is or was serving at the request of ACI as a director, officer, employee, agent or fiduciary of another corporation, partnership, joint venture, trust or other enterprise, or by reason of any action or inaction on the part of the employee while serving in such capacity. ACI also agrees, to the extent S1 maintains liability insurance applicable to directors, officers, employees, agents or fiduciaries, the employee will be covered by such policies as to provide the employee the same rights and benefits as are accorded to the most favorably similarly situated insured.

The above discussion of the DGCL and the Registrant's Amended and Restated Certificate of Incorporation is not intended to be exhaustive and is qualified in its entirety by such statute and Amended and Restated Certificate of Incorporation.

Item 21. Exhibits and Financial Statement Schedules.

(a) Exhibits.

See the Exhibit Index.

(b) Financial Statement Schedules.

None

(c) Reports, Opinions and Appraisals.

None.

Item 22. Undertakings

- (a) The undersigned registrant hereby undertakes:
- (a) (1) To file, during any period in which offers or sales are being made, a post-effective amendment to this registration statement:
 - (i) To include any prospectus/offer to exchange required by Section 10(a)(3) of the Securities Act of 1933;
- (ii) To reflect in the prospectus/offer to exchange any facts or events arising after the effective date of the registration statement (or the most recent post-effective amendment thereof) which, individually or in the aggregate, represent a fundamental change in the information set forth in the registration statement. Notwithstanding the foregoing, any increase or decrease in volume of securities offered (if the total dollar value of securities offered would not exceed that which was registered) and any deviation from the low or high end of the estimated maximum offering range may be reflected in the form of prospectus/offer to exchange filed with the SEC pursuant to Rule 424(b) promulgated under the Securities Act if, in the aggregate, the changes in volume and price represent no more than a 20 percent change in the maximum aggregate offering price set forth in the "Calculation of Registration Fee" table in the effective registration statement; and
- (iii) To include any material information with respect to the plan of distribution not previously disclosed in the registration statement or any material change to such information in the registration statement.
- (2) That, for the purpose of determining any liability under the Securities Act, each such post-effective amendment shall be deemed to be a new registration statement relating to the securities offered therein, and the offering of such securities at that time shall be deemed to be the initial bona fide offering thereof.
 - (3) To remove from registration by means of a post-effective amendment any of the securities being registered which remain unsold at the termination of the offering.
- (4) That, for the purpose of determining liability under the Securities Act, each prospectus filed pursuant to Rule 424(b) as part of a registration statement relating to an offering, other than registration statements relying on Rule 430B or other than prospectuses filed in reliance on Rule 430A, shall be deemed to be part of and included in the registration statement as of the date it is first used after effectiveness. Provided, however, that no statement made in a registration statement or prospectus that is part of the registration statement or made in a document incorporated or deemed incorporated by reference into the registration statement or prospectus that is part of the registration statement will, as to a purchaser with a time of contract of sale prior to such first use, supersede or modify any statement that was made in the registration statement or prospectus that was part of the registration statement or made in any such document immediately prior to such date of first use.
- (5) That, for the purpose of determining liability under the Securities Act, the undersigned registrant undertakes that in a primary offering of securities of the undersigned registrant pursuant to this registration statement, regardless of the underwriting method used to sell the securities to the purchaser, if the securities are offered or sold to such purchaser by means of any of the following communications, the undersigned registrant will be a seller to the purchaser and will be considered to offer or sell such securities to such purchaser:
 - (i) Any preliminary prospectus or prospectus of the undersigned registrant relating to the offering required to be filed pursuant to Rule 424;
 - (ii) Any free writing prospectus relating to the offering prepared by or on behalf of the undersigned registrant or used or referred to by the undersigned registrant;
 - (iii) The portion of any other free writing prospectus relating to the offering containing material information about the undersigned registrant or its securities provided by or on behalf of the undersigned registrant; and

- (iv) Any other communication that is an offer in the offering made by the undersigned registrant to the purchaser.
- (b) That, for purposes of determining any liability under the Securities Act, each filing of the registrant's annual report pursuant to Section 13(a) or 15(d) of the Securities Exchange of 1934 (and, where applicable, each filing of an employee benefit plan's annual report pursuant to Section 15(d) of the Exchange Act of 1934) that is incorporated by reference in the registration statement shall be deemed to be a new registration statement relating to the securities offered therein, and the offering of such securities at that time shall be deemed to be the initial bona fide offering thereof.
- (c) (1) That prior to any public reoffering of the securities registered hereunder through use of a prospectus which is a part of this registration statement, by any person or party who is deemed to be an underwriter within the meaning of Rule 145(c), the issuer undertakes that such reoffering prospectus will contain the information called for by the applicable registration form with respect to reofferings by persons who may be deemed underwriters, in addition to the information called for by the other Items of the applicable form.
- (2) That every prospectus (i) that is filed pursuant to paragraph (h)(1) immediately preceding, or (ii) that purports to meet the requirements of section 10(a)(3) of the Act and is used in connection with an offering of securities subject to Rule 415, will be filed as a part of an amendment to the registration statement and will not be used until such amendment is effective, and that, for purposes of determining any liability under the Securities Act of 1933, each such post-effective amendment shall be deemed to be a new registration statement relating to the securities offered therein, and the offering of such securities at that time shall be deemed to be the initial bona fide offering thereof.
- (d) Insofar as indemnification for liabilities arising under the Securities Act of 1933 may be permitted to directors, officers and controlling persons of the registrant pursuant to the foregoing provisions, or otherwise, the registrant has been advised that in the opinion of the SEC such indemnification is against public policy as expressed in the Securities Act and is, therefore, unenforceable. In the event that a claim for indemnification against such liabilities (other than the payment by the registrant of expenses incurred or paid by a director, officer or controlling person of the registrant in the successful defense of any action, suit or proceeding) is asserted by such director, officer or controlling person in connection with the securities being registered, the registrant will, unless in the opinion of its counsel the matter has been settled by controlling precedent, submit to a court of appropriate jurisdiction the question whether such indemnification by it is against public policy as expressed in the Securities Act and will be governed by the final adjudication of such issue.
- (e) To respond to requests for information that are incorporated by reference into the prospectus/offer to exchange pursuant to Item 4, 10(b), 11, or 13 of this form, within one business day of receipt of such request, and to send the incorporated documents by first class mail or other equally prompt means. This includes information contained in documents filed subsequent to the effective date of the Registration Statement through the date of responding to the request.
- (f) To supply by means of a post-effective amendment all information concerning a transaction, and the company being acquired involved therein, that was not the subject of and included in the Registration Statement when it became effective.

SIGNATURES

Pursuant to the requirements of the Securities Act of 1933, the Registrant has duly caused this Post-Effective Amendment No. 1 to the Registration Statement on Form S-4 to be signed on its behalf by the undersigned, thereunto duly authorized, in the City of New York, State of New York, on October 13, 2011.

ACI WORLDWIDE, INC.

/s/ Dennis P. Byrnes

Name: Dennis P. Byrnes
Title: Executive Vice President, General Counsel and Secretary

Pursuant to the requirements of the Securities Act of 1933, as amended, this Post-Effective Amendment No. 1 to the Registration Statement has been signed below by Dennis P. Byrnes for himself and as attorney-in-fact for each person named below in the capacities indicated on October 13, 2011:

<u>Signature</u>	Title
* Philip G. Heasley	President, Chief Executive Officer and Director (Principal Executive Officer)
* Scott W. Behrens	Senior Vice President, Chief Financial Officer and Chief Accounting Officer (Principal Financial Officer and Principal Accounting Officer)
*	Chairman of the Board of Directors and a Director
Harlan F. Seymour	
*	Director
Jan H. Suwinski	
*	Director
John D. Curtis	
*	Director
John M. Shay, Jr.	
*	Director
Alfred R. Berkeley, III	
*	Director
John E. Stokely	
*	Director
James C. McGroddy	
*By: /s/ Dennis P. Byrnes Name: Dennis P. Byrnes Attorney-in-Fact	

EXHIBIT INDEX

Exhibit Number	<u>Description</u>
3.1	Amended and Restated Certificate of Incorporation (incorporated by reference to Registrant's Current Report on Form 8-K filed July 30, 2007)
3.2	Amended and Restated Bylaws (incorporated by reference to Registrant's Current Report on Form 8-K filed December 18, 2008)
4.1	Form of Common Stock Certificate (incorporated by reference to Registrant's Registration Statement No. 33-88292 on Form S-1)
5.1**	Opinion of Jones Day regarding the validity of the Shares of Registrant's Common Stock to be issued pursuant to the Exchange Offer
21.1	List of Subsidiaries (incorporated by reference to Exhibit 21 to Registrant's Annual Report on Form 10-K for the year ended December 31, 2010, filed with the SEC on February 18, 2011)
23.1*	Consent of Deloitte & Touche LLP, an independent registered public accounting firm
23.2*	Consent of KPMG LLP, an independent registered public accounting firm
23.3*	Consent of PricewaterhouseCoopers LLP, an independent registered public accounting firm
23.4**	Consent of Jones Day (included in the opinion filed as Exhibit 5.1)
24.1**	Power of Attorney
99.1*	Form of Letter of Election and Transmittal
99.2*	Form of Notice of Guaranteed Delivery
99.3*	Form of Letter to Brokers, Dealers, Commercial Banks, Trust Companies and Other Nominees
99.4*	Form of Letter to Clients for Use by Brokers, Dealers, Commercial Banks, Trust Companies and Other Nominees

^{*} Filed herewith.

^{**} Previously filed

Consent of Independent Registered Public Accounting Firm

We consent to the incorporation by reference in this Post-Effective Amendment No. 1 to Registration Statement No. 333-176557 on Form S-4 of our reports dated February 18, 2011, relating to the consolidated financial statements of ACI Worldwide, Inc. and subsidiaries (the "Company") as of and for the years ended December 31, 2010 and 2009, and the effectiveness of the Company's internal control over financial reporting as of December 31, 2010, appearing in the Annual Report on Form 10-K of the Company for the year ended December 31, 2010, and to the reference to us under the heading "Experts" in the Prospectus, which is part of such Registration Statement.

/s/ Deloitte & Touche LLP

Omaha, Nebraska October 13, 2011

Consent of Independent Registered Public Accounting Firm

The Board of Directors ACI Worldwide, Inc.:

We consent to the use of our report dated March 3, 2009, with respect to the consolidated statements of operations, stockholders' equity and comprehensive income (loss), and cash flows of ACI Worldwide, Inc. and subsidiaries for the year ended December 31, 2008, incorporated herein by reference and to the reference to our firm under the heading "Experts" in the prospectus.

/s/ KPMG LLP

Omaha, Nebraska October 13, 2011

Consent of Independent Registered Public Accounting Firm

We hereby consent to the incorporation by reference in this Post-Effective Amendment No. 1 to Registration Statement No. 333-176557 on Form S-4 of ACI Worldwide, Inc. of our report dated March 11, 2011 relating to the financial statements, financial statement schedule and the effectiveness of internal control over financial reporting, which appears in S1 Corporation's Annual Report on Form 10-K for the year ended December 31, 2010. We also consent to the reference to us under the heading "Experts" in such Registration Statement.

/s/ PricewaterhouseCoopers LLP

Atlanta, Georgia October 13, 2011

LETTER OF ELECTION AND TRANSMITTAL

To Offer to Exchange Shares of Common Stock

S1 CORPORATION

for 0.3148 of a Share of ACI Common Stock

\$10.00 in Cash,

subject to the proration procedures described in the Prospectus/Offer to Exchange dated October 13, 2011 and this Letter of Election and Transmittal,

by

ANTELOPE INVESTMENT CO. LLC,

a wholly owned subsidiary of

ACI WORLDWIDE, INC.

THE EXCHANGE OFFER AND THE WITHDRAWAL RIGHTS WILL EXPIRE AT 5:00 P.M., EASTERN TIME, ON MONDAY, OCTOBER 31, 2011, UNLESS EXTENDED. SI SHARES TENDERED PURSUANT TO THE EXCHANGE OFFER MAY BE WITHDRAWN AT ANY TIME PRIOR TO THE EXPIRATION OF THE EXCHANGE OFFER, BUT NOT DURING ANY SUBSEQUENT OFFERING PERIOD.

DESCRIPTION OF SHARES TENDERED				
(see Instruction(s) 3 and 4)				
(See Instruction(3) 3 and 4)				
	(4 1	Shares Tendered		
		(Attach additional list, if necessary)		
	Share Certificate	Total Number		
Name(s) and Address(es) of Registered Holder(s)	Number(s) and/or	of Shares	Number of	
Manies) and Address(es) in Registered Houter(s) (Please fill in, if blank, exactly as name(s) appear	indicate Book-Entry		Shares	
(rease firm), transactives name(s) appear	Shares*	Represented by Certificate(s)*	Tendered**	
		0.00.00.00(0)		
Total Shares				

LOST CERTIFICATES: PLEASE CALL THE TRANSFER AGENT, AMERICAN STOCK TRANSFER & TRUST COMPANY, AT 800-937-5449 TO OBTAIN THE NECESSARY DOCUMENTS TO REPLACE YOUR LOST CERTIFICATES

^{*} If shares are held in book-entry form, you must indicate the number of shares you are tendering.

^{**} Unless otherwise indicated, it will be assumed that all shares represented by any certificates delivered to the Exchange Agent (as defined below) are being tendered. See Instruction 4.



By Mail
Wells Fargo Bank, N.A.
Shareowner Services
Voluntary Corporate Actions
P.O. Box 64854
St. Paul, MN 55164-0854

By Hand or Overnight Delivery: (Until 5:00 p.m. Eastern Time at the Expiration Time) Wells Fargo Bank, N.A. Shareowner Services Voluntary Corporate Actions 161 N. Concord Exchange South St. Paul, MN 55075-1139

DELIVERY OF THIS LETTER OF ELECTION AND TRANSMITTAL TO AN ADDRESS, OR TRANSMISSION OF INSTRUCTIONS VIA A FACSIMILE NUMBER, OTHER THAN AS SET FORTH BELOW, WILL NOT CONSTITUTE A VALID DELIVERY.

THE INSTRUCTIONS ACCOMPANYING THIS LETTER OF ELECTION AND TRANSMITTAL SHOULD BE READ CAREFULLY AND IN THEIR ENTIRETY BEFORE THIS LETTER OF ELECTION AND TRANSMITTAL IS COMPLETED.

This Letter of Election and Transmittal is to be used for the exchange of shares of common stock, par value \$0.01 per share (the "S1 Shares"), of S1 Corporation, a Delaware corporation ("S1"). Tendering S1 stockholders may use this form if certificates evidencing S1 Shares are to be forwarded herewith or, unless an Agent's Message (as defined in Instruction 2 below) is utilized, if delivery of S1 Shares is to be made by book-entry transfer to the account of Wells Fargo Bank, N.A. (the "Exchange Agent") at the book-entry transfer facility pursuant to the procedures set forth in the section of the prospectus/offer to exchange dated October 13, 2011 (the "Prospectus/Offer to Exchange") titled "The Exchange Offer — Procedure for Tendering." All capitalized terms used herein but not defined herein shall have the meanings ascribed to them in the Prospectus/Offer to Exchange

Holders whose certificates evidencing S1 Shares are not immediately available or who cannot deliver their certificates evidencing S1 Shares and all other required documents to the Exchange Agent on or prior to the expiration time of the Exchange Offer (as defined herein), or who cannot complete the procedure for book-entry transfer on a timely basis, must tender their S1 Shares according to the guaranteed delivery procedures set forth in the section of the Prospectus/Offer to Exchange titled "The Exchange Offer — Procedure for Tendering." See Instruction 2 below. Delivery of documents to the book-entry transfer facility does not constitute delivery to the Exchange Agent.

The undersigned hereby tenders for exchange the S1 Shares described under "Description of Shares Tendered" below pursuant to the terms and conditions of this Letter of Election and Transmittal and the Prospectus/Offer to Exchange. Such tenders are subject to the elections specified below under the heading "Exchange Offer Election" (the "Exchange Offer Elections"), or if no Exchange Offer Election is specified, will be deemed tendered with no election. See "The Exchange Offer — Consequences of Tendering with No Election" in the Prospectus/Offer to Exchange.

EXCHANGE OFFER ELECTION

(See Instruction 13)

NUMBER OF SHARES TENDERED FOR 0.3148 OF AN ACI SHARE, SUBJECT TO PRORATION AND THE PAYMENT OF CASH IN RESPECT OF FRACTIONAL ACI SHARES. NUMBER OF SHARES TENDERED FOR \$10.00 IN CASH, WITHOUT INTEREST, SUBJECT TO PRORATION.

ANY SHARES TENDERED BUT FOR WHICH NO ELECTION IS MADE ABOVE WILL BE DEEMED TO HAVE BEEN TENDERED WITHOUT AN ELECTION, AND THE TENDERED SHARES WILL BE TREATED AS DESCRIBED IN "THE EXCHANGE OFFER — CONSEQUENCES OF TENDERING WITH NO ELECTION" IN THE PROSPECTUS/OFFER TO EXCHANGE.

LOST CERTIFICATES: PLEASE CALL THE TRANSFER AGENT, AMERICAN STOCK TRANSFER, AT 800-937-5449 TO OBTAIN THE NECESSARY DOCUMENTS TO REPLACE YOUR LOST CERTIFICATES.

PLEASE READ THE ACCOMPANYING INSTRUCTIONS CAREFULLY AND IN THEIR ENTIRETY.

NOTE: THIS LETTER OF ELECTION AND TRANSMITTAL MUST BE SIGNED ON PAGE 9 BELOW.

PLEASE READ ACCOMPANYING INSTRUCTIONS CAREFULLY.

Ladies and Centlemen

The undersigned hereby tenders to Antelope Investment Co. LLC ("Offeror"), a Delaware limited liability company and a wholly owned subsidiary of ACI Worldwide, Inc., a Delaware corporation ("ACI"), the above described shares of common stock, par value \$0.01 per share (the "S1 Shares"), of S1 Corporation, a Delaware corporation ("S1"), pursuant to Offeror's offer to exchange for each outstanding S1 Share validly tendered and not properly withdrawn in the Exchange Offer (as defined below), at the election of the holder of such S1 Shares, either of the following:

- 0.3148 of a share of ACI common stock (the "ACI Shares"), par value \$0.005 per share (the "Stock Consideration"); or
- \$10.00 in cash, without interest (the "Cash Consideration"),

subject to the proration procedures described in the Prospectus/Offer to Exchange and this Letter of Election and Transmittal (together, as each may be amended, supplemented or otherwise modified from time to time, the "Exchange Offer").

The Exchange Offer is being made pursuant to the Transaction Agreement, dated as of October 3, 2011, among ACI, Offeror and S1 (as amended, the "Transaction Agreement"). The S1 Board unanimously recommends that the S1 stockholders accept the Exchange Offer by tendering their S1 Shares in the Exchange Offer.

S1 stockholders electing either the Cash Consideration or the Stock Consideration will be subject to proration such that not more than 66.2% of the S1 Shares tendered in the Exchange Offer can be exchanged for cash and not more than 33.8% of the S1 Shares tendered in the Exchange Offer can be exchanged for ACI Shares. S1 stockholders who otherwise would be entitled to receive a fractional ACI Share will instead receive cash in lieu of any fractional ACI Share such holder may have otherwise been entitled to receive. See "The Exchange Offer — Elections and Proration" in the Prospectus/Offer to Exchange for a detailed description of the proration procedure and "The Exchange Offer — Cash In Lieu of Fractional ACI Shares" in the Prospectus/Offer to Exchange for a detailed description of the treatment of fractional ACI Shares. The Prospectus/Offer to Exchange and this Letter of Election and Transmittal more fully describe the Exchange Offer. Capitalized terms used but not defined herein have the respective meanings given to them in the Prospectus/Offer to Exchange.

The undersigned elects to have his, her or its S1 Shares purchased or exchanged pursuant to one or more of the foregoing options, as indicated in the "Exchange Offer Election" section above.

IF THE UNDERSIGNED FAILS TO PROPERLY MAKE AN EXCHANGE OFFER ELECTION, THE UNDERSIGNED WILL BE DEEMED TO HAVE TENDERED HIS, HER OR ITS S1 SHARES WITH NO ELECTION, AND WILL BE DEEMED TO HAVE ELECTED THE CASH CONSIDERATION. SEE "THE EXCHANGE OFFER — CONSEQUENCES OF TENDERING WITH NO ELECTION" IN THE PROSPECTUS/OFFER TO EXCHANGE.

Upon the terms and subject to the conditions of the Exchange Offer and effective upon acceptance of the S1 Shares tendered herewith in accordance with the terms of the Exchange Offer, the undersigned hereby sells, assigns and transfers to, or upon the order of, Offeror all right, title and interest in and to all of the S1 Shares that are being tendered hereby (and any and all S1 Shares or other securities issued, paid or distributable in respect of such S1 Shares on or after October 3, 2011 (collectively, a "Distribution"), and irrevocably appoints the Exchange Agent the true and lawful agent, attorney-in-fact and proxy of the undersigned with respect to such S1 Shares (and any Distribution), with full power of substitution (such power of attorney being deemed to be an irrevocable power coupled with an interest), to (1) deliver Share Certificates (and any Distribution), or transfer ownership of such Shares (and any Distribution) on the account books maintained by the book-entry transfer facility, together, in any such case, with all accompanying evidences of transfer and authenticity, to or upon the order of ACI, (2) present such S1 Shares (and any Distribution) for transfer on the books of S1, and (3) receive all benefits and otherwise exercise all rights of beneficial ownership of such S1 Shares (and any Distribution), all in accordance with the terms of the Exchange Offer.

The undersigned hereby irrevocably appoints designees of Offeror as the agents, attorneys and proxies of the undersigned, each with full power of substitution, to exercise to the full extent the rights of the undersigned with respect to all of the S1 Shares tendered hereby (and any Distribution) which have been accepted for exchange by Offeror prior to

the time of any vote or other action. The assignees of Offeror will, for which such appointment is effective, be empowered to exercise such rights of the undersigned in such manner as each such agent, attorney and proxy or his or her substitute shall in his or her sole discretion deem proper, by written consent or otherwise. This proxy is irrevocable and shall be considered coupled with an interest and is granted in consideration of, and is effective upon, the acceptance for exchange of such S1 Shares by Offeror in accordance with the terms of the Exchange Offer. Such acceptance for exchange shall revoke any other proxy or written consent granted by the undersigned at any time with respect to such S1 Shares (and any Distribution), and no subsequent proxies will be given or written consents will be executed by the undersigned (and if given or executed, will not be deemed to be effective). Offeror reserves the right to require that, in order for the S1 Shares to be deemed validly tendered, immediately upon Offeror's acceptance of such S1 Shares, Offeror must be able to exercise full voting rights with respect to such S1 Shares (and any Distribution), including without limitation, voting at any meeting of S1 stockholders, subject to applicable law.

The undersigned hereby represents and warrants that the undersigned has full power and authority to tender, sell, assign and transfer the S1 Shares tendered hereby (and any Distribution) and that, when the same are accepted for exchange by Offeror, Offeror will acquire good, marketable and unencumbered title thereto, free and clear of all liens, restrictions, charges and encumbrances and not subject to any adverse claims and will not be transferred to Offeror in violation of any contractual or other restriction on the transfer thereof. The undersigned will, upon request, execute and deliver any additional documents deemed by the Exchange Agent or Offeror to be necessary or desirable to complete the sale, assignment and transfer of the S1 Shares tendered hereby (and any Distribution).

All authority herein conferred or agreed to be conferred shall survive the death or incapacity of the undersigned, and any obligation of the undersigned hereunder shall be binding upon the heirs, personal representatives, successors and assigns of the undersigned. Except as stated in the Prospectus/Offer to Exchange, this tender is irrevocable.

Tender of S1 Shares pursuant to any one of the procedures described in the Prospectus/Offer to Exchange under the caption "The Exchange Offer — Procedure for Tendering" and in the instructions hereto will constitute a binding agreement between the undersigned and Offeror upon the terms and subject to the conditions of the Exchange Offer, including the undersigned's representation that the undersigned owns the S1 Shares being tendered. The undersigned recognizes that under certain circumstances set forth in the Prospectus/Offer to Exchange, Offeror may not be required to accept for exchange any of the S1 Shares tendered hereby.

The delivery and surrender of S1 Shares tendered by the undersigned are not effective, and the risk of loss of S1 Shares does not pass to the Exchange Agent, until the Exchange Agent receives this Letter of Election and Transmittal, duly completed and signed, or an agent's message (as discussed in the Prospectus/Offer to Exchange in the section tilled "The Exchange Offer — Procedure for Tendering") in connection with a book-entry transfer of S1 Shares, together with all accompanying evidences of authority in form satisfactory to Offeror and any other required documents. The undersigned understands that all questions as to the form of documents (including notices of withdrawal) and the validity, form, eligibility (including time of receipt) and acceptance for exchange of any tender of S1 Shares will be determined by Offeror in its sole discretion and such determination shall be final and binding upon all tendering S1 stockholders. No tender of S1 Shares is valid until all defects and irregularities in tenders of S1 Shares have been cured or waived and none of ACI, Offeror, the Exchange Agent, the Information Agent or any other person is under any duty to give notification of any defects or irregularities in the tender of any S1 Shares or will incur any liability for failure to give any such notification.

Unless otherwise indicated below under "Special Payment Instructions," the undersigned hereby requests that the ACI Shares and/or a check for cash (including any cash paid in lieu of fractional ACI Shares), and the return of any S1 Shares not tendered or not accepted for exchange, be issued in the name(s) of the undersigned (and, in the case of S1 Shares tendered by book-entry transfer, by credit to the applicable account at the book-entry transfer facility designated above as tendering such S1 Shares). The undersigned recognizes that Offeror has no obligation, pursuant to the "Special Payment Instructions," to transfer any S1 Shares from the name of the registered holder(s) thereof if Offeror does not accept for exchange any of the S1 Shares so tendered.

Similarly, unless otherwise indicated below under "Special Delivery Instructions," the undersigned hereby requests that the ACI Shares and/or a check for cash (including any cash paid in lieu of fractional ACI Shares) and any share certificates not tendered or not accepted for exchange (and accompanying documents, as appropriate) be mailed to the undersigned at the address shown above in "Description of Shares Tendered."

SPECIAL ISSUANCE INSTRUCTIONS

(See Instructions 1, 5, 6 and 7)

To be completed ONLY if the certificates for ACI Shares (or, at Offeror's election, evidence of book-entry of ACI Shares), the check for cash payable pursuant to the Exchange Offer (less applicable withholding taxes and without interest), if applicable, are to be issued in the name of someone other than the registered holder(s) listed above in the box titled "Description of Shares Tendered."

Issue as follows:

o Check o Share Certificate(s) to:

Name:

(Please Print)

Address:

(Zip Code)

Area Code and Telephone Number:

Taxpayer Identification Number or Social Security Number: (See Form W-9 attached; foreign stockholder see appropriate Form W-8)

SPECIAL DELIVERY INSTRUCTIONS

To be completed **ONLY** if the certificates for ACI Shares (or, at Offeror's election, evidence of book-entry of ACI Shares), the check for cash payable pursuant to the Exchange Offer (less applicable withholding taxes and without interest), if applicable, are to be sent to an address other than the address(es) of the registered holders(s) listed above in the box titled "Description of Shares Tendered."

Mail as follows

Name:	
_	(Please Print)
Address:	
	(Zip Code)

Area Code and Telephone Number:

INSTRUCTIONS FORMING PART OF THE TERMS AND CONDITIONS OF THE OFFER

- 1. Guarantee of Signatures. No signature guarantee is required on this Letter of Election and Transmittal if (1) the Letter of Election and Transmittal is signed by the registered holder(s) (which term, for purposes of this document, includes any participant in the book-entry transfer facility whose name appears on a security position listing as the owner of the shares) of \$1\$ Shares and such holder(s) have not completed either the box titled "Special Issuance Instructions" on this Letter of Election and Transmittal or (2) if \$1\$ Shares are tendered for the account of a financial institution that is a member of the Securities Transfer Agents Medallion Signature Program or by any other "Eligible Guarantor Institution," as such term is defined in Rule 17Ad-15 under the Securities Exchange Act of 1934, as amended (each of the foregoing being referred to as an "Eligible Institution"). In all other cases, all signatures on this Letter of Election and Transmittal must be guaranteed by an Eligible Institution. See Instruction 5.
- 2. Requirements of Tender. This Letter of Election and Transmittal is to be completed by S1 stockholders either if certificates evidencing S1 Shares are to be forwarded herewith or, unless an Agent's Message is utilized, if delivery of the S1 Shares is to be made by book-entry transfer pursuant to the procedures set forth herein and in the Prospectus/Offer to Exchange. For an S1 stockholder to validly tender S1 Shares pursuant to the Exchange Offer, either (1) a properly completed and duly executed Letter of Election and Transmittal (or a manually signed facisinile thereof), together with any required signature guarantees on Agent's Message (in connection with book-entry transfer of the shares) and any other required documents, must be received by the Exchange Agent at one of its addresses set forth herein prior to the expiration time of the Exchange Offer and either (a) certificates evidencing S1 Shares must be received by the Exchange Agent at one of such addresses prior to the expiration time of the Exchange Offer or (b) all S1 Shares delivered electronically must be delivered pursuant to the procedures for book-entry transfer set forth herein and in the Prospectus/Offer to Exchange, and a book-entry confirmation must be received by the Exchange.

Stockholders whose certificates evidencing S1 Shares are not immediately available or who cannot deliver their certificates and all other required documents to the Exchange Agent prior to the expiration time of the Exchange Offer or who cannot comply with the book-entry transfer procedures on a timely basis may tender their S1 Shares by properly completing and duly executing the Notice of Guaranteed Delivery pursuant to the guaranteed delivery procedures set forth herein and in the Prospectus/Offer to Exchange.

Pursuant to such guaranteed delivery procedures, (1) such tender must be made by or through an Eligible Institution, (2) a properly completed and duly executed Notice of Guaranteed Delivery, substantially in the form made available by ACI and Offeror, must be received by the Exchange Agent prior to the expiration time of the Exchange Offer and (3) certificates evidencing S1 Shares, in proper form for transfer (or a book-entry confirmation with respect to all tendered S1 Shares), together with a properly completed and duly executed Letter of Election and Transmittal (or a manually signed facsimile thereof), with any required signature guarantees, or, in the case of a book-entry transfer, an Agent's Message, and other documents required by this Letter of Election and Transmittal must be received by the Exchange Agent within three NASDAQ trading days after the date of execution of such Notice of Guaranteed Delivery. A "trading day" is any day on which the NASDAQ is open for business.

The term "Agent's Message" means a message, transmitted by the book-entry transfer facility to, and received by, the Exchange Agent and forming a part of the book-entry confirmation which states that the book-entry transfer facility has received an express acknowledgement from the participant in the book-entry transfer facility tendering the S1 Shares that are the subject of such book-entry confirmation, that such participant has received and agrees to be bound by the terms of this Letter of Election and Transmittal and that ACI may enforce such agreement against the participant.

THE METHOD OF DELIVERY OF THE SI SHARES, THIS LETTER OF ELECTION AND TRANSMITTAL AND ALL OTHER REQUIRED DOCUMENTS, INCLUDING DELIVERY THROUGH THE BOOK-ENTRY TRANSFER FACILITY, IS AT THE OPTION AND RISK OF THE TENDERING SI STOCKHOLDER, AND THE DELIVERY WILL BE DEEMED MADE ONLY WHEN ACTUALLY RECEIVED BY THE EXCHANGE AGENT. IF DELIVERY IS BY MAIL, REGISTERED MAIL WITH RETURN RECEIPT REQUESTED, PROPERLY INSURED, IS RECOMMENDED. IN ALL CASES, SUFFICIENT TIME SHOULD BE ALLOWED TO ENSURE TIMELY DELIVERY.

No alternative, conditional or contingent tenders will be accepted and no fractional S1 Shares will be accepted for tender or exchanged. By execution of this Letter of Election and Transmittal (or a manually signed facsimile hereof), all tendering S1 stockholders waive any right to receive any notice of the acceptance of their S1 Shares for exchange.

- 3. Inadequate Space. If the space provided herein under "Description of Shares Tendered" is inadequate, the number of S1 Shares tendered and the certificate numbers evidencing such S1 Shares, if applicable, should be listed on a separate signed schedule and attached hereto.
- 4. Partial Tenders. If fewer than all S1 Shares evidenced by any certificate delivered to the Exchange Agent are to be tendered hereby, fill in the number of S1 Shares that are to be tendered in the box titled "Number of Shares Tendered." In such cases, new certificates evidencing the remainder of the S1 Shares that were evidenced by the old certificates, but that were not tendered, will be sent to the registered holder(s) at the address(es) listed above in the box titled "Description of Shares Tendered" as soon as practicable after the expiration time of the Exchange Offer or the termination of the Exchange Offer. All S1 Shares evidenced by certificates delivered to the Exchange Agent will be deemed to have been tendered unless otherwise indicated.
- 5. Signatures on Letter of Election and Transmittal; Stock Powers and Endorsements. If this Letter of Election and Transmittal is signed by the registered holder(s) of S1 Shares tendered hereby, the signature(s) must correspond with the name(s) as written on the face of the certificates evidencing such S1 Shares without alteration, enlargement or any other change whatsoever.
 - If any S1 Shares tendered hereby are held of record by two or more persons, all such persons must sign this Letter of Election and Transmittal.
- If any S1 Shares tendered hereby are registered in different names, it will be necessary to complete, sign and submit as many separate Letters of Election and Transmittal as there are different registrations of such S1 Shares.

If this Letter of Election and Transmittal is signed by the registered holder(s) of S1 Shares tendered hereby, no endorsements of certificates evidencing S1 Shares or separate stock powers are required, unless the ACI Shares are to be issued to, or certificates evidencing S1 Shares not tendered or not accepted for exchange are to be issued in the name of, a person other than the registered holder(s). If this Letter of Election and Transmittal is signed by a person other than the registered holder(s) of the certificate(s) evidencing S1 Shares tendered, the certificate(s) tendered hereby must be endorsed or accompanied by appropriate stock powers, in either case signed exactly as the name(s) of the registered holder(s) appear(s) on such certificate(s). Signatures on such certificate(s) or stock powers must be guaranteed by an Eligible Institution.

If this Letter of Election and Transmittal or any certificate evidencing S1 Shares or stock power is signed by a trustee, executor, administrator, guardian, attorney-in-fact, officer of a corporation or other person acting in a fiduciary or representative capacity, such person should so indicate when signing, and proper evidence satisfactory to Offeror of such person's authority so to act must be submitted.

- 6. Stock Transfer Taxes. Except as otherwise provided in this Instruction 6, Offeror will pay all stock transfer taxes with respect to the transfer of any S1 Shares to it, or by its order, pursuant to the Exchange Offer. If, however, issuance of ACI Shares is to be made to, or certificates (s) evidencing S1 Shares not tendered or not accepted for exchange are to be issued in the name of, any person other than the registered holder(s) or if tendered certificates evidencing S1 Shares are registered in the name of any person other than the person(s) signing the Letter of Election and Transmittal, the amount of any stock transfer taxes (whether imposed on the registered holder(s), or such other person, or otherwise) payable on account of such issuance or transfer to such other person will be deducted from the consideration to be received by such stockholder for the exchange of such S1 Shares in the Exchange Offer, unless evidence satisfactory to Offeror of the payment of such taxes, or exemption therefrom, is submitted.
- 7. Special Issuance and Delivery Instructions. If certificates (or, at Offeror's election, evidence of book-entry transfer) for ACI Shares and, if applicable, any S1 Shares not tendered or not accepted for exchange or a check for cash (including any cash in lieu of fractional ACI Shares) are to be issued in the name of and/or returned to, a person other than the registered holder(s) listed above in the box titled "Description of Shares Tendered," the appropriate boxes on this Letter of Election and Transmittal should be completed.

- 8. Backup Withholding; Substitute Form W-9; Form W-8. Each tendering S1 stockholder is required to provide the Exchange Agent with the S1 stockholder's correct taxpayer identification number ("TIN"), generally the S1 stockholder's social security or federal employer identification number, on the Substitute Form W-9, which is provided under "Important Tax Information" below or, alternatively, to establish another basis for exemption from backup withholding. An S1 stockholder must cross out item (2) in the Certification box on Substitute Form W-9 if such S1 stockholder is subject to backup withholding. Failure to provide the information on the form may subject the tendering S1 stockholder to a \$50 penalty imposed by the Internal Revenue Service and to applicable federal income tax backup withholding on payments made to the S1 stockholder or other payee with respect to the Exchange Offer. The box in Part III of the form should be checked if the tendering S1 stockholder has not been issued a TIN and has applied for a TIN or intends to apply for a TIN in the near future. If the box in Part III is checked and the Exchange Agent is not provided with a TIN, the Exchange Agent will withhold the applicable backup withholding amount from all payments with respect to the S1 Shares to be purchased until a TIN is provided to the Exchange Agent. In any case, 60 days after the payment date the Exchange Agent will remit the withheld amount to the IRS. Each tendering S1 stockholder that is a foreign person, including entities, must submit an appropriate properly completed Internal Revenue Service Form W-8 certifying, under penalties of perjury, to such S1 stockholder's foreign status in order to establish an exemption from backup withholding. An appropriate form W-8 can be obtained from the Exchange Agent.
- 9. Mutilated, Lost, Stolen or Destroyed Certificates. If the certificate(s) representing S1 Shares to be tendered have been mutilated, lost, stolen or destroyed, stockholders should immediately contact American Stock Transfer & Trust Company, the transfer agent for S1 Shares. The transfer agent will provide such holders with all necessary forms and instructions to replace any mutilated, lost, stolen or destroyed certificates. This Letter of Election and Transmittal and related documents cannot be processed until the mutilated, lost, stolen or destroyed certificates have been replaced and the replacement certificates have been delivered to the Exchange Agent with this Letter of Election and Transmittal.
- 10. Waiver of Conditions. Except as described in the Prospectus/Offer to Exchange, the conditions of the Exchange Offer may be waived, in whole or in part, by Offeror, in its discretion, at any time and from time to time, in the case of any \$1 Shares tendered.
- 11. Questions and Requests for Assistance or Additional Copies. Questions or requests for assistance may be directed to the Information Agent at its address and telephone numbers as set forth on the back page of this Letter of Election and Transmittal. Requests for additional copies of the Prospectus/Offer to Exchange, this Letter of Election and Transmittal, the Notice of Guaranteed Delivery and the IRS Form W-9 may be directed to the Information Agent. Stockholders may also contact their brokers, dealers, commercial banks, trust companies or other nominees for assistance concerning the Exchange Offer.
- 12. Withdrawal of Tenders. A tender of S1 Shares may be withdrawn at any time prior to the expiration date for the Exchange Offer (or such later time as described in the Prospectus/Offer to Exchange in the section titled "The Exchange Offer Withdrawal Rights") by delivery to the Exchange Agent at its address set forth on the cover of this Letter of Election and Transmittal of a written or facsimile notice of withdrawal in accordance with the instructions set forth in the Prospectus/Offer to Exchange. Notices of withdrawal must be received by the Exchange Agent prior to the Expiration Time.
- 13. Election Procedure; Revocation or Change of Election. The "Exchange Offer Election" section must be completed if you desire to elect the type of consideration to be received in exchange for the S1 Shares being tendered hereby. If an election is not properly made, the tendering S1 stockholder will be deemed to have been tendered without an election, and the tendered S1 Shares will be deemed to have elected the Cash Consideration as described in the Prospectus/Office to Exchange in the section titled "The Exchange Offer Consequences of Tendering with No Election".

Exchange Offer elections (whether actual or deemed) are irrevocable, except that S1 Shares tendered pursuant to the Exchange Offer may be withdrawn at any time prior to the Expiration Time. After an effective withdrawal, S1 Shares may be retendered with another election or with no election by submitting to the Exchange Agent a completed replacement of this Letter of Election and Transmittal (and any other documents required in connection with the Exchange Offer for properly tendering S1 Shares) prior to the Expiration Time.

14. Irregularities. All questions as to the validity, form, eligibility (including time of receipt) and acceptance for exchange of any tender of S1 Shares and any notice of withdrawal will be determined by Offeror, in its discretion, which

determination shall be final and binding to the fullest extent permitted by law. Offeror reserves the absolute right to reject any and all tenders of S1 Shares determined by it to not be in proper form or the acceptance of or exchange for which may, in the opinion of Offeror's counsel, be unlawful. Offeror also reserves the absolute right to waive any defect or irregularity in the tender of any S1 Shares of any particular S1 stockholder, whether or not similar defects or irregularities are waived in the case of other S1 stockholders. No tender of S1 Shares will be deemed to be validly made until all defects and irregularities with respect thereto have been cured or waived. None of ACI, Offeror or any of their affiliates or assigns, the Information Agent, the Exchange Agent or any other person is or will be obligated to give notice of any defect or irregularity in the tender of S1 Shares, or to waive any such defect or irregularity, and none of them will incur any liability for failure to give any such notice or waiver. Offeror's interpretation of the terms and conditions of the Exchange Offer, including the Letter of Election and Transmittal, will be final and binding to the fullest extent permitted by law.

IMPORTANT: THIS LETTER OF ELECTION AND TRANSMITTAL, TOGETHER WITH ANY SIGNATURE GUARANTEES, OR, IN THE CASE OF A BOOK-ENTRY TRANSFER, AN AGENT'S MESSAGE, AND ANY OTHER REQUIRED DOCUMENTS, MUST BE RECEIVED BY THE EXCHANGE AGENT OR 51 SHARES MUST BE DELIVERED PURSUANT TO THE EXCHANGE OFFER AND EITHER CERTIFICATES FOR TENDERED SHARES MUST BE RECEIVED BY THE EXCHANGE AGENT OR 51 SHARES MUST BE DELIVERED PURSUANT TO THE PROCEDURES FOR BOOK-ENTRY TRANSFER, IN EACH CASE PRIOR TO THE EXPIRATION TIME OF THE EXCHANGE OFFER, OR THE TENDERING S1 STOCKHOLDER MUST COMPLY WITH THE PROCEDURES FOR GUARANTEED DELIVERY.

Facsimiles of this Letter of Election and Transmittal, properly completed and duly signed, will be accepted. In such case, a Letter of Election and Transmittal and certificates evidencing S1 Shares and any other required documents should be sent or delivered by each stockholder or such stockholder's broker, dealer, commercial bank, trust company or other nominee to the Exchange Agent at one of its addresses.

ACI and Offeror are not aware of any jurisdiction where the making of the Exchange Offer or the tender of S1 Shares in connection therewith would not be in compliance with the laws of such jurisdiction. In any jurisdiction where the securities, blue sky or other laws require the Exchange Offer to be made by a licensed broker or dealer, the Exchange Offer shall be deemed to be made on behalf of ACI, through Offeror, by one or more registered brokers or dealers licensed under the laws of such jurisdiction.

IMPORTANT TAX INFORMATION

TO ENSURE COMPLIANCE WITH U.S. TREASURY DEPARTMENT CIRCULAR 230, S1 STOCKHOLDERS ARE HEREBY NOTIFIED THAT: (1) ANY DISCUSSION OF U.S. FEDERAL TAX ISSUES IN THIS LETTER OF ELECTION AND TRANSMITTAL IS NOT INTENDED OR WRITTEN TO BE RELIED UPON, AND CANNOT BE RELIED UPON BY S1 STOCKHOLDERS, FOR THE PURPOSE OF AVOIDING PENALTIES THAT MAY BE IMPOSED ON S1 STOCKHOLDERS UNDER THE U.S. INTERNAL REVENUE CODE; (2) SUCH DISCUSSION IS WRITTEN IN CONNECTION WITH THE PROMOTION OF MARKETING OF THE TRANSACTIONS OR MATTERS ADDRESSED HEREIN; AND (3) S1 STOCKHOLDERS SHOULD SEEK ADVICE BASED ON THEIR PARTICULAR CIRCUMSTANCES FROM THEIR OWN INDEPENDENT TAX ADVISORS.

Under federal income tax laws, an S1 stockholder whose tendered S1 Shares are accepted for payment is required to provide the Exchange Agent (as payor) with such S1 stockholder's correct TIN on the Substitute Form W-9 below or otherwise establish a basis for exemption from backup withholding. If such S1 stockholder is an individual, the TIN is his or her social security number or individual taxpayer identification number, as the case may be. If the Exchange Agent is not provided with the TIN, a \$50 penalty may be imposed on you by the Internal Revenue Service, and payments made with respect to S1 Shares purchased pursuant to the Exchange Offer may be subject to backup withholding. Failure to comply truthfully with the backup withholding requirements also may result in the imposition of severe criminal and/or civil fines and penalties.

Certain S1 stockholders (including, among others, corporations and certain foreign persons) are not subject to these backup withholding requirements. Exempt S1 stockholders (other than foreign persons) should furnish their TIN, check the box in Part II of the Substitute Form W-9, and sign, date and return the Substitute Form W-9 to the Exchange Agent. A foreign person, including an entity, may qualify as an exempt recipient by submitting to the Exchange Agent an appropriate, properly completed Internal Revenue Service Form W-8ECI, W-8ECI, W-8ECI or W-8IMY (a "Form W-8"), as the case may be, signed under penalties of perjury, certifying to that S1 stockholder's foreign status. An appropriate Form W-8 can be obtained from the Exchange Agent. See the enclosed "Guidelines for Certification of Taxpayer Identification Number on Substitute Form W-9" for additional instructions.

If backup withholding applies, the Exchange Agent is required to withhold the applicable backup withholding amount of any payments made to the S1 stockholder or other payee. Backup withholding is not an additional federal income tax. Rather, any federal income tax liability of persons subject to backup withholding will be reduced by the amount of tax withheld. If withholding results in an overpayment of taxes, a refund may be obtained from the Internal Revenue Service.

PURPOSE OF SUBSTITUTE FORM W-9

To prevent backup withholding on payments made to tendering S1 stockholders pursuant to the Exchange Offer, the S1 stockholder is required to provide the Exchange Agent with either (1) the S1 stockholder's correct TIN by completing the form below, certifying that the TIN provided on Substitute Form W-9 is correct (or that such S1 stockholder is awaiting a TIN) and that the S1 stockholder is not subject to backup withholding because (a) the S1 stockholder is exempt from backup withholding, (b) the S1 stockholder has not been notified by the Internal Revenue Service that the S1 stockholder is subject to backup withholding as a result of failure to report all interest or dividends or (c) the Internal Revenue Service has notified the S1 stockholder that the S1 stockholder is no longer subject to backup withholding; or (2) another adequate basis for exemption.

WHAT NUMBER TO GIVE THE EXCHANGE AGENT

The S1 stockholder is required to give the Exchange Agent the TIN (e.g., social security number, individual taxpayer identification number or employer identification number) of the registered S1 stockholder. If the S1 Shares are held in more than one name or are held not in the name of the actual owner, consult the enclosed "Guidelines for Certification of Taxpayer Identification Number on Substitute Form W-9" for additional guidance on which number to report.

SUBSTITUTE	PLEASE PROVIDE YOUR TIN IN THE BOX AT RIGHT AND CERTIFY BY SIGNING AND DATING BELOW	Part I — Social Security Number OR Employer Identification Number	
Form W-9			
Department of the Treasury Internal Revenue Service	Name	(If awaiting TIN, write "Applied For")	
Payer's Request for Taxpayer Identification Number (TIN)	Business Name Please check appropriate box o Individual/Sole Proprietor o Corporation o Partnership o Other	Part II — For Payees exempt from backup withholding, see the enclosed Guidelines for Certification of Taxpayer Identification Number on Substitute Form W-9, check the Exempt box below, and complete the Substitute Form W-9.	
	Address	Exempt o Part III	
	City, State, Zip Code	Awaiting TIN o Please complete the Certificate of Awaiting Taxpayer Number below.	
CERTIFICATION — UNDER PENALTIES OF PETJURY CERTIFICATION — UNDER PENALTIES OF PETJURY			

Under penalties of perjury, I certify that:

(1) The number shown on this form is my correct taxpayer identification number (or I am waiting for a number to be issued to me), and(2) I am not subject to backup withholding because (a) I am exempt from backup withholding, (b) I have not been notified by the Internal Revenue Service (IRS) that I am subject to backup withholding as a result of a failure to report all interest or dividends, or (c) the IRS has notified me that I am no longer subject to backup withholding, and(3) I am a U.S. person (including a U.S. resident alien).

Certification Instructions — You must cross out item (2) above if you have been notified by the IRS that you are subject to backup withholding because you have failed to report all interest and dividends on your tax return. (Also see instructions in the enclosed Guidelines for Certification of Taxpayer Identification Number on Substitute Form W-9).

SIGNATURE: DATE:

YOU MUST COMPLETE THE FOLLOWING CERTIFICATE IF YOU CHECKED THE BOX IN PART III OF THE SUBSTITUTE FORM W-9.

CERTIFICATE OF AWAITING TAXPAYER IDENTIFICATION NUMBER

I certify under penalties of perjury that a taxpayer identification number has not been issued to me and either (1) I have mailed or delivered an application to receive a taxpayer identification number to the appropriate Internal Revenue Service Center or Social Security Administration Office or (2) I intend to mail or deliver an application in the near future. I understand that if I do not provide a taxpayer identification number within 60 days, 28% of all reportable payments made to me will be withheld until I provide a taxpayer identification number.

SIGNATURE: DATE:

NOTE: FAILURE TO COMPLETE AND RETURN THIS FORM MAY RESULT IN BACKUP WITHHOLDING OF A PORTION OF ANY PAYMENT MADE TO YOU PURSUANT TO THE OFFER. PLEASE REVIEW THE ENCLOSED GUIDELINES FOR CERTIFICATION OF TAXPAYER IDENTIFICATION NUMBER ON SUBSTITUTE FORM W-9 FOR ADDITIONAL DETAILS.

GUIDELINES FOR CERTIFICATION OF TAXPAYER IDENTIFICATION NUMBER ON SUBSTITUTE FORM W-9

GUIDELINES FOR DETERMINING THE PROPER IDENTIFICATION NUMBER TO GIVE THE PAYER. — Social Security numbers have nine digits separated by two hyphens: i.e. 000-00-0000. Employer identification numbers have nine digits separated by only one hyphen: i.e. 00-0000000. The table below will help determine the number to give the payer.

or this	type of account:	Give the TAXPAYER IDENTIFICATION NUMBER of —
	An individual's account	The individual
	Two or more individuals (joint account)	The actual owner of the account or, if combined funds, the first individual on the account(1)
	Custodian account of a minor (Uniform Gift to Minors Act)	The minor(2)
	 The usual revocable savings trust account (grantor is also trustee) 	The grantor-trustee (1)
	 b. So-called trust account that is not a legal or valid trust under state law 	The actual owner(1)
	Sole proprietorship or single owner LLC account	The owner(3)
	A valid trust, estate, or pension trust	The legal entity(4)
	77 120, 50	
		Give the TAXPAYER
		IDENTIFICATION NUMBER

- For this type of account:

 7. Corporate or LLC electing corporate status on Form 8832 or Form 2553

 8. Association, club, religious, Charitable, educational, or other tax-exempt organization account

 9. Partnership or multi-member LLC account

 10. A broker or registered nominee

 11. Account with the Department of Agriculture in the name of a public entity (such as a state or local government, school district, or prison) that receives agricultural program payments

 12. Grantor trust filing under Optional Form 1099 Filing Method 1 (see Regulation section 1.671-4(b)(2)(i)(A))

 13. Grantor trust filing under the Form 1041 Filing Method or the Optional Form 1099 Filing Method 2 (see Regulation section 1.671-4(b)(2)(i)(B)) IDENTIFICATION N
 of —
 The corporation
 The organization
 The partnership
 The broker or nominee
 The public entity The grantor(5) The trust
- List first and circle the name of the person whose number you furnish. If only one person on a joint account has a Social Security number, that person's number must be furnished.

 Circle the minor's name and furnish the minor's Social Security number.

 If the owner is an individual, you must show your individual name, and you may also enter your business name, on the business name line and may use either your Social Security number or your employer identification number (if you have one). List first and circle the name of the legal trust, estate or pension trust. Do not furnish the taxpayer identification number of the personal representative or trustee unless the legal entity itself is not designated in the account title.

 Grantor also must provide a Form W-9 to trustee of trust.

If no name is circled when more than one name is listed, the number will be considered to be that of the first name listed.

RESIDENT ALIEN INDIVIDUALS: If you are a resident alien individual and you do not have, and are not eligible to get, a Social Security number, your taxpayer identification number is your individual taxpayer identification number ("ITIN") as issued by the Internal Revenue Service. Enter it on the portion of the Substitute Form W-9 where the Social Security number would otherwise be entered. If you do not have an ITIN, see "Obtaining a Number" below.

GUIDELINES FOR CERTIFICATION OF TAXPAYER IDENTIFICATION NUMBER ON SUBSTITUTE FORM W-9 Page 2

Obtaining a Number. If you do not have a taxpayer identification number or you don't know your number, obtain Form SS-5, Application for a Social Security Number Card (for individuals), or Form SS-4, Application for Employer Identification Number (for businesses and all other entities), at the local office of the Social Security Administration or the Internal Revenue Service (the "IRS") and apply for a number. You may also obtain such forms from the IRS's website at irs.gov. Resident alien individuals who are not eligible to get a Social Security number and need an ITIN should obtain Form W-7, Application for Individual Taxpayer Identification Number, from the IRS.

Payees and Payments Exempt from Backup Withholding. The following is a list of payees exempt from backup withholding and for which no information reporting is required. For interest and dividends, all listed payees are exempt except the payee in item (9). For broker transactions, (a) payees listed in items (2) through (13), (b) a "C corporation," and (c) a person registered under the Investment Advisers Act of 1940 who regularly acts as a broker are exempt. For barter exchange transactions and patronage dividends, payees listed in items (2) through (6) are exempt. Payments subject to reporting under sections 6041 and 6041A are generally exempt from backup withholding only if made to payees described in items (1) through (7). Unless otherwise indicated, all "section" references are to sections of the Internal Revenue Code of 1986, as amended (the "Code"):

- (1) A corporation.
- (2) An organization exempt from tax under section 501(a), or an IRA, or a custodial account under section 403(b)(7) if the account satisfies the requirements of section 401(f)(2).
- (3) The United States or any of its agencies or instrumentalities.
- (4) A state, the District of Columbia, a possession of the United States, or any of their political subdivisions or instrumentalities.
- (5) A foreign government or any of its political subdivisions, agencies or instrumentalities.
- (6) An international organization or any of its agencies or instrumentalities.
- (7) A foreign central bank of issue.
- (8) A dealer in securities or commodities required to register in the United States, the District of Columbia, or a possession of the United States.
- (9) A futures commission merchant registered with the Commodity Futures Trading Commission.
- (10) A real estate investment trust
- (11) An entity registered at all times during the tax year under the Investment Company Act of 1940.
- (12) A common trust fund operated by a bank under section 584(a).
- (13) A financial institution.
- (14) A middleman known in the investment community as a nominee or custodian.
- (15) A trust exempt from tax under section 664 or described in section 4947.

Payments of dividends and patronage dividends not generally subject to backup withholding include the following: payments to nonresident aliens subject to withholding under section 1441 of the Code; payments to partnerships not engaged in a trade or business in the U.S. and which have at least one nonresident partner; payments of patronage dividends where the amount received is not paid in money; payments made by certain foreign organizations; Section 404(k) distributions made by an ESOP.

Payments of interest not generally subject to backup withholding include the following: payments of interest on obligations issued by individuals (note: You may be subject to backup withholding if this interest is \$600 or more and is paid in the course of your trade or business and you have not provided your correct taxpayer identification number to the payer); payments of tax-exempt interest (including exempt-interest dividends under section 852); payments described in

section 6049(b)(5) of the Code to non-resident aliens; payments on tax-free covenant bonds under section 1451 of the Code; payments made by certain foreign organizations; payments of mortgage or student loan interest to you

Exempt payees described above should file Substitute Form W-9 to avoid possible erroneous backup withholding. FILE THIS FORM WITH THE PAYER, FURNISH YOUR TAXPAYER IDENTIFICATION NUMBER, WRITE "EXEMPT" ON THE FACE OF THE FORM, SIGN AND DATE THE FORM AND RETURN IT TO THE PAYER. IF YOU ARE A NON-RESIDENT ALIEN OR A FOREIGN ENTITY NOT SUBJECT TO BACKUP WITHHOLDING, FILE WITH PAYER AN APPROPRIATE COMPLETED INTERNAL REVENUE SERVICE FORM W-8.

Certain payments other than interest, dividends and patronage dividends, that are not subject to information reporting are also not subject to backup withholding. For details, see Sections 6041, 6041A, 6042, 6044, 6045, 6049, 6050A and 6050N of the Code and the Treasury regulations promulgated thereunder.

Privacy Act Notice. Section 6109 requires most recipients of dividend, interest, or other payments to give their correct taxpayer identification numbers to payers who must report the payments to the IRS. The IRS uses the numbers for identification purposes and to help verify the accuracy of your tax return. The IRS may also provide this information to the Department of Justice for civil and criminal litigation, and to cities, states, and the District of Columbia to carry out their tax laws. We may also disclose this information to other countries under a tax treaty, or to Federal and state agencies to enforce Federal nontax criminal laws and to combat terrorism. Payers must be given the numbers whether or not recipients are required to file tax returns. Payers must generally withhold 28% of taxable interest, dividend, and certain other payments to a payee who does not furnish a taxpayer identification number to a payer. Certain penalties may also apply.

Penalties

- (1) Penalty for Failure to Furnish Taxpayer Identification Number. If you fail to furnish your correct taxpayer identification number to a payer, you are subject to a penalty of \$50 for each such failure unless your failure is due to reasonable cause and not to willful neglect.
- (2) Civil Penalty for False Information With Respect to Withholding. If you make a false statement with no reasonable basis which results in no imposition of backup withholding, you are subject to a penalty of \$500.
 - (3) Criminal Penalty for Falsifying Information. Willfully falsifying certifications or affirmations may subject you to criminal penalties including fines and/or imprisonment.

FOR ADDITIONAL INFORMATION CONTACT YOUR TAX CONSULTANT OR THE INTERNAL REVENUE SERVICE.

Questions or requests for assistance may be directed to the Information Agent at its address and telephone numbers as set forth below. Requests for additional copies of the Prospectus/Offer to Exchange, this Letter of Election and Transmittal, the Notice of Guaranteed Delivery and the IRS Form W-9 may be directed to the Information Agent. S1 stockholders may also contact their brokers, dealers, commercial banks, trust companies or other nominees for assistance concerning the Exchange Offer.

The exchange agent for the Exchange Offer is:



By Mail
Wells Fargo Bank, N.A.
Shareowner Services
Voluntary Corporate Actions
P.O. Box 64854
St. Paul, MN 55164-0854

By Hand or Overnight Delivery:

(Until 5:00 p.m. Eastern Time at the Expiration Time) Wells Fargo Bank, N.A. Shareowner Services Voluntary Corporate Actions 161 N. Concord Exchange South St. Paul, MN 55075-1139

Any questions or requests for assistance may be directed to the information agent at the address or telephone numbers set forth below. Additional copies of the Prospectus/Offer to Exchange, this Letter of Election and Transmittal and the Notice of Guaranteed Delivery may be obtained from the information agent at its address and telephone numbers set forth below. S1 stockholders may also contact their brokers, dealers, commercial banks or trust companiess or other nominees for assistance concerning the Exchange Offer.

The information agent for the Exchange Offer is:



501 Madison Avenue, 20th Floor New York, New York 10022 Stockholders May Call Toll Free: (888) 750-5834 Banks and Brokers May Call Collect: (212) 750-5833

NOTICE OF GUARANTEED DELIVERY

To Tender Shares of Common Stock

S1 CORPORATION

for

0.3148 of a Share of ACI Common Stock

or

\$10.00 in Cash,

subject to the proration procedures described in the Prospectus/Offer to Exchange dated October 13, 2011 and the related Letter of Election and Transmittal,

h

ANTELOPE INVESTMENT CO. LLC,

a wholly owned subsidiary of

ACI WORLDWIDE, INC.

Pursuant to the Prospectus/Offer to Exchange dated October 13, 2011

(Not to be used for Signature Guarantees)

THE EXCHANGE OFFER AND THE WITHDRAWAL RIGHTS WILL EXPIRE AT 5:00 P.M., EASTERN TIME, ON MONDAY, OCTOBER 31, 2011, UNLESS EXTENDED. S1 SHARES TENDERED PURSUANT TO THE EXCHANGE OFFER MAY BE WITHDRAWN AT ANY TIME PRIOR TO THE EXPIRATION OF THE EXCHANGE OFFER, BUT NOT DURING ANY SUBSEQUENT OFFERING PERIOD.

The exchange agent for the Exchange Offer is:



By Mail
Wells Fargo Bank, N.A.
Shareowner Services
Voluntary Corporate Actions
P.O. Box 64854
St. Paul, MN 55164-0854

For Notice of Guaranteed Deliver (For Eligible Institutions Only) By Facsimile Transmission: (866) 734-9952 (FAX)

To Confirm Receipt of Notice of Guaranteed Delivery Only: (800) 468-9716 By Hand or Overnight Delivery: (Until 5:00 p.m. Eastern Time at the Expiration Time) Wells Fargo Bank, N.A. Shareowner Services Voluntary Corporate Actions 161 N. Concord Exchange South St. Paul, MN 55075-1139 This Notice of Guaranteed Delivery, or a form substantially equivalent to this form, must be used by stockholders of S1 Corporation, a Delaware corporation ("S1"), desiring to tender shares of common stock, par value \$0.01 per share (the "S1 Shares"), of S1 pursuant to the Exchange Offer (as defined below) if certificates evidencing S1 Shares are not immediately available or if the procedure for book-entry transfer cannot be completed on a timely basis or time will not permit all required documents to reach Wells Fargo Bank, N.A. (the "Exchange Agent") on or prior to the expiration time of the Exchange Offer.

To tender S1 Shares, this Notice of Guaranteed Delivery must be delivered to the Exchange Agent at one of its addresses set forth above by the expiration date of the Exchange Offer. Delivery of this Notice of Guaranteed Delivery to an address other than as set forth above or transmission to a facsimile number other than the one listed above will not constitute valid delivery to the Exchange Agent. Please see the section of the Prospectus/Offer to Exchange entitled "The Exchange Offer — Procedure for Tendering."

This Notice of Guaranteed Delivery is not to be used to guarantee signatures. If a signature on a Letter of Election and Transmittal is required to be guaranteed by an "eligible institution" under the instructions thereto, such signature guarantee must appear in the applicable space provided in the signature box on the Letter of Election and Transmittal.

The undersigned hereby tenders to Antelope Investment Co. LLC ("Offeror"), a Delaware limited liability company and a wholly owned subsidiary of ACI Worldwide, Inc., a Delaware corporation ("ACI"), upon the terms and subject to the conditions set forth in the Prospectus/Offer to Exchange and the related Letter of Election and Transmittal, each dated October 13, 2011 (which together, as amended, supplemented or modified from time to time, constitute the "Exchange Offer"), receipt of which is hereby acknowledged, the number of S1 Shares set forth below, pursuant to the guaranteed delivery procedures set forth in the section of the Prospectus/Offer to Exchange titled "The Exchange Offer — Procedure for Tendering." Such tenders are subject to the elections specified below under the heading "Exchange Offer Election," or, if no election is specified, will be deemed tendered with no election. See "The Exchange Offer — Consequences of Tendering with No Election" in the Prospectus/Offer to Exchange. Capitalized terms used but not defined herein have the respective meanings given to them in the Prospectus/Offer to Exchange.
Number of Shares:
Certificate Numbers (if available):
Name of Tendering Institution:
Name(s) of Record Holders:
Taxpayer Identification or Social Security Number:
Check this box if shares will be delivered by book-entry transfer: o
Account Number:
Address(es):
(Address)
(Zip Code) Area Code and Telephone Number(s):
Dated:
Signature(s) of Holder(s)):

Ladies and Gentlemen:

EXCHANGE OFFER ELECTION

	This section must be completed if you desire to elect the type of consideration to be received in exchange for the S1 Shares being tendered hereby. Please note that, as described in the Prospectus/Offer to Exchange		
The undersigned elects to receive the consideration indicated below. The undersigned acknowledges that failure to properly make an election will result in the undersigned being deemed to have made			
		NUMBER OF SHARES TENDERED FOR 0.3148 OF AN ACI SHARE , SUBJECT TO PRORATION AND THE PAYMENT OF CASH IN RESPECT OF FRACTIONAL ACI SHARES.	
		NUMBER OF SHARES TENDERED FOR \$10.00 IN CASH, WITHOUT INTEREST, SUBJECT TO PRORATION.	
ì	TENDERED S1 S	TENDERED BUT FOR WHICH NO ELECTION IS MADE ABOVE WILL BE DEEMED TO HAVE BEEN TENDERED WITHOUT AN ELECTION, AND THE HARES WILL BE DEEMED TO HAVE ELECTED THE CASH CONSIDERATION AS DESCRIBED IN "THE EXCHANGE OFFER — CONSEQUENCES OF TH NO ELECTION" IN THE PROSPECTUS/OFFER TO EXCHANGE.	

GUARANTEED DELIVERY PROCEDURES

S1 stockholders desiring to tender S1 Shares pursuant to the Exchange Offer but whose certificates are not immediately available or cannot otherwise be delivered with all other required documents to the Exchange Age

- the tender is by or through an eligible institution;
- a properly completed and duly executed form of this Notice of Guaranteed Delivery, or a form substantially equivalent to this form, is received by the Exchange Agent as provided below on or prior to the expiration date; and
- the certificates for all tendered S1 Shares (or a confirmation of a book-entry transfer of such shares into the Exchange Agent's account at the book-entry transfer facility), in proper form for transfer, together with a properly completed and duly executed Letter of Election and Transmittal with any required signature guarantees (or, in the case of a book-entry transfer, an agent's message) and all other documents required by the Letter of Election and Transmittal are received by the Exchange Agent at one of its addresses on the cover hereof within three (3) NASDAQ trading daysafter the date of execution of such notice of guaranteed delivery.

An S1 stockholder may deliver this Notice of Guaranteed Delivery by hand, facsimile transmission or mail to the Exchange Agent at one of its addresses on the cover hereof. The notice must include a guarantee by an e In all cases, Offeror will exchange S1 Shares tendered and accepted for exchange pursuant to the Exchange Offer only after timely receipt by the Exchange Agent of certificates for S1 Shares (or timely confirmation of

GUARANTEE (Not to be used for signature guarantee)

The undersigned, a member of the Securities Transfer Agents Medallion Program or an "Eligible Guarantor Institution," as such term is defined in Rule 17Ad-15 under the Securities Exchange Act of 1934, as amended The Eligible Institution that completes this form must communicate the guarantee to Exchange Agent and must deliver the Letter of Election and Transmittal and certificates for S1 Shares to the Exchange Agent within

Name of Firm			Authorized Signature
Address		Title ,	
	Zip Code	Name:	(Please Type or Print)
Area Code and Tel. No.		Dated:	

DO NOT SEND SHARE CERTIFICATES WITH THIS NOTICE. SHARE CERTIFICATES SHOULD BE SENT ONLY WITH YOUR LETTER OF ELECTION AND TRANSMITTAL.

LETTER TO BROKERS, DEALERS, COMMERCIAL BANKS, TRUST COMPANIES AND OTHER NOMINEES

To Tender Shares of Common Stock

S1 CORPORATION

for

0.3148 of a Share of ACI Common Stock

0

\$10.00 in Cash,

subject to the proration procedures described in the Prospectus/Offer to Exchange dated October 13, 2011 and the related Letter of Election and Transmittal,

bv

ANTELOPE INVESTMENT CO. LLC,

a wholly owned subsidiary of

ACI WORLDWIDE, INC.

THE EXCHANGE OFFER AND THE WITHDRAWAL RIGHTS WILL EXPIRE AT 5:00 P.M., EASTERN TIME, ON MONDAY, OCTOBER 31, 2011, UNLESS EXTENDED. SI

October 13, 2011

To Brokers, Dealers, Commercial Banks, Trust Companies and Other Nominees:

We have been engaged and appointed by ACI Worldwide, Inc., a Delaware corporation ("ACI"), and Antelope Investment Co. LLC ("Offeror"), a Delaware limited liability company and a wholly owned subsidiary of ACI, to

- $\bullet \quad 0.3148 \ of a \ share \ of \ ACI \ common \ stock \ (the "ACI \ Shares"), \ par \ value \ \$0.005 \ per \ share \ (the "Stock \ Consideration"); \ or \ an instance \ (the \ "Stock \ Consideration"), \ or \ an instance \ (the \ "Stock \ Consideration"); \ or \$
- \$10.00 in cash, without interest (the "Cash Consideration"),

subject to the proration procedures described in the Prospectus/Offer to Exchange, dated October 13, 2011, and in the related Letter of Election and Transmittal (which together, as amended, supplemented or modified from time to ti

As discussed in the Prospectus/Offer to Exchange, the Exchange Offer is not being made in any jurisdiction where the Exchange Offer would not be in compliance with the applicable laws of such jurisdiction.

YOUR PROMPT ACTION IS REQUESTED. WE URGE YOU TO CONTACT YOUR CLIENTS AS PROMPTLY AS POSSIBLE. PLEASE NOTE THAT THE EXCHANGE OFFER WILL EXPIRE AT 5:00 I

Enclosed herewith for your information and forwarding to your clients for whom you hold S1 Shares registered in your name or the name of your nominee are copies of the following documents:

- 1. The Prospectus/Offer to Exchange, dated October 13, 2011.
- 2. The Letter of Election and Transmittal for your use in accepting the Exchange Offer and tendering S1 Shares and for the information of your clients. Facsimile copies of the Letter of Election and Transmittal may be t
- 3. The Notice of Guaranteed Delivery to be used to accept the Exchange Offer if certificates evidencing S1 Shares are not immediately available or if you cannot deliver the certificates and all other required
- 4. A form of the letter that may be sent to your clients for whose accounts you hold S1 Shares registered in your name or in the name of your nominee, with space provided for obtaining such clients' instructions with re
- 5. A return envelope addressed to us.

Please note the following:

A. The consideration for each Share of S1 Common Stock is either of the following:

- · 0.3148 of an ACI Share (Stock Consideration); or
- \$10.00 in cash, without interest (Cash Consideration),

subject to the proration procedures described in the Prospectus/Offer to Exchange and the Letter of Election and Transmittal, and cash in lieu of any fractional ACI Shares, as described in the Prospectus/Offer to Exchange. IF THE

- B. The Exchange Offer is being made for all issued and outstanding S1 Shares. The Exchange Offer is being made pursuant to the Transaction Agreement, dated as of October 3, 2011, among ACI, Offeror and S1 (as amender
- C. The Exchange Offer and the withdrawal rights expire at 5:00 p.m., Eastern time, on Monday, October 31, 2011, unless extended as described in the Prospectus/Offer to Exchange (as extended, the "Expiration"
- D. The Exchange Offer is subject to the conditions set forth in the section of the Prospectus/Offer to Exchange offer Conditions of the Exchange Offer," which we urge you to review in detail. These of the Exchange offer is subject to the conditions set forth in the section of the Prospectus/Offer to Exchange Offer Conditions of the Exchange Offer," which we urge you to review in detail. These of the Exchange Offer is subject to the conditions set forth in the section of the Prospectus/Offer to Exchange Offer Conditions of the Exchange Offer, which we urge you to review in detail. These of the Exchange Offer is subject to the conditions set forth in the section of the Prospectus/Offer to Exchange Offer Conditions of the Exchange Offer is subject to the conditions set forth in the section of the Prospectus/Offer to Exchange Offer Conditions of the Exchange Offer is subject to the conditions set forth in the section of the Prospectus/Offer to Exchange Offer is subject to the conditions set forth in the section of the Prospectus/Offer to Exchange Offer is subject to the Exchange Offer is subject to the conditions set forth in the section of the Prospectus/Offer to Exchange Offer is subject to the Exchange Offer is subject to the
- S1 stockholders shall have validly tendered and not properly withdrawn prior to the Expiration Time at least that number of S1 Shares (together with the S1 Shares then owned by ACI, Offeror or any of ACI's other subsidiaries), shall constitute a majority of the S1 Shares issued and outstanding on a fully diluted basis. We refer to this condition as the "Minimum Tender Condition."
- The registration statement of which this prospectus/offer to exchange is a part shall have been declared effective under the Securities Act of 1933, as amended, and no stop order suspending the effectiveness of the registration statement shall have been issued and no proceedings for that purpose shall have been initiated or threatened by the SEC, and ACI shall have received all necessary state securities law or "blue sky" authorizations.
- Any applicable waiting period under the HSR Act, and, if applicable, any agreement with the Federal Trade Commission or the Antitrust Division of the U.S. Department of Justice to accept S1 Shares for exchange in the Exchange Offer, shall have expired or shall have been terminated prior to the Expiration Time (the "HSR Condition").

- Any clearance, approval, permit, authorization, waiver, determination, favorable review or consent of any Governmental Authority, other than the HSR Condition, shall have been obtained and such
 approvals shall be in full force and effect, or any applicable waiting periods for such clearances or approvals shall have expired, except for any failures that would not reasonably be expected to have a
 material adverse effect on ACI or S1.
- · Any of the following fail to be true:
 - (1) the representations and warranties of the S1 relating to organization, standing and power, authority, capital structure, absence of certain changes or events, brokers and vote required, as set forth in the Transaction Agreement were true and correct as of October 3, 2011 and will be true and correct on and as of the Expiration Time with the same force and effect as if made at the Expiration Time (in either case other than those representations and warranties which address matters only as of a particular date, which representations and warranties of S1 set forth in the Transaction Agreement were true and correct as of October 3, 2011 and will be true and correct on and as of the Expiration Time with the same force and effect as if made on the Expiration Time (in either case other than those representations and warranties which address matters only as of a particular date, which representations shall have been true and correct as of such particular date), except in either case contemplated by this clause (2) where the failure of such representations and warranties to be true and correct (disregarding all qualifications or limitations as to materiality, material adverse effect or words of similar import set forth therein) has not had and would not reasonably be expected to have a material adverse effect on S1;
 - S1 has performed or complied in all material respects with all agreements and covenants required by the Transaction Agreement to be performed or complied with by it on or prior to the Expiration Time; and
 - since October 3, 2011, there shall not have occurred any material adverse change in the business, financial condition or continuing results of S1 and its subsidiaries, taken as a whole (excluding certain events specified in the Transaction Agreement).

E. Exchange of S1 Shares pursuant to the Exchange Offer will be made only after timely receipt by us of (1) a properly completed and duly executed Letter of Election and Transmittal (or a manually signed facsimile thereof). F. Any stock transfer taxes applicable to the transfer of S1 Shares to Offeror pursuant to the Exchange Offer will be paid by Offeror, except as otherwise provided in the Prospectus/Offer to Exchange and the related Letter of I Neither ACI nor Offeror will pay any commissions or fees to any broker, dealer or other person, other than us, Innisfree M&A Incorporated (the "Information Agent") and other persons described in the section of the Prospect S1 stockholders who wish to tender their S1 Shares but whose certificates representing S1 Shares are not immediately available or who cannot deliver their certificates and all other required documents to us prior to the expira

Any inquiries you may have with respect to the Exchange Offer should be addressed to, and additional copies of the enclosed material may be obtained from, the Information Agent at the address and telephone numbers set fo

Very truly yours,

Wells Fargo Bank, N. A.

NOTHING CONTAINED HEREIN OR IN THE ENCLOSED DOCUMENTS SHALL CONSTITUTE YOU OR ANY OTHER PERSON THE AGENT OF ACI OR OFFEROR, THE INFORMATION AGENT OF ACID OR OFFEROR, THE INFORMATION AGENT OF ACID OR OFFEROR, THE OFFEROR OF ACID OR OFFEROR OFFE

LETTER TO CLIENTS

To Tender Shares of Common Stock of

S1 CORPORATION

for 0.3148 of a Share of ACI Common Stock

\$10.00 in Cash,

subject to the proration procedures described in the Prospectus/Offer to Exchange dated October 13, 2011 and the related Letter of Election and Transmittal,

ANTELOPE INVESTMENT CO. LLC
a wholly owned subsidiary of
ACI WORLDWIDE, INC.

THE EXCHANGE OFFER AND THE WITHDRAWAL RIGHTS WILL EXPIRE AT 5:00 P.M., EASTERN TIME, ON MONDAY, OCTOBER 31, 2011, UNLESS EXTENDED. 5

October 13, 2011

To Our Clients:

Enclosed for your consideration are a Prospectus/Offer to Exchange, dated October 13, 2011, and a related Letter of Election and Transmittal (which together, as amended, supplemented or modified from time to time, constit

- 0.3148 of a share of ACI common stock (the "ACI Shares"), par value \$0.005 per share (the "Stock Consideration"); or
- \$10.00 in cash, without interest (the "Cash Consideration"),

subject to the proration procedures described in the Prospectus/Offer to Exchange and the Letter of Election and Transmittal, and cash in lieu of any fractional ACI Shares, upon the terms and subject to the conditions of the Exchange

We (or our nominees) are the holder of record of S1 Shares held by us for your account. A tender of such S1 Shares can be made only by us as the holder of record and pursuant to your instructions. The enclosed

Accordingly, we request instructions as to whether you wish us to tender on your behalf the S1 Shares held by us for your account, pursuant to the terms and conditions set forth in the Exchange Offer:

Your attention is directed to the following:

- 1. The consideration for each S1 Share is either of the following:
- · 0.3148 of an ACI Share (Stock Consideration); or
- \$10.00 in cash, without interest (Cash Consideration),

subject to the proration procedures described in the Prospectus/Offer to Exchange and the Letter of Election and Transmittal, and cash in lieu of any fractional ACI Shares, as described in the Prospectus/Offer to Exchange. IF THE

- 2. The Exchange Offer is being made for all issued and outstanding S1 Shares. The Exchange Offer is being made pursuant to the Transaction Agreement, dated as of October 3, 2011, among ACI, Offeror and S1 (as amended
- 3. The Exchange Offer expires at 5:00 p.m., Eastern time, on Monday, October 31, 2011, unless extended as described in the Prospectus/Offer to Exchange (as extended, the "Expiration Time of the Exchange Offer States of States
- 4. The Exchange Offer is subject to the conditions set forth in the section of the Prospectus/Offer to Exchange Offer Conditions of the Exchange Offer," which we urge you to review in detail. These c
- S1 stockholders shall have validly tendered and not properly withdrawn prior to the Expiration Time at least that number of S1 Shares (together with the S1 Shares then owned by ACI, Offeror or any of ACI's other subsidiaries), shall constitute a majority of the S1 Shares issued and outstanding on a fully diluted basis. We refer to this condition as the "Minimum Tender Condition."
- The registration statement of which this prospectus/offer to exchange is a part shall have been declared effective under the Securities Act of 1933, as amended, and no stop order suspending the effectiveness of the registration statement shall have been issued and no proceedings for that purpose shall have been initiated or threatened by the SEC, and ACI shall have received all necessary state securities law or "blue sky" authorizations
- Any applicable waiting period under the HSR Act, and, if applicable, any agreement with the Federal Trade Commission or the Antitrust Division of the U.S. Department of Justice to accept S1 Shares for exchange in the Exchange Offer, shall have expired or shall have been terminated prior to the Expiration Time (the "HSR Condition").
- Any clearance, approval, permit, authorization, waiver, determination, favorable review or consent of any Governmental Authority, other than the HSR Condition, shall have been obtained and such
 approvals shall be in full force and effect, or any applicable waiting periods for such clearances or approvals shall have expired, except for any failures that would not reasonably be expected to have a
 material adverse effect on ACI or S1.
- · Any of the following fail to be true:
 - (1) the representations and warranties of the S1 relating to organization, standing and power, authority, capital structure, absence of certain changes or events, brokers and vote required, as set forth in the Transaction Agreement were true and correct as of October 3, 2011 and will be true and correct on and as of the Expiration Time with the same force and effect as if made at the Expiration Time (in either case other than those representations and warranties which address matters only as of a particular date, which representations and warranties shall have been true and correct as of such particular date), except in either case contemplated by this clause (1) for de minimis inaccuracies and (2) the other representations and warranties of S1 set forth in the Transaction Agreement were true and correct as of October 3, 2011 and will be true and correct on and as of the Expiration Time with the same force and effect as if made on the Expiration Time (in either case

other than those representations and warranties which address matters only as of a particular date, which representations shall have been true and correct as of such particular date), except in either case contemplated by this clause (2) where the failure of such representations and warranties to be true and correct (disregarding all qualifications or limitations as to materiality, material adverse effect or words of similar import set forth therein) has not had and would not reasonably be expected to have a material adverse effect on S1;

- S1 has performed or complied in all material respects with all agreements and covenants required by the Transaction Agreement to be performed or complied with by it on or prior to the Expiration Time; and
- since October 3, 2011, there shall not have occurred any material adverse change in the business, financial condition or continuing results of S1 and its subsidiaries, taken as a whole (excluding certain events specified in the Transaction Agreement).

5. Any stock transfer taxes applicable to the transfer of S1 Shares to Offeror pursuant to the Exchange Offer will be paid by Offeror, except as otherwise provided in the Prospectus/Offer to Exchange and the related Letter of I We urge you to read the enclosed Prospectus/Offer to Exchange and Letter of Election and Transmittal regarding the Exchange Offer carefully before instructing us to tender your S1 Shares.

The Exchange Offer is being made solely pursuant to the Prospectus/Offer to Exchange and the accompanying Letter of Election and Transmittal, and any amendments or supplements thereto, and is being made to all S1 stocl If you wish to tender any or all of the S1 Shares held by us for your account, please so instruct us by completing, executing, detaching and returning to us the instruction form set forth on the back page of this letter. You shoul

Instructions with Respect to the Offer to Exchange All Issued and Outstanding Shares of Common Stock

S1 Corporation

The undersigned acknowledge(s) receipt of your letter and the enclosed Prospectus/Offer to Exchange, dated October 13, 2011, and the related Letter of Election and Transmittal (which together, as amended, supplemented or

- 0.3148 of a share of ACI common stock (the "ACI Shares"), par value \$0.005 per share (the "Stock Consideration"); or
- \$10.00 in cash, without interest (the "Cash Consideration"),

subject to the proration procedures described in the Prospectus/Offer to Exchange and the Letter of Election and Transmittal, and cash in lieu of any fractional ACI Shares, upon the terms and subject to the conditions of the Exchange With respect to the Exchange Offer, the undersigned hereby instructs you (check appropriate box):

- o To TENDER some or all of the S1 Shares held by you for the account of the undersigned
- o NOT to TENDER any S1 Shares held by you for the account of the undersigned.

If the undersigned is instructing you to tender some or all of the S1 Shares held by you for the account of the undersigned, the undersigned agrees and acknowledges that you are authorized to tender the number of S1 Shares in NUMBER OF SHARES TENDERED FOR 0.3148 OF AN ACI SHARE, SUBJECT TO PRORATION AND THE PAYMENT OF CASH IN RESPECT OF FRACTIONAL ACI SHARES.

NUMBER OF SHARES TENDERED FOR \$10.00 IN CASH, WITHOUT INTEREST, SUBJECT TO PRORATION.

† Unless otherwise indicated above, it will be assumed that all S1 Shares held by us for your account will be tendered.

PLEASE SIGN AND COMPLETE		
Signature(s):		
Dated:		
(Please Print)		
Name of Beneficial Owner(s):		
Capacity(s) (full title)*:		
Address(es):		
Area Code and Telephone Number(s):		
Taxpayer Identification or Social Security Number(s):		
* Please provide if signature is by an attorney-in-fact, executor, administrator, trustee, guardian, officer of a corporation or other person acting in a fiduciary or representative capacity. PLEASE RETURN THIS FORM TO THE BROKERAGE FIRM MAINTAINING YOUR ACCOUNT, NOT TO THE EXCHANGE AGENT, INFORMATION AGENT, ACI OR OFFEROR.		

THE METHOD OF DELIVERY OF THIS DOCUMENT IS AT THE OPTION AND RISK OF THE ELECTING S1 STOCKHOLDER. IF DELIVERED BY MAIL, REGISTERED MAIL WITH RETURN RECEI